UNITED STATES SECURITIES AND EXCHANGE COMMISSION

Washington, D.C. 20549

FORM 10-K

(Mark One)

X ANNUAL REPORT PURSUANT TO SECTION 13 OR 15(d) OF THE SECURITIES EXCHANGE ACT OF 1934

to

For the Fiscal Year Ended February 1, 2014

TRANSITION REPORT PURSUANT TO SECTION 13 OR 15(d) OF THE SECURITIES EXCHANGE ACT OF 1934

for the transition period from

Commission File No. 1-3083

Genesco Inc.

(Exact name of registrant as specified in its charter)

Tennessee	62-0211340
(State or other jurisdiction of incorporation or organization)	(I.R.S. Employer Identification No.)
Genesco Park, 1415 Murfreesboro Road	
Nashville, Tennessee	37217-2895
(Address of principal executive offices)	(Zip Code)
Registrant's telephone number, including area code	e: (615) 367-7000
Securities Registered Pursuant to Section 12(b) of the Act:

Title of each class	Name of Exchange on which Registered
Common Stock, \$1.00 par value	New York
Preferred Share Purchase Rights	New York

Securities Registered Pursuant to Section 12(g) of the Act: Employees' Subordinated Convertible Preferred Stock

Indicate by check mark if the registrant is a well-known seasoned issuer, as defined in Rule 405 of the Securities Act. Yes x No 🗆

Indicate by check mark if the registrant is not required to file reports pursuant to Section 13 or Section 15(d) of the Act. Yes 🗆 No x

Indicate by check mark whether the registrant (1) has filed all reports required to be filed by Section 13 or 15(d) of the Securities Exchange Act of 1934 during the preceding 12 months (or for such shorter period that the registrant was required to file such reports) and (2) has been subject to such filing requirements for the past 90 days. Yes x No \Box

Indicate by check mark whether the registrant has submitted electronically and posted on its corporate Website, if any, every Interactive Data File required to be submitted and posted pursuant to Rule 405 of Regulation S-T (222-405 of this chapter) during the preceding 12 months (or for such shorter period that the registrant was required to submit and post such files). Yes x No

Indicate by check mark if disclosure of delinquent filers pursuant to Item 405 of Regulation S-K (§229.405 of this chapter) is not contained herein, and will not be contained, to the best of registrant's knowledge, in definitive proxy or information statements incorporated by reference in Part III of this Form 10-K or any amendment to this Form 10-K. x

Indicate by check mark whether the registrant is a large accelerated filer; an accelerated filer; a non-accelerated filer; or a smaller reporting company. See definitions of "large accelerated filer," "accelerated filer" and "smaller reporting company" in Rule 12b-2 of the Exchange Act.

Large accelerated filer	X	Accelerated filer	
Non-accelerated filer	□ (Do not check if smaller reporting company)	Smaller reporting company	

Indicate by check mark whether the registrant is a shell company (as defined in Rule 12b-2 of the Act.) Yes 🛛 No x

The aggregate market value of common stock held by nonaffiliates of the registrant as of August 3, 2013, the last business day of the registrant's most recently completed second fiscal quarter, was approximately \$1,738,000,000. The market value calculation was determined using a per share price of \$72.48, the price at which the common stock was last sold on the New York Stock Exchange on such date. For purposes of this calculation, shares held by nonaffiliates excludes only those shares beneficially owned by officers, directors, and shareholders owning 10% or more of the outstanding common stock (and, in each case, their immediate family members and affiliates).

As of March 14, 2014, 23,923,210 shares of the registrant's common stock were outstanding.

Documents Incorporated by Reference

Portions of Genesco's Annual Report to Shareholders for the fiscal year ended February 1, 2014 are incorporated into Part II by reference.

Portions of the proxy statement for the June 26, 2014 annual meeting of shareholders are incorporated into Part III by reference.

TABLE OF CONTENTS

		Page
	<u>PART I</u>	
<u>ltem 1.</u>	Business	<u>3</u>
ltem 1A.	Risk Factors	<u>8</u>
<u>tem 1B.</u>	Unresolved Staff Comments	<u>14</u>
<u>[tem 2.</u>	<u>Properties</u>	<u>14</u>
<u>tem 3.</u>	Legal Proceedings	<u>14</u>
<u>ltem 4.</u>	Mine Safety Disclosures	<u>16</u>
<u>ltem 4A.</u>	Executive Officers	<u>17</u>
	PART II	
<u>ltem 5.</u>	Market for Registrant's Common Equity, Related Stockholder Matters and Issuer Purchases of Equity Securities	<u>18</u>
<u>ltem 6.</u>	Selected Financial Data	<u>19</u>
l <u>tem 7.</u>	Management's Discussion and Analysis of Financial Condition and Results of Operations	<u>21</u>
<u>ltem 7A.</u>	Quantitative and Qualitative Disclosures about Market Risk	<u>39</u>
<u>ltem 8.</u>	Financial Statements and Supplementary Data	<u>40</u>
<u>ltem 9.</u>	Changes in and Disagreements with Accountants on Accounting and Financial Disclosure	<u>103</u>
<u>ltem 9A.</u>	Controls and Procedures	<u>103</u>
<u>(tem 9B.</u>	Other Information	<u>103</u>
	PART III	
(<u>tem 10.</u>	Directors, Executive Officers and Corporate Governance	<u>104</u>
[<u>tem 11.</u>	Executive Compensation	<u>104</u>
(<u>tem 12.</u>	Security Ownership of Certain Beneficial Owners and Management and Related Stockholder Matters	<u>104</u>
<u>ltem 13.</u>	Certain Relationships and Related Transactions, and Director Independence	<u>104</u>
<u>ltem 14.</u>	Principal Accounting Fees and Services	<u>104</u>

PART IV

Item 15. Exhibits and Financial Statement Schedules

2

<u>105</u>

PART I

ITEM 1, BUSINESS

General

Genesco Inc. ("Genesco" or the "Company") is a leading retailer and wholesaler of branded footwear, apparel and accessories with net sales for Fiscal 2014 of \$2.62 billion. During Fiscal 2014, the Company operated five reportable business segments (not including corporate): (i) Journeys Group, comprised of the Journeys, Journeys Kidz, Shi by Journeys and Underground by Journeys retail footwear chains, catalog and e-commerce operations; (ii) Schuh Group, comprised of the Schuh retail footwear chain and e-commerce operations; (iii) Lids Sports Group, comprised of (a) headwear and accessory stores under the Lids[®] name and other names in the U.S., Puerto Rico and Canada, (b) the Lids Locker Room and Lids Clubhouse businesses, consisting of sports-oriented fan shops featuring a broad array of licensed merchandise such as apparel, hats and accessories, sports decor and novelty products, operating under various trade names, (c) licensed team merchandise departments in Macy's department stores operated under the name Locker Room by Lids and on macys.com, under a license agreement with Macy's, (d) e-commerce operations, catalog and e-commerce operations and (e) an athletic team dealer business operating as Lids Team Sports; (iv) Johnston & Murphy Group, comprised of Johnston & Murphy retail operations, catalog and e-commerce operations and wholesale distribution of products under the Johnston & Murphy and Trask brands; and (v) Licensed Brands, comprised of Dockers[®] footwear, sourced and marketed under a license from Levi Strauss & Company; SureGrip[®]Footwear, occupational footwear primarily sold directly to consumers; and other brands.

At February 1, 2014, the Company operated 2,568 retail footwear, headwear and sports apparel and accessory stores and leased departments located primarily throughout the United States and in Puerto Rico, but also including 110 headwear stores and 38 footwear stores in Canada and 99 footwear stores in the United Kingdom and the Republic of Ireland. It currently plans to open a total of approximately 344 new retail stores, including 175 leased departments, and to close 47 retail stores in Fiscal 2015. At February 1, 2014, Journeys Group operated 1,168 stores, including 174 Journeys Kidz, 50 Shi by Journeys and 117 Underground by Journeys; Schuh Group operated 99 stores; Lids Sports Group operated 1,133 stores, including 930 Lids stores, 177 Lids Locker Room and Clubhouse stores and 26 Locker Room by Lids leased departments, and Johnston & Murphy Group operated 168 retail shops and factory stores.

The following table sets forth certain additional information concerning the Company's retail footwear, headwear and sports apparel and accessory stores and leased departments during the five most recent fiscal years:

	Fiscal 2010	Fiscal 2011	Fiscal 2012	Fiscal 2013	Fiscal 2014
Retail Stores					
Beginning of year	2,234	2,276	2,309	2,387	2,459
Opened during year	61	53	70	104	183
Acquired during year	38	58	85	33	15
Closed during year	(57)	(78)	(77)	(65)	(89)
End of year	2,276	2,309	2,387	2,459	2,568

The Company also designs, sources, markets and distributes footwear under its own Johnston & Murphy brand, the recently relaunched Trask brand, the licensed Dockers[®] brand and other brands that the Company licenses for men's footwear to over 1,000 retail accounts in the United States, including a number of leading department, discount, and specialty stores.

Shorthand references to fiscal years (e.g., "Fiscal 2014") refer to the fiscal year ended on the Saturday nearest January 31st in the named year (e.g., February 1, 2014). The terms "Company," "Genesco," "we," "our" or "us" as used herein and unless otherwise stated or indicated by context refer to Genesco Inc. and its subsidiaries. All information contained in "Management's Discussion and Analysis of Financial Condition and Results of Operations," which is referred to in Item 1 of this report, is incorporated by such reference in Item 1. This report contains forward-looking statements. Actual results may vary materially and adversely from the expectations reflected in these statements. For a discussion of some of the factors that may lead to different results, see Item 1A, "Risk Factors" and Item 7, "Management's Discussion and Analysis of Financial Condition and Results of Operations."

Available Information

The Company files reports with the Securities and Exchange Commission ("SEC"), including annual reports on Form 10-K, quarterly reports on Form 10-Q and other reports from time to time. The public may read and copy any materials we file with the SEC at the SEC's Public Reference Room at 100 F. Street, NE, Washington, DC 20549. The public may obtain

information on the operation of the Public Reference Room by calling the SEC at 1-800-SEC-0330. The Company is an electronic filer and the SEC maintains an Internet site at http://www.sec.gov that contains the reports, proxy and information statements, and other information filed electronically. The Company's website address is http://www.genesco.com. The Company's website address is provided as an inactive textual reference only. The Company makes available free of charge through the website annual reports on Form 10-K, quarterly reports on Form 10-Q, current reports on Form 8-K, and all amendments to those reports as soon as reasonably practicable after such material is electronically filed with or furnished to the SEC. Copies of the charters of each of the Company's Audit Committee, Compensation Committee and Nominating and Corporate Governance Committee, as well as the Company's Corporate Governance Guidelines and Code of Ethics along with position descriptions for the Board of Directors and Board committees are also available free of charge through the website. The information provided on the Company's website is not part of this report, and is therefore not incorporated by reference unless such information is otherwise specifically incorporated elsewhere in this report.

Segments

Journeys Group

The Journeys Group segment, including Journeys, Journeys Kidz, Shi by Journeys and Underground by Journeys retail stores, catalog and e-commerce operations, accounted for approximately 41% of the Company's net sales in Fiscal 2014. For Fiscal 2014, same store sales decreased 2%, comparable direct sales increased 18% and comparable sales, including both store and direct sales, decreased 1% from the prior fiscal year. Operating income attributable to Journeys Group was \$97.4 million in Fiscal 2014, with an operating margin of 9.0%. The Company believes that the Journeys Group's distinctive store formats, its mix of well-known brands and new product introductions, and its experienced management team provide significant competitive advantages for the Journeys Group.

At February 1, 2014, Journeys Group operated 1,168 stores, including 174 Journeys Kidz stores, 50 Shi by Journeys stores and 117 Underground by Journeys stores averaging approximately 1,875 square feet, throughout the United States and in Puerto Rico and Canada, selling footwear and accessories for young men, women and children.

Journeys stores target customers in the 13 to 22 year age group through the use of youth-oriented decor and multi-channel media. Journeys stores carry predominately branded merchandise across a wide range of prices. The Journeys Kidz retail footwear stores sell footwear and accessories primarily for younger children ages five to 12. Shi by Journeys retail footwear stores sell footwear and accessories to a target customer group consisting of fashion-conscious women in their early 20's to mid 30's. Underground by Journeys retail footwear stores sell footwear and accessories primarily for men and women in the 20 to 35 age group. In Fiscal 2014, the Journeys Group added eleven net new stores and plans to open approximately 17 net new stores in Fiscal 2015.

Lids Sports Group

The Lids Sports Group segment, as described above, accounted for approximately 31% of the Company's net sales in Fiscal 2014. Same store sales for Lids Sports Group decreased 1% for Fiscal 2014, while comparable direct sales increased 26% from the prior fiscal year. Comparable sales, including both store and direct sales, were flat for Fiscal 2014. Operating income attributable to Lids Sports Group was \$63.7 million in Fiscal 2014, with an operating margin of 7.8%.

At February 1, 2014, Lids Sports Group operated 1,133 stores, including 930 Lids stores, 177 Lids Locker Room and Clubhouse stores and 26 Locker Room by Lids leased departments, averaging approximately 1,200 square feet, throughout the United States and in Puerto Rico and Canada. Lids Sports Group added 80 net new stores and leased departments in Fiscal 2014, including 15 acquired stores, and plans to open or acquire approximately 260 net new stores in Fiscal 2015, which includes 175 Locker Room by Lids leased departments in Macy's department stores.

The core headwear stores and kiosks, located in malls, airports, street-level stores and factory outlet stores throughout the United States and in Puerto Rico and Canada, target customers in the early-teens to mid-20's age group. In general, the stores offer headwear from an assortment of college, MLB, NBA, NFL and NHL teams, as well as other specialty fashion categories. The Lids Locker Room and Lids Clubhouse stores, operating under a number of trade names, located in malls and other locations primarily in the United States, target sports fans of all ages. These stores offer headwear, apparel, accessories and novelties representing an assortment of college and professional teams. The Locker Room by Lids leased departments in Macy's department stores offer headwear, apparel, accessories and novelties representing an assortment of college and professional teams specific to that particular Macy's department store geographic location.

Δ

Schuh Group

The Schuh Group segment, including e-commerce operations, accounted for approximately 14% of the Company's net sales in Fiscal 2014. For Fiscal 2014, same store sales decreased 9%, comparable direct sales decreased 4% and comparable sales, including both store and direct sales, decreased 8%. Operating income attributable to Schuh Group was \$3.1 million in Fiscal 2014, with an operating margin of 0.8%. Operating income for Schuh included \$11.7 million in compensation expense related to a deferred purchase price obligation in connection with the Company's acquisition of Schuh during Fiscal 2012. For additional information, see Note 2 to the Consolidated Financial Statements included in Item 8.

At February 1, 2014, Schuh Group operated 95 Schuh stores, averaging approximately 5,000 square feet, which include both street-level and mall locations in the United Kingdom and the Republic of Ireland. Schuh Group opened its first Schuh Kids store in Fiscal 2013. As of February 1, 2014, Schuh Group operated four Schuh Kids stores averaging 2,425 square feet. Schuh Group opened seven net new stores in Fiscal 2014 and plans to open approximately 12 net new Schuh and Schuh Kids stores in Fiscal 2015. Schuh stores target men and women in the 15 to 30 age group, selling a broad range of branded casual and athletic footwear along with a meaningful private label offering.

Johnston & Murphy Group

The Johnston & Murphy Group segment, including retail stores, catalog and e-commerce operations and wholesale distribution, accounted for approximately 9% of the Company's net sales in Fiscal 2014. Same store sales for Johnston & Murphy retail operations increased 7%, comparable direct sales increased 15% and comparable sales, including both store and direct sales, increased 8% for Fiscal 2014. Operating income attributable to Johnston & Murphy Group was \$17.6 million in Fiscal 2014, with an operating margin of 7.2%. All of the Johnston & Murphy wholesale sales are of the Genesco-owned Johnston & Murphy brand and all of the group's retail sales are of Johnston & Murphy branded products.

Johnston & Murphy Retail Operations. At February 1, 2014, Johnston & Murphy operated 168 retail shops and factory stores throughout the United States and in Canada averaging approximately 1,825 square feet and selling footwear, apparel and accessories primarily for men in the 35 to 55 age group, targeting business and professional customers. Women's footwear and accessories are sold in select Johnston & Murphy locations. Johnston & Murphy retail shops are located primarily in better malls and airports nationwide and sell a broad range of men's dress and casual footwear, apparel and accessories. The Company also sells Johnston & Murphy products directly to consumers through an e-commerce website and a direct mail catalog. Retail prices for Johnston & Murphy footwear generally range from \$100 to \$275. Total footwear accounted for 66% of total Johnston & Murphy retail sales in Fiscal 2014, with the balance consisting primarily of apparel and accessories. Johnston & Murphy Group added eleven net new shops and factory stores, including two in Canada, and plans to open approximately eight net new shops and factory stores in Fiscal 2015.

Johnston & Murphy Wholesale Operations. Johnston & Murphy men's and women's footwear and accessories are sold at wholesale, primarily to better department and independent specialty stores. Johnston & Murphy's wholesale customers offer the brand's footwear for dress, dress casual, and casual occasions, with the majority of styles offered in these channels selling from \$100 to \$165. Additionally, the Company recently relaunched the Trask brand, with men's and women's footwear and leather accessories offered primarily through better independent retailers and department stores, an e-commerce website and catalog. Suggested retail prices for Trask footwear range from \$195 to \$495.

Licensed Brands

The Licensed Brands segment accounted for approximately 4% of the Company's net sales in Fiscal 2014. Operating income attributable to Licensed Brands was \$10.6 million in Fiscal 2014, with an operating margin of 9.7%. Licensed Brands sales are footwear marketed under the Dockers[®] brand, for which Genesco has had the exclusive men's footwear license in the United States since 1991. See "Licenses". Dockers footwear is marketed to men aged 30 to 55 through many of the same national retail chains that carry Dockers slacks and sportswear and in department and specialty stores across the country. Suggested retail prices for Dockers footwear generally range from \$50 to \$90. The Company acquired Keuka Footwear in the third quarter of Fiscal 2011 and subsequently launched its SureGrip Footwear line of slip-resistant, occupational footwear from that base. The Company sources and distributes the SureGrip line to employees in the hospitality, healthcare, and other industries.

For further information on the Company's business segments, see Note 14 to the Consolidated Financial Statements included in Item 8 and "Management's Discussion and Analysis of Financial Condition and Results of Operations."

Manufacturing and Sourcing

The Company relies on independent third-party manufacturers for production of its footwear products sold at wholesale. The Company sources footwear and accessory products from foreign manufacturers located in Brazil, Canada, China, Dominican Republic, France, India, Indonesia, Italy, Korea, Mexico, Netherlands, Pakistan, Peru, Thailand and Vietnam. The Company's retail operations source primarily branded products from third parties, who source primarily overseas.

Competition

Competition is intense in the footwear, headwear, sports apparel and accessory industries. The Company's retail footwear, headwear, sports apparel and accessory competitors range from small, locally owned stores to regional and national department stores, discount stores, specialty chains and online retailers. The Company also competes with hundreds of footwear wholesale operations in the United States and throughout the world, most of which are relatively small, specialized operations, but some of which are large, more diversified companies. Some of the Company's competitors have resources that are not available to the Company. The Company's success depends upon its ability to remain competitive with respect to the key factors of style, price, quality, comfort, brand loyalty, customer service, store location and atmosphere and the ability to offer distinctive products.

Licenses

The Company owns its Johnston & Murphy[®], H.S. Trask[®], Keuka[®] and SureGrip[®] brands and owns or licenses the trade names of its retail concepts either directly or through wholly-owned subsidiaries. The Dockers[®] brand footwear line, introduced in Fiscal 1993, is sold under a license agreement granting the Company the exclusive right to sell men's footwear under the trademark in the United States, Canada and Mexico and in certain other Latin American countries. The Dockers license agreement, as amended, expires on November 30, 2015, subject to extension for an additional 3-year term if certain conditions are met. Net sales of Dockers products were approximately \$85 million in Fiscal 2014 and approximately \$84 million in Fiscal 2013. The Company licenses certain of its footwear brands, mostly in foreign markets. License royalty income was not material in Fiscal 2014.

Wholesale Backlog

Most of the orders in the Company's wholesale divisions are for delivery within 150 days. Because most of the Company's business is at-once, the backlog at any one time is not necessarily indicative of future sales. As of March 1, 2014, the Company's wholesale operations had a backlog of orders, including unconfirmed customer purchase orders, amounting to approximately \$57.4 million, compared to approximately \$46.1 million on March 2, 2013. The backlog is somewhat seasonal, reaching a peak in spring. The Company maintains in-stock programs for selected product lines with anticipated high volume sales.

Employees

Genesco had approximately 22,250 employees at February 1, 2014, approximately 120 of whom were employed in corporate staff departments and the balance in operations. Retail stores employ a substantial number of part-time employees, and approximately 13,050 of the Company's employees were part-time.

Properties

At February 1, 2014, the Company operated 2,568 retail footwear, headwear and sports apparel and accessory stores and leased departments throughout the United States and in Puerto Rico, Canada, the United Kingdom and the Republic of Ireland. New shopping center store leases in the United States, Puerto Rico and Canada typically are for a term of approximately 10 years. Store leases in the United Kingdom and the Republic of Ireland typically have terms of between 10 and 20 years. Both typically provide for rent based on a percentage of sales against a fixed minimum rent based on the square footage leased.

Seasonality

The Company's business is seasonal with the Company's investment in inventory and accounts receivable normally reaching peaks in the spring and fall of each year.

The general location, use and approximate size of the Company's principal properties are set forth below:

Location	Owned/Leased	Segment	Use	Approximate Area Square Feet	
Lebanon, TN	Owned	Journeys Group	Distribution warehouse	320,000	
Indianapolis, IN	Leased	Lids Sports Group	Distribution warehouse and administrative offices	311,600	
Nashville, TN	Leased	Various	Executive & footwear operations offices	306,455	*
Indianapolis, IN	Leased	Lids Sports Group	Distribution warehouse and manufacturing	271,825	
Bathgate, Scotland	Owned	Schuh Group	Distribution warehouse	244,644	
Indianapolis, IN	Leased	Lids Sports Group	Distribution warehouse and administrative offices	195,080	
Chapel Hill, TN	Owned	Licensed Brands	Distribution warehouse	182,000	
Fayetteville, TN	Owned	Johnston & Murphy Group	Distribution warehouse	178,500	
Deans Industrial Estate, Livingston, Scotland	Owned	Schuh Group	Distribution warehouse and administrative offices	106,813	
Lake Katrine, NY	Leased	Lids Sports Group	Distribution warehouse and administrative offices	73,000	
Nashville, TN	Owned	Journeys Group	Distribution warehouse	63,000	
Houston Industrial Estate, Livingston, Scotland	Leased	Schuh Group	Distribution warehouse	51,012	
Mississauga, Ontario, Canada	Leased	Lids Sports Group	Distribution warehouse	28,392	
Anderson, IN	Leased	Lids Sports Group	Distribution Warehouse	18,463	
Indianapolis, IN	Leased	Lids Sports Group	Administrative offices	17,217	
Tigard, OR	Leased	Lids Sports Group	Administrative offices	7,231	

*

* The Company occupies approximately 85% of the building and subleases the remainder of the building.

The lease on the Company's Nashville office expires in April 2017, with an option to renew for an additional five years. The lease on the Indianapolis office expires in May 2015. The Company believes that all leases of properties that are material to its operations may be renewed, or that alternative properties are available, on terms not materially less favorable to the Company than existing leases.

Environmental Matters

The Company's former manufacturing operations and the sites of those operations as well as the sites of its current operations are subject to numerous federal, state, and local laws and regulations relating to human health and safety and the environment. These laws and regulations address and regulate, among other matters, wastewater discharge, air quality and the generation, handling, storage, treatment, disposal, and transportation of solid and hazardous wastes and regulations to require remediation of environmental conditions and, in the case of governmental agencies, to impose fines and penalties. Several of the facilities owned by the Company (currently or in the past) are located in industrial areas and have historically been used for extensive periods for industrial operations such as tanning, dyeing, and manufacturing. Some of these operations used materials and generated wastes that would be considered regulated substances under current

environmental laws and regulations. The Company currently is involved in certain administrative and judicial environmental proceedings relating to the Company's former facilities. See Item 3, Legal Proceedings and Note 13.

ITEM 1A, RISK FACTORS

Our business is subject to significant risks. You should carefully consider the risks and uncertainties described below and the other information in this Form 10-K, including our consolidated financial statements and the notes to those statements. The risks and uncertainties described below are not the only ones we face. Additional risks and uncertainties that we do not presently know about or that we currently consider immaterial may also affect our business operations and financial performance. If any of the events described below actually occur, our business, financial condition or results of operations could be adversely affected in a material way. This could cause the trading price of our stock to decline, perhaps significantly, and you may lose part or all of your investment.

Poor economic conditions and other factors can affect consumer spending and may significantly harm our business, affecting our financial condition, liquidity, and results of operations.

The success of our business depends to a significant extent upon the level of consumer spending. A number of factors may affect the level of consumer spending on merchandise that we offer, including, among other things:

- general economic, industry and weather conditions;
- energy costs, which affect gasoline and home heating prices;
- the level of consumer debt;
- pricing of products;
- interest rates;
- tax rates, refunds and policies;
- war, terrorism and other hostilities; and
- consumer confidence in future economic conditions.

Adverse economic conditions and any related decrease in consumer demand for discretionary items could have a material adverse effect on our business, results of operations and financial condition. The merchandise we sell generally consists of discretionary items. Reduced consumer confidence and spending may result in reduced demand for discretionary items and may force us to take inventory markdowns, decreasing sales and making expense leverage difficult to achieve. Demand can also be influenced by other factors beyond our control. For example, sales in the Lids Sports Group segment have historically been affected by developments in team sports, and could be adversely impacted by player strikes or other interruptions, as well as by the performance and reputation of certain teams.

Moreover, while the Company believes that its operating cash flows and its borrowing capacity under committed lines of credit will be more than adequate for its anticipated cash requirements, if the economy were to experience a renewed downturn, or if one or more of the Company's revolving credit banks were to fail to honor its commitments under the Company's credit lines, the Company could be required to modify its operations for decreased cash flow or to seek alternative sources of liquidity, and such alternative sources might not be available to the Company.

Our business involves a degree of fashion risk.

The majority of our businesses serve a fashion-conscious customer base and depend upon the ability of our buyers and merchandisers to react to fashion trends, to purchase inventory that reflects such trends, and to manage our inventories appropriately in view of the potential for sudden changes in fashion or in consumer taste. Failure to continue to execute any of these activities successfully could result in adverse consequences, including lower sales, product margins, operating income and cash flows.

Our business and results of operations are subject to a broad range of uncertainties arising out of world and domestic events.

Our business and results of operations are subject to uncertainties arising out of world and domestic events, which may impact not only consumer demand, but also our ability to obtain the products we sell, most of which are produced outside the countries in which we operate. These uncertainties may include a global economic slowdown, changes in consumer

spending or travel, increase in gasoline and natural gas prices, and the economic consequences of natural disasters, military action or terrorist activities and increased regulatory and compliance burdens related to governmental actions in response to a variety of factors, including but not limited to national security and anti-terrorism concerns and concerns about climate change. Any future events arising as a result of terrorist activity or other world events may have a material impact on our business, including the demand for and our ability to source products, and consequently on our results of operations and financial condition.

The increasing scope of our non-U.S. operations exposes our performance to risks including foreign economic conditions and exchange rate fluctuations.

Our performance depends in part on general economic conditions affecting all countries in which we do business. We are dependent on foreign manufacturers for the products we sell, and our inventory is subject to cost and availability of foreign materials and labor. In addition to the other risks disclosed herein, demand for our product offering in our non-U.S. operations is also subject to local market conditions. As a result, there can be no assurance that Schuh's future performance will not be adversely affected by economic conditions in its markets.

As we expand our international operations, we also increase our exposure to exchange rate fluctuations. Sales from stores outside the U.S. are denominated in the currency of the country in which these operations or stores are located and changes in foreign exchange rates affect the translation of the sales and earnings of these businesses into U.S. dollars for financial reporting purposes. Additionally, inventory purchase agreements may also be denominated in the currency of the country where the vendor resides.

Our business is intensely competitive and increased or new competition could have a material adverse effect on us.

The retail footwear, headwear, sports apparel and accessory markets are intensely competitive. We currently compete against a diverse group of retailers, including other regional and national specialty stores, department and discount stores, small independents and e-commerce retailers, which sell products similar to and often identical to those we sell. Our branded businesses, selling footwear at wholesale, also face intense competition, both from other branded wholesale vendors and from private label initiatives of their retailer customers. A number of different competitive factors could have a material adverse effect on our business, results of operations and financial condition, including:

- increased operational efficiencies of competitors;
- competitive pricing strategies;
- expansion by existing competitors;
- entry by new competitors into markets in which we currently operate; and
- adoption by existing retail competitors of innovative store formats or sales methods.

We are dependent on third-party vendors for the merchandise we sell.

We do not manufacture the merchandise we sell. This means that our product supply is subject to the ability and willingness of third-party suppliers to deliver merchandise we order on time and in the quantities and of the quality we need. In addition, a material portion of our retail footwear sales consists of products marketed under brands, belonging to unaffiliated vendors, which have fashion significance to our customers. Our core retail hat and sports apparel businesses are dependent upon products bearing sports and other logos, each generally controlled by a single licensee/vendor. If those vendors were to decide not to sell to us or to limit the availability of their products to us, or if they become unable because of economic conditions or any other reason to supply us with products, we could be unable to offer our customers the products they wish to buy and could lose their business to competitors.

An increase in the cost or a disruption in the flow of our imported products may significantly decrease our sales and profits.

Merchandise originally manufactured and imported from overseas makes up a large proportion of our total inventory. A disruption in the shipping of our imported merchandise or an increase in the cost of those products may significantly decrease our sales and profits. We may be unable to meet our customers' demands or pass on price increases to our customers. In addition, if imported merchandise becomes more expensive or unavailable, the transition to alternative sources may not occur in time to meet demand. Products from alternative sources may also be of lesser quality or more expensive than those we currently import. Risks associated with our reliance on imported products include:

- disruptions in the shipping and importation of imported products because of factors such as:
 - raw material shortages, work stoppages, strikes and political unrest;
 - problems with oceanic shipping, including shipping container shortages;
 - increased customs inspections of import shipments or other factors causing delays in shipments;
 - economic crises, natural disasters, international disputes and wars; and
 - increases in the cost of purchasing or shipping foreign merchandise resulting from:
 - denial by the United States of "most favored nation" trading status to or the imposition of quotas or other restriction on imports from a foreign country from which we purchase goods;
 - import duties, import quotas and other trade sanctions; and
 - increases in shipping rates.

A significant amount of the inventory we sell is imported from the People's Republic of China, which has historically been subject to efforts to increase duty rates or to impose restrictions on imports of certain products.

A small portion of the products we buy abroad are priced in foreign currencies and, therefore, we are affected by fluctuating currency exchange rates. In the past, we have entered into foreign currency exchange contracts with major financial institutions to hedge these fluctuations. We might not be able to effectively protect ourselves in the future against currency rate fluctuations, and our financial performance could suffer as a result. Even dollar-denominated foreign purchases may be affected by currency fluctuations, as suppliers seek to reflect appreciation in the local currency against the dollar in the price of the products that they provide. You should read "Management's Discussion and Analysis of Financial Condition and Results of Operations" for more information about our foreign currency exchange rate exposure and hedging activities.

Increased operating costs could have an adverse effect on our results.

Increased operating costs, including those resulting from potential increases in the minimum wage, store occupancy costs, and other expense items, may reduce our operating margin and, by making it more difficult to identify new store locations that we believe will meet our investment return requirements, slow our growth.

The operation of the Company's business is heavily dependent on its information systems.

We depend on a variety of information technology systems for the efficient functioning of our business and security of information. Much information essential to our business is maintained electronically, including competitively sensitive information and potentially sensitive personal information about customers and employees. Our insurance policies may not provide coverage for security breaches and similar incidents or may have coverage limits which may not be adequate to reimburse us for losses caused by security breaches. We also rely on certain hardware and software vendors to maintain and periodically upgrade many of these systems so that they can continue to support our business. The software programs supporting many of our systems were licensed to the Company by independent software developers. The inability of these developers or the Company to continue to maintain and upgrade these information systems and software programs could disrupt or reduce the efficiency of our operations. In addition, costs and potential problems and interruptions associated with the implementation of new or upgraded systems and technology or with maintenance or adequate support of existing systems could also disrupt or reduce the efficiency of our operations.

We also rely heavily on our information technology staff. If we cannot meet our staffing needs in this area, we may not be able to fulfill our technology initiatives or to provide maintenance on existing systems.

A privacy breach could have a material adverse effect on the Company's business and reputation.

We rely heavily on digital technologies for the successful operation of our business, including electronic messaging, digital marketing efforts and the collection and retention of customer data and employee information. We also rely on third parties to process credit card transactions, perform online e-commerce and social media activities and retain data relating to the Company's financial position and results of operations, strategic initiatives and other important information. Despite the security measures we have in place, our facilities and systems and those of our third-party service providers, may be vulnerable to cyber-security breaches, acts of vandalism, computer viruses, misplaced or lost data, programming and/or human errors or other similar events. Any misappropriation, loss or other unauthorized disclosure of confidential or personally identifiable information, whether by us or by our third-party service providers, could adversely affect our business

and operations, including loss of sales generated through our websites, severely damaging our reputation and our relationships with our customers, suppliers, employees and investors and expose us to risks of litigation and liability.

In addition, we may incur significant remediation costs in the event of a cyber-security breach or incident, including liability for stolen customer or employee information, repairing system damage or providing credit monitoring or other benefits to affected customers or employees. We may also incur increased costs to comply with various applicable laws or industry standards regarding use and/or unauthorized disclosure of personal information. These and other cyber-security-related compliance, prevention and remediation costs may adversely impact our financial condition and results of operations.

The loss of, or disruption in, one of our distribution centers and other factors affecting the distribution of merchandise, could have a material adverse effect on our business and operations.

Each of our operations uses a single distribution center to handle all or a significant amount of its merchandise. Most of our operations' inventory is shipped directly from suppliers to their distribution centers, where the inventory is then processed, sorted and shipped to our stores or to our wholesale customers. We depend on the orderly operation of this receiving and distribution process, which depends, in turn, on adherence to shipping schedules and effective management of the distribution centers. Although we believe that our receiving and distribution process is efficient and well positioned to support our current business and our expansion plans, we cannot offer assurance that we have anticipated all of the changing demands which our expanding operations will impose on our receiving and distribution system, or that events beyond our control, such as disruptions in operations due to fire or other catastrophic events, labor disagreements or shipping problems (whether in our own or in our third party vendors' or carriers' businesses), will not result in delays in the delivery of merchandise to our stores or to our wholesale customers. We also make changes in our distribution processes from time to time in an effort to improve efficiency, maximize capacity, etc. We cannot assure that these changes will not result in unanticipated delays or interruptions in distribution. We depend upon UPS for shipment of a significant amount of merchandise. An interruption in service by UPS for any reason could cause temporary disruptions in our business, a loss of sales and profits, and other material adverse effects.

Our freight cost is impacted by changes in fuel prices through surcharges. Fuel prices and surcharges affect freight cost both on inbound freight from vendors to our distribution centers and outbound freight from our distribution centers to our stores and wholesale customers. Increases in fuel prices and surcharges and other factors may increase freight costs and thereby increase our cost of goods sold.

Any acquisitions we make or new businesses we launch involve a degree of risk.

Acquisitions have been a component of the Company's growth strategy in recent years and we expect that we may continue to engage in acquisitions or launch new businesses to grow our revenues and meet our other strategic objectives. If any future acquisitions are not successfully integrated with our business, our ongoing operations could be adversely affected. Additionally, acquisitions or new businesses may not achieve desired profitability objectives or result in any anticipated successful expansion of the businesses or concepts. Although we review and analyze assets or companies we acquire, such reviews are subject to uncertainties and may not reveal all potential risks. Additionally, although we attempt to obtain protective contractual provisions, such as representations, warranties and indemnities, in connection with acquisitions, we cannot offer assurance that we can obtain such provisions in our acquisitions or that they will fully protect us from unforeseen costs of the acquisitions. We may also incur significant costs and diversion of management time and attention in connection with pursuing possible acquisitions even if the acquisition is not ultimately consummated.

We face a number of risks in opening new stores.

As part of our long-term growth strategy, we expect to open new stores, both in regional malls, where most of the operational experience of our U.S. businesses lies, and in other venues including outlet centers, major city street locations, airports and tourist destinations. We cannot offer assurances that we will be able to open as many stores as we have planned, that any new store will achieve similar operating results to those of our existing stores or that new stores opened in markets in which we operate will not have a material adverse effect on the revenues and profitability of our existing stores. The success of our planned expansion will be dependent upon numerous factors, many of which are beyond our control, including the following:

- our ability to identify suitable markets and individual store sites within those markets;
- the competition for suitable store sites;
- our ability to negotiate favorable lease terms for new stores and renewals (including rent and other costs) with landlords;

- our ability to obtain governmental and other third-party consents, permits and licenses needed to construct and operate our stores;
- the ability to build and remodel stores on schedule and at acceptable cost;
- the availability of employees to staff new stores and our ability to hire, train, motivate and retain store personnel;
- the availability of adequate management and financial resources to manage an increased number of stores;
- our ability to adapt our distribution and other operational and management systems to an expanded network of stores; and
- our ability to attract customers and generate sales sufficient to operate new stores profitably.

Additionally, the results we expect to achieve during each fiscal quarter are dependent upon opening new stores on schedule. If we fall behind, we will lose expected sales and earnings between the planned opening date and the actual opening and may further complicate the logistics of opening stores, possibly resulting in additional delays.

Our results of operations are subject to seasonal and quarterly fluctuations, which could have a material adverse effect on the market price of our stock.

Our business is seasonal, with a significant portion of our net sales and operating income generated during the fourth quarter, which includes the holiday shopping season. Because of this seasonality, we have limited ability to compensate for shortfalls in fourth quarter sales or earnings by changes in our operations or strategies in other quarters. A significant shortfall in results for the fourth quarter of any year could have a material adverse effect on our annual results of operations and on the market price of our stock. Our quarterly results of operations also may fluctuate significantly based on such factors as:

- the timing of new store openings and renewals;
- the amount of net sales contributed by new and existing stores;
- the timing of certain holidays and sales events;
- changes in our merchandise mix;
- general economic, industry and weather conditions that affect consumer spending; and
- actions of competitors, including promotional activity.

A failure to increase sales at our existing stores and in our e-commerce businesses may adversely affect our stock price and impact our results of operations.

A number of factors have historically affected, and will continue to affect, our comparable sales results, including:

- consumer trends, such as less disposable income due to the impact of economic conditions and tax policies;
- the lack of new fashion trends to drive demand in certain of our businesses;
- competition;
- timing of holidays including sales tax holidays and the timing of tax refunds;
- general regional and national economic conditions;
- inclement weather;
- changes in our merchandise mix;
- our ability to distribute merchandise efficiently to our stores;
- · timing and type of sales events, promotional activities or other advertising;
- other external events beyond our control;
- new merchandise introductions; and

• our ability to execute our business strategy effectively.

Our comparable sales have fluctuated in the past, and we believe such fluctuations may continue. The unpredictability of our comparable sales may cause our revenue and results of operations to vary from quarter to quarter, and an unanticipated change in revenues or operating income may cause our stock price to fluctuate significantly.

We are subject to regulatory proceedings and litigation that could have an adverse effect on our financial condition and results of operations.

We are party to certain lawsuits, governmental investigations, and regulatory proceedings, including the suits and proceedings arising out of alleged environmental contamination relating to historical operations of the Company and various suits involving current operations as disclosed in Note 13 to the Consolidated Financial Statements. If these or similar matters are resolved against us, our results of operations, our cash flows, or our financial condition could be adversely affected. The costs of defending such lawsuits and responding to such investigations and regulatory proceedings may be substantial and their potential to distract management from day-to-day business is significant. Moreover, with retail operations in 50 states, Puerto Rico, Canada, the United Kingdom and the Republic of Ireland, we are subject to federal, state, provincial, territorial, local and foreign regulations, which impose costs and risks on our business. Changes in regulations could make compliance more difficult and costly, and violations could result in liability for damages or penalties.

If we lose key members of management or are unable to attract and retain the talent required for our business, our operating results could suffer.

Our performance depends largely on the efforts and abilities of members of our management team. Our executives have substantial experience and expertise in our business and have made significant contributions to our growth and success. The unexpected future loss of services of one or more key members of our management team could have an adverse effect on our business. In addition, future performance will depend upon our ability to attract, retain and motivate qualified employees, including store personnel and field management. If we are unable to do so, our ability to meet our operating goals may be compromised. Finally, our stores are decentralized, are managed through a network of geographically dispersed management personnel and historically experience a high degree of turnover. If we are for any reason unable to maintain appropriate controls on store operations due to turnover or other reasons, including the ability to control losses resulting from inventory and cash shrinkage, our sales and operating margins may be adversely affected. There can be no assurance that we will be able to attract and retain the personnel we need in the future.

Goodwill recorded with acquisitions is subject to impairment which could reduce the Company's profitability.

Deterioration in the Company's market value, whether related to the Company's operating performance or to disruptions in the equity markets or deterioration in the operating performance of the business unit with which goodwill is associated, could require the Company to recognize the impairment of some or all of the \$288.1 million of goodwill on its Consolidated Balance Sheets at February 1, 2014, resulting in the reduction of net assets and a corresponding non-cash charge to earnings in the amount of the impairment.

In connection with acquisitions, the Company records goodwill on its Consolidated Financial Statements. This asset is not amortized but is subject to an impairment test at least annually, which consists of either a qualitative assessment on a reporting unit level, or a two-step impairment test if necessary, that is based on projected future cash flows from the acquired business discounted at a rate commensurate with the risk the Company considers to be inherent in its current business model. The Company performs the impairment test annually as of the close of its fiscal year, or more frequently if events or circumstances indicate that the value of the asset might be impaired.

As a result of the various acquisitions comprising the Lids Team Sports team dealer business, the Company carries goodwill at a value of \$14.2 million on its Consolidated Balance Sheets related to such acquisitions. The Company found that the result of its annual impairment test, which valued the business at approximately \$3.9 million in excess of its carrying value, indicated no impairment at that time. The Company may determine in future impairment tests that some or all of the carrying value of the goodwill may not be recoverable. Such a finding would require a write-off of the amount of the carrying value that is impaired, which would reduce the Company's profitability in the period of the impairment charge. Holding all other assumptions constant as of the measurement date, the Company noted that an increase in the weighted average cost of capital of 100 basis points would reduce the fair value of the Lids Team Sports business by \$5.9 million. Furthermore, the Company noted that a decrease in projected annual revenue by one percent would reduce the fair value of the Lids Team Sports business by \$0.4 million. However, if other assumptions do not remain constant, the fair value of the Lids Team Sports business may decrease by a greater amount. Since the maximum non-cash goodwill impairment charge would be \$14.2 million, the Company does not believe that any impairment charge related thereto would be material; however, there

can be no assurance that any future goodwill impairment will not have a material adverse effect on the Company's financial position.

ITEM 1B, UNRESOLVED STAFF COMMENTS

None.

ITEM 2, PROPERTIES

See Item 1, Business — Properties.

ITEM 3, LEGAL PROCEEDINGS

Environmental Matters

New York State Environmental Matters

In August 1997, the New York State Department of Environmental Conservation ("NYSDEC") and the Company entered into a consent order whereby the Company assumed responsibility for conducting a remedial investigation and feasibility study ("RIFS") and implementing an interim remedial measure ("IRM") with regard to the site of a knitting mill operated by a former subsidiary of the Company from 1965 to 1969. The Company undertook the IRM and RIFS voluntarily, without admitting liability or accepting responsibility for any future remediation of the site. The Company has completed the IRM and the RIFS. In the course of preparing the RIFS, the Company identified remedial alternatives with estimated undiscounted costs ranging from \$0.0 million to \$24.0 million, excluding amounts previously expended or provided for by the Company. The United States Environmental Protection Agency ("EPA"), which has assumed primary regulatory responsibility for the site from NYSDEC, issued a Record of Decision in September 2007. The Record of Decision requires a remedy of a combination of groundwater extraction and treatment and in-site chemical oxidation at an estimated present cost of approximately \$10.7 million.

In July 2009, the Company agreed to a Consent Order with the EPA requiring the Company to perform certain remediation actions, operations, maintenance and monitoring at the site. In September 2009, a Consent Judgment embodying the Consent Order was filed in the U.S. District Court for the Eastern District of New York.

The Village of Garden City, New York (the "Village"), has additionally asserted that the Company is liable for the costs associated with enhanced treatment required by the impact of the groundwater plume from the site on two public water supply wells, including historical total costs ranging from approximately \$1.8 million to in excess of \$2.5 million, and future operation and maintenance costs which the Village estimates at \$126,400 annually while the enhanced treatment continues. On December 14, 2007, the Village filed a complaint against the Company and the owner of the property under the Resource Conservation and Recovery Act ("RCRA"), the Safe Drinking Water Act, and the Comprehensive Environmental Response, Compensation and Liability Act ("CERCLA") as well as a number of state law theories in the U.S. District Court for the Eastern District of New York, seeking an injunction requiring the defendants to remediate contamination from the site and to establish their liability for future costs that may be incurred in connection with it, which the complaint alleges could exceed \$41 million, undiscounted, over a 70-year period. The Company has not verified the estimates of either historic or future costs based on a 70-year remediation period is unreasonable given the expected remedial period reflected in the EPA's Record of Decision. On May 23, 2008, the Company filed a motion to dismiss the Village's complaint on grounds including applicable statutes of limitation and preemption of certain claims by the NYSDEC's and the EPA's diligent prosecution of remediation. On January 27, 2009, the Court granted the motion to dismiss all counts of the plaintiff's complaint except for the CERCLA claim and a state law claim for indemnity for costs incurred after November 27, 2000. On September 23, 2009, on a motion for reconsideration by the Village, the Court reinstated the claims for injunctive relief under RCRA and for equitable relief under certain of the state law theories. The Company intends to continue to defend the

Whitehall Environmental Matters

The Company has performed sampling and analysis of soil, sediments, surface water, groundwater and waste management areas at the Company's former Volunteer Leather Company facility in Whitehall, Michigan.

In October 2010, the Company and the Michigan Department of Natural Resources and Environment entered into a Consent Decree providing for implementation of a remedial Work Plan for the facility site designed to bring the site into compliance with applicable regulatory standards. The Work Plan's implementation is substantially complete and the Company expects, based on its present understanding of the condition of the site, that its future obligations with respect to the site will be



limited to periodic monitoring and that future costs related to the site should not have a material effect on its financial condition or results of operations.

Accrual for Environmental Contingencies

Related to all outstanding environmental contingencies, the Company had accrued \$11.9 million as of February 1, 2014, \$11.9 million as of February 2, 2013 and \$13.0 million as of January 28, 2012. All such provisions reflect the Company's estimates of the most likely cost (undiscounted, including both current and noncurrent portions) of resolving the contingencies, based on facts and circumstances as of the time they were made. There is no assurance that relevant facts and circumstances will not change, necessitating future changes to the provisions. Such contingent liabilities are included in the liability arising from provision for discontinued operations on the accompanying Consolidated Balance Sheets because it relates to former facilities operated by the Company. The Company has made pretax accruals for certain of these contingencies, including approximately \$0.5 million reflected in Fiscal 2014, \$0.8 million reflected in Fiscal 2012. These charges are included in provision for discontinued operations, net in the Consolidated Statements of Operations and represent changes in estimates.

Other Matters

On December 10, 2010, the Company announced that it had suffered a criminal intrusion into the portion of its computer network that processes payments for transactions in certain of its retail stores. Visa, Inc., MasterCard Worldwide and American Express Travel Related Services Company, Inc. have asserted claims totaling approximately \$15.6 million in connection with the intrusion and the claims of two of the claimants have been collected by withholding payment card receivables of the Company. In the fourth quarter of Fiscal 2013, the Company recorded a \$15.4 million charge to earnings in connection with the disputed liability. On March 7, 2013, the Company filed an action in the U.S. District Court for the Middle District of Tennessee against Visa U.S.A. Inc., Visa Inc. and Visa International Service Association seeking to recover \$13.3 million in non-compliance fines and issuer reimbursement assessments collected from the Company in connection with the intrusion. The Company does not currently expect any future claims in connection with the intrusion to have a material effect on its financial condition, cash flows, or results of operations.

On January 5, 2012, a patent infringement action against the Company and numerous other defendants was filed in the U.S. District Court for the Eastern District of Texas, *GeoTag, Inc. v. Circle K Store, Inc., et al.*, alleging that features of certain of the Company's e-commerce websites infringe U.S. Patent No. 5,930,474, entitled "Internet Organizer for Accessing Geographically and Topically Based Information." The plaintiff sought relief including damages for the alleged infringement, costs, expenses and pre- and post-judgment interest and injunctive relief. Pursuant to a settlement agreement, the matter was dismissed on February 28, 2014. The settlement did not have a material effect on its financial condition or results of operations.

On June 13, 2012, a former vendor of a subsidiary of the Company filed an action, *Perfect Curve, Inc. v. Hat World, Inc.*, in U.S. District Court in Massachusetts, alleging patent, trademark, trade dress, and copyright infringement against the subsidiary based on the sale of a line of products developed by the subsidiary. The parties reached agreement to settle the matter and the action was dismissed pursuant to a Stipulated Dismissal dated March 13, 2014. The settlement did not have a material effect on its financial condition or results of operations.

On May 14, 2012, a putative class and collective action, *Maro v. Hat World, Inc.*, was filed in the U.S. District Court for the Northern District of Illinois. The action alleges that the Company failed to pay the plaintiff and other, similarly situated retail store employees of Hat World, Inc., for time spent making bank deposits of store collections, and seeks to recover unpaid wages, liquidated damages, statutory penalties, attorney's fees, and costs pursuant to the federal Fair Labor Standards Act, the Illinois Minimum Wage Law and the Illinois Wage Payment and Collection Act. On January 15, 2014, the court dismissed the *Maro* case with prejudice, based on the plaintiffs' failure to prosecute. On July 16, 2012 and July 30, 2012, additional putative class and collective actions, *Chavez v. Hat World, Inc.*, were filed in the same court, alleging that certain Hat World employees were misclassified as exempt from overtime pay, and seeking similar relief. The *Chavez* and *Dismukes* actions have been consolidated. The parties have reached an agreement in principle to resolve the matter, subject to documentation and court approval. The Company does not expect the matter or its settlement as proposed to have a material effect on its financial condition or results of operations.

On August 30, 2012, a former employee of a Company subsidiary filed a putative class and collective action, *Kershner v. Hat World, Inc.*, in the Philadelphia, Pennsylvania Court of Common Pleas alleging violations of the Pennsylvania Minimum Wage Act by the subsidiary. The Company has reached an agreement to resolve the matter, subject to approval by the court. On February 10, 2014, the court granted preliminary approval of the proposed settlement. The Company does not expect the matter or its settlement as proposed to have a material effect on its financial condition or results of operations.

On May 23, 2013, a former employee of the Company filed an action, *Everett v. Genesco Inc.*, in the U.S. District Court for the Middle District of Florida alleging violations of the Fair Labor Standards Act, seeking designation as a collective action and the award of allegedly unpaid minimum wages, overtime pay, liquidated damages, penalties, interest, attorneys' fees, and other relief. The Company disputes the material allegations in the action and intends to defend it.

On May 17, 2013, a former employee filed a putative class and representative action, *Garcia v. Genesco, Inc.* in the Superior Court of California for the County of Ventura, alleging various claims under the California Labor Code, including failure to provide meal and rest periods, failure to timely pay wages, failure to provide accurate itemized wage statements, and unfair competition and violation of the Private Attorneys' General Act of 2004, and seeking unspecified damages and penalties. On August 30, 2013, the Company removed the action to the United States District Court for the Central District of California. The Company disputes the material allegations in the complaint and is defending the matter.

In addition to the matters specifically described in this Item 3, Legal Proceedings, the Company is a party to other legal and regulatory proceedings and claims arising in the ordinary course of its business. While management does not believe that the Company's liability with respect to any of these other matters is likely to have a material effect on its financial position, cash flows, or results of operations, legal proceedings are subject to inherent uncertainties and unfavorable rulings could have a material adverse impact on the Company's business and results of operations.

ITEM 4, MINE SAFETY DISCLOSURES

Not applicable.

ITEM 4A, EXECUTIVE OFFICERS OF THE REGISTRANT

The officers of the Company are generally elected at the first meeting of the board of directors following the annual meeting of shareholders and hold office until their successors have been chosen and qualified. The name, age and office of each of the Company's executive officers and certain information relating to the business experience of each are set forth below:

Robert J. Dennis, 60, *Chairman, President and Chief Executive Officer*. Mr. Dennis joined the Company in 2004 as chief executive officer of the Company's acquired Hat World business. Mr. Dennis was named senior vice president of the Company in June 2004 and executive vice president and chief operating officer, with oversight responsibility for all the Company's operating divisions, in October 2005. Mr. Dennis was named president of the Company in October 2006 and chief executive officer in August 2008. Mr. Dennis was named chairman in February 2010, which became effective April 1, 2010. Mr. Dennis joined Hat World in 2001 from Asbury Automotive, where he was employed in senior management roles beginning in 1998. Mr. Dennis was with McKinsey and Company, an international consulting firm, from 1984 to 1997, and became a partner in 1990.

James S. Gulmi, 68, *Senior Vice President – Finance and Chief Financial Officer*. Mr. Gulmi joined the Company in 1971 as a financial analyst, appointed assistant treasurer in 1974 and named treasurer in 1979. He was elected a vice president in 1983 and assumed the responsibilities of chief financial officer in 1986. Mr. Gulmi was appointed senior vice president—finance in January 1996.

Jonathan D. Caplan, 60, *Senior Vice President*. Mr. Caplan rejoined the Company in 2002 as chief executive officer of the branded group and president of Johnston & Murphy and was named senior vice president of the Company in November 2003. Mr. Caplan first joined the Company in June 1982 and served as president of Genesco's Laredo-Code West division from December 1985 to May 1992. After that time, Mr. Caplan was president of Stride Rite's Children's Group and then its Ked's Footwear division, from 1992 to 1996. He was vice president, New Business Development and Strategy, for Service Merchandise Corporation from 1997 to 1998. Prior to rejoining Genesco in October 2002, Mr. Caplan served as president and chief executive officer of Hi-Tec Sports North America beginning in 1998.

James C. Estepa, 62, *Senior Vice President*. Mr. Estepa joined the Company in 1985 and in February 1996 was named vice president operations of Genesco Retail, which included the Jarman Shoe Company, Journeys, Boot Factory and General Shoe Warehouse. Mr. Estepa was named senior vice president operations of Genesco Retail in June 1998. He was named president of Journeys in March 1999. Mr. Estepa was named senior vice president of the Company in April 2000. He was named president and chief executive officer of the Genesco Retail Group in 2001, assuming additional responsibilities of overseeing the Company's former Underground Station segment.

Kenneth J. Kocher, 48, *Senior Vice President.* Mr. Kocher joined Hat World in 1997 as chief financial officer and was named president in October 2005. He was named senior vice president of the Company in October 2006 in addition to continuing his role as president of Hat World. Prior to joining Hat World, he served as a controller with several companies and was a certified public accountant with Edie Bailley, a public accounting firm.

Roger G. Sisson, 50, *Senior Vice President, Secretary and General Counsel.* Mr. Sisson joined the Company in 1994 as assistant general counsel and was elected secretary in February 1994. He was named general counsel in January 1996. Mr. Sisson was named vice president in November 2003. He was named senior vice president in October 2006.

Mimi Eckel Vaughn, 47, *Senior Vice President of Strategy and Shared Services*. Ms. Vaughn joined the Company in September 2003 as vice president of strategy and business development. She was named senior vice president, strategy and business development in October 2006 and senior vice president of strategy and shared services in April 2009. Prior to joining the Company, Ms. Vaughn was executive vice president of business development and marketing, and acting chief financial officer from 2000 to 2001 for Link2Gov Corporation in Nashville. From 1993 to 1999, she was a consultant at McKinsey and Company in Atlanta.

Paul D. Williams, 59, *Vice President and Chief Accounting Officer.* Mr. Williams joined the Company in 1977, was named director of corporate accounting and financial reporting in 1993 and chief accounting officer in April 1995. He was named vice president in October 2006.

Matthew N. Johnson, 49, *Vice President and Treasurer*. Mr. Johnson joined the Company in 1993 as manager, corporate finance and was elected assistant treasurer in December 1993. He was elected treasurer in June 1996. He was named vice president finance in October 2006 and renamed treasurer in April 2011 after a period of service as chief financial officer of one of the Company's divisions. Prior to joining the Company, Mr. Johnson was a vice president in the corporate and institutional banking division of The First National Bank of Chicago.

PART II

ITEM 5, MARKET FOR REGISTRANT'S COMMON EQUITY, RELATED STOCKHOLDER MATTERS AND ISSUER PURCHASES OF EQUITY SECURITIES

Market Information

The Company's common stock is listed on the New York Stock Exchange (Symbol: GCO). The following table sets forth for the periods indicated the high and low sales prices of the common stock as shown in the New York Stock Exchange Composite Transactions listed in the Wall Street Journal.

Fiscal Year ended February 2

	High	Low
2013 1st Quarter	\$ 78.97	\$ 60.02
2nd Quarter	78.78	55.65
3rd Quarter	74.93	55.40
4th Quarter	63.26	50.33

Fiscal Year ended February 1

	High	 Low
2014 1st Quarter	\$ 64.39	\$ 56.87
2nd Quarter	75.84	61.79
3rd Quarter	73.45	60.03
4th Quarter	79.32	65.70

There were approximately 2,675 common shareholders of record on March 14, 2014.

The Company has not paid cash dividends in respect of its common stock since 1973. The Company's ability to pay cash dividends in respect of its common stock is subject to various restrictions. See Notes 6 and 8 to the Consolidated Financial Statements included in Item 8 and "Management's Discussion and Analysis of Financial Condition and Results of Operations – Liquidity and Capital Resources – Sources of Liquidity" for information regarding restrictions on dividends and redemptions of capital stock.

Recent Sales of Unregistered Securities

None.

Issuer Purchases of Equity Securities

None.

Equity Compensation Plan Information

Refer to Part III, Item 12.

ITEM 6, SELECTED FINANCIAL DATA

Financial Summary

In Thousands except per common share data,				F	iscal Year End		
financial statistics and other data		2014	 2013		2012	 2011	 2010
Results of Operations Data							
Net sales	\$	2,624,972	\$ 2,604,817	\$	2,291,987	\$ 1,789,839	\$ 1,574,352
Depreciation and amortization		67,135	63,697		53,737	47,738	47,462
Earnings from operations		163,435	169,863		161,485	87,228	60,374
Earnings from continuing operations before income taxes		158,860	164,832		156,393	86,106	50,440
Earnings from continuing operations		92,982	112,897		93,451	55,244	29,059
Provision for discontinued operations,							
net		(329)	(462)		(1,025)	 (1,336)	 (273)
Net earnings	\$	92,653	\$ 112,435	\$	92,426	\$ 53,908	\$ 28,786
Per Common Share Data							
Earnings from continuing operations							
Basic	\$	3.99	\$ 4.78	\$	3.89	\$ 2.30	\$ 1.31
Diluted		3.94	4.69		3.83	2.27	1.28
Discontinued operations							
Basic		(0.01)	(0.02)		(0.05)	(0.06)	(0.02)
Diluted		(0.02)	(0.01)		(0.04)	(0.05)	(0.01)
Net earnings							
Basic		3.98	4.76		3.84	2.24	1.29
Diluted		3.92	4.68		3.79	2.22	 1.27
Balance Sheet Data							
Total assets	\$	1,439,284	\$ 1,326,072	\$	1,229,761	\$ 960,507	\$ 863,525
Long-term debt		33,730	50,682		40,704		—
Non-redeemable preferred stock		1,305	3,924		4,957	5,183	5,220
Common equity		914,885	817,936		721,774	620,038	577,299
Capital expenditures		98,456	71,737		49,456	29,299	33,825
Financial Statistics	-						
Earnings from operations as a percent of net sales		6.2%	6.5%		7.0%	4.9%	3.8%
Book value per share (common equity divided by common shares							
outstanding)	\$	38.25	\$ 34.09	\$	29.74	\$ 26.19	\$ 23.98
Working capital (in thousands)	\$	451,297	\$ 407,073	\$	291,990	\$ 279,595	\$ 280,621
Current ratio		2.5	2.5		2.0	2.2	2.7
Percent long-term debt to total capitalization		3.5%	5.8%		5.3%	 %	 %
Other Data (End of Year)							
Number of retail outlets*		2,568	2,459		2,387	2,309	2,276
Number of employees		22,250	22,700		21,475	15,200	13,925

* Includes 75 Schuh stores and concessions in Fiscal 2012 acquired June 23, 2011, 48 Sports Avenue stores in Fiscal 2011 acquired October 8, 2010, and 37 Sports Fan Attic stores in Fiscal 2010 acquired November 3, 2009.

Reflected in earnings from continuing operations for Fiscal 2012 was \$7.4 million in acquisition related expenses. See Note 2 to the Consolidated Financial Statements for additional information.

Reflected in earnings from continuing operations for Fiscal 2014, 2013, 2012, 2011 and 2010 were asset impairment and other charges of \$1.3 million, \$17.0 million, \$2.7 million, \$8.6 million and \$13.4 million, respectively. See Note 3 to the Consolidated Financial Statements for additional information regarding these charges.

Long-term debt includes current obligations. In January 2014, the Company entered into the third amended and restated credit agreement in the aggregate principal amount of \$400.0 million. During Fiscal 2010, the Company entered into separate exchange agreements whereby it acquired and retired all \$86.2 million in aggregate principal amount of its Debentures due June 15, 2023 in exchange for the issuance of 4,552,824 shares of its common stock. As a result of the exchange agreements and conversions, the Company recognized a loss on the early retirement of debt of \$5.5 million reflected in earnings from continuing operations. See Note 6 to the Consolidated Financial Statements for additional information regarding the Company's debt.

The Company has not paid dividends on its Common Stock since 1973. See Notes 6 and 8 to the Consolidated Financial Statements and "Management's Discussion and Analysis of Financial Condition and Results of Operations – Liquidity and Capital Resources – Sources of Liquidity" for a description of limitations on the Company's ability to pay dividends.

ITEM 7, MANAGEMENT'S DISCUSSION AND ANALYSIS OF FINANCIAL CONDITION AND RESULTS OF OPERATIONS

Forward Looking Statements

This discussion and the notes to the Consolidated Financial Statements, as well as Item 1, Business, include certain forward-looking statements, which include statements regarding our intent, belief or expectations and all statements other than those made solely with respect to historical fact. Actual results could differ materially from those reflected by the forward-looking statements in this discussion and a number of factors may adversely affect the forward looking statements and the Company's future results, liquidity, capital resources or prospects. These include, but are not limited to, the amount of required accruals related to the earn-out bonus potentially payable to Schuh management based on the achievement of certain performance objectives, the timing and amount of non-cash asset impairments related to retail store fixed assets or to intangible assets of acquired businesses, weakness in the consumer economy, competition in the Company's markets, inability of customers to obtain credit, fashion trends that affect the sales or product margins of the Company's retail product offerings, changes in buying patterns by significant wholesale customers, bankruptcies or deterioration in financial condition of significant wholesale customers, disruptions in product supply or distribution, unfavorable trends in fuel costs, foreign exchange rates, foreign labor and material costs, and other factors affecting the cost of products, the Company's ability to continue to complete and integrate acquisitions, expand its business and diversify its product base and changes in the timing of holidays or in the onset of seasonal weather affecting period-to-period sales comparisons. Additional factors that could affect the Company's prospects and cause differences from expectations include the ability to build, open, staff and support additional retail stores and to renew leases in existing stores and control occupancy costs, and to conduct required remodeling or refurbishment on schedule and at expected expense levels, deterioration in the performance of individual businesses or of the Company's market value relative to its book value, resulting in impairments of fixed assets or intangible assets or other adverse financial consequences, unexpected changes to the market for the Company's shares, variations from expected pensionrelated charges caused by conditions in the financial markets, and the outcome of litigation, investigations and environmental matters involving the Company. For a discussion of additional risk factors, see Item 1A, Risk Factors.

Overview

Description of Business

The Company's business includes the design and sourcing, marketing and distribution of footwear and accessories through retail stores, including Journeys[®], Journeys Kidz[®], Shi by Journeys[®], Underground by Journeys[®] and Johnston & Murphy[®] in the U.S., Puerto Rico and Canada and through Schuh[®] stores in the United Kingdom and the Republic of Ireland, and through e-commerce websites and catalogs, and at wholesale, primarily under the Company's Johnston & Murphy brand, the recently relaunched Trask brand, the licensed Dockers[®] brand, and other brands that the Company licenses for men's footwear. The Company's wholesale footwear brands are distributed to more than 1,000 retail accounts in the United States, including a number of leading department, discount, and specialty stores. The Company's business also includes Lids Sports, which operates (i) headwear and accessory stores under the Lids[®] name and other names in the U.S., Puerto Rico and Canada, (ii) the Lids Locker Room and Lids Clubhouse businesses, consisting of sports-oriented fan shops featuring a broad array of licensed merchandise such as apparel, hats and accessories, sports decor and novelty products, operating under various trade names, (iii) licensed team merchandise departments in Macy's department stores operated under the name Locker Room by Lids and on macys.com under a license agreement with Macy's, (iv) e-commerce operations and (v) an athletic team dealer business operating as Lids Team Sports. Including both the footwear businesses and the Lids Sports business, at February 1, 2014, the Company operated 2,568 retail stores and leased departments in the U.S., Puerto Rico, Canada, the United Kingdom and the Republic of Ireland.

During Fiscal 2014, the Company operated five reportable business segments (not including corporate): (i) Journeys Group, comprised of Journeys, Journeys Kidz, Shi by Journeys and Underground by Journeys retail footwear chains, catalog and e-commerce operations; (ii) Schuh Group, comprised of the Schuh retail footwear chain and e-commerce operations; (iii) Lids Sports Group, comprised as described in the preceding paragraph; (iv) Johnston & Murphy Group, comprised of Johnston & Murphy retail operations, catalog and e-commerce operations and wholesale distribution of products under the Johnston & Murphy and Trask brands; and (v) Licensed Brands, comprised of Dockers[®] Footwear, sourced and marketed under a license from Levi Strauss & Company; SureGrip[®]Footwear, occupational footwear primarily sold directly to consumers; and other brands.

The Journeys retail footwear stores sell footwear and accessories primarily for 13 to 22 year old men and women. The stores average approximately 1,975 square feet. The Journeys Kidz retail footwear stores sell footwear primarily for younger children, ages five to 12. These stores average approximately 1,425 square feet. Shi by Journeys retail footwear stores sell footwear and accessories to fashion-conscious women in their early 20's to mid 30's. These stores average approximately 2,125 square feet. The Underground by Journeys retail footwear stores sell footwear and accessories sell footwear and accessories primarily for stores sell footwear and accessories average approximately 2,125 square feet. The Underground by Journeys retail footwear stores sell footwear and accessories primarily for men and women in the 20 to 35 age group. These stores average approximately 1,825 square feet. The Journeys Group stores are

primarily in malls and factory outlet centers throughout the United States, Puerto Rico and Canada. The Journeys Group operates 31 stores in Canada. Journeys also sells footwear and accessories through direct-to-consumer catalog and e-commerce operations.

The Schuh retail footwear stores sell a broad range of branded casual and athletic footwear along with a meaningful private label offering primarily for 15 to 30 year old men and women. The stores, which average approximately 5,000 square feet, include both street-level and mall locations in the United Kingdom and the Republic of Ireland. During the third quarter of Fiscal 2013, the Schuh Group opened its first Schuh Kids store. As of February 1, 2014, the Company has opened four Schuh Kids stores that sell footwear primarily for younger children, ages five to 12, and average 2,425 square feet. The Schuh Group also sells footwear through e-commerce operations.

The Lids Sports Group includes stores and kiosks, primarily under the Lids banner, that sell licensed and branded headwear to men and women primarily in the early-teens to mid-20's age group. The Lids store locations average approximately 875 square feet and are primarily in malls, airports, street-level stores and factory outlet centers throughout the United States, Puerto Rico and Canada. The Lids Sports Group also operates Lids Locker Room and Lids Clubhouse stores under a number of trade names, selling licensed sports headwear, apparel and accessories to sports fans of all ages in locations averaging approximately 3,050 square feet in malls and other locations primarily in the United States. The Lids Sports Group operates 110 stores in Canada. The Lids Sports Group also operates Locker Room by Lids leased departments in Macy's department stores selling headwear, apparel, accessories and novelties from an assortment of college and professional teams specific to that particular Macy's department store geographic location. The Lids Sports Group also sells headwear and accessories through e-commerce operations. In addition, the Lids Sports Group operates Lids Team Sports, an athletic team dealer business.

Johnston & Murphy retail shops sell a broad range of men's footwear, apparel and accessories. Women's footwear and accessories are sold in select Johnston & Murphy retail locations. Johnston & Murphy shops average approximately 1,525 square feet and are located primarily in better malls and in airports throughout the United States and in Canada. Johnston & Murphy opened its first store in Canada during the fourth quarter of Fiscal 2012. As of February 1, 2014, Johnston & Murphy operated seven stores in Canada. The Company also has license and distribution agreements for wholesale and retail sales of Johnston & Murphy products in various non - U.S. jurisdictions. The Company also sells Johnston & Murphy footwear and accessories in factory stores, averaging approximately 2,350 square feet, located in factory outlet malls, and through a direct -to-consumer catalog and e-commerce operations. In addition, Johnston & Murphy shoes are distributed through the Company's wholesale operations to better department and independent specialty stores. Additionally, the Company recently relaunched the Trask brand, with men's and women's footwear and leather accessories distributed to better independent retailers and department stores.

The Licensed Brands segment markets casual and dress casual footwear under the licensed Dockers[®] brand to men aged 30 to 55 through many of the same national retail chains that carry Dockers slacks and sportswear and in department and specialty stores across the country. The Company entered into an exclusive license with Levi Strauss & Co. to market men's footwear in the United States under the Dockers brand name in 1991. Levi Strauss & Co. and the Company have subsequently added additional territories, including Canada and Mexico and certain other Latin American countries. The Dockers license agreement was renewed on July 23, 2012 for a term expiring November 30, 2015, subject to extension for an additional 3-year term if certain conditions are met. The Company acquired Keuka Footwear in the third quarter of Fiscal 2011 and subsequently launched its SureGrip[®] Footwear line of slip-resistant, occupational footwear from that base. The Company sources and distributes the SureGrip line to employees in the hospitality, healthcare, and other industries.

Strategy

The Company's long-term strategy has been to seek organic growth by: 1) increasing the Company's store base, 2) increasing retail square footage, 3) improving comparable sales, both in stores and digital commerce, 4) increasing operating margin and 5) enhancing the value of its brands. Most of the additional stores planned in North America are currently planned to be Lids Locker Room, Lids Clubhouse, Journeys Kidz and Johnston & Murphy stores, all of which management considers to be underpenetrated in many markets.

To further supplement its organic growth potential, the Company has made acquisitions, including the acquisition of the Schuh Group in June 2011 and several smaller acquisitions of businesses in the Lids Sports Group's markets, and expects to consider acquisition opportunities, either to augment its existing businesses or to enter new businesses that it considers compatible with its existing businesses, core expertise and strategic profile. Acquisitions involve a number of risks, including, among others, inaccurate valuation of the acquired business, the assumption of undisclosed liabilities, the failure to integrate the acquired business appropriately, and distraction of management from existing businesses. The Company seeks to mitigate these risks by applying appropriate financial metrics in its valuation analysis and developing and executing plans for due diligence and integration that are appropriate to each acquisition. The Company also seeks appropriate opportunities to

extend existing brands and retail concepts. For example, the Schuh Group opened its first Schuh Kids store in Scotland during the third quarter of Fiscal 2013. The Company typically tests such extensions on a relatively small scale to determine their viability and to refine their strategies and operations before making significant, long-term commitments.

More generally, the Company attempts to develop strategies to mitigate the risks it views as material, including those discussed under the caption "Forward Looking Statements," above, and those discussed in Item 1A, Risk Factors. Among the most important of these factors are those related to consumer demand. Conditions in the economy can affect demand, resulting in changes in sales and, as prices are adjusted to drive sales and manage inventories, in gross margins. Because fashion trends influencing many of the Company's target customers can change rapidly, the Company believes that its ability to react quickly to those changes has been important to its success. Even when the Company succeeds in aligning its merchandise offerings with consumer preferences, those preferences may affect results by, for example, driving sales of products with lower average selling prices or products which are more widely available in the marketplace and thus more subject to competitive pressures than the Company's typical offering. Moreover, economic factors, such as the relatively high level of unemployment and any future economic contraction and changes in tax policy, may reduce the consumer's disposable income or his or her willingness to purchase discretionary items, and thus may reduce demand for the Company's merchandise, regardless of the Company's skill in detecting and responding to fashion trends. The Company believes its experience and discipline in merchandising and the buying power associated with its relative size and importance in the industry segments in which it competes are important to its ability to mitigate risks associated with changing customer preferences and other changes in consumer demand.

Summary of Results of Operations

The Company's net sales increased 0.8% during Fiscal 2014 compared to Fiscal 2013. The increase reflected a 4% increase in Lids Sports Group sales, an 11% increase in Johnston & Murphy Group sales and a 1% increase in Licensed Brands sales, offset by a 3% decrease in Journeys Group sales and a 2% decrease in Schuh Group sales. Included in Fiscal 2013 was a 53rd week compared to a 52-week year for Fiscal 2014. Excluding the 53rd week from Fiscal 2013, sales would have increased 2.2% for Fiscal 2014. Gross margin decreased as a percentage of net sales during Fiscal 2014, reflecting gross margin decreases in Schuh Group, Lids Sports Group and Licensed Brands, partially offset by increased gross margin in Journeys Group and Johnston & Murphy Group. Selling and administrative expenses increased as a percentage of net sales during Fiscal 2014 in all of the Company's business segments except Licensed Brands. Earnings from operations decreased as a percentage of net sales from 6.5% in Fiscal 2013 to 6.2% in Fiscal 2014, reflecting decreased earnings in Journeys Group, Schuh Group and Lids Sports Group, partially offset by improved earnings from operations in Johnston & Murphy Group and Licensed Brands.

Significant Developments

Schuh Acquisition

On June 23, 2011, the Company, through its newly-formed, wholly-owned subsidiary Genesco (UK) Limited ("Genesco UK"), completed the acquisition of all the outstanding shares of Schuh Group Ltd. ("Schuh") for a total purchase price of approximately £100.0 million, less £29.5 million outstanding under existing Schuh credit facilities, which remain in place, less a £1.9 million working capital adjustment and plus £6.2 million net cash acquired, with £5.0 million withheld that was paid in June 2013. The Company financed the acquisition with borrowings under its existing credit facility and the balance from cash on hand. The purchase agreement also provides for deferred purchase price payments totaling £25 million, payable in installments of £15 million and £10 million on the third and fourth anniversaries of the closing, respectively, subject to the payees' not having terminated their employment with Schuh under certain specified circumstances. This amount will be recorded as compensation expense and not reported as a component of the cost of the acquisition.

Headquartered in Scotland, Schuh is a specialty retailer of casual and athletic footwear sold through 99 retail stores in the United Kingdom and the Republic of Ireland as of February 1, 2014. The Company completed the acquisition in order to enhance its strategic development and prospects for growth and provide the Company with an established retail presence in the United Kingdom and improved insight into global fashion trends. The results of Schuh's operations for Fiscal 2014 include net sales of \$364.7 million and operating earnings of \$3.1 million. The results of Schuh's operations for Fiscal 2013 include net sales of \$370.5 million and operating earnings of \$11.2 million. The results of Schuh's operations for the date of acquisition through January 28, 2012, including net sales of \$212.3 million and operating earnings of \$11.7 million, have been included in the Company's Consolidated Financial Statements for the fiscal year ended January 28, 2012. During the fiscal year ended January 28, 2012, the Company expensed \$7.4 million in costs related to the acquisition. These costs were recorded as selling and administrative expenses on the Consolidated Statement of Operations. Compensation expense related to the Schuh acquisition deferred purchase price obligation was \$11.7 million,

\$12.1 million and \$7.2 million in Fiscal 2014, 2013 and 2012, respectively. This expense is included in the operating earnings for the Schuh Group segment.

Other Acquisitions

In Fiscal 2014 and 2013, the Company completed other acquisitions of primarily small retail chains for a total purchase price of \$13.6 million and \$23.8 million, respectively. The stores acquired will be operated within the Lids Sports Group.

Network Intrusion

On December 10, 2010, the Company announced that it had suffered a criminal intrusion into the portion of its computer network that processes payments for transactions in certain of its retail stores. Visa, Inc., MasterCard Worldwide and American Express Travel Related Services Company, Inc. have asserted claims totaling approximately \$15.6 million in connection with the intrusion and the claims of two of the claimants have been collected by withholding payment card receivables of the Company. In the fourth quarter of Fiscal 2013, the Company recorded a \$15.4 million charge to earnings in connection with the disputed liability. On March 7, 2013, the Company filed an action in the U.S. District Court for the Middle District of Tennessee against Visa U.S.A. Inc., Visa Inc. and Visa International Service Association seeking to recover \$13.3 million in non-compliance fines and issuer reimbursement assessments collected from the Company in connection with the intrusion. The Company does not currently expect any future claims in connection with the intrusion to have a material effect on its financial condition, cash flows, or results of operations.

Asset Impairment and Other Charges

The Company recorded a pretax charge to earnings of \$1.3 million in Fiscal 2014, including \$3.3 million for network intrusion expenses, \$2.4 million for other legal matters, \$2.3 million for retail store asset impairments and \$1.6 million for a lease termination offset by an \$(8.3) million gain on the lease termination of a New York City Journeys store.

The Company recorded a pretax charge to earnings of \$17.0 million in Fiscal 2013, including \$15.6 million for network intrusion expenses, \$1.4 million for retail store asset impairments and \$0.1 million for other legal matters.

The Company recorded a pretax charge to earnings of \$2.7 million in Fiscal 2012, including \$1.1 million for retail store asset impairments, \$0.9 million for other legal matters and \$0.7 million for network intrusion expenses.

Postretirement Benefit Liability Adjustments

The return on pension plan assets was \$12.3 million for Fiscal 2014, compared to an expected return of \$6.7 million. The discount rate used to measure benefit obligations increased from 4.00% to 4.40% in Fiscal 2014. As a result of the increase in the return on plan assets and the increased discount rate, the pension liability reflected in the Consolidated Balance Sheets decreased to \$9.2 million compared to \$20.5 million in Fiscal 2013. There was a decrease in the pension liability adjustment of \$9.5 million (net of tax) in accumulated other comprehensive income in equity. Depending upon future interest rates and returns on plan assets and other factors, there can be no assurance that additional adjustments in future periods will not be required.

Discontinued Operations

In Fiscal 2014, the Company recorded an additional charge to earnings of \$0.5 million (\$0.3 million net of tax) reflected in discontinued operations primarily for anticipated costs of environmental remedial alternatives related to former facilities operated by the Company. For additional information, see Note 13 to the Consolidated Financial Statements.

In Fiscal 2013, the Company recorded an additional charge to earnings of \$0.8 million (\$0.5 million net of tax) reflected in discontinued operations primarily for anticipated costs of environmental remedial alternatives related to former facilities operated by the Company. For additional information, see Note 13 to the Consolidated Financial Statements.

In Fiscal 2012, the Company recorded an additional charge to earnings of \$1.7 million (\$1.0 million net of tax) reflected in discontinued operations, including \$1.8 million primarily for anticipated costs of environmental remedial alternatives related to former facilities operated by the Company, offset by a \$0.1 million gain for excess provisions to prior discontinued operations. For additional information, see Note 13 to the Consolidated Financial Statements.



Critical Accounting Policies

Inventory Valuation

As discussed in Note 1 to the Consolidated Financial Statements, the Company values its inventories at the lower of cost or market.

In its footwear wholesale operations, its Schuh Group segment and its Lids Sports Group wholesale operations, except for the Anaconda Sports wholesale division, cost is determined using the first-in, first-out ("FIFO") method. Market value is determined using a system of analysis which evaluates inventory at the stock number level based on factors such as inventory turn, average selling price, inventory level, and selling prices reflected in future orders. The Company provides reserves when the inventory has not been marked down to market value based on current selling prices or when the inventory is not turning and is not expected to turn at levels satisfactory to the Company.

The Lids Sports Group retail segment and its Anaconda Sports wholesale division employ the moving average cost method for valuing inventories and apply freight using an allocation method. The Company provides a valuation allowance for slow-moving inventory based on negative margins and estimated shrink based on historical experience and specific analysis, where appropriate.

In its retail operations, other than the Schuh Group and Lids Sports Group retail segments, the Company employs the retail inventory method, applying average cost-to-retail ratios to the retail value of inventories. Under the retail inventory method, valuing inventory at the lower of cost or market is achieved as markdowns are taken or accrued as a reduction of the retail value of inventories.

Inherent in the retail inventory method are subjective judgments and estimates, including merchandise mark-on, markups, markdowns, and shrinkage. These judgments and estimates, coupled with the fact that the retail inventory method is an averaging process, could produce a range of cost figures. To reduce the risk of inaccuracy and to ensure consistent presentation, the Company employs the retail inventory method in multiple subclasses of inventory with similar gross margins, and analyzes markdown requirements at the stock number level based on factors such as inventory turn, average selling price, and inventory age. In addition, the Company accrues markdowns as necessary. These additional markdown accruals reflect all of the above factors as well as current agreements to return products to vendors and vendor agreements to provide markdown support. In addition to markdown provisions, the Company maintains provisions for shrinkage and damaged goods based on historical rates.

Inherent in the analysis of both wholesale and retail inventory valuation are subjective judgments about current market conditions, fashion trends, and overall economic conditions. Failure to make appropriate conclusions regarding these factors may result in an overstatement or understatement of inventory value. A change of 10% from the recorded provisions for markdowns, shrinkage and damaged goods would have changed inventory by \$1.1 million at February 1, 2014.

Impairment of Long-Lived Assets

As discussed in Note 1 to the Consolidated Financial Statements, the Company periodically assesses the realizability of its long-lived assets, other than goodwill, and evaluates such assets for impairment whenever events or changes in circumstances indicate that the carrying amount of an asset may not be recoverable. Asset impairment is determined to exist if estimated future cash flows, undiscounted and without interest charges, are less than the carrying amount. Inherent in the analysis of impairment are subjective judgments about future cash flows. Failure to make appropriate conclusions regarding these judgments may result in an overstatement or understatement of the value of long-lived assets.

The goodwill impairment test involves performing a qualitative assessment, on a reporting unit level, based on current circumstances. If the results of the qualitative assessment indicate that it is more likely than not that the fair value of a reporting unit is greater than its carrying amount, a two-step impairment test will not be performed. However, if the results of the qualitative assessment indicate that it is more likely than not that the fair value of a reporting unit is less than its carrying amount, then a two-step impairment test is performed. Alternatively, the Company may elect to bypass the qualitative assessment and proceed directly to the two-step impairment test, on a reporting unit level. The first step is a comparison of the fair value and carrying value of the business unit with which the goodwill is associated. The Company estimates fair value using the best information available, and computes the fair value derived by an income approach utilizing discounted cash flow projections. The income approach uses a projection of a reporting unit's estimated operating results and cash flows that is discounted using a weighted-average cost of capital that reflects current market conditions. A key assumption in the Company's fair value estimate is the weighted average cost of capital utilized for discounting its cash flow projections in its income approach. The Company believes the rate it used in its annual test, which is completed in the fourth quarter each year, was consistent with the risks inherent in its business and with industry discount rates. The projection uses management's best estimates of economic and market conditions over the projected period including growth

rates in sales, costs, estimates of future expected changes in operating margins and cash expenditures. Other significant estimates and assumptions include terminal value growth rates, future estimates of capital expenditures and changes in future working capital requirements.

If the carrying value of the business unit is higher than its fair value, there is an indication that impairment may exist and the second step must be performed to measure the amount of impairment loss. The amount of impairment is determined by comparing the implied fair value of reporting unit goodwill to the carrying value of the goodwill in the same manner as if the reporting unit was being acquired in a business combination. Specifically, the Company would allocate the fair value to all of the assets and liabilities of the reporting unit, including any unrecognized intangible assets, in a hypothetical analysis that would calculate the implied fair value of goodwill. If the implied fair value of goodwill is less than the recorded goodwill, the Company would record an impairment charge for the difference.

As a result of the various acquisitions comprising the Lids Team Sports team dealer business, the Company carries goodwill at a value of \$14.2 million on its Consolidated Balance Sheets related to such acquisitions. The Company found that the result of its annual impairment test, which valued the business at approximately \$3.9 million in excess of its carrying value, indicated no impairment at that time. The Company may determine in future impairment tests that some or all of the carrying value of the goodwill may not be recoverable. Such a finding would require a write-off of the amount of the carrying value that is impaired, which would reduce the Company's profitability in the period of the impairment charge. Holding all other assumptions constant as of the measurement date, the Company noted that an increase in the weighted average cost of capital of 100 basis points would reduce the fair value of the Lids Team Sports business by \$5.9 million. Furthermore, the Company noted that a decrease in projected annual revenue by one percent would reduce the fair value of the Lids Team Sports business by \$0.4 million. However, if other assumptions do not remain constant, the fair value of the Lids Team Sports business may decrease by a greater amount. Since the maximum non-cash goodwill impairment charge would be \$14.2 million, the Company does not believe that any impairment charge related thereto would be material.

Environmental and Other Contingencies

The Company is subject to certain loss contingencies related to environmental proceedings and other legal matters, including those disclosed in Note 13 to the Company's Consolidated Financial Statements. The Company has made pretax accruals for certain of these contingencies, including approximately \$0.5 million reflected in Fiscal 2014, \$0.8 million reflected in Fiscal 2013 and \$1.8 million reflected in Fiscal 2012. These charges are included in provision for discontinued operations, net in the Consolidated Statements of Operations because they relate to former facilities operated by the Company. The Company monitors these matters on an ongoing basis and, on a quarterly basis, management reviews the Company's reserves and accruals in relation to each of them, adjusting provisions as management deems necessary in view of changes in available information. Changes in estimates of liability are reported in the periods when they occur. Consequently, management believes that its reserve in relation to each proceeding is a best estimate of probable loss connected to the proceeding, or in cases in which no best estimate is possible, the minimum amount in the range of estimated losses, based upon its analysis of the facts and circumstances as of the close of the most recent fiscal quarter. However, because of uncertainties and risks inherent in litigation generally and in environmental proceedings in particular, there can be no assurance that future developments will not require additional reserves, that some or all reserves will be adequate or that the amounts of any such additional reserves or any such inadequacy will not have a material adverse effect upon the Company's financial condition or results of operations.

Revenue Recognition

Retail sales are recorded at the point of sale and are net of estimated returns and exclude sales and value added taxes. Catalog and internet sales are recorded at time of delivery to the customer and are net of estimated returns and exclude sales and value added taxes. Wholesale revenue is recorded net of estimated returns and allowances for markdowns, damages and miscellaneous claims when the related goods have been shipped and legal title has passed to the customer. Shipping and handling costs charged to customers are included in net sales. Estimated returns are based on historical returns and claims. Actual amounts of markdowns have not differed materially from estimates. Actual returns and claims in any future period may differ from historical experience.

Income Taxes

As part of the process of preparing Consolidated Financial Statements, the Company is required to estimate its income taxes in each of the tax jurisdictions in which it operates. This process involves estimating actual current tax obligations together with assessing temporary differences resulting from differing treatment of certain items for tax and accounting purposes, such as depreciation of property and equipment and valuation of inventories. These temporary differences result in deferred tax assets and liabilities, which are included within the Consolidated Balance Sheets. The Company then assesses the

likelihood that its deferred tax assets will be recovered from future taxable income. Actual results could differ from this assessment if adequate taxable income is not generated in future periods. To the extent the Company believes that recovery of an asset is at risk, valuation allowances are established. To the extent valuation allowances are established or increased in a period, the Company includes an expense within the tax provision in the Consolidated Statements of Operations. These deferred tax valuation allowances may be released in future years when management considers that it is more likely than not that some portion or all of the deferred tax assets will be realized. In making such a determination, management will need to periodically evaluate whether or not all available evidence, such as future taxable income and reversal of temporary differences, tax planning strategies, and recent results of operations, provides sufficient positive evidence to offset any other potential negative evidence that may exist at such time. In the event the deferred tax valuation allowance is released, the Company would record an income tax benefit for the portion or all of the deferred tax valuation allowance of \$3.8 million.

Income tax reserves for uncertain tax positions are determined using the methodology required by the Income Tax Topic of the Accounting Standards Codification ("Codification"). This methodology requires companies to assess each income tax position taken using a two step process. A determination is first made as to whether it is more likely than not that the position will be sustained, based upon the technical merits, upon examination by the taxing authorities. If the tax position is expected to meet the more likely than not criteria, the benefit recorded for the tax position equals the largest amount that is greater than 50% likely to be realized upon ultimate settlement of the respective tax position. Uncertain tax positions require determinations and estimated liabilities to be made based on provisions of the tax law which may be subject to change or varying interpretation. If the Company's determinations and estimates prove to be inaccurate, the resulting adjustments could be material to its future financial results. See Note 9 to the Company's Consolidated Financial Statements for additional information regarding income taxes.

The Company recorded an effective income tax rate of 41.5% for Fiscal 2014 compared to 31.5% for 2013. The tax rate for Fiscal 2013 was lower primarily due to the reversal of previously recorded charges related to uncertain tax positions due to the expiration of the applicable statutes of limitations and a settlement with a state tax authority more favorable than anticipated related to other uncertain tax positions.

Postretirement Benefits Plan Accounting

Full-time employees who had at least 1,000 hours of service in calendar year 2004, except employees in the Lids Sports Group and Schuh Group segments, are covered by a defined benefit pension plan. The Company froze the defined benefit pension plan effective January 1, 2005. The Company also provides certain former employees with limited medical and life insurance benefits. The Company funds at least the minimum amount required by the Employee Retirement Income Security Act.

As required by the Compensation – Retirement Benefits Topic of the Codification, the Company is required to recognize the overfunded or underfunded status of postretirement benefit plans as an asset or liability in their Consolidated Balance Sheets and to recognize changes in that funded status in accumulated other comprehensive loss, net of tax, in the year in which the changes occur.

The Company accounts for the defined benefit pension plans using the Compensation-Retirement Benefits Topic of the Codification. As permitted under this topic, pension expense is recognized on an accrual basis over employees' approximate service periods. The calculation of pension expense and the corresponding liability requires the use of a number of critical assumptions, including the expected long-term rate of return on plan assets and the assumed discount rate, as well as the recognition of actuarial gains and losses. Changes in these assumptions can result in different expense and liability amounts, and future actual experience can differ from these assumptions.

Long Term Rate of Return Assumption – Pension expense increases as the expected rate of return on pension plan assets decreases. The Company estimates that the pension plan assets will generate a long-term rate of return of 7.75%. To develop this assumption, the Company considered historical asset returns, the current asset allocation and future expectations of asset returns. The expected long-term rate of return on plan assets is based on a long-term investment policy of 50% U.S. equities, 13% international equities, 35% U.S. fixed income securities and 2% cash equivalents. For Fiscal 2014, if the expected rate of return had been decreased by 1%, net pension expense would have increased by \$0.9 million, and if the expected rate of return had been increased by 1%, net pension expense would have decreased by \$0.9 million.

<u>Discount Rate</u> – Pension liability and future pension expense increase as the discount rate is reduced. The Company discounted future pension obligations using a rate of 4.40%, 4.00% and 4.35% for Fiscal 2014, 2013 and 2012, respectively. The discount rate at February 1, 2014 was determined based on a yield curve of high quality corporate bonds with cash flows matching the Company's plans' expected benefit payments. For Fiscal 2014, if the discount rate had been increased by 0.5%, net pension expense would have decreased by \$0.6 million, and if the discount rate had been decreased by 0.5%,

net pension expense would have increased by \$0.6 million. In addition, if the discount rate had been increased by 0.5%, the projected benefit obligation would have decreased by \$6.1 million. If the discount rate had been decreased by 0.5%, the projected benefit obligation would have been increased by \$6.8 million and the accumulated benefit obligation would have been increased by \$6.8 million and the accumulated benefit obligation would have been increased by \$6.8 million.

<u>Amortization of Gains and Losses</u> – The Company utilizes a calculated value of assets, which is an averaging method that recognizes changes in the fair values of assets over a period of five years. At the end of Fiscal 2014, the Company had unrecognized actuarial losses of \$27.1 million. Accounting principles generally accepted in the United States require that the Company recognize a portion of these losses when they exceed a calculated threshold. These losses might be recognized as a component of pension expense in future years and would be amortized over the average future service of employees, which is currently approximately six years. Future changes in plan asset returns, assumed discount rates and various other factors related to the pension plan will impact future pension expense and liabilities, including increasing or decreasing unrecognized actuarial gains and losses.

The Company recognized expense for its defined benefit pension plans of \$4.4 million, \$4.3 million and \$2.8 million in Fiscal 2014, 2013 and 2012, respectively. The Company's pension expense is expected to decrease in Fiscal 2015 by approximately \$1.5 million due to a smaller actuarial loss to be amortized.

Comparable Sales

During Fiscal 2013, the Company revised its presentation of comparable sales to include its e-commerce and direct mail catalog businesses. Prior year comparable sales have been adjusted to conform to the current year presentation. For purposes of this report, "comparable sales" are sales from stores open longer than one year, beginning in the fifty-third week of a store's operation (which we refer to in this report as "same store sales"), and sales from websites operated longer than one year and direct mail catalog sales (which we refer to in this report as "comparable direct sales"). Temporarily closed stores are excluded from the comparable sales calculation for every full week of the store closing. Expanded stores are excluded from the comparable sales calculation until the fifty-third week of operation in the expanded format.

Results of Operations—Fiscal 2014 Compared to Fiscal 2013

The Company's net sales for Fiscal 2014 (52 weeks) increased 0.8% to \$2.62 billion from \$2.60 billion in Fiscal 2013 (53 weeks). The increase in net sales was a result of increased sales in Lids Sports Group, Johnston & Murphy Group and Licensed Brands, offset by decreased sales in Journeys and Schuh Groups. Net sales for Fiscal 2013 included an estimated \$35.2 million of sales due to the fifty-third week. Gross margin was flat at \$1.30 billion, but decreased as a percentage of net sales from 49.9% to 49.5%, primarily reflecting decreased gross margin as a percentage of net sales in the Schuh Group, Lids Sports Group and Licensed Brands, offset slightly by increased gross margin as a percentage of net sales in the Schuh Group, Lids Sports Group and Licensed Brands, offset slightly by increased gross margin as a percentage of net sales from 42.7% to 43.2%, primarily reflecting expenses in Fiscal 2014 increased 2.0% from Fiscal 2013 and increased as a percentage of net sales from 42.7% to 43.2%, primarily reflecting expense increases in all the Company's business segments except Licensed Brands. The Company records buying and merchandising and occupancy costs in selling and administrative expense. Because the Company does not include these costs in cost of sales, the Company's gross margin may not be comparable to other retailers that include these costs in the calculation of gross margin. Explanations of the changes in results of operations are provided by business segment in discussions following these introductory paragraphs.

Earnings from continuing operations before income taxes ("pretax earnings") for Fiscal 2014 were \$158.9 million, compared to \$164.8 million for Fiscal 2013. Pretax earnings for Fiscal 2014 included asset impairment and other charges of \$1.3 million, including \$3.3 million for expenses related to the computer network intrusion announced in December 2010, \$2.4 million for other legal matters, \$2.3 million for retail store asset impairments and \$1.6 million for a lease termination offset by an \$(8.3) million gain on the lease termination of a New York City Journeys store. Pretax earnings for Fiscal 2013 included asset impairment and other charges of \$15.5 million for expenses related to the computer network intrusion, \$1.4 million for retail store asset impairments and \$0.1 million for other legal matters.

Net earnings for Fiscal 2014 were \$92.7 million (\$3.92 diluted earnings per share) compared to \$112.4 million (\$4.68 diluted earnings per share) for Fiscal 2013. Net earnings for Fiscal 2014 includes \$0.3 million (\$0.02 diluted loss per share) charge to earnings (net of tax), primarily for anticipated costs of environmental remedial alternatives related to former facilities operated by the Company. Net earnings for Fiscal 2013 includes \$0.5 million (\$0.01 diluted loss per share) charge to earnings (net of tax) primarily for anticipated costs of environmental remedial alternatives related to former facilities operated by the Company. Net earnings for Fiscal 2013 includes \$0.5 million (\$0.01 diluted loss per share) charge to earnings (net of tax) primarily for anticipated costs of environmental remedial alternatives related to former facilities operated by the Company. The Company recorded an effective federal income tax rate of 41.5% for Fiscal 2014 compared to 31.5% for Fiscal 2013. Fiscal 2013's lower effective tax rate reflects the reversal of previously recorded charges related to uncertain tax positions due to the expiration of the applicable statutes of limitations and a settlement with a state tax

authority more favorable than anticipated related to other uncertain tax positions. See Note 9 to the Consolidated Financial Statements for additional information.

Journeys Group

	Fiscal Year Ended				
	2014		2013	% Change	
	(dollars in thousands)				
Net sales	\$ 1,082,241	\$	1,111,490	(2.6)%	
Earnings from operations	\$ 97,377	\$	109,953	(11.4)%	
Operating margin	9.0%		9.9%		

Net sales from Journeys Group decreased 2.6% to \$1.08 billion for Fiscal 2014 from \$1.11 billion for Fiscal 2013. The decrease reflects primarily a 2% decrease in same store sales, an 18% increase in comparable direct sales and a 1% decrease in comparable sales, including both store and direct sales. The comparable store sales decrease reflected a 3% decrease in footwear unit comparable sales partially offset by a 1% increase in the average price per pair of shoes. Total unit sales decreased 4% during the same period. The store count for Journeys Group was 1,168 stores at the end of Fiscal 2014, including 174 Journeys Kidz stores, 50 Shi by Journeys stores, 117 Underground by Journeys stores and 31 Journeys stores in Canada, compared to 1,157 stores at the end of Fiscal 2013, including 156 Journeys Kidz stores, 51 Shi by Journeys stores, 130 Underground by Journeys stores and 24 Journeys stores in Canada.

Journeys Group earnings from operations for Fiscal 2014 decreased 11.4% to \$97.4 million, compared to \$110.0 million for Fiscal 2013. The decrease in earnings from operations was primarily due to decreased net sales and to increased expenses as a percentage of net sales, reflecting negative leverage from negative comparable sales, partially offset by decreased bonus accruals.

Schuh Group

		Fiscal Year Ended				
		2014		2013	% Change	
		ds)				
Net sales	\$	364,732	\$	370,480	(1.6)%	
Earnings from operations	\$	3,063	\$	11,209	(72.7)%	
Operating margin		0.8%		3.0%		

Net sales from the Schuh Group decreased 1.6% to \$364.7 million for Fiscal 2014, compared to \$370.5 million for Fiscal 2013. The sales decrease reflects a 9% decrease in same store sales, a 4% decrease in comparable direct sales, an 8% decrease in comparable sales, including both store and direct sales, and a \$2.1 million decrease in sales from changes in exchange rates, partially offset by a 13% increase in average stores operated (i.e. the sum of the number of stores open on the first day of the fiscal year and the last day of each fiscal month during the year divided by thirteen). Schuh Group operated 99 stores, including four Schuh Kids stores at the end of Fiscal 2014 compared to 79 stores, including three Schuh Kids stores, and 13 concessions at the end of Fiscal 2013.

Schuh Group earnings from operations were \$3.1 million for Fiscal 2014 compared to \$11.2 million for Fiscal 2013. Earnings included \$11.7 million this year and \$12.1 million last year in compensation expense related to a deferred purchase price obligation in connection with the acquisition. Earnings also included \$13.1 million this year and \$15.8 million last year related to accruals for a contingent bonus payment for Schuh employees provided for in the Schuh acquisition. The decreases in deferred purchase price expense and contingent bonus expense this year were offset by decreased gross margin and negative leverage from the negative comparable sales. The decreased gross margin reflected increased promotional activity and changes in product mix. See Note 2 to the Consolidated Financial Statements for additional information related to the Schuh acquisition.

Lids Sports Group

	Fiscal Year Ended				0/		
		2014		2013	% Change		
		(dollars i	n thousan	ds)			
Net sales	\$	820,996	\$	791,255	3.8 %		
Earnings from operations	\$	63,748	\$	82,867	(23.1)%		
Operating margin		7.8%		10.5%			

Net sales from the Lids Sports Group increased 3.8% to \$821.0 million for Fiscal 2014 from \$791.3 million for Fiscal 2013. The increase primarily reflects a 6% increase in average Lids Sports Group stores operated. Same store sales decreased 1%, comparable direct sales increased 26% and comparable sales, including both store and direct sales, were flat for Fiscal 2014. The comparable sales decrease reflected a 2% decrease in comparable store hat units sold while average price per hat was flat. Lids Sports Group operated 1,133 stores at the end of Fiscal 2014, including 110 Lids stores in Canada, 177 Lids Locker Room and Clubhouse stores and 26 Locker Room by Lids leased departments at Macy's, compared to 1,053 stores at the end of Fiscal 2013, including 98 Lids stores in Canada and 144 Lids Locker Room and Clubhouse stores.

Lids Sports Group earnings from operations for Fiscal 2014 decreased 23.1% to \$63.7 million compared to \$82.9 million for Fiscal 2013. The decrease in operating income was primarily due to decreased gross margin as a percentage of net sales, reflecting increased promotional activity and changes in product mix, and to increased expenses as a percentage of net sales, primarily reflecting negative leverage due to negative same store sales.

Johnston & Murphy Group

	 Fiscal Year Ended			0/	
	2014		2013	% Change	
	(dollars i	in thousan	ds)		
Net sales	\$ 245,941	\$	221,860	10.9%	
Earnings from operations	\$ 17,638	\$	15,696	12.4%	
Operating margin	7.2%)	7.1%		

Johnston & Murphy Group net sales increased 10.9% to \$245.9 million for Fiscal 2014 from \$221.9 million for Fiscal 2013. The increase reflected primarily a 7% increase in same store sales, a 15% increase in comparable direct sales and an 8% increase in comparable sales, including both store and direct sales, an 8% increase in Johnston & Murphy wholesale sales, a 5% increase in average stores operated for Johnston & Murphy retail operations and additional sales for the recently relaunched Trask brand. Unit sales for the Johnston & Murphy wholesale business increased 8% in Fiscal 2014, while the average price per pair of shoes was flat for the same period. Retail operations accounted for 71.9% of the Johnston & Murphy Group's sales in Fiscal 2014, up from 71.7% in Fiscal 2013. The comparable sales increase in Fiscal 2014 reflects a 5% increase in the average price per pair of shoes for Johnston & Murphy retail operations, primarily associated with increased sales of higher-priced dress shoes and increased prices in the casual lines and footwear unit comparable sales increased 3%. The store count for Johnston & Murphy retail operations at the end of Fiscal 2014 included 168 Johnston & Murphy shops and factory stores, including seven stores in Canada, compared to 157 Johnston & Murphy shops and factory stores, including five stores in Canada, at the end of Fiscal 2013.

Johnston & Murphy earnings from operations for Fiscal 2014 increased 12.4% to \$17.6 million from \$15.7 million for Fiscal 2013, primarily due to increased net sales and increased gross margin as a percentage of net sales, reflecting increased initial margins, offset by increased expenses as a percentage of net sales, due primarily to expenses associated with the relaunch of the Trask brand.

Licensed Brands

	 Fiscal Year Ended			
	 2014 2013		2013	% Change
	(dollars i	n thous	ands)	
Net sales	\$ 109,780	\$	108,498	1.2%
Earnings from operations	\$ 10,614	\$	10,078	5.3%
Operating margin	9.7%	,	9.3%	

Licensed Brands' net sales increased 1.2% to \$109.8 million for Fiscal 2014 from \$108.5 million for Fiscal 2013. The sales increase reflects an increase in sales of SureGrip Footwear and Dockers Footwear partially offset by decreased sales of the Chaps line of footwear. Unit sales for Dockers Footwear decreased 1% for Fiscal 2014, while the average price per pair of shoes increased 2% for the same period.

Licensed Brands' earnings from operations for Fiscal 2014 increased 5.3%, from \$10.1 million for Fiscal 2013 to \$10.6 million, primarily due to increased net sales and decreased expenses as a percentage of net sales, reflecting decreased bonus accruals.

Corporate, Interest Expenses and Other Charges

Corporate and other expense for Fiscal 2014 was \$29.0 million compared to \$59.9 million for Fiscal 2013. Corporate expense in Fiscal 2014 included \$1.3 million in asset impairment and other charges, primarily for network intrusion expenses, retail store asset impairments, other legal matters and a lease termination, partially offset by a gain on another lease termination. Corporate expense in Fiscal 2013 included \$17.0 million in asset impairment and other charges, retail store asset impairments and other charges, primarily for network intrusion expenses in Fiscal 2013 included \$17.0 million in asset impairment and other charges, primarily for network intrusion expenses, retail store asset impairments and other legal matters. Excluding the charges listed above, corporate and other expense decreased primarily due to decreased bonus accruals.

Interest expense decreased 9.5% from \$5.1 million in Fiscal 2013 to \$4.6 million in Fiscal 2014 primarily due to lower interest on the U.K. debt resulting from payments of the U.K. debt during Fiscal 2014 and Fiscal 2013.

Results of Operations—Fiscal 2013 Compared to Fiscal 2012

The Company's net sales for Fiscal 2013 (53 weeks) increased 13.6% to \$2.60 billion from \$2.29 billion in Fiscal 2012 (52 weeks). The increase in net sales was a result of the inclusion of Schuh Group for the full year in Fiscal 2013, an estimated \$35.2 million impact of sales for the fifty-third week and an increase in comparable sales in the Journeys Group, Schuh Group and Johnston & Murphy Group, combined with increased sales in Licensed Brands, offset by decreased comparable sales in Lids Sports Group. Gross margin increased 13.1% to \$1.30 billion in Fiscal 2013 from \$1.15 billion in Fiscal 2012, but decreased as a percentage of net sales from 50.1% to 49.9%, primarily reflecting decreased gross margin as a percentage of net sales in the Lids Sports and Johnston & Murphy Groups, offset slightly by increased gross margin as a percentage of net sales in the Lides Sports and Johnston & Murphy Groups, offset slightly by increased gross margin as a percentage of net sales in the Journeys while Licensed Brands' gross margin was flat. Selling and administrative expenses in Fiscal 2013 increased 13.0% from Fiscal 2012 but decreased as a percentage of net sales in the Johnston & Murphy and Journeys Groups due to positive comparable sales in the Journeys and Johnston & Murphy Groups and increased wholesale sales in the Johnston & Murphy Group. The Company records buying and merchandising and occupancy costs in selling and administrative expense. Because the Company does not include these costs in cost of sales, the Company's gross margin may not be comparable to other retailers that include these costs in the calculation of gross margin. Explanations of the changes in results of operations are provided by business segment in discussions following these introductory paragraphs.

Pretax earnings for Fiscal 2013 were \$164.8 million, compared to \$156.4 million for Fiscal 2012. Pretax earnings for Fiscal 2013 included asset impairment and other charges of \$17.0 million, including \$15.5 million for expenses related to the computer network intrusion announced in December 2010, \$1.4 million for retail store asset impairments and \$0.1 million for other legal matters. Pretax earnings for Fiscal 2012 included asset impairment and other charges of \$2.7 million, including \$1.1 million for retail store asset impairments, \$0.9 million for other legal matters and \$0.7 million for expenses related to the computer network intrusion.

Net earnings for Fiscal 2013 were \$112.4 million (\$4.68 diluted earnings per share) compared to \$92.4 million (\$3.79 diluted earnings per share) for Fiscal 2012. Net earnings for Fiscal 2013 includes \$0.5 million (\$0.01 diluted loss per share) charge to earnings (net of tax), primarily for anticipated costs of environmental remedial alternatives related to former

facilities operated by the Company. Net earnings for Fiscal 2012 includes \$1.0 million (\$0.04 diluted loss per share) charge to earnings (net of tax), including \$1.1 million primarily for anticipated costs of environmental remedial alternatives related to former facilities operated by the Company, offset by a \$0.1 million gain for excess provisions to prior discontinued operations. The Company recorded an effective federal income tax rate of 31.5% for Fiscal 2013 compared to 40.2% for Fiscal 2012. This year's lower effective tax rate of 31.5% reflects the reversal of charges previously recorded related to uncertain tax positions due to the expiration of the applicable statutes of limitations and a settlement with a state tax authority more favorable than anticipated related to other uncertain tax positions. See Note 9 to the Consolidated Financial Statements for additional information.

Journeys Group

	 Fiscal Year Ended			
	 2013 2012		2012	% Change
	(dollars i	n thousa	nds)	
Net sales	\$ 1,111,490	\$	1,020,116	9.0%
Earnings from operations	\$ 109,953	\$	89,045	23.5%
Operating margin	9.9%		8.7%	

Net sales from Journeys Group increased 9.0% to \$1.11 billion for Fiscal 2013 from \$1.02 billion for Fiscal 2012. The increase reflects primarily a 6% increase in same store sales, an 8% increase in comparable direct sales and a 6% increase in comparable sales, including both store and direct sales. The comparable store sales increase reflected a 6% increase in the average price per pair of shoes, offset by a 1% decrease in footwear unit comparable sales. Total unit sales increased 2% during the same period. The store count for Journeys Group was 1,157 stores at the end of Fiscal 2013, including 156 Journeys Kidz stores, 51 Shi by Journeys stores, 130 Underground by Journeys stores and 24 Journeys stores in Canada, compared to 1,154 stores at the end of Fiscal 2012, including 152 Journeys Kidz stores, 53 Shi by Journeys stores, 137 Underground by Journeys stores and 13 Journeys stores in Canada.

Journeys Group earnings from operations for Fiscal 2013 increased 23.5% to \$110.0 million, compared to \$89.0 million for Fiscal 2012. The increase in earnings from operations was primarily due to increased net sales, increased gross margin as a percentage of net sales, reflecting lower markdowns, and to decreased expenses as a percentage of net sales, reflecting leverage of store related costs, including occupancy costs and depreciation.

Schuh Group

	Fiscal Year Ended				0/	
		2013		2012	% Change	
		(dollars in thousands)				
Net sales	\$	370,480	\$	212,262	74.5%	
Earnings from operations	\$	11,209	\$	11,711	(4.3)%	
Operating margin		3.0%		5.5%		

Net sales from the Schuh Group increased 74.5% to \$370.5 million for Fiscal 2013, compared to \$212.3 million for Fiscal 2012. Net sales for Schuh Group in Fiscal 2012 included sales only for the seven months after the Company acquired Schuh on June 23, 2011. The sales increase also reflects a 7% increase in same store sales, a 13% increase in comparable direct sales and an 8% increase in comparable sales, including both store and direct sales for the seven months ended February 2, 2013. Schuh Group operated 79 stores, including three Schuh Kids stores, and 13 concessions at the end of Fiscal 2012 compared to 64 stores and 14 concessions at the end of Fiscal 2012.

Schuh Group earnings from operations were \$11.2 million for Fiscal 2013 compared to \$11.7 million for Fiscal 2012. The earnings included \$12.1 million this year and \$7.2 million last year in compensation expense related to a deferred purchase price obligation in connection with the acquisition. The earnings also included \$15.8 million this year and \$4.9 million last year related to accruals for a contingent bonus payment for Schuh employees provided for in the Schuh acquisition. The decrease in earnings is due to the increase in expense associated with both the deferred purchase price and contingent bonus payment. See Note 2 to the Consolidated Financial Statements for additional information related to the Schuh acquisition.

Lids Sports Group

	 Fiscal	0/		
	 2013 2		2012	% Change
	(dollars i	n thous	ands)	
Net sales	\$ 791,255	\$	759,324	4.2 %
Earnings from operations	\$ 82,867	\$	86,037	(3.7)%
Operating margin	10.5%		11.3%	

Net sales from the Lids Sports Group increased 4.2% to \$791.3 million for Fiscal 2013 from \$759.3 million for Fiscal 2012. The increase primarily reflects a 3% increase in average Lids Sports Group stores operated. Same store sales decreased 4%, comparable direct sales increased 9% and comparable sales, including both store and direct sales, decreased 3% for Fiscal 2013. The comparable sales decrease reflected a 1% decrease in average price per hat and a 1% decrease in comparable store hat units sold. The comparable sales decrease reflects the current popularity of adjustable "snap-back" hat styles, which have displaced some demand for fitted merchandise. Management believes that the relative ease of merchandising non-fitted hats has enabled a variety of non-headwear retailers to carry the adjustable styles, increasing competition in the category. Lids Sports Group operated 1,053 stores at the end of Fiscal 2013, including 98 Lids stores in Canada and 144 Lids Locker Room and Clubhouse stores, compared to 1,002 stores at the end of Fiscal 2012, including 82 Lids stores in Canada and 120 Lids Locker Room and Clubhouse stores.

Lids Sports Group earnings from operations for Fiscal 2013 decreased 3.7% to \$82.9 million compared to \$86.0 million for Fiscal 2012. The decrease in operating income was primarily due to increased expenses as a percentage of net sales, primarily reflecting negative leverage due to negative comparable sales and lower gross margin due to lowered prices in the snap-back category.

Johnston & Murphy Group

	Fiscal Year Ended			
	2013 2012		2012	% Change
	(dollars i	n thousar	nds)	
Net sales	\$ 221,860	\$	201,725	10.0%
Earnings from operations	\$ 15,696	\$	13,743	14.2%
Operating margin	7.1%		6.8%	

Johnston & Murphy Group net sales increased 10.0% to \$221.9 million for Fiscal 2013 from \$201.7 million for Fiscal 2012. The increase reflected primarily a 3% increase in same store sales, a 13% increase in comparable direct sales and a 4% increase in comparable sales, including both store and direct sales, and a 21% increase in Johnston & Murphy wholesale sales slightly offset by a 1% decrease in average stores operated for Johnston & Murphy retail operations. Unit sales for the Johnston & Murphy wholesale business increased 25% in Fiscal 2013, while the average price per pair of shoes decreased 3% for the same period. Retail operations accounted for 71.7% of the Johnston & Murphy Group's sales in Fiscal 2013, down from 74.3% in Fiscal 2012. The comparable sales increase in Fiscal 2013 reflects a 4% increase in the average price per pair of shoes for Johnston & Murphy retail operations, primarily associated with increased sales of higher-priced dress shoes, while footwear unit comparable sales were flat. The store count for Johnston & Murphy retail operations at the end of Fiscal 2013 included 157 Johnston & Murphy shops and factory stores, including five stores in Canada, compared to 153 Johnston & Murphy shops and factory stores, including five stores in Canada, compared to 153 Johnston & Murphy shops and factory stores, including one store in Canada, at the end of Fiscal 2012.

Johnston & Murphy earnings from operations for Fiscal 2013 increased 14.2% to \$15.7 million from \$13.7 million for Fiscal 2012, primarily due to increased net sales.

Licensed Brands

	 Fiscal Year Ended				
	 2013 2012		2012	% Change	
	(dollars i	sands)			
Net sales	\$ 108,498	\$	97,444	11.3%	
Earnings from operations	\$ 10,078	\$	9,451	6.6%	
Operating margin	9.3%)	9.7%		

Licensed Brands' net sales increased 11.3% to \$108.5 million for Fiscal 2013 from \$97.4 million for Fiscal 2012. The sales increase reflects \$5.6 million of increased sales from Dockers Footwear as well as increased sales of SureGrip Footwear and the Chaps line of footwear. Unit sales for Dockers Footwear increased 4% for Fiscal 2013 and the average price per pair of shoes increased 3% for the same period.

Licensed Brands' earnings from operations for Fiscal 2013 increased 6.6%, from \$9.5 million for Fiscal 2012 to \$10.1 million, primarily due to increased net sales, partially offset by higher bonus accruals.

Corporate, Interest Expenses and Other Charges

Corporate and other expense for Fiscal 2013 was \$59.9 million compared to \$48.5 million for Fiscal 2012. Corporate expense in Fiscal 2013 included \$17.0 million in asset impairment and other charges, primarily for network intrusion expenses, retail store asset impairments and other legal matters. Corporate expense in Fiscal 2012 included \$2.7 million in asset impairment and other charges, primarily for retail store asset impairments, other legal matters and network intrusion expenses and \$7.4 million in acquisition related expenses. Excluding the charges listed above, corporate and other expense increased primarily due to increases in bonus accruals, professional fees, restricted stock expense and salary expense.

Interest expense decreased 0.6% from \$5.2 million in Fiscal 2012 to \$5.1 million in Fiscal 2013.

Liquidity and Capital Resources

The following table sets forth certain financial data at the dates indicated.

	_	Feb. 1, 2014	Feb. 2, 2013		Jan. 28, 2012
			(dollars in millions)		
Cash and cash equivalents	9	59.4	\$ 59.8	8 \$	53.8
Working capital	9	451.3	\$ 407.1	. \$	292.0
Long-term debt (includes current maturities)	9	33.7	\$ 50.7	7 \$	40.7

Working Capital

The Company's business is seasonal, with the Company's investment in inventory and accounts receivable normally reaching peaks in the spring and fall of each year. Historically, cash flow from operations has been generated principally in the fourth quarter of each fiscal year.

Cash provided by operating activities was \$140.0 million in Fiscal 2014 compared to \$123.2 million in Fiscal 2013. The \$16.8 million increase from operating activities from Fiscal 2013 reflects an increase in cash flow from changes in accounts payable of \$37.8 million, partially offset by decreased earnings of \$19.8 million. The \$37.8 million increase in cash flow from accounts payable reflects changes in buying patterns and payment terms negotiated with individual vendors.

The \$58.4 million increase in inventories at February 1, 2014 from February 2, 2013 levels reflects increases in retail inventory, reflecting slower than expected sales and a 6.4% increase in square footage, and increased wholesale inventory in Licensed Brands and Lids Team Sports.

Accounts receivable at February 1, 2014 increased \$3.7 million compared to February 2, 2013, due primarily to increased wholesale sales in the Licensed Brands business and increased tenant allowance and other receivables in retail.

Cash provided by operating activities was \$123.2 million in Fiscal 2013 compared to \$145.0 million in Fiscal 2012. The \$21.8 million decrease from operating activities from Fiscal 2012 reflects a decrease in cash flow from changes in accounts receivable, inventory, accounts payable, prepaids and other current assets and other accrued liabilities of \$8.8 million, \$18.7

million, \$16.8 million, \$9.8 million and \$16.0 million, respectively, partially offset by improved earnings and a \$43.8 million increase from changes in other assets and liabilities. The \$8.8 million decrease in cash flow from accounts receivable was due to increased footwear wholesale sales. The \$18.7 million decrease in cash flow from prepaids and other current assets was due to changes in prepaid income taxes. The \$16.8 million decrease in cash flow from accounts payable reflects changes in buying patterns and payment terms negotiated with individual vendors. The \$16.0 million decrease in cash flow from other accrued liabilities reflects decreased bonus accruals and decreased income tax accruals in Fiscal 2013 compared to Fiscal 2012. The \$43.8 million increase in cash flow from other assets and liabilities reflects increases and liabilities reflects increased accruals for the deferred purchase price and bonus earn-out related to Schuh and an increase in the bonus bank liability and a decrease in long-term receivables related to the network intrusion.

The \$61.0 million increase in inventories at February 2, 2013 from January 28, 2012 levels reflects increases in retail inventory, reflecting slower than expected sales and a 5.5% increase in square footage, and increased inventory in Licensed Brands to support growth initiatives.

Accounts receivable at February 2, 2013 increased \$5.8 million compared to January 28, 2012, due primarily to increased wholesale sales reflecting growth in the Johnston & Murphy wholesale business and Licensed Brands business.

Sources of Liquidity

The Company has three principal sources of liquidity: cash from operations, cash and cash equivalents on hand and the Credit Facility discussed below. The Company believes that cash and cash equivalents on hand, cash from operations and availability under its Credit Facility will be sufficient to cover its working capital and capital expenditures for the foreseeable future.

On January 31, 2014, the Company entered into a Third Amended and Restated Credit Agreement (the "Credit Facility") with the lenders party thereto and Bank of America, N.A., as agent, providing for a revolving credit facility in the aggregate principal amount of \$400.0 million, including a \$70.0 million sublimit for the issuance of letters of credit and a domestic swingline subfacility of up to \$40.0 million, a revolving credit subfacility for the benefit of GCO Canada, Inc. in an aggregate amount not to exceed \$25.0 million, which includes a \$5.0 million sublimit for the issuance of letters of credit, and revolving credit subfacility for the benefit of Genesco (UK) Limited in an aggregate amount not to exceed \$50.0 million, which includes a \$10.0 million sublimit for the issuance of letters of credit and a swingline subfacility of up to \$10.0 million. The facility has a five-year term. Any swingline loans and any letters of credit and borrowings under the Canadian facilities will reduce the availability under the Credit Facility on a dollar-for-dollar basis.

The Company has the option, from time to time, to increase the availability under the Credit Facility by an aggregate amount of up to \$150.0 million subject to, among other things, the receipt of commitments for the increased amount. In connection with this increased facility, the Canadian revolving credit facility may be increased up to no more than \$40.0 million.

Genesco (UK) Limited has a one-time option to increase the availability of its subfacility under the Credit Facility by an additional amount of up to \$50.0 million.

The aggregate amount of the loans made and letters of credit issued under the Credit Facility shall at no time exceed the lesser of the facility amount (\$400.0 million or, if increased as described above, up to \$550.0 million or \$600.0 million, respectively) or the "Borrowing Base", which generally is based on 90% of eligible inventory plus 85% of eligible wholesale receivables (50% of eligible wholesale receivables of the Lids Team Sports business) plus 90% of eligible credit card and debit card receivables less applicable reserves (the "Loan Cap"). The relevant assets of Genesco (UK) Limited will be included in the Borrowing Base if the additional \$50.0 million sublimit increase is exercised, provided that amounts borrowed by Genesco (UK) Limited based solely on its own borrowing base will be limited to \$50.0 million and the total outstanding to Genesco (UK) Limited will not exceed 30% of the Loan Cap.

The Credit Facility also provides that a first-in, last-out tranche could be added to the revolving credit facility at the option of the Company subject to, among other things, the receipt of commitments for such tranche. For additional information on the Company's Credit Facility, see Note 6 to the Consolidated Financial Statements included in Item 8, Financial Statements and Supplementary Data.

In connection with the Schuh acquisition, Schuh entered into an amended and restated Senior Term Facilities Agreement and Working Capital Facility Letter (collectively, the "UK Credit Facilities"), which provide for term loans of up to £29.5 million (a £15.5 million A term loan and £14.0 million B term loan) and a working capital facility of £5.0 million. The Working Capital Facility Letter was allowed to lapse in June 2012. The A term loan bears interest at LIBOR plus 2.50% per annum. The B term loan bears interest at LIBOR plus 3.75% per annum. The Company is not required to make any payments on the B term loan until it expires October 31, 2015, unless the Company's Schuh Group segment has Excess Cash Flow

Table of Contents

(as defined in the UK Credit Facilities). The Company paid less than £0.1 million, £4.8 million and £4.5 million on the B term loan in Fiscal 2014, 2013 and 2012, respectively.

In November 2013, Schuh Group Limited entered into an Amended and Restated Facilities Agreement to provide for an additional term loan of up to £12.5 million ("C term loan"). The C term loan bears interest at LIBOR plus 2.50% per annum.

There were \$33.7 million in UK term loans outstanding at February 1, 2014. The UK Credit Facilities contains certain covenants at the Schuh level including a minimum interest coverage covenant initially set at 4.25x and increasing to 4.50x in January 2012 and thereafter, a maximum leverage covenant initially set at 2.75x declining over time at various rates to 2.25x beginning in July 2012 and a minimum cash flow coverage of 1.10x. The Company was in compliance with all the covenants at February 1, 2014. The UK Credit Facilities are secured by a pledge of all the assets of Schuh and its subsidiaries.

Revolving credit borrowings averaged \$38.5 million during Fiscal 2014 and \$30.5 million during Fiscal 2013, as cash on hand, cash generated from operations and revolver borrowings primarily funded seasonal working capital requirements, capital expenditures and stock repurchases for Fiscal 2014.

There were \$14.5 million of letters of credit outstanding and no revolver borrowings outstanding under the Credit Facility at February 1, 2014. The Company is not required to comply with any financial covenants under the Credit Facility unless Excess Availability (as defined in the Credit Agreement) is less than the greater of \$25.0 million or 10.0% of the Loan Cap. If and during such time as Excess Availability is less than the greater of \$25.0 million or 10.0% of the Loan Cap, the Credit Facility requires the Company to meet a minimum fixed charge coverage ratio of (a) an amount equal to consolidated EBITDA less capital expenditures and taxes paid in cash, in each case for such period, to (b) fixed charges for such period, of not less than 1.0:1.0. Excess Availability was \$358.0 million at February 1, 2014. Because Excess Availability exceeded \$25.0 million or 10.0% of the Loan Cap, the Company was not required to comply with this financial covenant at February 1, 2014.

The Company's Credit Facility prohibits the payment of dividends and other restricted payments unless as of the date of the making of any Restricted Payment (as defined in the Credit Facility) or consummation of any Acquisition (as defined in the Credit Facility), (a) no Default (as defined in the Credit Facility) or Event of Default (as defined in the Credit Facility)exists or would arise after giving effect to such Restricted Payment or Acquisition, and (b) either (i) the Borrowers (as defined in the Credit Facility) have pro forma projected Excess Availability for the following six month period equal to or greater than 25% of the Loan Cap, after giving pro forma effect to such Restricted Payment or Acquisition, or (ii) (A) the Borrowers have pro forma projected Excess Availability for the following six month period of less than 25% of the Loan Cap but equal to or greater than 15% of the Loan Cap, after giving pro forma effect to the Restricted Payment or Acquisition, and (B) the Fixed Charge Coverage Ratio (as defined in the Credit Facility), on a pro-forma basis for the twelve months preceding such Restricted Payment or Acquisition, will be equal to or greater than 1.0:1.0 and (c) after giving effect to such Restricted Payment or Acquisition, the Borrowers are Solvent (as defined in the Credit Facility). The Company's management does not expect availability under the Credit Facility to fall below the requirements listed above during Fiscal 2015. The Company's UK Credit Facilities prohibit the payment of any dividends by Schuh or its subsidiaries to the Company.

The Company issued a mandatory notice of redemption effective April 30, 2013, to its holders of Subordinated Serial Preferred Stock \$2.30 Series 1, \$4.75 Series 3 and \$4.75 Series 4, and on its \$1.50 Subordinated Cumulative Preferred Stock during the first quarter of Fiscal 2014. The total cost of the redemption was \$1.5 million. As a result, all of these preferred issues of stock were either converted to common stock or redeemed in the first quarter of Fiscal 2014 and there are no outstanding shares remaining. Therefore, there is no longer an annual dividend requirement.



Contractual Obligations

The following tables set forth aggregate contractual obligations and commitments as of February 1, 2014.

(in thousands)	 Payments Due by Period								
Contractual Obligations	Total		Less than 1 year		1 - 3 years		3 - 5 years		More than 5 years
Capital Lease Obligations	\$ 11	\$	1	\$	3	\$	3	\$	4
Long-Term Debt Obligations ⁽¹⁾	33,730		6,793		16,439		4,117		6,381
Operating Lease Obligations	1,306,479		235,049		405,584		284,021		381,825
Purchase Obligations ⁽²⁾	621,533		621,533				_		_
Long-Term Obligations – Schuh ⁽³⁾	104,844		25,800		78,368		553		123
Other Long-Term Liabilities	1,254		176		351		351		376
Total Contractual Obligations ⁽⁴⁾	\$ 2,067,851	\$	889,352	\$	500,745	\$	289,045	\$	388,709

(in	thousands)
-----	------------

(in thousands)		Amount	or Com	mitment Expiration	Per Pe	riod	 	
Commercial Commitments	l Amounts ommitted	Less than 1 year		1 - 3 years		3 - 5 years	More than 5 years	
Letters of Credit	\$ 14,466	\$ 14,466	\$	_	\$	—	\$	_
Total Commercial Commitments	\$ 14,466	\$ 14,466	\$		\$	_	\$	_

ant of Committee ant Equivation Dev Devia

(1) Excludes interest on revolver borrowings due to uncertainty of timing of payments.

(2) Open purchase orders for inventory.

(3) Includes deferred purchase price payments and earn-out bonus payments related to the Schuh acquisition and interest on the UK term loans. For additional information, see Notes 2 and 6 to the Consolidated Financial Statements included in Item 8.

(4) Excludes unrecognized tax benefits of \$10.7 million due to their uncertain nature in timing of payments, if any.

Capital Expenditures

Capital expenditures were \$98.5 million, \$71.7 million and \$49.5 million for Fiscal 2014, 2013 and 2012, respectively. The \$26.8 million increase in Fiscal 2014 capital expenditures as compared to Fiscal 2013 reflected an increase in retail store capital expenditures due to the construction of 183 new stores and leased departments opened in Fiscal 2014, compared to 104 stores in Fiscal 2013. The \$22.2 million increase in Fiscal 2013 capital expenditures as compared to Fiscal 2012 reflected an increase in retail store capital expenditures due to the construction of 104 new stores opened in Fiscal 2013, compared to 70 stores in Fiscal 2012, and increased major renovations due to lease renewals.

Total capital expenditures in Fiscal 2015 are expected to be approximately \$149 million. These include retail capital expenditures of approximately \$134 million to open approximately 25 Journeys stores, including 5 in Canada, 25 Journeys Kidz stores, 15 Schuh stores, including three Schuh Kids, 11 Johnston & Murphy shops and factory stores, and 268 Lids Sports Group stores and leased departments, including 45 Lids stores, with 15 stores in Canada, 48 Lids Locker Room and Lids Clubhouse stores, and 175 Locker Room by Lids leased departments in Macy's department stores, and to complete approximately 164 major store renovations. The planned amount of capital expenditures in Fiscal 2015 for wholesale operations and other purposes is approximately \$15 million, including approximately \$8 million for new systems to improve customer service and support the Company's growth.

Future Capital Needs

The Company expects that cash on hand and cash provided by operations and borrowings under its Credit Facility will be sufficient to support seasonal working capital and capital expenditure requirements during Fiscal 2015. The approximately \$7.3 million of costs associated with discontinued operations that are expected to be paid during the next twelve months are expected to be funded from cash on hand, cash generated from operations and borrowings under the Credit Facility during Fiscal 2015.

The Company had total available cash and cash equivalents of \$59.4 million and \$59.8 million as of February 1, 2014 and February 2, 2013, respectively, of which approximately \$39.4 million and \$38.5 million was held by the Company's foreign subsidiaries as of February 1, 2014 and February 2, 2013, respectively. The Company's strategic plan does not require the repatriation of foreign cash in order to fund its operations in the U.S., and it is the Company's current intention to permanently reinvest its foreign cash and cash equivalents outside of the U.S. If the Company were to repatriate foreign cash to the U.S., it would be required to accrue and pay U.S. taxes in accordance with applicable U.S. tax rules and regulations as a result of the repatriation.

Common Stock Repurchases

The Company repurchased 337,665 shares of common stock at a cost of \$20.7 million during Fiscal 2014. During the third quarter of Fiscal 2014, the Company's board of directors increased the share repurchase authorization to \$75.0 million. Shares repurchased during the third quarter of Fiscal 2014, at a cost of \$9.5 million, will be applied to the new repurchase authorization. Therefore, the Company has \$65.5 million remaining under its current \$75.0 million share repurchase authorization. The Company has \$65.5 million during Fiscal 2013. The Company did not repurchase any shares during Fiscal 2012.

Environmental and Other Contingencies

The Company is subject to certain loss contingencies related to environmental proceedings and other legal matters, including those disclosed in Note 13 to the Company's Consolidated Financial Statements. The Company has made pretax accruals for certain of these contingencies, including approximately \$0.5 million reflected in Fiscal 2014, \$0.8 million reflected in Fiscal 2013 and \$1.8 million reflected in Fiscal 2012. These charges are included in provision for discontinued operations, net in the Consolidated Statements of Operations because they relate to former facilities operated by the Company. The Company monitors these matters on an ongoing basis and, on a quarterly basis, management reviews the Company's reserves and accruals in relation to each of them, adjusting provisions as management deems necessary in view of changes in available information. Changes in estimates of liability are reported in the periods when they occur. Consequently, management believes that its reserve in relation to each proceeding is a best estimate of the probable loss connected to the proceeding, or in cases in which no best estimate is possible, the minimum amount in the range of estimated losses, based upon its analysis of the facts and circumstances as of the close of the most recent fiscal quarter. However, because of uncertainties and risks inherent in litigation generally and in environmental proceedings in particular, there can be no assurance that future developments will not require additional reserves, that some or all reserves may not be adequate or that the amounts of any such additional reserves or any such inadequacy will not have a material adverse effect upon the Company's financial condition or results of operations.

Financial Market Risk

The following discusses the Company's exposure to financial market risk related to changes in interest rates.

Outstanding Debt of the Company – The Company has \$33.7 million of outstanding U.K. term loans at a weighted average interest rate of 3.40% as of February 1, 2014. A 100 basis point adverse change in interest rates would increase interest expense by \$0.3 million on the \$33.7 million term loans.

Cash and Cash Equivalents – The Company's cash and cash equivalent balances are invested in financial instruments with original maturities of three months or less. The Company did not have significant exposure to changing interest rates on invested cash at February 1, 2014. As a result, the Company considers the interest rate market risk implicit in these investments at February 1, 2014 to be low.

Accounts Receivable – The Company's accounts receivable balance at February 1, 2014 is concentrated in two of its footwear wholesale businesses, which sell primarily to department stores and independent retailers across the United States and its Lids Team Sports wholesale businesses, which sells primarily to colleges and high school athletic teams and their fan bases. Including both footwear wholesale and Lids Team Sports wholesale businesses, three customers each accounted for 5% and no other customer accounted for more than 4% of the Company's total trade receivables balance as of February 1, 2014. The Company monitors the credit quality of its customers and establishes an allowance for doubtful accounts based upon factors surrounding credit risk of specific customers, historical trends and other information, as well as customer specific factors; however, credit risk is affected by conditions or occurrences within the economy and the retail industry, as well as company-specific information.

Summary – Based on the Company's overall market interest rate exposure at February 1, 2014, the Company believes that the effect, if any, of reasonably possible near-term changes in interest rates on the Company's consolidated financial position, results of operations or cash flows for Fiscal 2015 would not be material.

New Accounting Principles

In February 2013, the Financial Accounting Standards Board ("FASB") issued Accounting Standards Update ("ASU") No. 2013-02, "Reporting of Amounts Reclassified Out of Accumulated Other Comprehensive Income" ("AOCI"), which sets forth additional disclosure requirements for items reclassified out of AOCI and into net income, and is effective for annual and interim reporting periods beginning after December 15, 2012. The Company adopted ASU No. 2013-02 in the first quarter of Fiscal 2014 by presenting amounts reclassified out of AOCI as a separate disclosure in Note 1 to the Consolidated Financial Statements. Amounts reclassified out of AOCI were related to amortization of net actuarial loss associated with the Company's pension and postretirement plans.

Inflation

The Company does not believe inflation has had a material impact on sales or operating results during periods covered in this discussion.

ITEM 7A, QUANTITATIVE AND QUALITATIVE DISCLOSURES ABOUT MARKET RISK

The Company incorporates by reference the information regarding market risk appearing under the heading "Financial Market Risk" in Item 7, Management's Discussion and Analysis of Financial Condition and Results of Operations.

Table of Contents

ITEM 8, FINANCIAL STATEMENTS AND SUPPLEMENTARY DATA

INDEX TO FINANCIAL STATEMENTS

	Page
Report of Independent Registered Public Accounting Firm on Internal Control over Financial Reporting	<u>41</u>
Report of Independent Registered Public Accounting Firm on Consolidated Financial Statements	<u>42</u>
Consolidated Balance Sheets, February 1, 2014 and February 2, 2013	<u>43</u>
Consolidated Statements of Operations, each of the three fiscal years ended 2014, 2013 and 2012	<u>45</u>
Consolidated Statements of Comprehensive Income, each of the three fiscal years ended 2014, 2013 and 2012	<u>46</u>
Consolidated Statements of Cash Flows, each of the three fiscal years ended 2014, 2013 and 2012	<u>47</u>
Consolidated Statements of Equity, each of the three fiscal years ended 2014, 2013 and 2012	<u>48</u>
Notes to Consolidated Financial Statements	<u>49</u>

Report of Independent Registered Public Accounting Firm On Internal Control over Financial Reporting

The Board of Directors and Shareholders Genesco Inc.

We have audited Genesco Inc. and Subsidiaries' internal control over financial reporting as of February 1, 2014, based on criteria established in Internal Control—Integrated Framework issued by the Committee of Sponsoring Organizations of the Treadway Commission (1992 Framework) (the COSO criteria). Genesco Inc. and Subsidiaries' management is responsible for maintaining effective internal control over financial reporting, and for its assessment of the effectiveness of internal control over financial reporting included in the accompanying Management's Report on Internal Control over Financial Reporting. Our responsibility is to express an opinion on the company's internal control over financial reporting based on our audit.

We conducted our audit in accordance with the standards of the Public Company Accounting Oversight Board (United States). Those standards require that we plan and perform the audit to obtain reasonable assurance about whether effective internal control over financial reporting was maintained in all material respects. Our audit included obtaining an understanding of internal control over financial reporting, assessing the risk that a material weakness exists, testing and evaluating the design and operating effectiveness of internal control based on the assessed risk, and performing such other procedures as we considered necessary in the circumstances. We believe that our audit provides a reasonable basis for our opinion.

A company's internal control over financial reporting is a process designed to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles. A company's internal control over financial reporting includes those policies and procedures that (1) pertain to the maintenance of records that, in reasonable detail, accurately and fairly reflect the transactions and dispositions of the assets of the company; (2) provide reasonable assurance that transactions are recorded as necessary to permit preparation of financial statements in accordance with generally accepted accounting principles, and that receipts and expenditures of the company are being made only in accordance with authorizations of management and directors of the company; and (3) provide reasonable assurance regarding prevention or timely detection of unauthorized acquisition, use, or disposition of the company's assets that could have a material effect on the financial statements.

Because of its inherent limitations, internal control over financial reporting may not prevent or detect misstatements. Also, projections of any evaluation of effectiveness to future periods are subject to the risk that controls may become inadequate because of changes in conditions, or that the degree of compliance with the policies or procedures may deteriorate.

In our opinion, Genesco Inc. and Subsidiaries maintained, in all material respects, effective internal control over financial reporting as of February 1, 2014, based on the COSO criteria.

We also have audited, in accordance with the standards of the Public Company Accounting Oversight Board (United States), the consolidated balance sheets of Genesco Inc. and Subsidiaries as of February 1, 2014 and February 2, 2013, and the related consolidated statements of operations, comprehensive income, cash flows, and equity for each of the three fiscal years in the period ended February 1, 2014, and our report dated April 2, 2014 expressed an unqualified opinion thereon. Our audits also included the financial statement schedule listed in the Index at Item 15.

/s/ Ernst & Young LLP

Nashville, Tennessee April 2, 2014

Report of Independent Registered Public Accounting Firm on Consolidated Financial Statements

The Board of Directors and Shareholders Genesco Inc.

We have audited the accompanying consolidated balance sheets of Genesco Inc. and Subsidiaries (the "Company") as of February 1, 2014 and February 2, 2013, and the related consolidated statements of operations, comprehensive income, cash flows and equity for each of the three fiscal years in the period ended February 1, 2014. Our audits also included the financial statement schedule listed in the Index at Item 15. These financial statements and schedule are the responsibility of the Company's management. Our responsibility is to express an opinion on these financial statements and schedule based on our audits.

We conducted our audits in accordance with the standards of the Public Company Accounting Oversight Board (United States). Those standards require that we plan and perform the audit to obtain reasonable assurance about whether the financial statements are free of material misstatement. An audit includes examining, on a test basis, evidence supporting the amounts and disclosures in the financial statements. An audit also includes assessing the accounting principles used and significant estimates made by management, as well as evaluating the overall financial statement presentation. We believe that our audits provide a reasonable basis for our opinion.

In our opinion, the financial statements referred to above present fairly, in all material respects, the consolidated financial position of Genesco Inc. and Subsidiaries at February 1, 2014 and February 2, 2013, and the consolidated results of their operations and their cash flows for each of the three fiscal years in the period ended February 1, 2014, in conformity with U.S. generally accepted accounting principles. Also, in our opinion, the related financial statement schedule, when considered in relation to the basic financial statements taken as a whole, presents fairly in all material respects the information set forth therein.

We also have audited, in accordance with the standards of the Public Company Accounting Oversight Board (United States), the Company's internal control over financial reporting as of February 1, 2014, based on criteria established in Internal Control – Integrated Framework issued by the Committee of Sponsoring Organizations of the Treadway Commission (1992 Framework), and our report dated April 2, 2014 expressed an unqualified opinion thereon.

/s/ Ernst & Young LLP

Nashville, Tennessee April 2, 2014

Genesco Inc. and Subsidiaries Consolidated Balance Sheets In Thousands, except share amounts

		As of Fiscal Year End					
Assets	Feb	February 2, 2013					
Current Assets:							
Cash and cash equivalents	\$	59,447	\$	59,795			
Accounts receivable, net of allowances of \$4,420 at February 1,							
2014 and \$6,082 at February 2, 2013		52,646		48,214			
Inventories		567,261		505,344			
Deferred income taxes		23,089		23,725			
Prepaids and other current assets		54,432		45,193			
Total current assets		756,875		682,271			
Property and equipment:							
Land		6,169		6,128			
Buildings and building equipment		20,474		20,390			
Computer hardware, software and equipment		131,110		120,757			
Furniture and fixtures		173,992		148,903			
Construction in progress		35,623		8,702			
Improvements to leased property		335,287		318,376			
Property and equipment, at cost		702,655		623,256			
Accumulated depreciation		(422,618)		(381,587)			
Property and equipment, net		280,037		241,669			
Deferred income taxes		3,342		18,731			
Goodwill		288,100		273,827			
Trademarks, net of accumulated amortization of \$4,312 at							
February 1, 2014 and \$3,350 at February 2, 2013		77,571		77,408			
Other intangibles, net of accumulated amortization of \$20,645 at							
February 1, 2014 and \$17,220 at February 2, 2013		9,082		11,598			
Other noncurrent assets		24,277		20,568			
Total Assets	\$	1,439,284	\$	1,326,072			

Genesco Inc. and Subsidiaries Consolidated Balance Sheets In Thousands, except share amounts

	As of Fise	As of Fiscal Year End						
Liabilities and Equity	February 1, 2014	February 2, 2013						
Current Liabilities:								
Accounts payable	\$ 145,483	\$ 118,350						
Accrued employee compensation	49,078	55,241						
Accrued other taxes	26,247	25,985						
Accrued income taxes	2,188	2,096						
Current portion – long-term debt	6,793	5,675						
Other accrued liabilities	68,526	60,659						
Provision for discontinued operations	7,263	7,192						
Total current liabilities	305,578	275,198						
Long-term debt	26,937	45,007						
Pension liability	9,223	20,514						
Deferred rent and other long-term liabilities	175,311	157,407						
Provision for discontinued operations	4,112	4,159						
Total liabilities	521,161	502,285						
Commitments and contingent liabilities								
Equity								
Non-redeemable preferred stock	1,305	3,924						
Common equity:								
Common stock, \$1 par value:								
Authorized: 80,000,000 shares								
Issued/Outstanding:								
February 1, 2014 – 24,407,724/23,919,260								
February 2, 2013 – 24,484,915/23,996,451	24,408	24,485						
Additional paid-in capital	190,568	170,360						
Retained earnings	734,533	669,189						
Accumulated other comprehensive loss	(16,767)	(28,241						
Treasury shares, at cost (488,464 shares)	(17,857)	(17,857						
Total Genesco equity	916,190	821,860						
Noncontrolling interest – non-redeemable	1,933	1,927						
Total equity	918,123	823,787						
Total Liabilities and Equity	\$ 1,439,284	\$ 1,326,072						

The accompanying Notes are an integral part of these Consolidated Financial Statements

Genesco Inc. and Subsidiaries Consolidated Statements of Operations In Thousands, except per share amounts

	Fiscal Year					
		2014	2013	2012		
Net sales	\$	2,624,972 \$	2,604,817 \$	2,291,987		
Cost of sales		1,325,922	1,306,200	1,143,632		
Selling and administrative expenses		1,134,274	1,111,717	984,193		
Asset impairments and other, net		1,341	17,037	2,677		
Earnings from operations		163,435	169,863	161,485		
Interest expense, net:						
Interest expense		4,641	5,126	5,157		
Interest income		(66)	(95)	(65)		
Total interest expense, net		4,575	5,031	5,092		
Earnings from continuing operations before income taxes		158,860	164,832	156,393		
Income tax expense		65,878	51,935	62,942		
Earnings from continuing operations		92,982	112,897	93,451		
Provision for discontinued operations, net		(329)	(462)	(1,025)		
Net Earnings	\$	92,653 \$	112,435 \$	92,426		
Basic earnings per common share:						
Continuing operations	\$	3.99 \$	4.78 \$	3.89		
Discontinued operations		(0.01)	(0.02)	(0.05)		
Net earnings	\$	3.98 \$	4.76 \$	3.84		
Diluted earnings per common share:						
Continuing operations	\$	3.94 \$	4.69 \$	3.83		
Discontinued operations		(0.02)	(0.01)	(0.04)		
Net earnings	\$	3.92 \$	4.68 \$	3.79		

The accompanying Notes are an integral part of these Consolidated Financial Statements.

Genesco Inc. and Subsidiaries Consolidated Statements of Comprehensive Income In Thousands, except as noted

	Fiscal Year				
		2014	2013	2012	
Net earnings	\$	92,653 \$	112,435 \$	92,426	
Other comprehensive income (loss):					
Gain (loss) on foreign currency forward contract,					
net of tax of \$0.0 million for each period		—	42	(35)	
Pension liability adjustment net of tax of \$6.2 million					
and \$2.4 million for 2014 and 2013, respectively, and					
net of tax benefit of \$3.1 million for 2012		9,510	3,657	(4,670)	
Postretirement liability adjustment net of tax benefit of					
\$0.3 million for 2014 and \$0.1 million for 2013 and 2012		(542)	(79)	(109)	
Foreign currency translation adjustments		2,506	1,105	(3,847)	
Total other comprehensive income (loss)		11,474	4,725	(8,661)	
Comprehensive Income	\$	104,127 \$	117,160 \$	83,765	

The accompanying Notes are an integral part of these Consolidated Financial Statements.

Genesco Inc. and Subsidiaries Consolidated Statements of Cash Flows In Thousands

		Fi	scal Year	
		2014	2013	2012
CASH FLOWS FROM OPERATING ACTIVITIES:				
Net earnings	\$	92,653 \$	112,435 \$	92,426
Adjustments to reconcile net earnings to net cash				
provided by operating activities:				
Depreciation and amortization		67,135	63,697	53,737
Amortization of deferred note expense and debt discount		801	792	708
Deferred income taxes		14,983	(17,618)	9,661
Provision for losses on accounts receivable		(525)	1,325	2,004
Impairment of long-lived assets		2,347	1,396	1,119
Restricted stock and share-based compensation		12,295	10,508	7,660
Provision for discontinued operations		543	796	1,692
Tax benefit of stock options and restricted stock exercised		(3,784)	(4,820)	(4,744)
Other		1,301	1,327	1,005
Effect on cash from changes in working capital and other				
assets and liabilities, before acquisitions:				
Accounts receivable		(3,684)	(5,821)	3,011
Inventories		(58,386)	(61,049)	(42,324)
Prepaids and other current assets		(8,885)	(4,524)	5,286
Accounts payable		19,850	(17,953)	(1,201)
Other accrued liabilities		(10,093)	(6,624)	9,384
Other assets and liabilities		13,448	49,343	5,536
Net cash provided by operating activities		139,999	123,210	144,960
CASH FLOWS FROM INVESTING ACTIVITIES:				
Capital expenditures		(98,456)	(71,737)	(49,456)
Acquisitions, net of cash acquired		(13,567)	(23,818)	(92,985)
Proceeds from asset sales		75	81	27
Net cash used in investing activities		(111,948)	(95,474)	(142,414)
CASH FLOWS FROM FINANCING ACTIVITIES:				
Payments of capital leases		(2)	(2)	(22)
Payments of long-term debt		(6,428)	(13,581)	(25,321)
Proceeds from issuance of long-term debt		15,124	_	—
Borrowings under revolving credit facility		402,200	439,600	299,800
Payments on revolving credit facility		(429,900)	(416,900)	(294,800)
Tax benefit of stock options and restricted stock exercised		3,784	4,820	4,744
Shares repurchased		(20,676)	(37,650)	_
Change in overdraft balances		6,025	(2,925)	2,931
Redemption of preferred shares		(1,462)	_	
Dividends paid on non-redeemable preferred stock		(33)	(147)	(193)
Exercise of stock options and issue shares - Employee Stock Purchase Plan		3,230	4,965	9,820
Other		(1,788)	4	(939)
Net cash used in financing activities		(29,926)	(21,816)	(3,980)
Effect of foreign exchange rate fluctuations on cash		1,527	85	(710)
Net (Decrease) Increase in Cash and Cash Equivalents		(348)	6,005	(2,144)
Cash and cash equivalents at beginning of period		59,795	53,790	55,934
Cash and cash equivalents at end of period	\$	59,447 \$	59,795 \$	53,790
Supplemental Cash Flow Information:				, -
Net cash paid for:				
Interest	\$	3,769 \$	4,391 \$	4,789
Income taxes	*	52,618	81,607	50,254
he accompanying Notes are an integral part of these Consolidated Financial Statements.		5_,510	51,507	20,204

The accompanying Notes are an integral part of these Consolidated Financial Statements.

Genesco Inc. and Subsidiaries Consolidated Statements of Equity

In Thousands	Total Non- Redeemable Preferred Stock	Common Stock	Additional Paid-In Capital	Retained Earnings	Accumulated Other Comprehensive Loss	Treasury Shares	Non Controlling Interest Non- Redeemable	Total Equity
Balance January 29, 2011	\$ 5,183	\$ 24,163	\$ 131,910	\$ 506,127	\$ (24,305)	\$ (17,857)	\$ 2,503	\$ 627,724
Net earnings	—	_	_	92,426	_	_	_	92,426
Other comprehensive loss	—	_	_	_	(8,661)	_	_	(8,661)
Dividends paid on non-redeemable preferred stock	_		_	(193)	_	_	_	(193)
Exercise of stock options	_	390	9,297	(155)	_			9,687
Issue shares – Employee Stock Purchase								
Plan Employee and non-employee restricted	_	3	130	_	_	_	_	133
stock	_	_	7,659	_	_	_	_	7,659
Share-based compensation	_	—	1	_	_	_	_	1
Restricted stock issuance	_	304	(304)	_	_	_	_	_
Restricted shares withheld for taxes	_	(93)	(4,034)	_	_	_	_	(4,127)
Tax benefit of stock options and								
restricted stock exercises	—	—	4,585	—	_	—	—	4,585
Shares repurchased	_	_	_	_	_	_	_	_
Other	(226)	(9)	235	—	_	—	—	_
Noncontrolling interest – loss					_		(254)	(254)
Balance January 28, 2012	4,957	24,758	149,479	598,360	(32,966)	(17,857)	2,249	728,980
Net earnings	—	—	_	112,435	—	_	_	112,435
Other comprehensive income	—	_	_	_	4,725	_	_	4,725
Dividends paid on non-redeemable preferred stock	_		_	(147)	_	_		(147)
Exercise of stock options	_	224	4,584	(1.7) 	_	_		4,808
Issue shares – Employee Stock Purchase Plan	_	2	155		_	_		157
Employee and non-employee restricted stock	_	_	10,508	_	_	_	_	10,508
Restricted stock issuance	_	194	(194)	_	_	_	_	
Restricted shares withheld for taxes	_	(76)	(15.)	(4,455)	_	_		(4,531)
Tax benefit of stock options and		()		(1,122)				(1,000)
restricted stock exercises	_	_	4,820	_	_	_	_	4,820
Shares repurchased		(646)	.,	(37,004)	_	_		(37,650)
Other	(1,033)	29	1,008	(07,001)	_	_	_	4
Noncontrolling interest – loss	(1,000)			_	_	_	(322)	(322)
Balance February 2, 2013	3,924	24,485	170,360	669,189	(28,241)	(17,857)	1,927	823,787
	5,524	24,403	170,300	92,653	(20,241)	(17,057)	1,527	
Net earnings Other comprehensive income			—	92,055	11,474	—	_	92,653 11,474
Dividends paid on non-redeemable			—		11,474	—	_	
preferred stock	—	—	—	(33)	_	_	—	(33)
Exercise of stock options	_	130	2,904	_	—	_	_	3,034
Issue shares – Employee Stock Purchase Plan	_	3	193	_	_	_	—	196
Employee and non-employee restricted stock	_		12,295	_	_	_	_	12,295
Restricted stock issuance		214	(214)					
Restricted shares withheld for taxes		(105)	105	(6,938)				(6,938)
Tax benefit of stock options and		(103)	103	(0,550)				(0,000)
restricted stock exercises	_	_	3,784	_	_	_	_	3,784
Shares repurchased	—	(338)	5,704	(20,338)	—	—	—	(20,676)
Redemption of preferred shares	(1,462)	(330)	_	(20,000)		_	-	(1,462)
Other	(1,402)		1,141	_	_	_	_	(1,402)
Noncontrolling interest – income	(1,137)	19	1,141	—	—	—	6	5 6
inconcontroning interest - income		_	_				0	U

The accompanying Notes are an integral part of these Consolidated Financial Statements.

Note 1 <u>Summary of Significant Accounting Policies</u>

Nature of Operations

The Company's business includes the design and sourcing, marketing and distribution of footwear and accessories through retail stores in the U.S., Puerto Rico and Canada primarily under the Journeys, Journeys Kidz, Shi by Journeys, Underground by Journeys and Johnston & Murphy banners and under the Schuh banner in the United Kingdom and the Republic of Ireland; through e-commerce websites including journeys.com, journeyskidz.com, shibyjourneys.com, schuh.co.uk, johnstonmurphy.com and trask.com and catalogs, and at wholesale, primarily under the Company's Johnston & Murphy brand, the recently relaunched Trask brand, the licensed Dockers brand and other brands that the Company licenses for footwear, and the Company's SureGrip[®] line of slip-resistant, occupational footwear. The Company's business also includes Lids Sports Group, which operates headwear and accessory stores in the U.S. and Canada primarily under the Lids, Hat World and Hat Shack banners; the Lids Locker Room and Lids Clubhouse businesses, consisting of sports-oriented fan shops featuring a broad array of licensed merchandise such as apparel, hats and accessories, sports decor and novelty products, operating under various trade names; licensed team merchandise departments in Macy's department stores operated under the name of Locker Room by Lids and on macys.com, under a license agreement with Macy's; certain e-commerce operations including lids.com, lids.ca, lidslockerroom.com and lidsclubhouse.com; and an athletic team dealer business operating as Lids Team Sports. Including both the footwear businesses and the Lids Sports Group business, at February 1, 2014, the Company operated 2,568 retail stores in the U.S., Puerto Rico, Canada, the United Kingdom and the Republic of Ireland.

During Fiscal 2014, the Company operated five reportable business segments (not including corporate): (i) Journeys Group, comprised of the Journeys, Journeys Kidz, Shi by Journeys and Underground by Journeys retail footwear chains, catalog and e-commerce operations; (ii) Schuh Group, comprised of the Schuh retail footwear chain and e-commerce operations; (iii) Lids Sports Group, comprised as described in the preceding paragraph; (iv) Johnston & Murphy Group, comprised of Johnston & Murphy retail operations, catalog and e-commerce operations and wholesale distribution of products under the Johnston & Murphy and Trask brands; and (v) Licensed Brands, comprised of Dockers[®] Footwear, sourced and marketed under a license from Levi Strauss & Company; SureGrip[®]Footwear, occupational footwear primarily sold directly to consumers; and other brands.

Principles of Consolidation

All subsidiaries are consolidated in the consolidated financial statements. All significant intercompany transactions and accounts have been eliminated.

Fiscal Year

The Company's fiscal year ends on the Saturday closest to January 31. As a result, Fiscal 2014 was a 52-week year with 364 days, Fiscal 2013 was a 53-week year with 371 days and Fiscal 2012 was a 52-week year with 364 days. Fiscal 2014 ended on February 1, 2014, Fiscal 2013 ended on February 2, 2013 and Fiscal 2012 ended on January 28, 2012.

Note 1 <u>Summary of Significant Accounting Policies, Continued</u>

Use of Estimates

The preparation of financial statements in conformity with U.S. generally accepted accounting principles requires management to make estimates and assumptions that affect the reported amounts of assets and liabilities and disclosure of contingent assets and liabilities at the date of the financial statements and the reported amounts of revenues and expenses during the reporting period. Actual results could differ from those estimates.

Significant areas requiring management estimates or judgments include the following key financial areas:

Inventory Valuation

The Company values its inventories at the lower of cost or market.

In its footwear wholesale operations, its Schuh Group segment and its Lids Sports Group wholesale operations, except for the Anaconda Sports wholesale division, cost is determined using the FIFO method. Market value is determined using a system of analysis which evaluates inventory at the stock number level based on factors such as inventory turn, average selling price, inventory level, and selling prices reflected in future orders. The Company provides reserves when the inventory has not been marked down to market value based on current selling prices or when the inventory is not turning and is not expected to turn at levels satisfactory to the Company.

The Lids Sports Group retail segment and its Anaconda Sports wholesale division employ the moving average cost method for valuing inventories and apply freight using an allocation method. The Company provides a valuation allowance for slow-moving inventory based on negative margins and estimated shrink based on historical experience and specific analysis, where appropriate.

In its retail operations, other than the Schuh Group and Lids Sports Group retail segments, the Company employs the retail inventory method, applying average cost-to-retail ratios to the retail value of inventories. Under the retail inventory method, valuing inventory at the lower of cost or market is achieved as markdowns are taken or accrued as a reduction of the retail value of inventories.

Inherent in the retail inventory method are subjective judgments and estimates, including merchandise mark-on, markups, markdowns, and shrinkage. These judgments and estimates, coupled with the

fact that the retail inventory method is an averaging process, could produce a range of cost figures. To reduce the risk of inaccuracy and to ensure consistent presentation, the Company employs the retail inventory method in multiple subclasses of inventory with similar gross margins, and analyzes markdown requirements at the stock number level based on factors such as inventory turn, average selling price, and inventory age. In addition, the Company accrues markdowns as necessary. These additional markdown accruals reflect all of the above factors as well as current agreements to return

products to vendors and vendor agreements to provide markdown support. In addition to markdown provisions, the Company maintains provisions for shrinkage and damaged goods based on historical rates.

Note 1 <u>Summary of Significant Accounting Policies, Continued</u>

Inherent in the analysis of both wholesale and retail inventory valuation are subjective judgments about current market conditions, fashion trends, and overall economic conditions. Failure to make appropriate conclusions regarding these factors may result in an overstatement or understatement of inventory value.

Impairment of Long-Lived Assets

The Company periodically assesses the realizability of its long-lived assets, other than goodwill, and evaluates such assets for impairment whenever events or changes in circumstances indicate that the carrying amount of an asset may not be recoverable. Asset impairment is determined to exist if estimated future cash flows, undiscounted and without interest charges, are less than the carrying amount. Inherent in the analysis of impairment are subjective judgments about future cash flows. Failure to make appropriate conclusions regarding these judgments may result in an overstatement or understatement of the value of long-lived assets. See also Notes 3 and 5.

The goodwill impairment test involves performing a qualitative assessment, on a reporting unit level, based on current circumstances. If the results of the qualitative assessment indicate that it is more likely than not that the fair value of a reporting unit is greater than its carrying amount, a two-step impairment test will not be performed. However, if the results of the qualitative assessment indicate that it is more likely than not that the fair value of a reporting unit is less than its carrying amount, then a two-step impairment test is performed. Alternatively, the Company may elect to bypass the qualitative assessment and proceed directly to the two-step impairment test, on a reporting unit level. The first step is a comparison of the fair value and carrying value of the business unit with which the goodwill is associated. The Company estimates fair value using the best information available, and computes the fair value derived by an income approach utilizing discounted cash flow projections. The income approach uses a projection of a reporting unit's estimated operating results and cash flows that is discounted using a weighted-average cost of capital that reflects current market conditions. A key assumption in the Company's fair value estimate is the weighted average cost of capital utilized for discounting its cash flow projections in its income approach. The Company believes the rate it used in its latest annual test, which was completed in the fourth quarter, was consistent with the risks inherent in its business and with industry discount rates. The projection uses management's best estimates of economic and market conditions over the projected period including growth rates in sales, costs, estimates of future expected changes in operating margins and cash expenditures.

Other significant estimates and assumptions include terminal value growth rates, future estimates of capital expenditures and changes in future working capital requirements.

If the carrying value of the reporting unit is higher than its fair value, there is an indication that impairment may exist and the second step must be performed to measure the amount of impairment loss. The amount of impairment is determined by comparing the implied fair value of reporting

unit goodwill to the carrying value of the goodwill in the same manner as if the reporting unit was being acquired in a business combination.

Note 1 <u>Summary of Significant Accounting Policies, Continued</u>

Specifically, the Company would allocate the fair value of the reporting unit to all of the assets and liabilities of the reporting unit, including any unrecognized intangible assets, in a hypothetical analysis that would calculate the implied fair value of goodwill. If the implied fair value of goodwill is less than the recorded goodwill, the Company would record an impairment charge for the difference.

Environmental and Other Contingencies

The Company is subject to certain loss contingencies related to environmental proceedings and other legal matters. The Company has made pretax accruals for certain of these contingencies, including approximately \$0.5 million in Fiscal 2014, \$0.8 million in Fiscal 2013 and \$1.8 million in Fiscal 2012. These charges are included in provision for discontinued operations, net in the Consolidated Statements of Operations because they relate to former facilities operated by the Company. The Company monitors these matters on an ongoing basis and, on a quarterly basis, management reviews the Company's reserves and accruals, adjusting provisions as management deems necessary in view of changes in available information. Changes in estimates of liability are reported in the periods when they occur. Consequently, management believes that its reserve in relation to each proceeding is a best estimate of probable loss connected to the proceeding, or in cases in which no best estimate is possible, the minimum amount in the range of estimated losses, based upon its analysis of the facts and circumstances as of the close of the most recent fiscal quarter. However, because of uncertainties and risks inherent in litigation generally and in environmental proceedings in particular, there can be no assurance that future developments will not require additional reserves, that some or all reserves will be adequate or that the amounts of any such additional reserves or any such inadequacy will not have a material adverse effect upon the Company's financial condition, cash flows, or results of operations. See also Notes 3 and 13.

Revenue Recognition

Retail sales are recorded at the point of sale and are net of estimated returns and exclude sales and value added taxes. Catalog and internet sales are recorded at estimated time of delivery to the customer and are net of estimated returns and exclude sales and value added taxes. Wholesale revenue is recorded net of estimated returns and allowances for markdowns, damages and miscellaneous claims when the related goods have been shipped and legal title has passed to the customer. Shipping and handling costs charged to customers are included in net sales. Estimated returns are based on historical returns and claims. Actual amounts of markdowns have not differed materially from estimates. Actual returns and claims in any future period may differ from historical experience.

Income Taxes

As part of the process of preparing the Consolidated Financial Statements, the Company is required to estimate its income taxes in each of the tax jurisdictions in which it operates. This process involves estimating actual current tax obligations together with assessing temporary differences resulting from differing treatment of certain items for tax and accounting purposes, such as depreciation of property and equipment and valuation of inventories. These temporary differences result in deferred tax assets and liabilities, which are included within the Consolidated Balance Sheets. The Company then assesses the likelihood that its deferred tax assets will be recovered from future taxable income or other sources. Actual results could differ from this assessment if adequate taxable income is not

Note 1 <u>Summary of Significant Accounting Policies, Continued</u>

generated in future periods. To the extent the Company believes that recovery of an asset is at risk, valuation allowances are established or increased in a period, the Company includes an expense within the tax provision in the Consolidated Statements of Operations. These deferred tax valuation allowances may be released in future years when management considers that it is more likely than not that some portion or all of the deferred tax assets will be realized. In making such a determination, management will need to periodically evaluate whether or not all available evidence, such as future taxable income and reversal of temporary differences, tax planning strategies, and recent results of operations, provides sufficient positive evidence to offset any potential negative evidence that may exist at such time. In the event the deferred tax valuation allowance is released, the Company would record an income tax benefit for the portion or all of the deferred tax valuation allowance of \$3.8 million.

Income tax reserves for uncertain tax positions are determined using the methodology required by the Income Tax Topic of the Accounting Standards Codification ("Codification"). This methodology requires companies to assess each income tax position taken using a two step process. A determination is first made as to whether it is more likely than not that the position will be sustained, based upon the technical merits, upon examination by the taxing authorities. If the tax position is expected to meet the more likely than not criteria, the benefit recorded for the tax position equals the largest amount that is greater than 50% likely to be realized upon ultimate settlement of the respective tax position. Uncertain tax positions require determinations and estimated liabilities to be made based on provisions of the tax law which may be subject to change or varying interpretation. If the Company's determinations and estimates prove to be inaccurate, the resulting adjustments could be material to its future financial results.

The Company recorded an effective income tax rate of 41.5% for Fiscal 2014 compared to 31.5% for Fiscal 2013 and 40.2% for Fiscal 2012. The tax rate for Fiscal 2013 was lower compared to Fiscal 2014 and Fiscal 2012 primarily due to the reversal of previously recorded charges related to uncertain tax positions due to the expiration of the applicable statutes of limitations and a settlement with a state tax authority more favorable than anticipated related to other uncertain tax positions.

Postretirement Benefits Plan Accounting

Full-time employees who had at least 1,000 hours of service in calendar year 2004, except employees in the Lids Sports Group and Schuh Group segments, are covered by a defined benefit pension plan. The Company froze the defined benefit pension plan effective January 1, 2005. The Company also provides certain former employees with limited medical and life insurance benefits. The Company funds at least the minimum amount required by the Employee Retirement Income Security Act.

Note 1 <u>Summary of Significant Accounting Policies, Continued</u>

As required by the Compensation – Retirement Benefits Topic of the Codification, the Company is required to recognize the overfunded or underfunded status of postretirement benefit plans as an asset or liability, respectively, in their Consolidated Balance Sheets and to recognize changes in that funded status in accumulated other comprehensive loss, net of tax, in the year in which the changes occur.

The Company accounts for the defined benefit pension plans using the Compensation-Retirement Benefits Topic of the Codification. As permitted under this topic, pension expense is recognized on an accrual basis over employees' approximate service periods. The calculation of pension expense

and the corresponding liability requires the use of a number of critical assumptions, including the expected long-term rate of return on plan assets and the assumed discount rate, as well as the recognition of actuarial gains and losses. Changes in these assumptions can result in different expense and liability amounts, and future actual experience can differ from these assumptions.

Cash and Cash Equivalents

The Company had total available cash and cash equivalents of \$59.4 million and \$59.8 million as of February 1, 2014 and February 2, 2013, respectively. Included in cash and cash equivalents at February 1, 2014 and February 2, 2013 are cash equivalents of \$0.0 million and \$0.2 million, respectively. Cash equivalents are highly-liquid financial instruments having an original maturity of three months or less.

At February 1, 2014, substantially all of the Company's domestic cash was invested in deposit accounts at FDIC-insured banks. The majority of payments due from banks for domestic customer credit card transactions process within 24 - 48 hours and are accordingly classified as cash and cash equivalents in the Consolidated Balance Sheets.

At February 1, 2014 and February 2, 2013, outstanding checks drawn on zero-balance accounts at certain domestic banks exceeded book cash balances at those banks by approximately \$42.1 million and \$36.1 million, respectively. These amounts are included in accounts payable in the Consolidated Balance Sheets.

Concentration of Credit Risk and Allowances on Accounts Receivable

The Company's footwear wholesale businesses sell primarily to independent retailers and department stores across the United States. Receivables arising from these sales are not collateralized. Customer credit risk is affected by conditions or occurrences within the economy and the retail industry as well as by customer specific factors. The Company's Lids Team Sports wholesale business sells primarily to colleges and high school athletic teams and their fan bases. Including both footwear wholesale and Lids Team Sports wholesale businesses, three customers each accounted for 5% of the Company's total trade receivables balance, while no other customer accounted for more than 4% of the Company's total trade receivables balance as of February 1, 2014.

The Company establishes an allowance for doubtful accounts based upon factors surrounding the credit risk of specific customers, historical trends and other information, as well as customer specific factors. The Company also establishes allowances for sales returns, customer deductions and co-op advertising based on specific circumstances, historical trends and projected probable outcomes.

Note 1 <u>Summary of Significant Accounting Policies, Continued</u>

Property and Equipment

Property and equipment are recorded at cost and depreciated or amortized over the estimated useful life of related assets. Depreciation and amortization expense are computed principally by the straight-line method over the following estimated useful lives:

Buildings and building equipment	20-45 years
Computer hardware, software and equipment	3-10 years
Furniture and fixtures	10 years

Leases

Leasehold improvements and properties under capital leases are amortized on the straight-line method over the shorter of their useful lives or their related lease terms and the charge to earnings is included in selling and administrative expenses in the Consolidated Statements of Operations.

Certain leases include rent increases during the initial lease term. For these leases, the Company recognizes the related rental expense on a straight-line basis over the term of the lease (which

includes any rent holidays and the pre-opening period of construction, renovation, fixturing and merchandise placement) and records the difference between the amounts charged to operations and amounts paid as deferred rent.

The Company occasionally receives reimbursements from landlords to be used towards construction of the store the Company intends to lease. Leasehold improvements are recorded at their gross costs including items reimbursed by landlords. The reimbursements are amortized as a reduction of rent expense over the initial lease term.

Goodwill and Other Intangibles

Under the provisions of the Intangibles – Goodwill and Other Topic of the Codification, goodwill and intangible assets with indefinite lives are not amortized, but are tested at least annually, during the fourth quarter, for impairment. The Company will update the tests between annual tests if events or

circumstances occur that would more likely than not reduce the fair value of the business unit with which the goodwill is associated below its carrying amount. It is also required that intangible assets with finite lives be amortized over their respective lives to their estimated residual values, and reviewed for impairment in accordance with the Property, Plant and Equipment Topic of the Codification.

Intangible assets of the Company with indefinite lives are primarily goodwill and identifiable trademarks, net of amortization, acquired in connection with the acquisition of Schuh Group Ltd. in

June 2011 and Hat World Corporation in April 2004. The Consolidated Balance Sheets include goodwill of \$182.4 million for the Lids Sports Group, \$104.9 million for the Schuh Group and \$0.8 million for Licensed Brands at February 1, 2014, and \$172.3 million for the Lids Sports Group, \$100.7 million for the Schuh Group and \$0.8 million for Licensed Brands at February 2, 2013. The Company tests for impairment of intangible assets with an indefinite life, relying on a number of factors including operating results, business plans, projected future cash flows and observable market data. The impairment test for identifiable assets not subject to amortization consists of a comparison of the fair value of the

Note 1 <u>Summary of Significant Accounting Policies, Continued</u>

intangible asset with its carrying amount. The Company has not recorded an impairment charge for intangible assets.

In connection with acquisitions, the Company records goodwill on its Consolidated Financial Statements. This asset is not amortized but is subject to an impairment test at least annually, based on projected future cash flows from the acquired business discounted at a rate commensurate with the risk the Company considers to be inherent in its current business model. The Company performs the impairment test annually as of the close of its fiscal year, or more frequently if events or circumstances indicate that the value of the asset might be impaired.

As a result of the various acquisitions comprising the Lids Team Sports team dealer business, the Company carries goodwill at a value of \$14.2 million on its Consolidated Balance Sheets related to such acquisitions. The Company found that the result of its annual impairment test, which valued the business at approximately \$3.9 million in excess of its carrying value, indicated no impairment at that time. The Company may determine in future impairment tests that some or all of the carrying value of the goodwill may not be recoverable. Such a finding would require a write-off of the amount of the carrying value that is impaired, which would reduce the Company noted that an increase in the weighted average cost of capital of 100 basis points would reduce the fair value of the Lids Team Sports business by \$5.9 million. Furthermore, the Company noted that a decrease in projected annual revenue by one percent would reduce the fair value of the Lids Team Sports business by \$0.4 million. However, if other assumptions do not remain constant, the fair value of the Lids Team Sports business may decrease by a greater amount.

Identifiable intangible assets of the Company with finite lives are trademarks, customer lists, in-place leases, non-compete agreements and a vendor contract. They are subject to amortization based upon

their estimated useful lives. Finite-lived intangible assets are evaluated for impairment using a process similar to that used to evaluate other definite-lived long-lived assets, a comparison of the fair value of the intangible asset with its carrying amount. An impairment loss is recognized for the amount

by which the carrying value exceeds the fair value of the asset.

Fair Value of Financial Instruments

The carrying amounts and fair values of the Company's financial instruments at February 1, 2014 and February 2, 2013 are:

In thousands	February 1, 2014				February 2, 2013			
		arrying mount	Fair Value		Carrying Amount	Fair Value		
U.S. Revolver Borrowings	\$	_ \$		\$	27,700	\$	27,742	
UK Term Loans		33,730	33,840		22,982		22,982	

Note 1 <u>Summary of Significant Accounting Policies, Continued</u>

Debt fair values were determined using a discounted cash flow analysis based on current market interest rates for similar types of financial instruments and would be classified in Level 2 as defined in Note 5.

Carrying amounts reported on the Consolidated Balance Sheets for cash, cash equivalents, receivables and accounts payable approximate fair value due to the short-term maturity of these instruments.

Cost of Sales

For the Company's retail operations, the cost of sales includes actual product cost, the cost of transportation to the Company's warehouses from suppliers and the cost of transportation from the Company's warehouses to the stores. Additionally, the cost of its distribution facilities allocated to its retail operations is included in cost of sales.

For the Company's wholesale operations, the cost of sales includes the actual product cost and the cost of transportation to the Company's warehouses from suppliers.

Selling and Administrative Expenses

Selling and administrative expenses include all operating costs of the Company excluding (i) those related to the transportation of products from the supplier to the warehouse, (ii) for its retail operations, those related to the transportation of products from the warehouse to the store and (iii) costs of its distribution facilities which are allocated to its retail operations. Wholesale and unallocated retail costs of distribution are included in selling and administrative expenses in the amounts of \$8.7 million, \$8.2 million and \$9.2 million for Fiscal 2014, 2013 and 2012, respectively.

Gift Cards

The Company has a gift card program that began in calendar 1999 for its Lids Sports operations and calendar 2000 for its footwear operations. The gift cards issued to date do not expire. As such, the Company recognizes income when: (i) the gift card is redeemed by the customer; or (ii) the likelihood of the gift card being redeemed by the customer for the purchase of goods in the future is remote and there are no related escheat laws (referred to as "breakage"). The gift card breakage rate is based upon historical redemption patterns and income is recognized for unredeemed gift cards in proportion to those historical redemption patterns.

Gift card breakage is recognized in revenues each period. Gift card breakage recognized as revenue was \$0.8 million, \$0.7 million and \$0.6 million for Fiscal 2014, 2013 and 2012, respectively. The Consolidated Balance Sheets include an accrued liability for gift cards of \$14.4 million and \$13.1 million at February 1, 2014 and February 2, 2013, respectively.

Buying, Merchandising and Occupancy Costs

The Company records buying, merchandising and occupancy costs in selling and administrative expense. Because the Company does not include these costs in cost of sales, the Company's gross margin may not be comparable to other retailers that include these costs in the calculation of gross margin. Retail occupancy costs recorded in selling and administrative expense were \$381.6 million, \$359.3 million and \$314.6 million for Fiscal 2014, 2013 and 2012, respectively.

Note 1 <u>Summary of Significant Accounting Policies, Continued</u>

Shipping and Handling Costs

Shipping and handling costs related to inventory purchased from suppliers are included in the cost of inventory and are charged to cost of sales in the period that the inventory is sold. All other shipping and handling costs are charged to cost of sales in the period incurred except for wholesale and unallocated retail costs of distribution, which are included in selling and administrative expenses on the Consolidated Statements of Operations.

Preopening Costs

Costs associated with the opening of new stores are expensed as incurred, and are included in selling and administrative expenses on the Consolidated Statements of Operations.

Store Closings and Exit Costs

From time to time, the Company makes strategic decisions to close stores or exit locations or activities. If stores or operating activities to be closed or exited constitute components, as defined by the Property, Plant and Equipment Topic of the Codification, and will not result in a migration of customers and cash flows, these closures will be considered discontinued operations when the related assets meet the criteria to be classified as held for sale, or at the cease-use date, whichever occurs first. The results of operations of discontinued operations are presented retroactively, net of tax, as a separate component on the Consolidated Statements of Operations, if material individually or cumulatively. In each of the years presented, no store closings meeting the discontinued operations criteria have been material individually or cumulatively.

Assets related to planned store closures or other exit activities are reflected as assets held for sale and recorded at the lower of carrying value or fair value less costs to sell when the required criteria, as

defined by the Property, Plant and Equipment Topic of the Codification, are satisfied. Depreciation ceases on the date that the held for sale criteria are met.

Assets related to planned store closures or other exit activities that do not meet the criteria to be classified as held for sale are evaluated for impairment in accordance with the Company's normal impairment policy, but with consideration given to revised estimates of future cash flows. In any event, the remaining depreciable useful lives are evaluated and adjusted as necessary.

Exit costs related to anticipated lease termination costs, severance benefits and other expected charges are accrued for and recognized in accordance with the Exit or Disposal Cost Obligations Topic of the Codification.

Advertising Costs

Advertising costs are predominantly expensed as incurred. Advertising costs were \$56.9 million, \$48.3 million and \$42.5 million for Fiscal 2014, 2013 and 2012, respectively. Direct response advertising costs for catalogs are capitalized in accordance with the Other Assets and Deferred Costs Topic for Capitalized Advertising Costs of the Codification. Such costs are amortized over the estimated future period as revenues are realized from such advertising, not to exceed six months.

Note 1 <u>Summary of Significant Accounting Policies, Continued</u>

The Consolidated Balance Sheets include prepaid assets for direct response advertising costs of \$2.3 million and \$1.4 million at February 1, 2014 and February 2, 2013.

Consideration to Resellers

In its wholesale businesses, the Company does not have any written buy-down programs with retailers, but the Company has provided certain retailers with markdown allowances for obsolete and slow moving products that are in the retailer's inventory. The Company estimates these allowances and provides for them as reductions to revenues at the time revenues are recorded. Markdowns are negotiated with retailers and changes are made to the estimates as agreements are reached. Actual amounts for markdowns have not differed materially from estimates.

Cooperative Advertising

Cooperative advertising funds are made available to most of the Company's wholesale footwear customers. In order for retailers to receive reimbursement under such programs, the retailer must meet specified advertising guidelines and provide appropriate documentation of expenses to be reimbursed. The Company's cooperative advertising agreements require that wholesale customers present documentation or other evidence of specific advertisements or display materials used for the Company's products by submitting the actual print advertisements presented in catalogs, newspaper inserts or other advertising circulars, or by permitting physical inspection of displays. Additionally, the Company's cooperative advertising agreements require that the amount of reimbursement requested for such advertising or materials be supported by invoices or other evidence of the actual costs incurred by the retailer. The Company accounts for these cooperative advertising costs as selling and administrative

expenses, in accordance with the Revenue Recognition Topic for Customer Payments and Incentives of the Codification.

Cooperative advertising costs recognized in selling and administrative expenses were \$3.2 million, \$3.5 million and \$3.3 million for Fiscal 2014, 2013 and 2012, respectively. During Fiscal 2014, 2013 and 2012, the Company's cooperative advertising reimbursements paid did not exceed the fair value of the benefits received under those agreements.

Vendor Allowances

From time to time, the Company negotiates allowances from its vendors for markdowns taken or expected to be taken. These markdowns are typically negotiated on specific merchandise and for specific amounts. These specific allowances are recognized as a reduction in cost of sales in the

period in which the markdowns are taken. Markdown allowances not attached to specific inventory on hand or already sold are applied to concurrent or future purchases from each respective vendor.

The Company receives support from some of its vendors in the form of reimbursements for cooperative advertising and catalog costs for the launch and promotion of certain products. The reimbursements are agreed upon with vendors and represent specific, incremental, identifiable costs incurred by the Company in selling the vendor's specific products. Such costs and the related reimbursements are accumulated and monitored on an individual vendor basis, pursuant to the respective cooperative advertising agreements with vendors. Such cooperative advertising reimbursements are recorded as a reduction of selling and administrative expenses in the same period in which the associated expense is

Note 1 <u>Summary of Significant Accounting Policies, Continued</u>

incurred. If the amount of cash consideration received exceeds the costs being reimbursed, such excess amount would be recorded as a reduction of cost of sales.

Vendor reimbursements of cooperative advertising costs recognized as a reduction of selling and administrative expenses were \$2.8 million, \$3.8 million and \$3.0 million for Fiscal 2014, 2013 and 2012, respectively. During Fiscal 2014, 2013 and 2012, the Company's cooperative advertising reimbursements received were not in excess of the costs incurred.

Environmental Costs

Environmental expenditures relating to current operations are expensed or capitalized as appropriate. Expenditures relating to an existing condition caused by past operations, and which do not contribute to current or future revenue generation, are expensed. Liabilities are recorded when environmental assessments and/or remedial efforts are probable and the costs can be reasonably estimated and are evaluated independently of any future claims for recovery. Generally, the timing of these accruals coincides with completion of a feasibility study or the Company's commitment to a formal plan of action. Costs of future expenditures for environmental remediation obligations are not discounted to their present value.

Earnings Per Common Share

Basic earnings per share excludes dilution and is computed by dividing income available to common shareholders by the weighted average number of common shares outstanding for the period. Diluted

earnings per share reflects the potential dilution that could occur if securities to issue common stock were exercised or converted to common stock (see Note 11).

Share-Based Compensation

The Company has share-based compensation covering certain members of management and non-employee directors. The Company recognizes compensation expense for share-based payments based on the fair value of the awards as required by the Compensation - Stock Compensation Topic of the Codification. The Company has not granted any stock options since the first quarter of Fiscal 2008. The fair value of employee restricted stock is determined based on the closing price of the Company's stock on the date of grant. The benefits of tax deductions in excess or recognized compensation expense are reported as a financing cash flow (see Note 12).

Other Comprehensive Income

The Comprehensive Income Topic of the Codification requires, among other things, the Company's pension liability adjustment, postretirement liability adjustment, unrealized gains or losses on foreign currency forward contracts and foreign currency translation adjustments to be included in other comprehensive income net of tax. Accumulated other comprehensive loss at February 1, 2014 consisted

of \$16.5 million of cumulative pension liability adjustment, net of tax, a cumulative post retirement liability adjustment of \$0.9 million, net of tax, offset by a cumulative foreign currency translation adjustment of \$0.6 million.

Note 1 <u>Summary of Significant Accounting Policies, Continued</u>

The following table summarizes the components of accumulated other comprehensive loss for the year ended February 1, 2014:

	Foreign Currency I Translation		Unrecognized Pension/Postretirement Benefit Costs	Total Accumulated Other Comprehensive Income (Loss)		
(In thousands)						
Balance February 2, 2013	\$	(1,931) \$	6 (26,310) 5	\$ (28,241)		
Other comprehensive income (loss) before reclassifications:						
Foreign currency translation adjustment		2,506	_	2,506		
Net actuarial gain			8,581	8,581		
Amounts reclassified from AOCI:						
Amortization of net actuarial loss (1)			6,257	6,257		
Amortization reclassified from AOCI, before tax			6,257	6,257		
Income tax expense			5,870	5,870		
Current period other comprehensive income, net of tax		2,506	8,968	11,474		
Balance February 1, 2014	\$	575 \$	6 (17,342) 9	\$ (16,767)		

(1) Amount is included in net periodic benefit cost, which is recorded in selling and administrative expense on the Consolidated Statements of Operations.

Business Segments

The Segment Reporting Topic of the Codification requires that companies disclose "operating segments" based on the way management disaggregates the Company's operations for making internal operating decisions (see Note 14).

New Accounting Principles

In February 2013, the FASB issued ASU No. 2013-02, "Reporting of Amounts Reclassified Out of Accumulated Other Comprehensive Income" ("AOCI"), which sets forth additional disclosure requirements for items reclassified out of AOCI and into net income, and is effective for annual and interim reporting periods beginning after December 15, 2012. The Company adopted ASU No. 2013-02 in the first quarter of Fiscal 2014 by presenting amounts reclassified out of AOCI as a separate disclosure in Note 1 to the Consolidated Financial Statements. Amounts reclassified out of AOCI were related to amortization of net actuarial loss associated with the Company's pension and postretirement plans.

Note 2 Acquisitions and Intangible Assets

Schuh Acquisition

On June 23, 2011, the Company, through its newly-formed, wholly-owned subsidiary Genesco (UK) Limited ("Genesco UK"), completed the acquisition of all the outstanding shares of Schuh Group Ltd. ("Schuh") for a total purchase price of approximately £100.0 million, less £29.5 million outstanding under existing Schuh credit facilities, which remain in place, less a £1.9 million working capital adjustment and plus £6.2 million net cash acquired, with £5.0 million withheld that was paid in June 2013. The Company financed the acquisition with borrowings under its existing credit facility and the balance from cash on hand. The purchase agreement also provides for deferred purchase price payments totaling £25 million, payable in installments of £15 million and £10 million on the third and fourth anniversaries of the closing, respectively, subject to the payees' not having terminated their employment with Schuh under certain specified circumstances. This amount will be recorded as compensation expense and not reported as a component of the cost of the acquisition.

Headquartered in Scotland, Schuh is a specialty retailer of casual and athletic footwear sold through 99 retail stores in the United Kingdom and the Republic of Ireland as of February 1, 2014. The Company completed the acquisition in order to enhance its strategic development and prospects for growth and provide the Company with an established retail presence in the United Kingdom and improved insight into global fashion trends. The results of Schuh's operations for Fiscal 2014 include net sales of \$364.7 million and operating earnings of \$3.1 million. The results of Schuh's operations for Fiscal 2013 include net sales of \$364.7 million and operating earnings of \$11.2 million. The results of Schuh's operations for Fiscal 2013 include net sales of \$370.5 million and operating earnings of \$11.2 million. The results of Schuh's operations for the fiscal year from the date of acquisition through January 28, 2012, including net sales of \$212.3 million and operating earnings of \$11.7 million, have been included in the Company's Consolidated Financial Statements for the fiscal year ended January 28, 2012. During the fiscal year ended January 28, 2012, the Company expensed \$7.4 million in costs related to the acquisition. These costs were recorded as selling and administrative expenses on the Consolidated Statement of Operations. Compensation expense related to the Schuh acquisition deferred purchase price obligation was \$11.7 million, \$12.1 million and \$7.2 million in Fiscal 2014, 2013 and 2012, respectively. This expense is included in the operating earnings for the Schuh Group segment.

Note 2 <u>Acquisitions and Intangible Assets, Continued</u>

The acquisition has been accounted for using the purchase method in accordance with the Business Combinations Topic of the Codification. Accordingly, the total purchase price has been allocated to the assets acquired and liabilities assumed based on their estimated fair values at acquisition as follows (amounts in thousands):

At June	23,	2011
---------	-----	------

Cash	\$ 24,836
Accounts receivable	4,673
Inventories	32,179
Other current assets	7,565
Property and equipment	30,314
Other non-current assets	6,977
Deferred taxes	4,197
Trademarks	27,224
Other intangibles	4,995
Goodwill	102,907
Accounts payable	(16,196)
Other current liabilities	(24,718)
Long-term debt (includes current portion)	(62,562)
Other non-current liabilities	 (26,637)
Net Assets Acquired	\$ 115,754

The trademarks acquired include the concept names and are deemed to have an indefinite life. Other intangibles include a \$1.7 million customer list, a \$2.5 million asset to reflect the adjustment of acquired leases to market and a vendor contract of \$0.8 million. The weighted average amortization period for the asset to adjust acquired leases to market is 2.7 years. The weighted average amortization period for customer lists is 4.6 years.

Goodwill is calculated as the excess of the consideration transferred over the net assets recognized and represents the future economic benefits arising from other assets acquired that could not be individually identified and separately recognized. Specifically, the goodwill recorded as part of the acquisition of Schuh includes the expected purchasing synergies and other benefits that result from combining the Schuh business with the Company, improved insight into global fashion trends, any intangible assets that do not qualify for separate recognition and an acquired assembled workforce. The goodwill related to the Schuh acquisition is not deductible for tax purposes.

Note 2 Acquisitions and Intangible Assets, Continued

The following pro forma information presents the results of operations of the Company as if the Schuh acquisition had taken place at the beginning of Fiscal 2011 or January 31, 2010. Pro forma adjustments have been made to reflect additional interest expense from the \$89.0 million in debt associated with the acquisition, interest expense on the acquired debt, amortization of intangible assets and the related income tax effects. Pro forma earnings for the twelve months ended January 28, 2012 have been adjusted to exclude \$7.4 million of costs related to the acquisition.

	Twelve Months Ended - Pro forma
In thousands, except per share data	 January 28, 2012
Net sales	\$ 2,384,267
Earnings from continuing operations	96,845
Earnings per share:	
Basic	\$ 4.03
Diluted	\$ 3.97

The pro forma results have been prepared for comparative purposes only and do not purport to be indicative of the results of operations that would have occurred had the Schuh acquisition occurred at the beginning of Fiscal 2011.

Intangible Assets

Other intangibles by major classes were as follows:

	Leases Customer Lists		ists	Other*		Total		
x .a	 Feb. 1, 2014	Feb. 2, 2013	Feb. 1, 2014	Feb. 2, 2013	Feb. 1, 2014	Feb. 2, 2013	Feb. 1, 2014	Feb. 2, 2013
In thousands	-		-		-		-	
Gross other intangibles	\$ 13,104 \$	12,584 \$	14,381 \$	14,116 \$	2,242 \$	2,118 \$	29,727 \$	28,818
Accumulated amortization	(11,997)	(10,800)	(7,354)	(5,312)	(1,294)	(1,108)	(20,645)	(17,220)
Net Other Intangibles	\$ 1,107 \$	1,784 \$	7,027 \$	8,804 \$	948 \$	1,010 \$	9,082 \$	11,598

*Includes non-compete agreements, vendor contract and backlog.

The amortization of intangibles, including trademarks, was \$3.2 million, \$3.4 million and \$3.2 million for Fiscal 2014, 2013 and 2012, respectively. The amortization of intangibles, including trademarks, will be \$2.8 million, \$2.1 million, \$1.6 million, \$1.0 million and \$0.9 million for Fiscal 2015, 2016, 2017, 2018 and 2019, respectively.

Note 3 Asset Impairments and Other Charges and Discontinued Operations

Asset Impairments and Other Charges

In accordance with Company policy, assets are determined to be impaired when the revised estimated future cash flows are insufficient to recover the carrying costs. Impairment charges represent the excess of the carrying value over the fair value of those assets.

Asset impairment charges are reflected as a reduction of the net carrying value of property and equipment, and in asset impairment and other, net in the accompanying Consolidated Statements of Operations.

The Company recorded a pretax charge to earnings of \$1.3 million in Fiscal 2014, including \$3.3 million for network intrusion expenses, \$2.4 million for other legal matters, \$2.3 million for retail store asset impairments and \$1.6 million for a lease termination offset by an \$(8.3) million gain on the lease termination of a New York City Journeys store.

The Company recorded a pretax charge to earnings of \$17.0 million in Fiscal 2013, including \$15.6 million for network intrusion expenses, \$1.4 million for retail store asset impairments and \$0.1 million for other legal matters.

The Company recorded a pretax charge to earnings of \$2.7 million in Fiscal 2012, including \$1.1 million for retail store asset impairments, \$0.9 million for other legal matters and \$0.7 million for network intrusion expenses.

Discontinued Operations

In Fiscal 2014, the Company recorded an additional charge to earnings of \$0.5 million (\$0.3 million net of tax) reflected in discontinued operations, primarily for anticipated costs of environmental remedial alternatives related to former facilities operated by the Company (see Note 13).

In Fiscal 2013, the Company recorded an additional charge to earnings of \$0.8 million (\$0.5 million net of tax) reflected in discontinued operations, primarily for anticipated costs of environmental remedial alternatives related to former facilities operated by the Company (see Note 13).

In Fiscal 2012, the Company recorded an additional charge to earnings of \$1.7 million (\$1.0 million net of tax) reflected in discontinued operations, including \$1.8 million primarily for anticipated costs of environmental remedial alternatives related to former facilities operated by the Company, offset by a \$0.1 million gain for excess provisions to prior discontinued operations (see Note 13).



Note 3 <u>Asset Impairments and Other Charges and Discontinued Operations, Continued</u>

Accrued Provision for Discontinued Operations

In thousands	Facility Shutdown Costs
Balance January 29, 2011	\$ 15,035
Additional provision Fiscal 2012	1,692
Charges and adjustments, net	(4,210)
Balance January 28, 2012	12,517
Additional provision Fiscal 2013	796
Charges and adjustments, net	(1,962)
Balance February 2, 2013	11,351
Additional provision Fiscal 2014	543
Charges and adjustments, net	(519)
Balance February 1, 2014*	11,375
Current provision for discontinued operations	7,263
Total Noncurrent Provision for Discontinued Operations	\$ 4,112

*Includes a \$11.9 million environmental provision, including \$7.8 million in current provision for discontinued operations.

Note 4 <u>Inventories</u>

In thousands	February 1, 2014	February 2, 2013
Raw materials	\$ 26,115	\$ 24,223
Wholesale finished goods	64,357	57,161
Retail merchandise	476,789	423,960
Total Inventories	\$ 567,261	505,344

Note 5 <u>Fair Value</u>

The Fair Value Measurements and Disclosures Topic of the Codification defines fair value, establishes a framework for measuring fair value in accordance with generally accepted accounting principles and expands disclosures about fair value measurements. This Topic defines fair value as the exchange price that would be received for an asset or paid to transfer a liability (an exit price) in the principal or most advantageous market for the asset or liability in an orderly transaction between market participants on the measurement date. It also establishes a fair value hierarchy which requires an entity to maximize the use of observable inputs and minimize the use of unobservable inputs when measuring fair value. The standard describes three levels of inputs that may be used to measure fair value:

Level 1 - Quoted prices in active markets for identical assets or liabilities.

Level 2 - Observable inputs other than Level 1 prices such as quoted prices for similar assets or liabilities; quoted prices in markets that are not active; or other inputs that are observable or can be corroborated by observable market data for substantially the full term of the assets or liabilities.

Level 3 - Unobservable inputs that are supported by little or no market activity and that are significant to the fair value of the assets or liabilities.

A financial asset or liability's classification within the hierarchy is determined based on the lowest level input that is significant to the fair value measurement.

The following table presents the Company's assets and liabilities measured at fair value on a nonrecurring basis as of February 1, 2014 aggregated by the level in the fair value hierarchy within which those measurements fall (in thousands):

	0	-Lived Assets Ield and Used	Level 1	Level 2	Level 3	Impairment Charges
Measured as of May 4, 2013	\$	191	\$ _	\$ _	\$ 191	\$ 1,208
Measured as of August 3, 2013		93	_	_	93	209
Measured as of November 2, 2013		514	_	_	514	350
Measured as of February 1, 2014		448	_	_	448	580
Total Asset Impairment Fiscal 2014						\$ 2,347

In accordance with the Property, Plant and Equipment Topic of the Codification, the Company recorded \$2.3 million of impairment charges as a result of the fair value measurement of its long-lived assets held and used and tested on a nonrecurring basis during the twelve months ended February1, 2014. These charges are reflected in asset impairments and other, net on the Consolidated Statements of Operations.

The Company used a discounted cash flow model to estimate the fair value of these long-lived assets. Discount rate and growth rate assumptions are derived from current economic conditions, expectations of management and projected trends of current operating results. As a result, the Company has determined that the majority of the inputs used to value its long-lived assets held and used are unobservable inputs that fall within Level 3 of the fair value hierarchy.

Note 6 <u>Long-Term Debt</u>

In thousands	Fel	February 1, 2014		February 2, 2013		
Revolver borrowings	\$	_	\$	27,700		
UK term loans		33,730		22,982		
Total long-term debt		33,730		50,682		
Current portion		6,793		5,675		
Total Noncurrent Portion of Long-Term Debt	\$ 26,937		\$	45,007		

Long-term debt maturing during each of the next five years ending in January each year is \$6.8 million, \$14.4 million, \$2.1 million, \$2.1 million and \$2.1 million, respectively, and \$6.2 million thereafter.

The Company did not have any revolver borrowings outstanding under the Credit Facility at February 1, 2014 and had \$33.7 million in term loans outstanding under the U.K. Credit Facilities (described below) at February 1, 2014. The Company had outstanding letters of credit of \$14.5 million under the Credit Facility at February 1, 2014. These letters of credit support product purchases and lease and insurance indemnifications.

Credit Facility:

On January 31, 2014, the Company entered into a Third Amended and Restated Credit Agreement (the "Credit Facility") by and among the Company, certain subsidiaries of the Company party thereto, as other borrowers, the lenders party thereto and Bank of America, N.A., as agent (the "Agent"). The Credit Facility provides revolving credit in the aggregate principal amount of \$400.0 million and replaces the previous \$375.0 million revolving credit facility. The Credit Facility expires January 31, 2019.

Deferred financing costs incurred of \$1.6 million related to the Credit Facility were capitalized and are being amortized over five years. In addition, the remaining deferred financing costs of \$1.5 million related to the previous Amendment are being amortized over five years. These costs are included in other non-current assets on the Consolidated Balance Sheets.

The material terms of the Credit Facility are as follows:

<u>Availability</u>

The Credit Facility is a revolving credit facility in the aggregate principal amount of \$400.0 million, including a \$70.0 million sublimit for the issuance of letters of credit and a domestic swingline subfacility of up to \$40.0 million, a revolving credit subfacility for the benefit of GCO Canada, Inc. in an aggregate amount not to exceed \$25.0 million, which includes a \$5.0 million sublimit for the issuance of letters of credit, and revolving credit subfacility for the benefit of Genesco (UK) Limited in an aggregate amount not to exceed \$50.0 million, which includes a \$10.0 million sublimit for the issuance of letters of credit and a five-year term. Any

Note 6 Long-Term Debt, Continued

swingline loans and any letters of credit and borrowings under the Canadian facilities and UK facilities will reduce the availability under the Credit Facility on a dollar-for-dollar basis.

The Company has the option, from time to time, to increase the availability under the Credit Facility by an aggregate amount of up to \$150.0 million subject to, among other things, the receipt of commitments for the increased amount. In connection with this increased facility, the Canadian revolving credit facility may be increased up to no more than \$40.0 million.

Genesco (UK) Limited has a one-time option to increase the availability of its subfacility under the Credit Facility by an additional amount of up to \$50.0 million.

The aggregate amount of the loans made and letters of credit issued under the Credit Facility shall at no time exceed the lesser of the facility amount (\$400.0 million or, if increased as described above, up to \$550.0 million or \$600.0 million, respectively) or the "Borrowing Base", which generally is based on 90% of eligible inventory plus 85% of eligible wholesale receivables (50% of eligible wholesale receivables of the Lids Team Sports business) plus 90% of eligible credit card and debit card receivables less applicable reserves (the "Loan Cap"). The relevant assets of Genesco (UK) Limited will be included in the Borrowing Base if the additional \$50.0 million sublimit increase is exercised, provided that amounts borrowed by Genesco (UK) Limited based solely on its own borrowing base will be limited to \$50.0 million and the total outstanding to Genesco (UK) Limited will not exceed 30% of the Loan Cap.

The Credit Facility also provides that a first-in, last-out tranche could be added to the revolving credit facility at the option of the Company subject to, among other things, the receipt of commitments for such tranche.

<u>Collateral</u>

The loans and other obligations under the Credit Facility are secured by a perfected first priority lien and security interest in all tangible and intangible assets and excludes real estate and leaseholds of the Company and certain subsidiaries of the Company, including a pledge of 65% of the Company's interest in Genesco (UK) Limited. The assets of Genesco (UK) Limited will not be pledged as collateral unless the additional \$50.0 million sublimit increase is exercised and once pledged, will only serve to secure the obligations of GCO Canada, Inc. and Genesco (UK) Limited and their respective subsidiaries.

Interest and Fees

The Company's borrowings under the Credit Facility bear interest at varying rates that, at the Company's option, can be based on:

Domestic Facility:

(a) LIBOR plus the applicable margin (as defined and based on average Excess Availability during the prior quarter), or (b) the domestic Base Rate (defined as the higher of (i) the Bank of America prime rate, (ii) the federal funds rate plus 0.50% or (iii) LIBOR for an interest period of thirty days plus 1.0%) plus the applicable margin.

Note 6 Long-Term Debt, Continued

Canadian Sub-Facility:

(a) For loans made in Canadian dollars, the bankers' acceptances ("BA") rate plus the applicable margin, or (b) the Canadian Prime Rate (defined as the highest of the (i) Bank of America Canadian Prime Rate, (ii) the Bank of America (Canada Branch) overnight rate plus 0.50%, and (iii) the BA rate for a one month interest period plus 1.0%) plus the applicable margin.

(a) For loans made in U.S. dollars, LIBOR plus the applicable margin, or (b) the U.S. Index Rate (defined as the highest of the (i) Bank of America (Canada branch) U.S. dollar base rate, (ii) the Federal Funds rate plus 0.50%, and (iii) LIBOR for an interest period of thirty days plus 1.0%) plus the applicable margin.

UK Sub-Facility: LIBOR plus the applicable margin.

Swingline Loans:

Domestic swingline loans - domestic Base Rate plus the applicable margin.

UK swingline loans - UK Base Rate (being the "base rate" of the local Bank of America branch in the jurisdiction of the currency chosen) plus the applicable margin.

The initial applicable margin for Base Rate loans and U.S. Index rate loans and Canadian Prime Rate loans is 0.50% and the initial applicable margin for LIBOR loans, BA equivalent loans and UK swingline loans is 1.50%. Thereafter, the applicable margin will be subject to adjustment based on "Excess Availability" for the prior quarter. The term "Excess Availability" means, as of any given date, the excess (if any) of the Loan Cap (being the lesser of the total commitments and the Borrowing Base) over the outstanding credit extensions under the Credit Facility.

Interest on the Company's borrowings is payable monthly in arrears for domestic Base Rate loans (including domestic swingline loans), U.S. Index rate loans, Canadian Prime Rate loans and UK swingline loans and at the end of each interest rate period (but not less often than quarterly) for LIBOR loans and BA equivalent loans.

The Company is also required to pay a commitment fee on the actual daily unused portions of the Credit Facility at a rate of 0.25% per annum.

<u>Currency</u>

Loans to GCO Canada, Inc. may be made in U.S. dollars or Canadian dollars. Loans to Genesco (UK) Limited may be made in U.S. dollars, Euros, Pounds Sterling or any other freely transferable currencies approved by the Agent and applicable lenders.

Certain Covenants

The Company is not required to comply with any financial covenants unless Excess Availability is less than the greater of \$25.0 million or 10.0% of the Loan Cap. If and during such time as Excess Availability is less than the greater of \$25.0 million or 10.0% of the Loan Cap, the Credit Facility requires the

Note 6 Long-Term Debt, Continued

Company to meet a minimum fixed charge coverage ratio of (a) an amount equal to consolidated EBITDA less capital expenditures and taxes paid in cash, in each case for such period, to (b) fixed charges for such period, of not less than 1.0:1.0. Excess Availability was \$358.0 million at February 1,

2014. Because Excess Availability exceeded \$25.0 million or 10.0% of the Loan Cap, the Company was not required to comply with this financial covenant at February 1, 2014.

The Credit Facility also permits the Company to incur up to \$500.0 million of senior debt provided that certain terms and conditions are met.

In addition, the Credit Facility contains certain covenants that, among other things, restrict additional indebtedness, liens and encumbrances, loans and investments, acquisitions, dividends and other restricted payments, transactions with affiliates, asset dispositions, mergers and consolidations, prepayments or material amendments of other indebtedness and other matters customarily restricted in such agreements.

Cash Dominion

The Credit Facility also contains cash dominion provisions that apply in the event that the Company's Excess Availability is less than the greater of \$30.0 million or 12.5% of the Loan Cap or there is an event of default under the Credit Facility.

Events of Default

The Credit Facility contains customary events of default, including, without limitation, payment defaults, breaches of representations and warranties, covenant defaults, cross-defaults to certain other material indebtedness in excess of specified amounts and to agreements which would have a material adverse effect if breached, certain events of bankruptcy and insolvency, certain ERISA events, judgments in excess of specified amounts and change in control.

Certain of the lenders under the Credit Facility or their affiliates have provided and may in the future provide certain commercial banking, financial advisory, and investment banking services in the ordinary course of business for the Company, its subsidiaries and certain of its affiliates, for which they receive customary fees and commissions.

U.K. Credit Facility

In connection with the Schuh acquisition, Schuh entered into an amended and restated Senior Term Facilities Agreement and Working Capital Facility Letter, (collectively, the "UK Credit Facilities")

which provide for term loans of up to £29.5 million (a £15.5 million A term loan and £14.0 million B term loan) and a working capital facility of £5.0 million. The Working Capital Facility Letter was allowed to lapse in June 2012. The A term loan bears interest at LIBOR plus 2.50% per annum. The B term loan bears interest at LIBOR plus 3.75% per annum. The Company is not required to make any payments on the B term loan until it expires October 31, 2015, unless the Company's Schuh Group segment has Excess Cash Flow (as defined in the UK Credit Facilities). The Company paid less than£0.1 million, £4.8 million and £4.5 million on the B term loan in Fiscal 2014, 2013 and 2012, respectively.

Note 6 <u>Long-Term Debt, Continued</u>

In November 2013, Schuh Group Limited entered into an Amended and Restated Facilities Agreement to provide for an additional term loan of up to £12.5 million ("C term loan"). The C term loan bears interest at LIBOR plus 2.50% per annum and expires September 30, 2019.

The UK Credit Facilities contain certain covenants at the Schuh level including a minimum interest coverage covenant initially set at 4.25x and increasing to 4.50x in January 2012 and thereafter, a maximum leverage covenant initially set at 2.75x declining over time at various rates to 2.25x beginning in July 2012 and a minimum cash flow coverage of 1.10x. The Company was in compliance with all the covenants at February 1, 2014. The UK Credit Facilities are secured by a pledge of all the assets of Schuh and its subsidiaries.

Note 7 <u>Commitments Under Long-Term Leases</u>

Operating Leases

The Company leases its office space and all of its retail store locations and transportation equipment under various noncancelable operating leases. The leases have varying terms and expire at various dates through 2030. The store leases in the United States, Puerto Rico and Canada typically have initial terms of approximately 10 years. The stores leases in the United Kingdom and the Republic of Ireland typically have initial terms of between 10 and 20 years. Generally, most of the leases require the Company to pay taxes, insurance, maintenance costs and contingent rentals based on sales. Approximately 3% of the Company's leases contain renewal options.

Rental expense under operating leases of continuing operations was:

In thousands	2014		2013		2012	
Minimum rentals	\$	227,880	\$	215,516	\$	192,175
Contingent rentals		9,667		14,786		12,918
Sublease rentals		(663)		(667)		(686)
Total Rental Expense	\$	236,884	\$	229,635	\$	204,407

Minimum rental commitments payable in future years are:

Fiscal Years	Ir	n thousands
	2015 \$	235,049
	2016	216,870
	2017	188,714
	2018	157,295
	2019	126,726
Later years		381,825
Total Minimum Rental Commitments	\$	1,306,479

For leases that contain predetermined fixed escalations of the minimum rentals, the related rental expense is recognized on a straight-line basis and the cumulative expense recognized on the straight-line basis in excess of the cumulative payments is included in deferred rent and other long-term liabilities on the Consolidated Balance Sheets. The Company occasionally receives reimbursements from landlords to be used towards construction of the store the Company intends to lease. Leasehold improvements are recorded at their gross costs including items reimbursed by landlords. The reimbursements are recorded as deferred rent and amortized as a reduction of rent expense over the initial lease term. Tenant allowances of \$24.2 million and \$20.0 million for Fiscal 2014 and 2013, respectively, and deferred rent of \$41.6 million and \$37.9 million for Fiscal 2014 and 2013, respectively, are included in deferred rent and other long-term liabilities on the Consolidated Balance Sheets.

Note 8 <u>Equity</u> Non-Redeemable Preferred Stock

		N	umber of Share	s	А	mo	ounts in Thousands			Common	No. of
Class (In order of preference)*	Shares Authorized	2014	2013	2012	2014		2013		2012	Convertible Ratio	Votes per share
Subordinated Serial Preferred (Cumulative)											
Aggregate	3,000,000 **	_	—	_	_		_			N/A	N/A
\$2.30 Series 1	64,368	_	16,203	30,368	\$		\$ 648	\$	1,215	.83	1
\$4.75 Series 3	40,449	_	7,398	11,643	_		740		1,164	2.11	2
\$4.75 Series 4	53,764	_	3,247	3,397	_		325		340	1.52	1
Series 6	800,000	_		_							100
\$1.50 Subordinated											
Cumulative Preferred	5,000,000		30,067	30,067	—		902		902		1
		_	56,915	75,475	_		2,615		3,621		
Employees' Subordinated Convertible Preferred	5,000,000	46,069	46,852	47,922	1,382		1,405		1,437	1.00 ***	1
Stated Value of Issued Shares					1,382		4,020		5,058		
Employees' Preferred Stock Purchase Accounts					(77)	(96)		(101)		
Total Non-Redeemable Preferred Stock					\$ 1,305	_	\$ 3,924	\$	4,957		

* In order of preference for liquidation and dividends.

** The Company's charter permits the board of directors to issue Subordinated Serial Preferred Stock in as many series, each with as many shares and such rights and preferences as the board may designate.

*** Also convertible into one share of \$1.50 Subordinated Cumulative Preferred Stock.

Note 8 <u>Equity, Continued</u>

Preferred Stock Transactions

In thousands	Non-Redeemable Preferred Stock		Non-Redeemable Employees' Preferred Stock		Employees' Preferred Stock Purchase Accounts	Total Non-Redeemable Preferred Stock
Balance January 29, 2011	\$ 3,81	16	\$ 1,476	\$	(109)	\$ 5,183
Other	(19	95)	(39)		8	 (226)
Balance January 28, 2012	3,62	21	1,437		(101)	4,957
Other	(1,00)6)	(32)		5	(1,033)
Balance February 2, 2013	2,61	15	1,405		(96)	3,924
Preferred stock redemptions	(1,46	52)	—		—	(1,462)
Other stock conversions	(1,15	53)	(23)	_	19	 (1,157)
Balance February 1, 2014	\$ -	_	\$ 1,382		(77)	\$ 1,305

Subordinated Serial Preferred Stock (Cumulative):

The Company issued a notice of mandatory redemption effective April 30, 2013, to its holders of Subordinated Serial Preferred Stock \$2.30 Series 1, \$4.75 Series 3 and \$4.75 Series 4 during the first quarter of Fiscal 2014. The Series 1 preferred stock was redeemed at \$40 per share plus accumulated dividends. During Fiscal 2014, 13,713 shares of Series 1 preferred stock were converted to common stock and 2,490 shares of Series 1 preferred stock were redeemed. The Series 3 and 4 preferred stocks were redeemed at \$100 per share plus accumulated dividends. During Fiscal 2014, 6,046 shares of Series 3 preferred stock were converted to common stock and 1,352 shares of Series 3 preferred stock were redeemed. During Fiscal 2014, 3,247 shares of Series 4 preferred stock were redeemed. The total cost of the redemption for Series 1, 3 and 4 preferred stock was \$0.6 million in Fiscal 2014.

The Company's shareholders' rights plan grants to common shareholders the right to purchase, at a specified exercise price, a fraction of a share of subordinated serial preferred stock, Series 6, in the event of an acquisition of, or an announced tender offer for, 15% or more of the Company's outstanding common stock. Upon any such event, each right also entitles the holder (other than the person making such acquisition or tender offer) to purchase, at the exercise price, shares of common stock having a market value of twice the exercise price. In the event the Company is acquired in a transaction in which the Company is not the surviving corporation, each right would entitle its holder to purchase, at the exercise price, shares of the acquiring company having a market value of twice the exercise price. The rights expire in March 2020, are redeemable under certain circumstances for \$.01 per right and are subject to exchange for one share of common stock or an equivalent amount of preferred stock at any time after the event which makes the rights exercisable and before a majority of the Company's common stock is acquired.



Note 8 <u>Equity, Continued</u>

\$1.50 Subordinated Cumulative Preferred Stock:

The Company issued a notice of mandatory redemption effective April 30, 2013, to its holders of \$1.50 Subordinated Cumulative Preferred Stock during the first quarter of Fiscal 2014. The \$1.50 Subordinated Cumulative Preferred Stock was redeemed at \$30 per share plus accumulated dividends. During Fiscal 2014, 30,067 shares of \$1.50 Subordinated Cumulative Preferred Stock were redeemed. The total cost of the redemption for the \$1.50 Subordinated Cumulative Preferred Stock was \$0.9 million in Fiscal 2014.

Employees' Subordinated Convertible Preferred Stock:

Stated and liquidation values are 88 times the average quarterly per share dividend paid on common stock for the previous eight quarters (if any), but in no event less than \$30 per share.

Common Stock:

Common stock-\$1 par value. Authorized: 80,000,000 shares; issued: February 1, 2014 – 24,407,724 shares; February 2, 2013 – 24,484,915 shares. There were 488,464 shares held in treasury at February 1, 2014 and February 2, 2013. Each outstanding share is entitled to one vote. At February 1, 2014, common shares were reserved as follows: 46,069 shares for conversion of preferred stock; 60,000 shares for the 1996 Stock Incentive Plan; 70,854 shares for the 2005 Stock Incentive Plan; 1,204,662 shares for the 2009 Amended and Restated Stock Incentive Plan; and 310,292 shares for the Genesco Employee Stock Purchase Plan.

For the year ended February 1, 2014, 130,051 shares of common stock were issued for the exercise of stock options at an average weighted exercise price of \$23.33, for a total of \$3.0 million; 199,392 shares of common stock were issued as restricted shares as part of the 2009 Amended and Restated Equity Incentive Plan; 3,146 shares of common stock were issued for the purchase of shares under the Employee Stock Purchase Plan at an average weighted market price of \$62.30, for a total of \$0.2 million; 14,435 shares were issued to directors for no consideration; 105,193 shares were withheld for taxes on restricted stock vested in Fiscal 2014; 6,279 shares of restricted stock were forfeited in Fiscal 2014; and 24,922 shares were issued in miscellaneous conversions of Series 1, 3 and Employees' Subordinated Convertible Preferred Stock. The 130,051 options exercised were all fixed stock options (see Note 12). In addition, the Company repurchased and retired 337,665 shares of common stock at an average weighted market price of \$61.23 for a total of \$20.7 million.

For the year ended February 2, 2013, 223,618 shares of common stock were issued for the exercise of stock options at an average weighted exercise price of \$21.50, for a total of \$4.8 million; 194,232 shares of common stock were issued as restricted shares as part of the 2009 Amended and Restated Equity Incentive Plan; 2,463 shares of common stock were issued for the purchase of shares under the Employee Stock Purchase Plan at an average weighted market price of \$63.84, for a total of \$0.2 million; 10,224 shares were issued to directors for no consideration; 75,552 shares were withheld for taxes on restricted stock vested in Fiscal 2013; 4,020 shares of restricted stock were forfeited in Fiscal 2013; and 22,028 shares were issued in miscellaneous conversions of Series 1, 3, 4 and Employees' Subordinated Convertible Preferred Stock. The 223,618 options exercised were all fixed stock options (see Note 12).

Note 8 <u>Equity, Continued</u>

In addition, the Company repurchased and retired 645,904 shares of common stock at an average weighted market price of \$58.29 for a total of \$37.6 million.

For the year ended January 28, 2012, 390,357 shares of common stock were issued for the exercise of stock options at an average weighted exercise price of \$24.82, for a total of \$9.7 million; 289,407 shares of common stock were issued as restricted shares as part of the 2009 Equity Incentive Plan; 2,717 shares of common stock were issued for the purchase of shares under the Employee Stock Purchase Plan at an average weighted market price of \$48.95, for a total of \$0.1 million; 14,643 shares were issued to directors for no consideration; 93,089 shares were withheld for taxes on restricted stock vested in Fiscal 2012; 14,081 shares of restricted stock were forfeited in Fiscal 2012; and 5,238 shares were issued in miscellaneous conversions of Series 1, 3, 4 and Employees' Subordinated Convertible Preferred Stock. The 390,357 options exercised were all fixed stock options (see Note 12).

Restrictions on Dividends and Redemptions of Capital Stock:

The Company's charter provides that no dividends may be paid and no shares of capital stock acquired for value if there are dividend or redemption arrearages on any senior or equally ranked stock. Exchanges of subordinated serial preferred stock for common stock or other stock junior to such exchanged stock are permitted.

The Company's Credit Facility prohibits the payment of dividends and other restricted payments unless as of the date of the making of any Restricted Payment or consummation of any Acquisition, (a) no Default or Event of Default exists or would arise after giving effect to such Restricted Payment or Acquisition, and (b) either (i) the Borrowers have pro forma projected Excess Availability for the following six month period equal to or greater than 25% of the Loan Cap, after giving pro forma effect to such Restricted Payment or Acquisition, or (ii) (A) the Borrowers have pro forma projected Excess Availability for the following six month period but equal to or greater than 15% of the Loan Cap, after giving pro forma effect to the Restricted Payment or Acquisition, and (B) the Fixed Charge Coverage Ratio, on a pro forma basis for the twelve months preceding such Restricted Payment or Acquisition, will be equal to or greater than 1.0:1.0, and (c) after giving effect to such Restricted Payment or Acquisition, the Borrowers are Solvent. The Company's management does not expect availability under the Credit Facility to fall below the requirements listed above during Fiscal 2015. The Company's UK Credit Facility prohibits the payment of any dividends by Schuh or its subsidiaries to the Company.

The Company issued a mandatory notice of redemption effective April 30, 2013, to its holders of Subordinated Serial Preferred Stock \$2.30 Series 1, \$4.75 Series 3 and \$4.75 Series 4 and on its \$1.50 Subordinated Cumulative Preferred Stock during the first quarter of Fiscal 2014. The total cost of the redemption was \$1.5 million. As a result, all of these preferred issues of stock were either converted to common stock or redeemed in Fiscal 2014, and there are no outstanding shares remaining. Therefore, there is no longer an annual dividend requirement. Dividends paid during Fiscal 2014 were less than \$0.1 million.

Note 8 <u>Equity, Continued</u>

Changes in the Shares of the Company's Capital Stock

	Common Stock	Non- Redeemable Preferred Stock	Employees' Preferred Stock
Issued at January 29, 2011	24,162,634	79,306	49,192
Exercise of options	390,357	—	
Issue restricted stock	304,050		—
Issue shares—Employee Stock Purchase Plan	2,717	—	
Shares repurchased	0		—
Other	(101,932)	(3,831)	(1,270)
Issued at January 28, 2012	24,757,826	75,475	47,922
Exercise of options	223,618	_	_
Issue restricted stock	204,456	_	_
Issue shares—Employee Stock Purchase Plan	2,463	_	_
Shares repurchased	(645,904)	_	_
Other	(57,544)	(18,560)	(1,070)
Issued at February 2, 2013	24,484,915	56,915	46,852
Exercise of options	130,051	_	_
Issue restricted stock	213,827	_	_
Issue shares—Employee Stock Purchase Plan	3,146	_	_
Shares repurchased	(337,665)	_	_
Other	(86,550)	(56,915)	(783)
Issued at February 1, 2014	24,407,724		46,069
Less shares repurchased and held in treasury	488,464	_	
Outstanding at February 1, 2014	23,919,260		46,069

Note 9 <u>Income Taxes</u>

The components of earnings from continuing operations before income taxes is comprised of the following:

In thousands	2014	2013	2012
United States	\$ 152,832	\$ 152,457	\$ 139,174
Foreign	6,028	12,375	17,219
Total Earnings from Continuing Operations before Income Taxes	\$ 158,860	\$ 164,832	\$ 156,393

Income tax expense from continuing operations is comprised of the following:

In thousands	 2014 2013		 2012	
Current				
U.S. federal	\$ 35,463	\$	50,859	\$ 42,103
International	7,293		9,853	2,226
State	8,139		8,841	8,952
Total Current Income Tax Expense	 50,895		69,553	53,281
Deferred				
U.S. federal	14,078		(7,924)	5,579
International	(1,813)		(6,379)	4,370
State	2,718		(3,315)	(288)
Total Deferred Income Tax Expense (Benefit)	 14,983		(17,618)	9,661
Total Income Tax Expense – Continuing Operations	\$ 65,878	\$	51,935	\$ 62,942

Discontinued operations were recorded net of income tax benefit of approximately \$(0.2) million, \$(0.3) million and \$(0.7) million in Fiscal 2014, 2013 and 2012, respectively.

As a result of the exercise of stock options and vesting of restricted stock during Fiscal 2014, 2013 and 2012, the Company realized an additional income tax benefit of approximately \$3.8 million, \$4.8 million and \$4.6 million, respectively. These tax benefits are reflected as an adjustment to additional paid-in capital.

Note 9 <u>Income Taxes, Continued</u>

Deferred tax assets and liabilities are comprised of the following:

	February 1,			February 2,
In thousands		2014		2013
Identified intangibles	\$	(28,468)	\$	(28,076)
Prepaids		(3,063)		(2,943)
Convertible bonds		(3,001)		(3,001)
Total deferred tax liabilities		(34,532)		(34,020)
Options		448		965
Deferred rent		4,986		5,847
Pensions		4,253		8,321
Expense accruals		15,673		16,766
Uniform capitalization costs		13,750		12,539
Book over tax depreciation		2,839		13,783
Provisions for discontinued operations and restructurings		4,731		4,745
Inventory valuation		2,115		2,015
Tax net operating loss and credit carryforwards		2,396		3,535
Allowances for bad debts and notes		761		1,598
Deferred compensation and restricted stock		6,606		6,382
Other		4,320		3,500
Gross deferred tax assets		62,878		79,996
Deferred tax asset valuation allowance		(3,771)		(3,541)
Deferred tax asset net of valuation allowance		59,107		76,455
Net Deferred Tax Assets	\$	24,575	\$	42,435

The deferred tax balances have been classified in the Consolidated Balance Sheets as follows:

	2014	2013
Net current asset	\$ 23,089	\$ 23,725
Net non-current asset	3,342	18,731
Net non-current liability	(1,856)	(21)
Net Deferred Tax Assets	\$ 24,575	\$ 42,435

Note 9 Income Taxes, Continued

Reconciliation of the United States federal statutory rate to the Company's effective tax rate from continuing operations is as follows:

	2014	2013	2012
U. S. federal statutory rate of tax	35.00 %	35.00 %	35.00 %
State taxes (net of federal tax benefit)	4.62	3.11	3.62
Foreign rate differential	(1.24)	(1.98)	(1.71)
Change in valuation allowance	0.05	(0.17)	0.60
Permanent items	2.18	1.85	2.27
Uncertain federal, state and foreign tax positions	0.21	(5.73)	_
Other	0.65	(0.57)	0.47
Effective Tax Rate	41.47 %	31.51 %	40.25 %

The provision for income taxes resulted in an effective tax rate for continuing operations of 41.47% for Fiscal 2014, compared with an effective tax rate of 31.51% for Fiscal 2013. The tax rate for Fiscal 2013 was lower primarily due to the reversal of previously recorded charges related to uncertain tax positions due to the expiration of the applicable statutes of limitations and a settlement with a state tax authority more favorable than anticipated related to other uncertain tax positions.

As of February 1, 2014, February 2, 2013 and January 28, 2012, the Company had a federal net operating loss carryforward, which was assumed in one of the prior year acquisitions, of \$1.3 million, \$1.5 million and \$1.6 million, respectively, which expire in fiscal years 2025 through 2030.

As of February 1, 2014, February 2, 2013 and January 28, 2012, the Company had state net operating loss carryforwards of \$0.0 million, \$0.1 million and \$0.1 million, respectively, which expire in fiscal years 2016 through 2031.

As of February 1, 2014, February 2, 2013 and January 28, 2012, the Company had state tax credits of \$0.7 million, \$0.9 million and \$0.6 million, respectively. These credits expire in fiscal years 2014 through 2019.

As of February 1, 2014, February 2, 2013 and January 28, 2012, the Company had foreign tax credits of \$0.0 million, \$0.0 million and \$0.1 million, respectively. These credits will expire in fiscal year 2022.

As of February 1, 2014, February 2, 2013 and January 28, 2012, the Company had foreign net operating losses of \$7.5 million, \$10.4 million and \$28.8 million, respectively, which have no expiration.

As of February 1, 2014, as part of the Schuh acquisition, the Company has provided a valuation allowance of approximately \$3.8 million on deferred tax assets associated primarily with foreign net operating losses and foreign fixed assets for which management has determined it is more likely than

Note 9 Income Taxes, Continued

not that the deferred tax assets will not be realized. The \$0.3 million net increase in the valuation allowance during Fiscal 2014 from the \$3.5 million provided for as of February 2, 2013 determined in

accordance with the Income Tax Topic of the Codification relates to foreign net operating losses arising in Fiscal 2012 and increases in fixed asset-related deferred tax assets that will more likely than not

never be realized. Management believes that it is more likely than not that the remaining deferred tax assets will be fully realized.

As of February 1, 2014, the Company has not provided for withholding or United States federal income taxes on approximately \$26.4 million of accumulated undistributed earnings of its foreign subsidiaries as they are considered by management to be permanently reinvested. If these undistributed earnings were not considered to be permanently reinvested, the related U.S. tax liability may be reduced by foreign income taxes paid on those earnings. Because of the complexities involved with the hypothetical tax calculation, a determination of the unrecognized deferred tax liability related to these undistributed earnings is not practicable.

The methodology in the Income Tax Topic of the Codification prescribes that a company should use a more-likely-than-not recognition threshold based on the technical merits of the tax position taken. Tax positions that meet the more-likely-than-not recognition threshold should be measured in order to determine the tax benefit to be recognized in the financial statements.

The following is a tabular reconciliation of the total amounts of unrecognized tax benefits for Fiscal 2014, 2013 and 2012.

In thousands	2014			2013	2012	
Unrecognized Tax Benefit – Beginning of Period	\$	10,437	\$	20,467	\$	14,167
Gross Increases (Decreases) – Tax Positions in a Prior Period		139		(2,464)		(29)
Gross Increases – Tax Positions in a Current Period		1,452		133		6,986
Settlements		(340)		(449)		(533)
Lapse of Statutes of Limitations		(728)		(7,250)		(124)
Unrecognized Tax Benefit – End of Period	\$	10,960	\$	10,437	\$	20,467

The amount of unrecognized tax benefits as of February 1, 2014, February 2, 2013 and January 28, 2012, which would impact the annual effective rate if recognized were \$1.3 million, \$2.4 million and \$12.6 million, respectively. The Company believes it is reasonably possible that there will be a \$0.1 million decrease in the gross tax liability for uncertain tax positions within the next 12 months based upon the expiration of statutes of limitation.

The Company recognizes interest expense and penalties related to the above unrecognized tax benefits within income tax expense on the Consolidated Statements of Operations. Related to the uncertain tax benefits noted above, the Company recorded interest and penalties of approximately \$(0.1) million

Note 9 Income Taxes, Continued

expense and \$(0.1) million, respectively, during Fiscal 2014, \$(1.2) million expense and \$0.1 million, respectively, during Fiscal 2013 and \$0.5 million expense and \$0.0 million, respectively, during Fiscal 2012. The Company recognized a liability for accrued interest and penalties of \$0.9 million and \$0.1 million, respectively, as of February 1, 2014, \$1.1 million and \$0.2 million, respectively, as of February 2, 2013 and \$2.3 million and \$0.2 million, respectively, as of January 28, 2012. The long-term portion of the unrecognized tax benefits and related accrued interest and penalties are included in deferred rent and other long-term liabilities on the Consolidated Balance Sheets.

Income tax reserves are determined using the methodology required by the Income Tax Topic of the Codification.

The Company and its subsidiaries file income tax returns in federal and in many state and local jurisdictions as well as foreign jurisdictions. With few exceptions, the Company's U.S. federal and state and local income tax returns for fiscal years ended January 29, 2011 and beyond remain subject to examination. In addition, the Company has subsidiaries in various foreign jurisdictions that have statutes of limitation generally ranging from two to six years.

Note 10

Defined Benefit Pension Plans and Other Postretirement Benefit Plans

Defined Benefit Pension Plans

The Company sponsored a non-contributory, defined benefit pension plan. As of January 1, 1996, the Company amended the plan to change the pension benefit formula to a cash balance formula from the then existing benefit calculation based upon years of service and final average pay. The benefits accrued under the old formula were frozen as of December 31, 1995. Upon retirement, the participant will receive this accrued benefit payable as an annuity. In addition, the participant will receive as a lump sum (or annuity if desired) the amount credited to the participant's cash balance account under the new formula. Effective January 1, 2005, the Company froze the defined benefit cash balance plan which prevents any new entrants into the plan as of that date as well as affects the amounts credited to the participants' accounts as discussed below.

Under the cash balance formula, beginning January 1, 1996, the Company credits each participants' account annually with an amount equal to 4% of the participant's compensation plus 4% of the participant's compensation in excess of the Social Security taxable wage base. Beginning December 31, 1996 and annually thereafter, the account balance of each active participant was credited with 7% interest calculated on the sum of the balance as of the beginning of the plan year and 50% of the amounts credited to the account, other than interest, for the plan year. The account balance of each participants' cash balance plan account will be credited annually only with interest at the 30 year Treasury rate, not to exceed 7%, until the participant retires. The amount credited each year will be based on the rate at the end of the prior year.

Note 10 Defined Benefit Pension Plans and Other Postretirement Benefit Plans, Continued

Other Postretirement Benefit Plans

The Company provides health care benefits for early retirees and life insurance benefits for certain retirees not covered by collective bargaining agreements. Under the health care plan, early retirees are eligible for benefits until age 65. Employees who meet certain requirements are eligible for life insurance benefits upon retirement. The Company accrues such benefits during the period in which the employee renders service.

Obligations and Funded Status

Change in Benefit Obligation

	Pension Benefits					Other Benefits			
In thousands		2014		2013		2014		2013	
Benefit obligation at beginning of year	\$	119,126	\$	118,644	\$	4,487	\$	3,908	
Service cost		350		350		428		356	
Interest cost		4,584		4,961		159		157	
Plan participants' contributions		_				86		74	
Benefits paid		(9,000)		(9,038)		(436)		(221)	
Actuarial (gain) loss		(3,927)		4,209		990		213	
Benefit Obligation at End of Year	\$	111,133	\$	119,126	\$	5,714	\$	4,487	

Change in Plan Assets

		Benef	Other	Other Benefits			
In thousands	2014 2013		2014	2014 2013			
Fair value of plan assets at beginning of year	\$	98,612	\$	96,443			_
Actual gain on plan assets		12,298		11,207	—		—
Employer contributions		_			350		147
Plan participants' contributions		_		—	86		74
Benefits paid		(9,000)		(9,038)	(436)		(221)
Fair Value of Plan Assets at End of Year	\$	101,910	\$	98,612	_		_
Funded Status at End of Year	\$	(9,223)	\$	(20,514)	\$ (5,714)	\$	(4,487)

Amounts recognized in the Consolidated Balance Sheets consist of:

	Pension Benefits Other H					Benefits		
In thousands		2014		2013		2014		2013
Noncurrent assets	\$	_	\$	—	\$	—	\$	_
Current liabilities		—				(208)		(160)
Noncurrent liabilities		(9,223)		(20,514)		(5,506)		(4,327)
Net Amount Recognized	\$	(9,223)	\$	(20,514)	\$	(5,714)	\$	(4,487)

Note 10 Defined Benefit Pension Plans and Other Postretirement Benefit Plans, Continued

Amounts recognized in accumulated other comprehensive income consist of:

	Pension Benefits				Other Benefits			
In thousands		2014		2013		2014		2013
Prior service cost	\$	_	\$	_	\$	_	\$	
Net loss		27,147		42,879		1,459		566
Total Recognized in Accumulated Other Comprehensive Loss	\$	27,147	\$	42,879	\$	1,459	\$	566

Amounts for projected and accumulated benefit obligation and fair value of plan assets are as follows:

	F	ebruary 1,		
In thousands		2014	Feb	ruary 2, 2013
Projected benefit obligation	\$	111,133	\$	119,126
Accumulated benefit obligation		111,133		119,126
Fair value of plan assets		101,910		98,612

Components of Net Periodic Benefit Cost

Net Periodic Benefit Cost

	Pension Benefits						Other Benefits					
In thousands		2014		2013		2012		2014		2013		2012
Service cost	\$	350	\$	350	\$	250	\$	428	\$	356	\$	166
Interest cost		4,584		4,961		5,597		159		157		174
Expected return on plan assets		(6,654)		(7,003)		(7,807)		_		_		_
Amortization:												
Prior service cost		_		4		4		_		_		_
Losses		6,160		6,032		4,728		97		84		79
Net amortization	\$	6,160	\$	6,036	\$	4,732	\$	97	\$	84	\$	79
Net Periodic Benefit Cost	\$	4,440	\$	4,344	\$	2,772	\$	684	\$	597	\$	419

Reconciliation of Accumulated Other Comprehensive Income

	Pens	Other Benefits		
In thousands		2014		2014
Net loss (gain)	\$	(9,571)	\$	990
Amortization of prior service cost		—		—
Amortization of net actuarial loss		(6,160)		(97)
Total Recognized in Other Comprehensive Income	\$	(15,731)	\$	893
Total Recognized in Net Periodic Benefit Cost and Other Comprehensive Income	\$	(11,291)	\$	1,577

Note 10 <u>Defined Benefit Pension Plans and Other Postretirement Benefit Plans, Continued</u>

The estimated net loss and prior service cost for the defined benefit pension plans that will be amortized from accumulated other comprehensive income into net periodic benefit cost over the next fiscal year

are \$3.9 million and \$0.0, respectively. The estimated net loss for the other postretirement benefit plans that will be amortized from accumulated other comprehensive income into net periodic benefit cost over the next fiscal year is \$0.1 million.

Weighted-average assumptions used to determine benefit obligations

	Pension B	enefits	Other Benefits		
	2014	2013	2014	2013	
Discount rate	4.40%	4.00%	4.40%	4.01%	
Rate of compensation increase	NA	NA	—		

For Fiscal 2014 and 2013, the discount rate was based on a yield curve of high quality corporate bonds with cash flows matching the Company's planned expected benefit payments.

Weighted-average assumptions used to determine net periodic benefit costs

	Pe	ension Benefits		C	ther Benefits		
-	2014	2013	2012	2014	2013	2012	
Discount rate	4.00%	4.35%	5.25%	4.01%	4.17%	5.25%	
Expected long-term rate of return on plan assets	7.75%	7.75%	8.25%	_	_	_	
Rate of compensation increase	NA	NA	NA	_	_	_	

The weighted average discount rate used to measure the benefit obligation for the pension plan increased from 4.00% to 4.40% from Fiscal 2013 to Fiscal 2014. The increase in the rate decreased the accumulated benefit obligation by \$3.9 million and decreased the projected benefit obligation by \$3.9 million. The weighted average discount rate used to measure the benefit obligation for the pension plan decreased from 4.35% to 4.00% from Fiscal 2012 to Fiscal 2013. The decrease in the rate increased the accumulated benefit obligation by \$4.3 million and increased the projected benefit obligation by \$4.3 million.

To develop the expected long-term rate of return on assets assumption, the Company considered historical asset returns, the current asset allocation and future expectations. Considering this information, the Company selected a 7.75% long-term rate of return on assets assumption.

Assumed health care cost trend rates

	2014	2013
Health care cost trend rate assumed for next year	8.0%	7.0%
Rate to which the cost trend rate is assumed to decline (the ultimate trend rate)	5%	5%
Year that the rate reaches the ultimate trend rate	2019	2017

Note 10 <u>Defined Benefit Pension Plans and Other Postretirement Benefit Plans, Continued</u>

The effect on disclosed information of one percentage point change in the assumed health care cost trend rate for each future year is shown below.

In thousands	1% Increase in Rates			1% Decrease in Rates		
Aggregated service and interest cost	\$	134	\$	106		
Accumulated postretirement benefit obligation	\$	626	\$	735		

Plan Assets

The Company's pension plan weighted average asset allocations as of February 1, 2014 and February 2, 2013, by asset category are as follows:

	Plan A	ssets
	February 1, 2014	February 2, 2013
Asset Category		
Equity securities	65%	66%
Debt securities	35%	34%
Other	0%	0%
Total	100%	100%

The investment strategy of the Trust is to ensure over the long-term an asset pool, that when combined with Company contributions, will support benefit obligations to participants, retirees and beneficiaries. Investment management responsibilities of plan assets are delegated to outside investment advisers and overseen by an Investment Committee comprised of members of the Company's senior management that are appointed by the Board of Directors. The Company has an investment policy that provides direction on the implementation of this strategy.

The investment policy establishes a target allocation for each asset class and investment manager. The actual asset allocation versus the established target is reviewed at least quarterly and is maintained within a +/- 5% range of the target asset allocation. Target allocations are 50% domestic equity, 13% international equity, 35% fixed income and 2% cash investments.

All investments are made solely in the interest of the participants and beneficiaries for the exclusive purposes of providing benefits to such participants and their beneficiaries and defraying the expenses related to administering the Trust as determined by the Investment Committee. All assets shall be properly diversified to reduce the potential of a single security or single sector of securities having a disproportionate impact on the portfolio.

The Committee utilizes an outside investment consultant and investment managers to implement its various investment strategies. Performance of the managers is reviewed quarterly and the investment objectives are consistently evaluated.

Note 10 <u>Defined Benefit Pension Plans and Other Postretirement Benefit Plans, Continued</u>

At February 1, 2014 and February 2, 2013, there were no Company related assets in the plan.

Generally, quoted market prices are used to value pension plan assets. Equities, some fixed income securities, publicly traded investment funds and U.S. government obligations are valued at the closing price reported on the active market on which the individual security is traded.

The following tables present the pension plan assets by level within the fair value hierarchy as of February 1, 2014 and February 2, 2013.

February 1, 2014	Level 1	Level 2	Level 3	Total
Equity Securities:				
International securities	\$ 13,026	\$ —	\$ 	\$ 13,026
U.S. securities	53,187	_		53,187
Fixed Income Securities	35,481	—	—	35,481
Other:				
Cash Equivalents	235	—	—	235
Other (includes receivables and payables)	(19)	—	—	(19)
Total Pension Plan Assets	\$ 101,910	\$ _	\$ _	\$ 101,910
February 2, 2013	Level 1	Level 2	Level 3	Total
Equity Securities:				
International securities	\$ 13,757	\$ —	\$ —	\$ 13,757
U.S. securities	51,011	_	_	51,011
Fixed Income Securities	33,633	_	_	33,633
Other:				
Cash Equivalents	235	_	_	235
Other (includes receivables and payables)	(24)	—	—	(24)
Total Pension Plan Assets	\$ 98,612	\$	\$	\$ 98,612

Cash Flows

Return of Assets

There was no return of assets from the plan to the Company in Fiscal 2014 and no plan assets are projected to be returned to the Company in Fiscal 2015.

Contributions

There was no Employee Retirement Income Security Act ("ERISA") cash requirement for the plan in 2013 and none is projected to be required in 2014. It is the Company's policy to contribute enough cash to maintain at least an 80% funding level.

Note 10 <u>Defined Benefit Pension Plans and Other Postretirement Benefit Plans, Continued</u>

Estimated Future Benefit Payments

Expected benefit payments from the trust, including future service and pay, are as follows:

Estimated future payments	Pension Benefits (\$ in millio		Other Benefits (\$ in millions)		
2014	\$	8.7	\$	0.2	
2015		8.5		0.2	
2016		8.3		0.2	
2017		8.1		0.3	
2018		8.0		0.3	
2019 – 2023		36.7		1.8	

Section 401(k) Savings Plan

The Company has a Section 401(k) Savings Plan available to employees who have completed one full year of service and are age 21 or older.

Since January 1, 2005, the Company has matched 100% of each employee's contribution of up to 3% of salary and 50% of the next 2% of salary. In addition, for those employees hired before December 31, 2004, who were eligible for the Company's cash balance retirement plan before it was frozen, the Company annually makes an additional contribution of 2 1/2 % of salary to each employee's account. In calendar 2005 and future years, participants are immediately vested in their contributions and the Company's matching contribution plus actual earnings thereon. The contribution expense to the Company for the matching program was approximately \$5.0 million for Fiscal 2014, \$5.3 million for Fiscal 2013 and \$4.2 million for Fiscal 2012.

Note 11 <u>Earnings Per Share</u>

			r the Year Ended ebruary 1, 2014			or the Year Er February 2, 20			For the Year Ended January 28, 2012						
(In thousands, except per share amounts)	Incom (Numerat		Shares (Denominator)	r-Share mount	Income Numerator)	Shares (Denominate	or)	Per-Sh Amou			income umerator)	Shares (Denominator)		-Share nount
Earnings from continuing operations	\$ 92	2,982		 	\$ 112,897					\$	93,451				
Less: Preferred stock dividends and income from participating securities		(33)			(147)						(3,338)				
Basic EPS from continuing operations															
Income from continuing operations available to common shareholders	92	2,949	23,297	\$ 3.99	112,750	23,	,584	\$	4.78		90,113	23,1	79	\$	3.89
Effect of Dilutive Securities from continuing operations															
Plus: Income from participating securities		_			_						43				
Options and restricted stock			272				372					2	33		
Convertible preferred stock ⁽¹⁾		_	_		88		34				141		55		
Employees' preferred stock ⁽²⁾			46				47						48		
Diluted EPS from continuing operations															
Income from continuing operations available to common shareholders plus assumed conversions	\$ 92	2,949	23,615	\$ 3.94	\$ 112,838	24	,037	\$	4.69	\$	90,297	23,5	65	\$	3.83

(1) As a result of the Company issuing a notice of mandatory redemption to the holders of Series 1, 3 and 4 preferred stock in the first quarter of Fiscal 2014, there were no remaining convertible preferred stock of that series outstanding as of February 1, 2014. Therefore, convertible preferred stocks were not included in diluted earnings per share for Fiscal 2014. The amount of the dividend on the convertible preferred stock per common share obtainable on conversion of the convertible preferred stock was less than basic earnings per share for Series 1, 3 and 4 preferred stocks for Fiscal 2013 and 2012. Therefore, conversion of these convertible preferred stocks were included in diluted earnings per share for Fiscal 2013 and 2012.

(2) The Company's Employees' Subordinated Convertible Preferred Stock is convertible one for one to the Company's common stock. Because there are no dividends paid on this stock, these shares are assumed to be converted.

All outstanding options to purchase shares of common stock at the end of Fiscal 2014, 2013 and 2012 were included in the computation of diluted earnings per share because the options' exercise prices were less than the average market price of the common shares.

The weighted shares outstanding reflects the effect of stock buy back programs. The Company repurchased 337,665 shares at a cost of \$20.7 million during Fiscal 2014. The Company has \$65.5 million remaining under its current \$75.0 million share repurchase authorization. The Company repurchased 645,904 shares at a cost of \$37.7 million during Fiscal 2013. The Company did not repurchase any shares during Fiscal 2012.

Note 12 Share-Based Compensation Plans

The Company's stock-based compensation plans, as of February 1, 2014, are described below. The Company recognizes compensation expense for share-based payments based on the fair value of the awards as required by the Compensation – Stock Compensation Topic of the Codification.

Stock Incentive Plans

The Company has two fixed stock incentive plans. Under the 2009 Amended and Restated Equity Incentive Plan (the "2009 Plan"), effective as of June 22, 2011, the Company may grant options, restricted shares, performance awards and other stock-based awards to its employees, consultants and directors for up to 2.5 million shares of common stock. Under the 2005 Equity Incentive Plan (the "2005 Plan"), effective as of June 23, 2005, the Company was permitted to grant options, restricted shares and other stock-based awards to its employees and consultants as well as directors for up to 2.5 million shares of common stock. There will be no future awards under the 2005 Equity Incentive Plan. Under both plans, the exercise price of each option equals the market price of the Company's stock on the date of grant, and an option's maximum term is 10 years. Options granted under both plans vest 25% per year over four years.

For Fiscal 2014, 2013 and 2012, the Company recognized stock option related share-based compensation of \$0.0, \$0.0 and less than \$1,000, respectively, for its fixed stock incentive plans included in selling and administrative expenses in the accompanying Consolidated Statements of Operations. The Company did not capitalize any share-based compensation cost.

The Compensation—Stock Compensation Topic of the Codification requires that the cash flows resulting from tax benefits for tax deductions in excess of the compensation cost recognized for those options (excess tax benefit) be classified as financing cash flows. Accordingly, the Company classified excess tax benefits of \$3.8 million, \$4.8 million and \$4.7 million as financing cash inflows rather than as operating cash inflows on its Consolidated Statement of Cash Flows for Fiscal 2014, 2013 and 2012, respectively.

The Company did not grant any fixed stock options in Fiscal 2014, 2013 or 2012.

Note 12 Share-Based Compensation Plans, Continued

A summary of fixed stock option activity and changes for Fiscal 2014, 2013 and 2012 is presented below:

	Options	Weighted-Average Exercise Price	Weighted-Average Remaining Contractual Term	Va	gate Intrinsic alue (in 1sands) ⁽¹⁾
Outstanding, January 29, 2011	877,130	\$ 24.75			
Granted	—	—			
Exercised	(390,357)	24.82			
Forfeited	—	—			
Outstanding, January 28, 2012	486,773	\$ 24.70			
Granted		_			
Exercised	(223,618)	21.50			
Forfeited	_				
Outstanding, February 2, 2013	263,155	\$ 27.43			
Granted		_			
Exercised	(130,051)	23.33			
Forfeited	(2,250)	17.50			
Outstanding, February 1, 2014	130,854	\$ 31.67	1.58	\$	5,045
Exercisable, February 1, 2014	130,854	\$ 31.67	1.58	\$	5,045

(1) Based upon the difference between the closing market price of the Company's common stock on the last trading day of the year and the grant price of in-the-money options.

The total intrinsic value, which represents the difference between the underlying stock's market price and the option's exercise price, of options exercised during Fiscal 2014, 2013 and 2012 was \$6.1 million, \$11.5 million and \$10.3 million, respectively.

As of February 1, 2014, the Company does not have any nonvested options of its fixed stock incentive plans.

As of February 1, 2014, there was no unrecognized compensation costs related to stock options under the 2009 Plan. Cash received from option exercises under all share-based payment arrangements for Fiscal 2014, 2013 and 2012 was \$3.0 million, \$4.8 million and \$9.7 million, respectively.

Restricted Stock Incentive Plans

Director Restricted Stock

The 2009 Plan permits grants to non-employee directors on such terms as the Company's board of directors may approve. Restricted stock awards were made to independent directors on the date of the annual meeting of shareholders in each of Fiscal 2014, 2013 and 2012. The shares granted in each award vested on the first anniversary of the grant date, subject to the director's continued service through that date. The board of directors also approved a grant of 365 additional shares in Fiscal 2014 and 336 additional shares in Fiscal 2013 to a newly elected director each year on the annual meeting date in

Note 12 Share-Based Compensation Plans, Continued

Fiscal 2014 and 2013 on the same terms as the Fiscal 2014 and 2013 grant to all outside directors. In all cases, the director is restricted from selling, transferring, pledging or assigning the shares for three years from the grant date unless he or she earlier leaves the board.

The Fiscal 2012 grants were valued at \$70,000 for each director and Fiscal 2013 and 2014 grants were each valued at \$80,000 per director based on the average closing price of the stock for the first five trading days of the month in which they were granted and vested on the first anniversary of the grant date. For Fiscal 2014, 2013 and 2012, the Company issued 9,280 shares, 9,888 shares and 14,643 shares, respectively, of director restricted stock.

For Fiscal 2014, 2013 and 2012, the Company recognized \$1.0 million, \$0.9 million and \$0.8 million, respectively, of director restricted stock related share-based compensation in selling and administrative expenses in the accompanying Consolidated Statements of Operations.

Employee Restricted Stock

Under the 2009 Plan, the Company issued 199,392 shares, 194,232 shares and 289,407 shares of employee restricted stock in Fiscal 2014, 2013 and 2012, respectively. Shares of employee restricted stock issued in Fiscal 2012, 2013 and 2014 vest 25% per year over four years, provided that on such date the grantee has remained continuously employed by the Company since the date of grant. The fair value of employee restricted stock is charged against income as compensation cost over the vesting period. Compensation cost recognized in selling and administrative expenses in the accompanying Consolidated Statements of Operations for these shares was \$11.3 million, \$9.6 million and \$6.9 million for Fiscal 2014, 2013 and 2012, respectively.

Note 12 Share-Based Compensation Plans, Continued

A summary of the status of the Company's nonvested shares of its employee restricted stock as of February 1, 2014 is presented below:

Nonvested Restricted Shares	Shares	Č	ghted-Average Grant-Date Fair Value
Nonvested at January 29, 2011	818,119	\$	23.95
Granted	289,407		45.14
Vested	(227,691)		22.58
Withheld for federal taxes	(93,089)		22.42
Forfeited	(14,081)		27.38
Nonvested at January 28, 2012	772,665		32.41
Granted	194,232		57.58
Vested	(195,203)		29.95
Withheld for federal taxes	(75,552)		29.97
Forfeited	(3,360)		38.96
Nonvested at February 2, 2013	692,782		40.59
Granted	199,392		65.11
Vested	(199,428)		34.31
Withheld for federal taxes	(105,193)		34.42
Forfeited	(6,279)		46.48
Nonvested at February 1, 2014	581,274	\$	52.21

As of February 1, 2014, there was \$23.1 million of total unrecognized compensation costs related to nonvested share-based compensation arrangements for restricted stock discussed above. That cost is expected to be recognized over a weighted average period of 1.25 years.

Employee Stock Purchase Plan

Under the Employee Stock Purchase Plan, the Company is authorized to issue up to 1.0 million shares of common stock to qualifying full-time employees whose total annual base salary is less than \$90,000, effective October 1, 2002. Prior to October 1, 2002, the total annual base salary was limited to \$100,000. Under the terms of the Plan, employees could choose each year to have up to 15% of their annual base earnings or \$8,500, whichever is lower, withheld to purchase the Company's common stock. The purchase price of the stock was 85% of the closing market price of the stock on either the exercise date or the grant date, whichever was less. The Company's board of directors amended the Company's Employee Stock Purchase Plan effective October 1, 2005 to provide that participants may acquire shares under the Plan at a 5% discount from fair market value on the last day of the Plan year. Employees can choose each year to have up to 15% of their annual base earnings or \$9,500, whichever is lower, withheld to purchase the Company's common stock. Under the Compensation – Stock Compensation Topic of the Codification, shares issued under the Plan as amended are non-compensatory. Under the Plan, the Company sold 3,146 shares, 2,463 shares and 2,717 shares to employees in Fiscal 2014, 2013 and 2012, respectively.

Note 13 <u>Legal Proceedings</u>

Environmental Matters

New York State Environmental Matters

In August 1997, the New York State Department of Environmental Conservation ("NYSDEC") and the Company entered into a consent order whereby the Company assumed responsibility for conducting a remedial investigation and feasibility study ("RIFS") and implementing an interim remedial measure ("IRM") with regard to the site of a knitting mill operated by a former subsidiary of the Company from 1965 to 1969. The Company undertook the IRM and RIFS voluntarily, without admitting liability or accepting responsibility for any future remediation of the site. The Company has completed the IRM and the RIFS. In the course of preparing the RIFS, the Company identified remedial alternatives with estimated undiscounted costs ranging from \$0.0 million to \$24.0 million, excluding amounts previously expended or provided for by the Company. The United States Environmental Protection Agency ("EPA"), which has assumed primary regulatory responsibility for the site from NYSDEC, issued a Record of Decision in September 2007. The Record of Decision requires a remedy of a combination

of groundwater extraction and treatment and in-site chemical oxidation at an estimated present cost of approximately \$10.7 million.

In July 2009, the Company agreed to a Consent Order with the EPA requiring the Company to perform certain remediation actions, operations, maintenance and monitoring at the site. In September 2009, a Consent Judgment embodying the Consent Order was filed in the U.S. District Court for the Eastern District of New York.

The Village of Garden City, New York (the "Village"), has additionally asserted that the Company is liable for the costs associated with enhanced treatment required by the impact of the groundwater plume from the site on two public water supply wells, including historical total costs ranging from approximately \$1.8 million to in excess of \$2.5 million, and future operation and maintenance costs which the Village estimates at \$126,400 annually while the enhanced treatment continues. On December 14, 2007, the Village filed a complaint against the Company and the owner of the property under the Resource Conservation and Recovery Act ("RCRA"), the Safe Drinking Water Act, and the Comprehensive Environmental Response, Compensation and Liability Act ("CERCLA") as well as a number of state law theories in the U.S. District Court for the Eastern District of New York, seeking an injunction requiring the defendants to remediate contamination from the site and to establish their liability for future costs that may be incurred in connection with it, which the complaint alleges could exceed \$41 million, undiscounted, over a 70-year period. The Company has not verified the estimates of either historic or future costs asserted by the Village, but believes that an estimate of future costs based on a 70-year remediation period is unreasonable given the expected remedial period reflected in the EPA's Record of Decision. On May 23, 2008, the Company filed a motion to dismiss the Village's complaint on grounds including applicable statutes of limitation and preemption of certain claims by the NYSDEC's and the EPA's diligent prosecution of remediation. On January 27, 2009, the Court granted the motion to dismiss all counts of the plaintiff's complaint except for the CERCLA claim and a state law claim for indemnity for costs incurred after November 27, 2000. On September 23, 2009, on a motion for reconsideration by the Village, the Court reinstated the claims for injunctive relief under RCRA and for equitable relief under certain of the state law theories. The Company intends to continue to defend the action if an acceptable settlement agreement cannot be reached.

Note 13 <u>Legal Proceedings, Continued</u>

Whitehall Environmental Matters

The Company has performed sampling and analysis of soil, sediments, surface water, groundwater and waste management areas at the Company's former Volunteer Leather Company facility in Whitehall, Michigan.

In October 2010, the Company and the Michigan Department of Natural Resources and Environment entered into a Consent Decree providing for implementation of a remedial Work Plan for the facility site designed to bring the site into compliance with applicable regulatory standards. The Work Plan's implementation is substantially complete and the Company expects, based on its present understanding of the condition of the site, that its future obligations with respect to the site will be limited to periodic monitoring and that future costs related to the site should not have a material effect on its financial condition or results of operations.

Accrual for Environmental Contingencies

Related to all outstanding environmental contingencies, the Company had accrued \$11.9 million as of February 1, 2014, \$11.9 million as of February 2, 2013 and \$13.0 million as of January 28, 2012. All such provisions reflect the Company's estimates of the most likely cost (undiscounted, including both current and noncurrent portions) of resolving the contingencies, based on facts and circumstances as of the time they were made. There is no assurance that relevant facts and circumstances will not change, necessitating future changes to the provisions. Such contingent liabilities are included in the liability

arising from provision for discontinued operations on the accompanying Consolidated Balance Sheets because it relates to former facilities operated by the Company. The Company has made pretax accruals for certain of these contingencies, including approximately \$0.5 million reflected in Fiscal 2014, \$0.8 million reflected in Fiscal 2013 and \$1.8 million reflected in Fiscal 2012. These charges are included in provision for discontinued operations, net in the Consolidated Statements of Operations and represent changes in estimates.

Other Matters

On December 10, 2010, the Company announced that it had suffered a criminal intrusion into the portion of its computer network that processes payments for transactions in certain of its retail stores. Visa, Inc., MasterCard Worldwide and American Express Travel Related Services Company, Inc. have asserted claims totaling approximately \$15.6 million in connection with the intrusion and the claims of two of the claimants have been collected by withholding payment card receivables of the Company. In the fourth quarter of Fiscal 2013, the Company recorded a \$15.4 million charge to earnings in connection with the disputed liability. On March 7, 2013, the Company filed an action in the U.S. District Court for the Middle District of Tennessee against Visa U.S.A. Inc., Visa Inc. and Visa International Service Association seeking to recover \$13.3 million in non-compliance fines and issuer reimbursement assessments collected from the Company in connection with the intrusion. The Company does not currently expect any future claims in connection with the intrusion to have a material effect on its financial condition, cash flows, or results of operations.

Note 13 <u>Legal Proceedings, Continued</u>

On January 5, 2012, a patent infringement action against the Company and numerous other defendants was filed in the U.S. District Court for the Eastern District of Texas, *GeoTag, Inc. v. Circle K Store, Inc., et al.*, alleging that features of certain of the Company's e-commerce websites infringe U.S. Patent No. 5,930,474, entitled "Internet Organizer for Accessing Geographically and Topically Based Information." The plaintiff sought relief including damages for the alleged infringement, costs, expenses and pre- and post-judgment interest and injunctive relief. Pursuant to a settlement agreement, the matter was dismissed on February 28, 2014. The settlement did not have a material effect on its financial condition or results of operations.

On June 13, 2012, a former vendor of a subsidiary of the Company filed an action, *Perfect Curve, Inc. v. Hat World, Inc.*, in U.S. District Court in Massachusetts, alleging patent, trademark, trade dress, and copyright infringement against the subsidiary based on the sale of a line of products developed by the subsidiary. The parties reached agreement to settle the matter and the action was dismissed pursuant to a Stipulated Dismissal dated March 13, 2014. The settlement did not have a material effect on its financial condition or results of operations.

On May 14, 2012, a putative class and collective action, *Maro v. Hat World, Inc.*, was filed in the U.S. District Court for the Northern District of Illinois. The action alleges that the Company failed to pay

the plaintiff and other, similarly situated retail store employees of Hat World, Inc., for time spent making bank deposits of store collections, and seeks to recover unpaid wages, liquidated damages, statutory penalties, attorney's fees, and costs pursuant to the federal Fair Labor Standards Act, the Illinois Minimum Wage Law and the Illinois Wage Payment and Collection Act. On January 15, 2014, the court dismissed the *Maro* case with prejudice, based on the plaintiffs' failure to prosecute. On July 16, 2012 and July 30, 2012, additional putative class and collective actions, *Chavez v. Hat World, Inc.* and *Dismukes v. Hat World, Inc.*, were filed in the same court, alleging that certain Hat World employees were misclassified as exempt from overtime pay, and seeking similar relief. The *Chavez* and *Dismukes* actions have been consolidated. The parties have reached an agreement in principle to resolve the matter, subject to documentation and court approval. The Company does not expect the matter or its settlement as proposed to have a material effect on its financial condition or results of operations.

On August 30, 2012, a former employee of a Company subsidiary filed a putative class and collective action, *Kershner v. Hat World, Inc.*, in the Philadelphia, Pennsylvania Court of Common Pleas alleging violations of the Pennsylvania Minimum Wage Act by the subsidiary. The Company has reached an agreement to resolve the matter, subject to approval by the court. On February 10, 2014, the court granted preliminary approval of the proposed settlement. The Company does not expect the matter or its settlement as proposed to have a material effect on its financial condition or results of operations.

On May 23, 2013, a former employee of the Company filed an action, *Everett v. Genesco Inc.*, in the U.S. District Court for the Middle District of Florida alleging violations of the Fair Labor Standards Act, seeking designation as a collective action and the award of allegedly unpaid minimum wages, overtime pay, liquidated damages, penalties, interest, attorneys' fees, and other relief. The Company disputes the material allegations in the action and intends to defend it.

Note 13 <u>Legal Proceedings, Continued</u>

On May 17, 2013, a former employee filed a putative class and representative action, *Garcia v. Genesco, Inc.* in the Superior Court of California for the County of Ventura, alleging various claims under the California Labor Code, including failure to provide meal and rest periods, failure to timely pay wages, failure to provide accurate itemized wage statements, and unfair competition and violation of the Private Attorneys' General Act of 2004, and seeking unspecified damages and penalties. On August 30, 2013, the Company removed the action to the United States District Court for the Central District of California. The Company disputes the material allegations in the complaint and is defending the matter.

In addition to the matters specifically described in this Note, the Company is a party to other legal and regulatory proceedings and claims arising in the ordinary course of its business. While management does not believe that the Company's liability with respect to any of these other matters is likely to have a material effect on its financial position, cash flows, or results of operations, legal proceedings are subject to inherent uncertainties and unfavorable rulings could have a material adverse impact on the Company's business and results of operations.

Note 14 <u>Business Segment Information</u>

During Fiscal 2014, the Company operated five reportable business segments (not including corporate): (i) Journeys Group, comprised of the Journeys, Journeys Kidz, Shi by Journeys and Underground by Journeys retail footwear chains, catalog and e-commerce operations; (ii) Schuh Group, comprised of the Schuh retail footwear chain and e-commerce operations; (iii) Lids Sports Group, comprised primarily of the Lids, Hat World and Hat Shack retail headwear stores, the Lids Locker Room and Lids Clubhouse fan shops (operated under various trade names), licensed team merchandise departments in Macy's department stores operated under the name of Locker Room by Lids under a license agreement with Macy's, the Lids Team Sports business and certain e-commerce operations; (iv) Johnston & Murphy Group, comprised of Johnston & Murphy retail operations, catalog and e-commerce operations and wholesale distribution of products under the Johnston & Murphy and Trask brands; and (v) Licensed Brands, comprised of Dockers[®] Footwear, sourced and marketed under a license from Levi Strauss &

Company; SureGrip[®] Footwear, occupational footwear primarily sold directly to consumers; and other brands.

The accounting policies of the segments are the same as those described in the summary of significant accounting policies.

The Company's reportable segments are based on management's organization of the segments in order to make operating decisions and assess performance along types of products sold. Journeys Group, Schuh Group and Lids Sports Group sell primarily branded products from other companies while Johnston & Murphy Group and Licensed Brands sell primarily the Company's owned and licensed

brands.

Note 14 <u>Business Segment Information, Continued</u>

Corporate assets include cash, domestic prepaid rent expense, prepaid income taxes, deferred income taxes, deferred note expense and corporate fixed assets. The Company charges allocated retail costs of distribution to each segment. The Company does not allocate certain costs to each segment in order

to make decisions and assess performance. These costs include corporate overhead, interest expense, interest income, asset impairment charges and other, including major litigation and major lease terminations.

Fiscal 2014

In thousands	urneys roup		Schuh Group	L	ids Sports Group	8	Johnston & Murphy Group	Licensed Brands	Corporate & Other	Consolidated
Sales	\$ 1,082,241	\$	364,732	\$	821,779	\$	245,941	\$ 109,989	\$ 1,282	\$ 2,625,964
Intercompany sales	—	—	—		(783)		—	(209)	—	(992)
Net sales to external customers	\$ 1,082,241	\$	364,732	\$	820,996	\$	245,941	\$ 109,780	\$ 1,282	\$ 2,624,972
Segment operating income (loss)	\$ 97,377	\$	3,063	\$	63,748	\$	17,638	\$ 10,614	\$ (27,664)	\$ 164,776
Asset Impairments and other*	_		_		—		_	_	(1,341)	(1,341)
Earnings (loss) from operations	97,377		3,063		63,748		17,638	10,614	(29,005)	163,435
Interest expense	_		—		—		_	_	(4,641)	(4,641)
Interest income	—		—		—		—	—	66	66
Earnings (loss) from continuing operations before income taxes	\$ 97,377	\$	3,063	\$	63,748	\$	17,638	\$ 10,614	\$ (33,580)	\$ 158,860
Total assets**	\$ 298,105	\$	268,514	\$	574,664	\$	97,532	\$ 50,955	\$ 149,514	\$ 1,439,284
Depreciation and amortization	19,400		11,339		28,345		4,002	468	3,581	67,135
Capital expenditures	20,223		29,673		35,193		9,178	1,452	2,737	98,456

*Asset Impairments and other includes a \$2.3 million charge for asset impairments, of which \$1.4 million is in the Lids Sports Group, \$0.6 million is in the Journeys Group and \$0.3 million is in the Johnston & Murphy Group, a \$3.3 million charge for network intrusion costs, a \$2.4 million charge for other legal matters and a \$1.6 million charge for a lease termination partially offset by a gain of \$(8.3) million for the lease termination of a New York City Journeys store.

**Total assets for the Lids Sports Group, Schuh Group and Licensed Brands include \$182.4 million, \$104.9 million and \$0.8 million of goodwill, respectively. Goodwill for Lids Sports Group includes \$10.1 million of additions in Fiscal 2014 resulting from small acquisitions and the Schuh Group goodwill increased by \$4.2 million due to foreign currency translation adjustment.

<u>Note 14</u> <u>Business Segment Information, Continued</u>

Fiscal										
In thousands	urneys oup	Sc	huh Group	L	ids Sports Group	Johnston & Murphy Group	Licensed Brands	Corporate & Other	C	Consolidated
Sales	\$ 1,111,490		370,480	\$	793,016	\$ 221,870	\$ 108,808	\$ 1,234	\$	2,606,898
Intercompany sales	_		—		(1,761)	(10)	(310)	_		(2,081)
Net sales to external customers	\$ 1,111,490	\$	370,480	\$	791,255	\$ 221,860	\$ 108,498	\$ 1,234	\$	2,604,817
Segment operating income (loss)	\$ 109,953	\$	11,209	\$	82,867	\$ 15,696	\$ 10,078	\$ (42,903)	\$	186,900
Asset Impairments and other*	_		—		_	—	—	(17,037)		(17,037)
Earnings (loss) from operations	109,953		11,209		82,867	15,696	10,078	(59,940)		169,863
Interest expense	_		—		_	—	—	(5,126)		(5,126)
Interest income	_		—		_	—	—	95		95
Earnings (loss) from continuing operations before income taxes	\$ 109,953	\$	11,209	\$	82,867	\$ 15,696	\$ 10,078	\$ (64,971)	\$	164,832
Total assets**	\$ 280,396		231,323	\$	519,006	\$ 89,505	\$ 43,212	\$ 162,630	\$	1,326,072
Depreciation and amortization	20,190		10,040		26,892	3,738	366	2,471		63,697
Capital expenditures	21,852		16,873		21,448	6,680	1,255	3,629		71,737

*Asset Impairments and other includes a \$1.4 million charge for asset impairments, of which \$0.9 million is in the Lids Sports Group, \$0.4 million is in the Journeys Group, and \$0.1 million is in the Johnston & Murphy Group, a \$15.6 million charge for network intrusion costs and a \$0.1 million charge for other legal matters.

**Total assets for the Lids Sports Group, Schuh Group and Licensed Brands include \$172.3 million, \$100.7 million and \$0.8 million of goodwill, respectively. Goodwill for Lids Sports Group includes \$13.2 million of additions in Fiscal 2013 resulting from small acquisitions and the Schuh Group goodwill increased by \$0.8 million due to foreign currency translation adjustment.

Note 14 Business Segment Information, Continued

Fiscal 20	Fiscal 2012											
In thousands		urneys roup	Sc	huh Group	L	ids Sports Group		Johnston & Murphy Group		Licensed Brands	Corporate & Other	Consolidated
Sales	\$	1,020,116	\$	212,262	\$	759,671	\$	201,725	\$	97,721	\$ 1,116	\$ 2,292,611
Intercompany sales		—				(347)		—		(277)	—	(624)
Net sales to external customers	\$	1,020,116	\$	212,262	\$	759,324	\$	201,725	\$	97,444	\$ 1,116	\$ 2,291,987
Segment operating income (loss)	\$	89,045	\$	11,711	\$	86,037	\$	13,743	\$	9,451	\$ (45,825)	\$ 164,162
Asset Impairments and other*		—		—		—		—		—	(2,677)	(2,677)
Earnings (loss) from operations		89,045		11,711		86,037		13,743		9,451	(48,502)	161,485
Interest expense		—		—		—		—		—	(5,157)	(5,157)
Interest income											65	65
Earnings (loss) from continuing operations before income taxes	\$	89,045	\$	11,711	\$	86,037	\$	13,743	\$	9,451	\$ (53,594)	\$ 156,393
Total assets**	\$	259,331	\$	205,313	\$	489,512	\$	79,321	\$	34,974	\$ 161,310	\$ 1,229,761
Depreciation and amortization		20,742		4,602		22,541		3,538		285	2,029	53,737
Capital expenditures		11,125		7,406		24,497		1,894		718	3,816	49,456

*Asset Impairments and other includes a \$1.1 million charge for asset impairments, of which \$0.6 million is in the Journeys Group, \$0.3 million is in the Lids Sports Group and \$0.2 million is in the Johnston & Murphy Group, a \$0.7 million charge for network intrusion costs and a \$0.9 million charge for other legal matters.

**Total assets for the Lids Sports Group, Schuh Group and Licensed Brands include \$159.1 million, \$99.9 million and \$0.8 million of goodwill, respectively. Goodwill for the Lids Sports Group includes \$6.5 million of additions in Fiscal 2012 resulting from small acquisitions and the Schuh Group goodwill is due to the acquisition of Schuh in the second quarter of Fiscal 2012 of \$102.9 million which has been decreased by \$3.0 million due to foreign currency translation adjustment.

Note 15 **Quarterly Financial Information (Unaudited)**

(In thousands,	 1st (Quar	ter	_	 2n	d Qua	arter		 3rc	l Qua	rter	_	41	th Qu	arter	_	Fisca	ıl Year
except per share amounts)	2014		2013		2014		2013		2014		2013		2014 ^(a)		2013		2014 ^(b)	2013
Net sales	\$ 591,388	-	\$ 600,144		\$ 574,746	_	\$ 543,522	_	\$ 666,332	-	\$ 664,458		\$ 792,506	-	\$ 796,693	-	\$ 2,624,972	\$ 2,604,817
Gross margin	298,437		306,906		282,808		273,059		332,161		334,412		385,644		384,240		1,299,050	1,298,617
Earnings from continuing operations before income taxes	24,685	(1)	35,886	(3)	14,388	(5)	14,638	(7)	45,789	(8)	52,907	(9)	73,998	(11)	61,401	(13)	158,860	164,832
Earnings from continuing operations	14,509		21,754		8,465		10,009		27,796		42,221		42,212		38,913		92,982	112,897
Net earnings	14,410	(2)	21,577	(4)	8,340	(6)	9,968		27,750		42,127	(10)	42,153	(12)	38,763	(14)	92,653	112,435
Diluted earnings per common share:																		
Continuing operations	0.61		0.88		0.36		0.41		1.18		1.76		1.79		1.64		3.94	4.69
Net earnings	 0.61	_	0.87		 0.35		0.40		 1.18	_	1.76		1.79	_	1.63	_	3.92	4.68

Includes a net asset impairment and other charge of \$1.3 million (see Note 3). (a) 13 week period vs. 14 (1)

(2)Includes a loss of \$0.1 million, net of tax, from discontinued operations (see Note 3). weeks in prior period.

Includes a net asset impairment and other charge of \$0.1 million (see Note 3). (b) 52 week period vs. 53 (3)

(4) Includes a loss of \$0.2 million, net of tax, from discontinued operations (see Note 3). weeks in prior period.

Includes a net asset impairment and other credit of \$(7.1) million (see Note 3). (5)

(6) Includes a loss of \$0.1 million, net of tax, from discontinued operations (see Note 3).

(7) Includes a net asset impairment and other charge of \$0.4 million (see Note 3).

(8) Includes a net asset impairment and other charge of \$1.5 million (see Note 3).

Includes a net asset impairment and other charge of \$0.4 million (see Note 3). (9)

Includes a loss of \$0.1 million, net of tax, from discontinued operations (see Note 3). (10)Includes a net asset impairment and other charge of \$5.6 million (see Note 3).

(11)

Includes a loss of \$0.1 million, net of tax, from discontinued operations (see Note 3). (12)

(13) Includes a net asset impairment and other charge of \$16.1 million (see Note 3).

Includes a loss of \$0.2 million, net of tax, from discontinued operations (see Note 3). (14)

ITEM 9, CHANGES IN AND DISAGREEMENTS WITH ACCOUNTANTS ON ACCOUNTING AND FINANCIAL DISCLOSURE

None.

ITEM 9A, CONTROLS AND PROCEDURES

Evaluation of disclosure controls and procedures.

We have established disclosure controls and procedures to ensure that material information relating to the Company, including its consolidated subsidiaries, is made known to the officers who certify the Company's financial reports and to other members of senior management and Board of Directors.

Based on their evaluation as of February 1, 2014, the principal executive officer and principal financial officer of the Company have concluded that the Company's disclosure controls and procedures as defined in Rules 13a-15(e) and 15d-15(e) under the Securities Exchange Act of 1934, as amended (the "Exchange Act"), were effective to ensure that the information required to be disclosed by the Company in the reports that it files or submits under the Exchange Act is (i) recorded, processed, summarized and reported, within the time periods specified in the SEC's rules and forms, and (ii) accumulated and communicated to management, including the principal executive and principal financial officers, or persons performing similar functions, as appropriate, to allow timely decisions regarding required disclosure.

Management's report on internal control over financial reporting.

Management of the Company is responsible for establishing and maintaining effective internal control over financial reporting as defined in Rule 13a-15(f) under the Exchange Act. The Company's internal control over financial reporting is a process designed to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles.

Because of its inherent limitations, internal control over financial reporting may not prevent or detect misstatements. Therefore, even those systems determined to be effective can provide only reasonable assurance with respect to financial statement preparation and presentation.

Management assessed the effectiveness of the Company's internal control over financial reporting as of February 1, 2014. In making this assessment, management used the criteria set forth in *Internal Control – Integrated Framework* (1992) drafted by the Committee of Sponsoring Organizations of the Treadway Commission (COSO). Based on this assessment, management believes that, as of February 1, 2014, the Company's internal control over financial reporting was effective based on those criteria.

Ernst & Young LLP, the independent registered public accounting firm who also audited the Company's Consolidated Financial Statements, has issued an attestation report on the Company's effectiveness of internal control over financial reporting which is included herein.

Changes in internal control over financial reporting.

There were no changes in the Company's internal control over financial reporting that occurred during the Company's last fiscal quarter that have materially affected or are reasonable likely to materially affect the Company's internal control over financial reporting.

ITEM 9B, OTHER INFORMATION

Not applicable.

PART III

ITEM 10, DIRECTORS, EXECUTIVE OFFICERS AND CORPORATE GOVERNANCE

Certain information required by this item is incorporated herein by reference to the sections entitled "Election of Directors," "Corporate Governance" and "Section 16(a) Beneficial Ownership Reporting Compliance" in the Company's definitive proxy statement for its annual meeting of shareholders to be held June 26, 2014, to be filed with the Securities and Exchange Commission. Pursuant to General Instruction G(3), certain information concerning the executive officers of the Company appears under the caption "Executive Officers of the Registrant" in this report following Item 4 of Part I.

The Company has a code of ethics (the "Code of Ethics") that applies to all of its directors, officers (including its chief executive officer, chief financial officer and chief accounting officer) and employees. The Company has made the Code of Ethics available and intends to post any legally required amendments to, or waivers of, such Code of Ethics on its website at http://www.genesco.com. Our website address is provided as an inactive textual reference only. The information provided on our website is not a part of this report, and therefore is not incorporated herein by reference.

ITEM 11, EXECUTIVE COMPENSATION

The information required by this item is incorporated herein by reference to the sections entitled "Director Compensation," "Compensation Committee Report" and "Executive Compensation" in the Company's definitive proxy statement for its annual meeting of shareholders to be held June 26, 2014, to be filed with the Securities and Exchange Commission.

ITEM 12, SECURITY OWNERSHIP OF CERTAIN BENEFICIAL OWNERS AND MANAGEMENT AND RELATED STOCKHOLDER MATTERS

Certain information required by this item is incorporated herein by reference to the section entitled "Security Ownership of Officers, Directors and Principal Shareholders" in the Company's definitive proxy statement for its annual meeting of shareholders to be held June 26, 2014, to be filed with the Securities and Exchange Commission.

The following table provides certain information as of February 1, 2014 with respect to our equity compensation plans:

EQUITY COMPENSATION PLAN INFORMATION*

Plan Category	(a) Number of securities to be issued upon exercise of outstanding options, warrants and rights	(b) Weighted-average exercise price of outstanding options, warrants and rights	(c) Number of securities remaining available for future issuance under equity compensation plans (excluding securities reflected in column (a)) (1)
Equity compensation plans approved by security holders	130,854	\$ 31.67	1,514,954
Equity compensation plans not approved by security holders			
Total	130,854	\$ 31.67	1,514,954

(1) Such shares may be issued as restricted shares or other forms of stock-based compensation pursuant to our stock incentive plans.

For additional information concerning our equity compensation plans, see the discussion in Note 1 in the Notes to Consolidated Financial Statements —Summary of Significant Accounting Policies – Share-Based Compensation and Note 12 Share-Based Compensation Plans.

ITEM 13, CERTAIN RELATIONSHIPS AND RELATED TRANSACTIONS, AND DIRECTOR INDEPENDENCE

The information required by this item is incorporated herein by reference to the section entitled "Election of Directors" in the Company's definitive proxy statement for its annual meeting of shareholders to be held June 26, 2014, to be filed with the Securities and Exchange Commission.

ITEM 14, PRINCIPAL ACCOUNTING FEES AND SERVICES

The information required by this item is incorporated herein by reference to the section entitled "Audit Matters" in the Company's definitive proxy statement for its annual meeting of shareholders to be held June 26, 2014, to be filed with the Securities and Exchange Commission.

PART IV

ITEM 15, EXHIBITS AND FINANCIAL STATEMENT SCHEDULES

Financial Statements

The following consolidated financial statements of Genesco Inc. and Subsidiaries (the "Company") are filed as part of this report under Item 8, Financial Statements and Supplementary Data

Report of Independent Registered Public Accounting Firm on Internal Control over Financial Reporting

Report of Independent Registered Public Accounting Firm on Financial Statements

Consolidated Balance Sheets, February 1, 2014 and February 2, 2013

Consolidated Statements of Operations, each of the three fiscal years ended 2014, 2013 and 2012

Consolidated Statements of Comprehensive Income, each of the three fiscal years ended 2014, 2013 and 2012

Consolidated Statements of Cash Flows, each of the three fiscal years ended 2014, 2013 and 2012

Consolidated Statements of Equity, each of the three fiscal years ended 2014, 2013 and 2012

Notes to Consolidated Financial Statements

Financial Statement Schedules

Schedule 2 — Valuation and Qualifying Accounts, each of the three fiscal years ended 2014, 2013 and 2012

All other schedules are omitted because the required information is either not applicable or is presented in the financial statements or related notes. These schedules begin on page 111.

Exhibits

(2)	a.	Agreement and Plan of Merger, dated as of February 5, 2004, by and among Genesco Inc., HWC Merger Sub, Inc. and Hat World Corporation. Incorporated by reference to Exhibit (2)a to the current report on Form 8-K filed April 9, 2004 (File No. 1-3083).
	b.	Stock Purchase Agreement, dated December 9, 2006, by and among Hat World, Inc., Hat Shack, Inc. and all the shareholders of Hat Shack, Inc. Incorporated by reference to Exhibit 10.1 to the current report on Form 8-K filed December 12, 2006 (File No. 1-3083).
	c.	Sale and Purchase Agreement, dated as of June 23, 2011, by and among Genesco Inc., Schuh Group Limited, Genesco (UK) Limited and the persons listed on Schedule 1 thereto. (Pursuant to Item 601(b)(2) of Regulation S-K, the schedules and exhibits from this agreement are omitted, but will be provided supplementally to the Commission upon request.) Incorporated by reference to Exhibit 2.1 to the current report on Form 8-K filed June 28, 2011 (File No. 1-3083).
	d.	£25 million Loan Note Instrument of Genesco (UK) Limited dated June 23, 2011. Incorporated by reference to Exhibit 2.2 to the current report on Form 8-K filed June 28, 2011 (File No. 1-3083).
(3)	a.	Amended and Restated Bylaws of Genesco Inc. Incorporated by reference to Exhibit 3.1 to the current report on Form 8-K filed December 19, 2007 (File No. 1-3083).
	b.	Restated Charter of Genesco Inc., as amended. Incorporated by reference to Exhibit 1 to the Genesco Inc. Registration Statement on Form 8-A/A filed with the SEC on May 1, 2003 (File No.1-3083).
(4)	a.	Second Amended and Restated Rights Agreement dated as of April 18, 2010. Incorporated by reference to Exhibit 4.1 to the current report on Form 8-K filed April 9, 2010 (File No. 1-3083).
	b.	Form of Certificate for the Common Stock. Incorporated by reference to Exhibit 3 to the Genesco Inc. Registration Statement on Form 8-A/A filed with the SEC on May 1, 2003 (File No.1-3083).
		105

Table of Contents

(10)

- a. Third Amended and Restated Credit Agreement, dated as of January 31, 2014, by and among Genesco Inc., certain subsidiaries of the Genesco Inc. party thereto, as other domestic borrowers, GCO Canada Inc., Genesco (UK)
 Limited, the lenders party thereto and Bank of America, N.A., as Agent. Incorporated by reference to Exhibit 10.1 to the current report on Form 8-K filed February 5, 2014 (File No. 1-3083).
 - b. Amendment and Restatement Agreement dated November 1, 2013 between Schuh Group Limited as Parent and others as Borrowers and Guarantors, Lloyds Bank PLC as Arranger, Agent and Security Trustee.
 - c. Form of Split-Dollar Insurance Agreement with Executive Officers. Incorporated by reference to Exhibit (10)a to the Company's Annual Report on Form 10-K for the fiscal year ended February 1, 1997 (File No.1-3083).
 - d. 1996 Stock Incentive Plan Amended and Restated as of October 24, 2007. Form of Option Agreement. Incorporated by reference to Exhibit (10)c to the Company's Annual Report on Form 10-K for the fiscal year ended February 3, 2007 (File No.1-3083).
 - e. Genesco Inc. 2005 Equity Incentive Plan Amended and Restated as of October 24, 2007. Incorporated by reference to Exhibit (10)d to the Company's Annual Report on Form 10-K for the fiscal year ended February 2, 2008 (File No.1-3083).
 - f. Genesco Inc. 2009 Equity Incentive Plan. Incorporated by reference to Exhibit A to the Company's definitive proxy statement dated May 15, 2009. Amended and Restated Genesco Inc. 2009 Equity Incentive Plan. Incorporated by reference to Exhibit A to the Company's definitive proxy statement dated May 13, 2011.
 - g. Amended and Restated EVA Incentive Compensation Plan. Incorporated by reference to Exhibit (10)a to the Company's Quarterly Report on Form 10-Q for the quarter ended April 30, 2011 (File No. 1-3083).
 - h. Amended and Restated EVA Incentive Compensation Plan. Incorporated by reference to Exhibit (10)a to the Company's Quarterly Report on Form 10-Q for the quarter ended April 28, 2012 (File No. 1-3083).
 - i. Form of Incentive Stock Option Agreement. Incorporated by reference to Exhibit (10)c to the Company's Quarterly Report on Form 10-Q for the quarter ended October 29, 2005 (File No.1-3083).
 - j. Form of Non-Qualified Stock Option Agreement. Incorporated by reference to Exhibit (10)d to the Company's Quarterly Report on Form 10-Q for the quarter ended October 29, 2005 (File No.1-3083).
 - k. Form of Restricted Share Award Agreement for Executive Officers. Incorporated by reference to Exhibit (10)e to the Company's Quarterly Report on Form 10-Q for the quarter ended October 29, 2005 (File No.1-3083).
 - 1. Form of Restricted Share Award Agreement for Officers and Employees. Incorporated by reference to Exhibit (10)f to the Company's Quarterly Report on Form 10-Q for the quarter ended October 29, 2005 (File No.1-3083).
 - m. Form of Restricted Share Award Agreement. Incorporated by reference to Exhibit (10)a to the Company's Quarterly Report on Form 10-Q for the quarter ended August 1, 2009 (File No. 1-3083).
 - n. Form of Indemnification Agreement For Directors. Incorporated by reference to Exhibit (10)m to the Company's Annual Report on Form 10-K for the fiscal year ended January 31, 1993 (File No.1-3083).
 - o. Form of Non-Executive Director Indemnification Agreement. Incorporated by reference to Exhibit (10.1) to the current report on Form 8-K filed November 3, 2008 (File No. 1-3083).
 - p. Form of Officer Indemnification Agreement. Incorporated by reference to Exhibit (10.2) to the Company's Quarterly Report on Form 10-Q for the quarter ended November 1, 2008 (File No.1-3083).
 - q. Supplemental Pension Agreement dated as of October 18, 1988 between the Company and William S. Wire II, as amended January 9, 1993. Incorporated by reference to Exhibit (10)p to the Company's Annual Report on Form 10-K for the fiscal year ended January 31, 1993 (File No.1-3083).
 - r. Deferred Compensation Trust Agreement dated as of February 27, 1991 between the Company and NationsBank of Tennessee for the benefit of William S. Wire, II, as amended January 9, 1993. Incorporated by reference to Exhibit (10)q to the Company's Annual Report on Form 10-K for the fiscal year ended January 31, 1993 (File No.1-3083).
 - s. Form of Employment Protection Agreement between the Company and certain executive officers dated as of February 26, 1997. Incorporated by reference to Exhibit (10)p to the Company's Annual Report on Form 10-K for the fiscal year ended February 1, 1997 (File No.1-3083).

- t. First Amendment to Form of Employment Protection Agreement. Incorporated by reference to Exhibit (10)s to the Company's Annual Report on Form 10-K for the fiscal year ended January 30, 2010 (File No.1-3083).
- u. Employment Agreement dated as of March 29, 2010 between the Company and Hal N. Pennington. Incorporated by reference to Exhibit (10)t to the Company's Annual Report on Form 10-K for the fiscal year ended January 30, 2010 (File No.1-3083).
- v. Trademark License Agreement, dated August 9, 2000, between Levi Strauss & Co. and Genesco Inc. Incorporated by reference to Exhibit (10.1) to the Company's Quarterly Report on Form 10-Q for the quarter ended October 30, 2004 (File No.1-3083).*
- w. Amendment No. 1 (Renewal) to Trademark License Agreement, dated October 18, 2004, between Levi Strauss & Co. and Genesco Inc. Incorporated by reference to Exhibit (10.2) to the Company's Quarterly Report on Form 10-Q for the quarter ended October 30, 2004 (File No.1-3083).*
- x. Amendment No. 2 (Renewal) to Trademark License Agreement, dated November 1, 2006, between Levi Strauss & Co. and Genesco. Inc. Incorporated by reference to Exhibit (10.1) to the Company's Quarterly Report on Form 10-Q for the quarter ended October 28, 2006 (File No.1-3083).*
- y. Amendment No. 4 (Renewal) to Trademark License Agreement, dated May 15, 2009, between Levi Strauss & Co. and Genesco Inc. Incorporated by reference to Exhibit (10)b to the Company's Quarterly Report on Form 10-Q for the quarter ended August 1, 2009 (File No.1-3083).*
- z. Amendment No. 5 (Renewal) to Trademark License Agreement, dated July 23, 2012, between Levi Strauss & Co. and Genesco Inc. Incorporated by reference to Exhibit (10.1) to the Company's Current Report on Form 8-K filed July 25, 2012 (File No. 1-3083).*
- aa. Genesco Inc. Deferred Income Plan dated as of July 1, 2000. Incorporated by reference to Exhibit (10)p to the Company's Annual Report on Form 10-K for the fiscal year ended January 29, 2005. Amended and Restated Deferred Income Plan dated August 22, 2007. Incorporated by reference to Exhibit (10)r to the Company's Annual Report on Form 10-K for the fiscal year ended February 2, 2008 (File No.1-3083).
- bb. Non-Employee Director and Named Executive Officer Compensation. Incorporated by reference to Exhibit (10)b to the Company's Quarterly Report on Form 10-Q for the quarter ended October 29, 2005 (File No.1-3083).
- cc. The Schuh Group Limited 2015 Management Bonus Scheme. Incorporated by reference to Exhibit (10)a to the Company's Quarterly Report on Form 10-Q for the quarter ended July 30, 2011 (File No.1-3083).
- dd. 1996 Employee Stock Purchase Plan. Incorporated by reference to Registration Statement on Form S-8 filed September 14, 1995 (File No. 333-62653).
- ee. Amended and Restated Genesco Employee Stock Purchase Plan dated August 22, 2007. Incorporated by reference to Exhibit (10)u to the Company's Annual Report on Form 10-K for the fiscal year ended February 2, 2008 (File No.1-3083).
- ff. Basic Form of Exchange Agreement (Restricted Stock). Incorporated by reference to Exhibit 10.1 to the current report on Form 8-K filed April 29, 2009 (File No. 1-3083).
- gg. Basic Form of Exchange Agreement (Unrestricted Stock). Incorporated by reference to Exhibit 10.2 to the current report on Form 8-K filed April 29, 2009 (File No. 1-3083).
- hh. Form of Conversion Agreement. Incorporated by reference to Exhibit 10.1 to the current report on Form 8-K filed November 2, 2009 (File No. 1-3083).
- ii. Form of Conversion Agreement. Incorporated by reference to Exhibit 10.1 to the current report on Form 8-K filed November 6, 2009 (File No. 1-3083).
- jj. Settlement Agreement, dated as of March 3, 2008, by and among UBS Securities LLC and UBS Loan Finance LLC, The Finish Line, Inc. and Headwind, Inc. and Genesco Inc. Incorporated by reference to Exhibit 10.1 to the current report on Form 8-K filed March 4, 2008 (File No. 1-3083).
- (21) Subsidiaries of the Company
- (23) Consent of Ernst & Young LLP, Independent Registered Public Accounting Firm included on page 109.
- (24) Power of Attorney
- (31.1) Certification of the Chief Executive Officer Pursuant to Section 302 of the Sarbanes-Oxley Act of 2002.

(31.2)	Certification of the Chief Financial Officer Pursuant to Section 302 of the Sarbanes-Oxley Act of 2002.
(32.1)	Certification of the Chief Executive Officer Pursuant to 18 U.S.C. Section 1350, as Adopted Pursuant to Section 906 of the Sarbanes-Oxley Act of 2002.
(32.2)	Certification of the Chief Financial Officer Pursuant to 18 U.S.C. Section 1350, as Adopted Pursuant to Section 906 of the Sarbanes-Oxley Act of 2002.
(99)	Financial Statements and Report of Independent Registered Public Accounting Firm with respect to the Genesco Employee Stock Purchase Plan being filed herein in lieu of filing Form 11-K pursuant to Rule 15d-21(filed with the Original Filing).
101.INS	XBRL Instance Document
101.SCH	XBRL Schema Document
101.CAL	XBRL Calculation Linkbase Document
101.DEF	XBRL Definition Linkbase Document
101.LAB	XBRL Label Linkbase Document
101.PRE	XBRL Presentation Linkbase Document

Exhibits (10)c through (10)m, (10)s through (10)u and (10)aa through (10)ee are Management Contracts or Compensatory Plans or Arrangements required to be filed as Exhibits to this Form 10-K.

* Certain information has been omitted and filed separately with the Securities and Exchange Commission. Confidential treatment has been granted with respect to the omitted portion.

A copy of any of the above described exhibits will be furnished to the shareholders upon written request, addressed to Director, Corporate Relations, Genesco Inc., Genesco Park, Room 498, P.O. Box 731, Nashville, Tennessee 37202-0731, accompanied by a check in the amount of \$15.00 payable to Genesco Inc.

Consent of Independent Registered Public Accounting Firm

We consent to the incorporation by reference in the following Registration Statements:

- (1) Registration statement (Form S-8 No. 033-62653) of Genesco Inc.,
- (2) Registration statement (Form S-8 No. 333-08463) of Genesco Inc.,
- (3) Registration statement (Form S-8 No. 333-104908) of Genesco Inc.,
- (4) Registration statement (Form S-8 No. 333-40249) of Genesco Inc.,
- (5) Registration statement (Form S-8 No. 333-128201) of Genesco Inc.,
- (6) Registration statement (Form S-8 No. 333-160339) of Genesco Inc.,
- (7) Registration statement (Form S-8 No. 333-180463) of Genesco Inc., and
- (8) Registration statement (Form S-3 No. 333-109019) of Genesco Inc.,

of our reports dated April 2, 2014, with respect to the consolidated financial statements and schedule of Genesco Inc. and Subsidiaries and the effectiveness of internal control over financial reporting of Genesco Inc. and Subsidiaries included in this Annual Report (Form 10-K) of Genesco Inc. for the year ended February 1, 2014.

We also consent to the incorporation by reference in the Registration Statement (Form S-8 No. 333-62653) pertaining to the 1996 Employee Stock Purchase Plan of Genesco Inc. of our report dated April 2, 2014, with respect to the financial statements of the Genesco Inc. Employee Stock Purchase Plan included as an exhibit to this Annual Report (Form 10-K) of Genesco Inc. for the year ended February 1, 2014.

/s/ Ernst & Young LLP

Nashville, Tennessee April 2, 2014

SIGNATURES

Pursuant to the requirements of Section 13 or 15(d) of the Securities Exchange Act of 1934, the registrant has duly caused this report to be signed on its behalf by the undersigned, thereunto duly authorized.

GENESCO INC.

By: /s/James S. Gulmi

James S. Gulmi Senior Vice President – Finance and Chief Financial Officer

Date: April 2, 2014

Pursuant to the requirements of the Securities Exchange Act of 1934, this report has been signed below by the following persons on behalf of the registrant and in the capacities indicated on the 2nd day of April, 2014.

/s/Robert J. Dennis	Chairman, President, Chief Executive Officer
Robert J. Dennis	and a Director
/s/James S. Gulmi	Senior Vice President – Finance and
James S. Gulmi	Chief Financial Officer
	(Principal Financial Officer)
/s/Paul D. Williams	Vice President and Chief Accounting Officer
Paul D. Williams	
Directors:	
Joanna Barsh*	Matthew C. Diamond *
James S. Beard*	Marty G. Dickens *
Leonard L. Berry *	Thurgood Marshall, Jr. *
William F. Blaufuss, Jr.*	Kathleen Mason *
James W. Bradford*	

*By /s/Roger G. Sisson

Roger G. Sisson Attorney-In-Fact

Genesco Inc. and Subsidiaries Financial Statement Schedule February 1, 2014

Genesco Inc. and Subsidiaries Valuation and Qualifying Accounts

Year Ended February 1, 2014

In Thousands	Beginning Balance		Charged to Profit and Loss		Increases (Decreases)		Ending Balance	
Reserves deducted from assets in the balance sheet:								
Accounts Receivable Allowances	\$	6,082	\$	(525)	\$	(1,137)	\$	4,420
<u>Year Ended February 2, 2013</u>								

In Thousands	Beginning Balance		Charged to Profit and Loss		Increases (Decreases)		Ending Balance	
Reserves deducted from assets in the balance sheet:								
Accounts Receivable Allowances	\$	6,900	\$	1,325	\$	(2,143)	\$	6,082

Year Ended January 28, 2012

In Thousands		Beginning Balance		Charged to Profit and Loss		Increases (Decreases)		Ending Balance	
Reserves deducted from assets in the balance sheet:									
Accounts Receivable Allowances	\$	3,301	\$	2,004	\$	1,595	\$	6,900	

AMENDMENT AND RESTATEMENT AGREEMENT

dated November 1, 2013 between

SCHUH GROUP LIMITED as Parent

and others as Borrowers and Guarantors

LLOYDS BANK PLC (FORMERLY LLOYDS TSB BANK PLC) as Arranger

LLOYDS BANK PLC (FORMERLY LLOYDS TSB BANK PLC)

as Agent

and

LLOYDS BANK PLC (FORMERLY LLOYDS TSB BANK PLC)

as Security Trustee

M

Dickson Minto W.S. Edinburgh THIS AGREEMENT is dated November 1, 2013 and made between:

- (1) **SCHUH GROUP LIMITED**, a company incorporated in Scotland with registered number SC379625 and having its registered office at 1 Neilson Square, Deans Industrial Estate, Livingston, West Lothian EH54 8RQ (the "**Parent**");
- (2) **THE SUBSIDIARIES** of the Parent listed in Part I of Schedule 1 as borrowers (the "**Borrowers**");
- (3) **THE SUBSIDIARIES** of the Parent listed in Part II of Schedule 1 as guarantors (the "Guarantors");
- (4) LLOYDS BANK PLC (FORMERLY LLOYDS TSB BANK PLC) (in this capacity the "Arranger");
- (5) LLOYDS BANK PLC (FORMERLY LLOYDS TSB BANK PLC) (in this capacity the "Original Lender");
- (6) LLOYDS BANK PLC (FORMERLY LLOYDS TSB BANK PLC) as agent of the other Finance Parties (the "Agent"); and
- (7) LLOYDS BANK PLC (FORMERLY LLOYDS TSB BANK PLC) as security trustee for the Secured Parties (the "Security Trustee").

WHEREAS:

- (A) The Parent, the Guarantors, the Borrowers, the Agent, the Arranger, the Security Trustee and the Original Lender entered into a senior term facilities agreement dated 10 November 2010 as amended and restated on 23 June 2011 (as amended, restated, novated or varied from time to time the "Facilities Agreement") in terms of which certain facilities were made available to the Borrowers (as defined therein);
- (B) the parties wish to amend and restate the Facilities Agreement in accordance with the terms of this Agreement.

NOW IT IS HEREBY AGREED as follows:

1. DEFINITIONS AND INTERPRETATION

1.1. In this Agreement:

"Amended and Restated Facilities Agreement" means the Facilities Agreement as amended and restated in the form set out in Schedule 3 (Form of Amended and Restated Facilities Agreement).

"Effective Date" means the date on which the Agent confirms in writing to the Parent and the Original Lender that it has received (or has waived the requirement to receive) the documents and/or evidence listed in Schedule 2 (Conditions Precedent), in each case in form and substance satisfactory to the Agent.

- 1.2. In this Agreement a term defined in the Facilities Agreement has the same meaning when used in this Agreement and Clause 1 (Definitions and Interpretation) of the Facilities Agreement shall apply hereto except that references in such Clause to the Facilities Agreement are to be construed as references to this Agreement.
- 1.3. Unless expressly provided to the contrary in this Agreement, a person who is not a party to this Agreement has no right under the Contracts (Rights of Third Parties) Act 1999 to enforce or

enjoy the benefit of any term of this Agreement. Notwithstanding any term of any Finance Document, the consent of any person who is not a party to this Agreement is not required to rescind or vary this Agreement at any time.

1. AMENDMENT OF THE FACILITIES AGREEMENT

- 1.1. With effect from the Effective Date, the Facilities Agreement shall be amended and restated in the form set out as Schedule 3 to this Agreement.
- 1.2. Subject to the terms of this Agreement, the Facilities Agreement shall remain in full force and effect. With effect from the Effective Date, this Agreement and the Facilities Agreement shall be read and construed as one document and all references in the Facilities Agreement and in each of the Finance Documents to the Facilities Agreement shall be read and construed as references to the Amended and Restated Facilities Agreement.
- 1.3. Save to the extent expressly provided in this Agreement, the Finance Documents shall remain in full force and effect.
- 1.4. If the Effective Date has not occurred by 5pm (UK time) on 30 November 2013, this Agreement shall lapse and shall cease to have any effect other than Clause 5 (Fees and expenses) which shall remain in full force and effect.

1. CONSENTS

The Agent (on behalf of the Lenders) hereby consents to the following with effect from the Effective Date:

- (i) the acquisition of the Additional Livingston Property; and
- (ii) making and repayment of a loan by Genesco, Inc to Schuh Limited in order to fund repayment of certain loan notes issued to the management of the Group.

1. **REPRESENTATIONS AND WARRANTIES**

- 1.5. Each Obligor represents and warrants to each Finance Party on the date of this Agreement and on the Effective Date in terms of the Repeating Representations, in each case as if references to "this Agreement" and "the Finance Documents" in those representations are construed as references to this Agreement and (on the Effective Date) the Amended and Restated Facilities Agreement.
- 1.6. Each Obligor represents and warrants to the Agent and the other Finance Parties on the date of this Agreement and on the Effective Date that:
 - (i) the Parent has complied with the terms of Clause 22 (Financial Covenants) of the Facilities Agreement at all times prior to the date hereof; and
 - (ii) on each date within the next 12 months on which any of the financial covenants contained in Clause 22 (Financial Covenants) of the Facilities Agreement falls to be tested, the Parent shall comply with the terms of Clause 22 (Financial Covenants) of the Facilities Agreement.

2. FEES AND EXPENSES

2.1. In consideration for the Lenders agreeing to the amendments to the Facilities Agreement provided for in this Agreement, on the Effective Date the Parent shall pay to the Agent for the account of the Lenders an arrangement fee of ££125,000, which shall be debited to the current account of the Parent with the Agent.

3

2.2. The Parent shall promptly on demand pay to the Agent and the Security Trustee the amount of all costs and expenses (together with any VAT or similar taxes thereon) reasonably incurred by them or by the Lenders in connection with this Agreement and the documents contemplated by this Agreement (including, without limitation, legal fees).

3. ACCESSION OF SCHUH LIMITED

- 3.1. With effect from the Effective Date, Schuh Limited agrees to become an Additional Borrower and to be bound by the terms of the Facilities Agreement and the other Finance Documents as an Additional Borrower pursuant to Clause 26.2 (Additional Borrowers) of the Facilities Agreement. Schuh Limited is a company duly incorporated under the laws of Scotland and is a limited liability company and registered number SC125327.
- 3.2. The Parent confirms that no Default would occur as a result of Schuh Limited becoming an Additional Borrower.

4. PRESERVATION OF GUARANTEES AND SECURITY

Each Obligor:

- (i) confirms its consent to the amendment and restatement of the terms of the Facilities Agreement as contemplated by this Agreement; and
- (ii) confirms that the Transaction Security Documents granted by it on or prior to the date of this Agreement, the guarantee granted by it pursuant to Clause 19 (Guarantee and Indemnity) of the Facilities Agreement and the security rights constituted or evidenced by the Transaction Security Documents are and remain in full force and effect and apply as from the Effective Date to, inter alia, the Amended and Restated Facilities Agreement and the guarantees granted pursuant to Clause 19 (Guarantee and Indemnity) of the Facilities Agreement will continue to be legal, valid, binding and enforceable in accordance with their respective terms.

5. GENERAL

- 5.1. The provisions of Clauses 36 (Remedies and Waivers), 35 (Partial Invalidity), 33 (Notices) and 41 (Enforcement) of the Facilities Agreement shall be deemed to be incorporated in this Agreement (with such conforming amendments as the context requires) as if set out in this Agreement.
- 5.2. Neither the execution of this Agreement nor the making of any amount available under the Facilities Agreement or the Amended and Restated Facilities Agreement amounts to a waiver of any outstanding Event of Default.
- 5.3. This Agreement is designated by the Parent and the Agent as a Finance Document.
- 5.4. Each Obligor hereby confirms that the amendment and restatement of the Facilities Agreement pursuant to this Agreement is an amendment and restatement within the contemplation of the Finance Documents.
- 5.5. The Parent shall, and shall procure that each other Obligor will, at the request of the Agent and at its own expense, do all such acts and things necessary to give effect to the amendments effected or to be effected pursuant to this Agreement.

1. COUNTERPARTS

This Agreement may be executed in any number of counterparts, and this has the same effect as if the signatures on the counterparts were on a single copy of this Agreement.

1. GOVERNING LAW

This Agreement is, and any non-contractual obligations arising out of or in connection with it are, governed by English law.

This Agreement has been entered into on the date stated at the beginning of this Agreement.

SCHEDULE 1

5

PART I THE BORROWERS

Schuh Group Limited (Reg. No. SC379625) Schuh (Holdings) Limited (Reg. No. SC265833)

Schuh Limited (Reg. No. SC125327)

PART II THE GUARANTORS

Schuh Group Limited (Reg. No. SC379625) Schuh (Holdings) Limited (Reg. No. SC265833) Schuh Limited (Reg. No. SC125327) Schuh (ROI) Limited (Reg. No. 272987)

SCHEDULE 2

CONDITIONS PRECEDENT

1. Obligors

- (a) A copy of the Constitutional Documents and of the constitutional documents of each other Obligor, or a certificate of an authorised signatory of the Parent confirming that there has been no change to such documents since 23 June 2011.
- (b) A copy of a resolution of the board or, if applicable, a committee of the board of directors of each Obligor:

6

- (i) approving the terms of, and the transactions contemplated by, this Agreement and resolving that it execute, deliver and perform this Agreement;
- (ii) authorising a specified person, on its behalf to sign and/or despatch all documents and notices to be signed and/or despatched by it under or in connection with this Agreement.
- (c) A certificate of an authorised signatory of the Parent certifying that each copy document relating to it specified in this Schedule 2 is true and complete and in full force and effect as at a date no earlier than the date of this Agreement.

2. Legal Opinions

Legal opinions of Dickson Minto W.S., legal advisers to the Agent as to Scots law and English law each addressed to the Finance Parties.

3. Other documents and evidence

- (a) This Agreement executed by the Obligors.
- (b) An updated copy of the Group Structure Chart.
- (c) A list of the directors of the Parent as at the Effective Date.
- (d) Any information and evidence required by the Finance Parties in respect of any Obligor to comply with its know your customer or anti money laundering procedures.
- (e) A copy of any other Authorisation or other document, opinion or assurance which the Agent considers to be necessary or desirable (if it has notified the Parent accordingly) in connection with the entry into and performance of the transactions contemplated by this Agreement or for the validity and enforceability of this Agreement.
- (f) A standard security (the "Additional Standard Security") granted by the Company over the subjects known as 240 (and formerly known as Plot D, Phase II), J4M8 Distribution Park, Bathgate, EH48 2EA being the subjects (IN THE FIRST PLACE) registered in the Land Register of Scotland under Title Number WLN39760; and (IN THE SECOND PLACE) the subjects at J4M8 Distribution Park lying to the north west of Inchmuir Road, Whitehill Industrial Estate, by Bathgate, shown tinted pink on the plan 2 contained in Schedule 4 of the Amended and Restated Facilities Agreement and forming part of the subjects registered under Title Number WLN6795 (the "Additional Livingston Property").

- (g) a letter of obligation from Morton Fraser LLP in form satisfactory to the Agent covering the grant of a standard security over the Additional Livingston Property;
- (h) a solicitor's undertaking in form and substance satisfactory to the Agent undertaking to hold all title documents in relation to the Additional Livingston Property to the order of the Security Trustee;
- (i) A valuation by CBRE addressed to Lloyds Bank plc in form and substance satisfactory to the Agent in respect of the Additional Livingston Property.
- (j) The certificate of title by Morton Fraser LLP in respect of the Additional Livingston Property addressed to the Agent in form and substance satisfactory to the Agent.
- (k) A search in the Register of Charges and Company File in respect of Schuh Limited as the granter of the Additional Standard Security, such Search disclosing no entries prejudicial to the grant of the Additional Standard Security and brought down to a date being as close as practicable to the date of completion.
- (l) A form 12 in respect of the Additional Livingston Property brought down to a date as near as practicable to the date of this Agreement disclosing no entries adverse to the grantor of the Additional Standard Security's interest in the property secured.
- (m) A cheque made payable to Dickson Minto W.S. in reimbursement of Registers of Scotland registration dues payable on the Additional Standard Security, which will have been paid on behalf of Schuh Limited.
- (n) An updated business plan and financial projections in form and substance satisfactory to the Agent.
- (o) An assignment in security granted by the Company in favour of the Security Trustee of the contractor's collateral warranty dated 17 February 2011 between RP IOM GP Limited and RP IOM Nominees Limited and Twintec Limited, as assigned to the Company on or around the date hereof, together with all notices required thereto.
- (p) Evidence satisfactory to the Agent that there has been no material and adverse change in the financial condition of any member of the Group since the date of the most recent audited consolidated financial statements which have been delivered to the Agent in accordance with clause 21.1(a) of the Facilities Agreement.

SCHEDULE 3

8

FORM OF AMENDED AND RESTATED FACILITIES AGREEMENT

SENIOR TERM FACILITIES AGREEMENT

£24,357,000

FACILITIES AGREEMENT

for

SCHUH GROUP LIMITED as Parent

arranged by

LLOYDS BANK PLC (FORMERLY LLOYDS TSB BANK PLC) as Mandated Lead Arranger

with

LLOYDS BANK PLC (FORMERLY LLOYDS TSB BANK PLC) acting as Agent

and

LLOYDS BANK PLC (FORMERLY LLOYDS TSB BANK PLC) acting as Security Trustee

CONTENTS

1. DEFINITIONS AND INTERPRETATION 12 2. THE FACILITIES 40 3.PURPOSE 43 4.CONDITIONS OF UTILISATION 43 5.UTILISATION – LOANS 45 6.REPAYMENT 47 7.ILLEGALITY, VOLUNTARY PREPAYMENT AND CANCELLATION 48 8.MANDATORY PREPAYMENT 50 9.RESTRICTIONS 54 10.INTEREST 56 11.INTEREST PERIODS 56 12.CHANGES TO THE CALCULATION OF INTEREST 58 13.FEES 59 14. TAX GROSS UP AND INDEMNITIES 60 15.INCREASED COSTS 67 **16.OTHER INDEMNITIES** 68 17.MITIGATION BY THE LENDERS 70 18.COSTS AND EXPENSES 71 19.GUARANTEE AND INDEMNITY 72 20.REPRESENTATIONS 76 21.INFORMATION UNDERTAKINGS 83 22.FINANCIAL COVENANTS 88 23.GENERAL UNDERTAKINGS 96 24. EVENTS OF DEFAULT 105 25.CHANGES TO THE LENDERS 110 26. CHANGES TO THE OBLIGORS 116

27,ROLE OF THE AGENT, THE ARRANGER AND OTHERS 120
28,ROLE OF THE SECURITY TRUSTEE 129
29,CONDUCT OF BUSINESS BY THE FINANCE PARTIES 136
30,SHARING AMONG THE FINANCE PARTIES 137
31,PAYMENT MECHANICS 139
32,SET-OFF 142
33,NOTICES 143
34,CALCULATIONS AND CERTIFICATES 145
35,PARTIAL INVALIDITY 145

<u>36.</u>REMEDIES AND WAIVERS 145

37. AMENDMENTS AND WAIVERS 145

38.CONFIDENTIALITY 149

<u>39.</u>COUNTERPARTS 153

40.GOVERNING LAW 156

41.ENFORCEMENT 156

SCHEDULE 1 THE ORIGINAL PARTIES 157

SCHEDULE 2 CONDITIONS PRECEDENT159

SCHEDULE 3 REQUEST165

ADDITIONAL LIVINGSTON PROPERTY 168

SCHEDULE 5 FORM OF TRANSFER CERTIFICATE 169

SCHEDULE 6 FORM OF ASSIGNMENT AGREEMENT172

SCHEDULE 7 FORM OF ACCESSION LETTER 176

SCHEDULE 9 FORM OF COMPLIANCE CERTIFICATE178

SCHEDULE 10 FORM OF INCREASE CONFIRMATION179

THIS AGREEMENT is the amended and restated form of a facility agreement originally dated 10 November 2010 and in its amended and restated form is made between:

- (1) **SCHUH GROUP LIMITED** a company incorporated in Scotland with registered number SC379625 and with its registered office at 5th Floor, Quartermile Two, 2 Lister Square, Edinburgh EH3 9GL (the "**Parent**");
- (2) **THE SUBSIDIARY** of the Parent listed in Part I of Schedule 1 (The Original Parties) as borrower (together with the Parent being the "**Borrowers**");
- (3) THE SUBSIDIARIES of the Parent listed in Part I of Schedule 1 (The Original Parties) as guarantors (together with the Parent, the "Guarantors");
- (4) **LLOYDS BANK PLC (FORMERLY LLOYDS TSB BANK PLC)** with registered number 00002065 and having its registered office at 25 Gresham Street, London EC2V 7HN as mandated lead arranger (the "**Arranger**");
- (5) THE FINANCIAL INSTITUTIONS listed in Part II of Schedule 1 (The Original Parties) as lenders (the "Original Lenders");
- (6) **LLOYDS BANK PLC (FORMERLY LLOYDS TSB BANK PLC)** with registered number 00002065 and having its registered office at 25 Gresham Street, London EC2V 7HN as agent of the other Finance Parties (the "**Agent**"); and
- (7) **LLOYDS BANK PLC (FORMERLY LLOYDS TSB BANK PLC)** with registered number 00002065 and having its registered office at 25 Gresham Street, London EC2V 7HN as security trustee for the Secured Parties (the "Security Trustee").

IT IS AGREED as follows:

SECTION 1

INTERPRETATION

1. DEFINITIONS AND INTERPRETATION

1.1. Definitions

In this Agreement:

"Acceptable Bank" means:

- (a) a bank or financial institution duly authorised under applicable laws to carry on the business of banking (including, without limitation, the business of making deposits) which is approved in writing by the Agent; or
- (b) any Finance Party.

"Accession Letter" means a document substantially in the form set out in Schedule 7 (Form of Accession Letter).

"Accounting Period" means, in respect of the period up to and including P6 2012, each of the thirteen periods in any Financial Year for which the Parent is to prepare consolidated four weekly management accounts for the purposes of this Agreement and each such period shall be referred to herein as "P" followed by the relevant number and Financial Year in respect thereof (for the avoidance of doubt and by way of example "P9 2011" means the ninth trading period in the period of thirteen Accounting Periods from 29 March 2010 to 27 March 2011) and, thereafter, each of the twelve periods in any Financial Year for which the Parent is to prepare monthly management accounts for the purposes of this Agreement.

"Accounting Principles" means generally accepted accounting principles in the UK, including IFRS.

"Accounting Reference Date" means the accounting reference date for the purposes of section 391 of the Companies Act 2006, being 31 January in the case of each member of the Group, (with their financial year ending not more than 7 days after, or less than 7 days before, such date).

"**Acquisition Agreement**" means the sale and purchase agreement dated on or about the Restatement Date setting out the terms on which the UK Acquisition Company is to acquire the shares in the Parent.

"Additional Borrower" means a company which becomes a Borrower in accordance with Clause 26 (Changes to the Obligors).

"Additional Guarantor" means a company which becomes a Guarantor in accordance with Clause 26 (Changes to the Obligors).

"Additional Livingston Property" means the subjects known as 240 (and formerly known as Plot D, Phase II), J4M8 Distribution Park, Bathgate, EH48 2EA being the subjects (IN THE FIRST PLACE) registered in the Land Register of Scotland under Title Number WLN39760; and (IN THE SECOND PLACE) the subjects at J4M8 Distribution Park lying to the north west of Inchmuir Road, Whitehill Industrial Estate, by Bathgate, shown tinted pink on the plan 2 contained in Schedule 4 and forming part of the subjects registered under Title Number WLN6795.

"Additional Obligor" means an Additional Borrower or an Additional Guarantor.

"Additional Standard Security" means the standard security by the Company in favour of the Security Trustee in respect of the Additional Livingston Property dated on or around the Second Restatement Date.

"Affiliate" means, in relation to any person, a Subsidiary of that person or a Holding Company of that person or any other Subsidiary of that Holding Company.

"Amendment and Restatement Agreement" means the amendment and restatement agreement in respect of this Agreement dated on the Restatement Date.

"**Annual Financial Statements**" means the financial statements for a Financial Year delivered pursuant to paragraph (a) of Clause 21.1 (Financial Statements).

"Assignment Agreement" means an agreement substantially in the form set out in Schedule 6 (Form of Assignment Agreement) or any other form agreed between the relevant assignor and assignee.

"Auditors" means one of PricewaterhouseCoopers, Ernst & Young, KPMG, Deloitte & Touche or such other firm approved in advance by the Majority Lenders (such approval not to be unreasonably withheld or delayed).

"Authorisation" means an authorisation, consent, approval, resolution, licence, exemption, filing, notarisation or registration.

"**Availability Period**" means, in respect of Facility A and Facility B, the period from and including the date of this Agreement to and including the Termination Date in relation to the relevant Facility and, in respect of Facility C, the period from and including the Second Restatement Date to 30 June 2014.

"Available Commitment" means, in relation to a Facility, a Lender's Commitment under that Facility minus (subject as set out below):

- (a) the amount of its participation in any outstanding Utilisations under that Facility; and
- (b) in relation to any proposed Utilisation, the amount of its participation in any other Utilisations that are due to be made under that Facility on or before the proposed Utilisation Date.

"Available Facility" means, in relation to a Facility, the aggregate for the time being of each Lender's Available Commitment in respect of that Facility.

"**Borrower**" means a Borrower listed in Part I of Schedule 1 (the Original Parties) or an Additional Borrower unless it has ceased to be a Borrower in accordance with Clause 26 (Changes to the Obligors).

"Break Costs" means the amount (if any) by which:

(a) the interest (excluding Margin) which a Lender should have received for the period from the date of receipt of all or any part of its participation in a Loan or Unpaid Sum to the last day of the current Interest Period in respect of that Loan or Unpaid Sum, had the principal amount or Unpaid Sum received been paid on the last day of that Interest Period;

exceeds:

(b) the amount which that Lender would be able to obtain by placing an amount equal to the principal amount or Unpaid Sum received by it on deposit with a leading bank in the Relevant Interbank Market for a period starting on the Business Day following receipt or recovery and ending on the last day of the current Interest Period.

"Budget" any budget delivered by the Parent to the Agent pursuant to Clause 21.4 (Budget).

"Business Acquisition" means the acquisition of a company or any shares or securities or a business or undertaking (or, in each case, any interest in any of them) or the incorporation of a company.

"Business Day" means a day (other than a Saturday or Sunday) on which banks are open for general business in London.

"Business Plan" means the business plan prepared by the Parent and delivered to the Agent on or around 23 July 2010 including the agreed form financial projections of the Group.

"**Capital Expenditure**" means any expenditure or obligation (other than expenditure or obligations in respect of Business Acquisitions) in respect of expenditure which, in accordance with the Accounting Principles, is treated as capital expenditure (and including the capital element of any expenditure or obligation incurred in connection with a Finance Lease).

"**Cash**" means, at any time, cash in hand or at bank or amounts on deposit with a Lender (which, for the avoidance of any doubt, includes any cash held by way of cash cover for any reason) which are freely transferable and freely convertible and accessible by a member of the Group within 7 days together with (without double counting) cash in transit and in any such case is not subject to any Security (other than the Transaction Security).

"Cashflow" has the meaning given to that term in Clause 22.2 (Financial definitions).

"**Change of Control**" means (A) any person or group of persons acting in concert gains direct or indirect control of Genesco Inc. after the Restatement Date, (B) Genesco Inc. ceases to have legal and beneficial ownership of the entire issued share capital of UK Acquisition Company, (C) UK Acquisition Company ceases to have legal and beneficial ownership of the entire issued share capital of the Parent or (D) any person or group of persons acting in concert gains direct or indirect control of the Parent after the Restatement Date. For the purposes of this definition:

- (a) "**control**" of the Parent means:
 - (i) the power (whether by way of ownership of shares, proxy, contract, agency or otherwise) to:
 - (A) cast, or control the casting of, more than 50% of the maximum number of votes that might be cast at a general meeting of the Parent; or
 - (B) appoint or remove all, or the majority, of the directors or other equivalent officers of the Parent; or
 - (C) give directions with respect to the operating and financial policies of the Parent with which the directors or other equivalent officers of the Parent are obliged to comply; and/or
 - (ii) the holding beneficially of more than 50% of the issued share capital of the Parent (excluding any part of that issued share capital that carries no right to participate beyond a specified amount in a distribution of either profits or capital);
- (b) "**control**" of Genesco Inc. means:
 - (i) any "person" or "group" (as such terms are used in Sections 13(d) and 14(d) of the Securities Exchange Act of 1934, but excluding any employee benefit plan of such person or its subsidiaries, and any person or entity acting in its capacity as trustee, agent or other fiduciary or administrator of any such plan) becomes the "beneficial owner" (as defined in Rules 13d-3 and 13d-5 under

16

the Securities Exchange Act of 1934, except that a person or group shall be deemed to have "beneficial ownership" of all securities that such person or group has the right to acquire, whether such right is exerciseable immediately or only after the passage of time (such right being an "option right") directly or indirectly, of more than 50% of the equity securities of Genesco Inc. entitled to vote for members of the board of directors or equivalent governing body of Genesco Inc. on a fully-diluted basis (and taking into account all such securities that such person or group has the right to acquire pursuant to any option right); and/or

- (ii) during any period of 12 consecutive months, a majority of the members of the board of directors or other equivalent body of Genesco Inc. cease to be composed of individuals (i) who were members of that board or equivalent governing body on the first day of such period, (ii) whose election or nomination to that board or equivalent governing body was approved by individuals referred to in paragraph (i) above constituting at the time of such election or nomination at least a majority of that board or equivalent governing body or (iii) whose election or nomination to that board or equivalent governing body was approved by individuals referred to in paragraphs (i) and (ii) above constituting at the time of such election or nomination at least a majority of that board or equivalent governing body;
- (C) "acting in concert" means, a group of persons who, pursuant to an agreement or understanding (whether formal or informal), actively cooperate, through the acquisition directly or indirectly of shares in the Parent by any of them, either directly or indirectly, to obtain or consolidate control of the Parent.

"Charged Property" means all of the assets of the Obligors which from time to time are, or are expressed to be, the subject of the Transaction Security.

"Commitment" means a Facility A Commitment, a Facility B Commitment or a Facility C Commitment.

"**Collateral Warranty Assignment**" means the assignment in security granted by the Company in favour of the Security Trustee of the contractor's collateral warranty dated 17 February 2011 between RP IOM GP Limited and RP IOM Nominees Limited and Twintec Limited, as assigned to the Company on or around the Second Restatement Date.

"**Company**" means Schuh Limited, a company incorporated in Scotland with registered number SC125327 and with its registered office at 1 Neilson Square, Deans Industrial Estate, Livingston EH54 8RQ.

"Compliance Certificate" means a certificate substantially in the form set out in Schedule 9 (Form of Compliance Certificate).

"**Confidential Information**" means all information relating to the Parent, the Company, any Obligor, the Group, the Finance Documents or a Facility of which a Finance Party becomes aware in its capacity as, for the purpose of becoming, a Finance Party or which is received by a Finance Party in relation to, or for the purpose of becoming a Finance Party under, the Finance Documents or a Facility from either:

- (a) any member of the Group or any of its advisers; or
- (b) another Finance Party, if the information was obtained by that Finance Party directly or indirectly from any member of the Group or any of its advisers,

17

in whatever form, and includes information given orally and any document, electronic file or any other way of representing or recording information which contains or is derived or copied from such information but excludes information that:

- (i) is or becomes public information other than as a direct or indirect result of any breach by that Finance Party of Clause 38 (Confidentiality); or
- (ii) is identified in writing at the time of delivery as non-confidential by any member of the Group or any of its advisers;
- (iii) is known by that Finance Party before the date the information is disclosed to it in accordance with paragraphs (a) or (b) above or is lawfully obtained by that Finance Party after that date, from a source which is, as far as that Finance Party is aware, unconnected with the Group and which, in either case, as far as that Finance Party is aware, has not been obtained in breach of, and is not otherwise subject to, any obligation of confidentiality.

"**Confidentiality Undertaking**" means a confidentiality undertaking substantially in a recommended form of the LMA or in any other form agreed between the Parent and the Agent.

"**Constitutional Documents**" means the certificate of incorporation, the certificate of incorporation on change of name (if any) and the memorandum and articles of association of the Parent in the agreed form at the Restatement Date.

"CTA" means the Corporation Tax Act 2009.

"**Deal Costs**" means all fees, costs and expenses, stamp, registration and other Taxes incurred by the Parent or any other member of the Group in connection with the Transaction Documents.

"Debenture" means the debenture by Schuh (ROI) Limited in favour of the Security Trustee dated on or around the date of this Agreement.

"**Debt Purchase Transaction**" means, in relation to a person, a transaction where such person:

- (i) purchases by way of assignment or transfer;
- (ii) enters into sub-participation in respect of; or
- (iii) enters into any other agreement or arrangement having an economic effect substantially similar to a sub-participation in respect of,

any Commitment or amount outstanding under this Agreement.

"**Default**" means an Event of Default or any event or circumstance specified in Clause 24 (Events of Default) which would (with the expiry of a grace period, the giving of notice, the making of any determination under the Finance Documents or any combination of any of the foregoing) be an Event of Default.

"Defaulting Lender" means any Lender:

(a) which has failed to make its participation in a Loan available or has notified the Agent that it will not make its participation in a Loan available by the Utilisation Date of that Loan in accordance with Clause 5.4 (Lenders' participation);

(b)which has rescinded or repudiated a Finance Document; or

(c) with respect to which an Insolvency Event has occurred and is continuing,

unless, in the case of paragraph (a) above:

- (i) its failure to pay is caused by:
 - (A) administrative or technical error; or
 - (B) a Disruption Event; and

payment is made within 5 Business Days of its due date; or

- (ii) the Lender is disputing in good faith whether it is contractually obliged to make the payment in question;
- (iii) it is unlawful in any relevant jurisdiction for the Lender to make that payment (provided that this shall not prejudice the rights of the Borrowers under Clause 7.1 (Illegality);
- (iv) the failure to make that payment is caused by the negligence or wilful default of a third party beyond its control;
- (v) the failure to make that payment is caused by an administrative or technical error experienced by a third party beyond its control; or
- (vi) the Agent is an Impaired Agent and a Borrower has failed to notify the Lenders by giving not less than 3 Business Days prior notice of alternative arrangements for that payment.

"Delegate" means any delegate, agent, attorney or co-trustee appointed by the Security Trustee.

"Disruption Event" means either or both of:

- (a) a material disruption to those payment or communications systems or to those financial markets which are, in each case, required to operate in order for payments to be made in connection with the Facilities (or otherwise in order for the transactions contemplated by the Finance Documents to be carried out) which disruption is not caused by, and is beyond the control of, any of the Parties; or
- (b) the occurrence of any other event which results in a disruption (of a technical or systems-related nature) to the treasury or payments operations of a Party preventing that, or any other Party:
 - (i) from performing its payment obligations under the Finance Documents; or
 - (ii) from communicating with other Parties in accordance with the terms of the Finance Documents,

and which (in either such case) is not caused by, and is beyond the control of, the Party whose operations are disrupted.

"**Dormant Subsidiary**" means a member of the Group which does not trade (for itself or as agent for any person) and does not own, legally or beneficially, assets (including, without limitation, indebtedness owed to it) which in aggregate have a value of £5,000 or more or its equivalent in other currencies.

"Environment" means humans, animals, plants and all other living organisms including the ecological systems of which they form part and the following media:

- (a) air (including, without limitation, air within natural or man-made structures, whether above or below ground);
- (b) water (including, without limitation, territorial, coastal and inland waters, water under or within land and water in drains and sewers); and
- (c) land (including, without limitation, land under water).

19

"Environmental Claim" means any claim, proceeding, notice or investigation by any person in respect of any Environmental Law.

"Environmental Law" means any applicable law or regulation which relates to:

- (a) the pollution or protection of the Environment;
- (b) the conditions of the workplace; or
- (C) any emission, generation, handling, storage, disposal, removal, use, release, spillage or discharge of any substance which, alone or in combination with any other, is capable of causing harm to the Environment including, without limitation, any waste.

"Environmental Permits" means any permit and other Authorisation and the filing of any notification, report or assessment required under any Environmental Law for the operation of the business of any member of the Group conducted on or from the properties owned or used by any member of the Group.

"Escrow Account" means an interest-bearing account:

- (a) held in Scotland by a Borrower with the Agent or the Security Trustee;
- (b) identified in a letter between the Parent and the Agent as the Escrow Account;
- (c) subject to Security in favour of the Security Trustee, which Security is in form and substance satisfactory to the Agent and the Security Trustee; and
- (d) from which no withdrawals may be made by any members of the Group except as contemplated by this Agreement,

(as the same may be re-designated, substituted or replaced from time to time).

"Escrow Account Certificate" means, in respect of any Quarter Date, a certificate signed by two directors of the Parent (one of whom must be the chief financial officer of the Group) confirming that as at such Quarter Date:

- (a) no Default is continuing;
- (b) the Parent has complied with each of the financial covenants set out in Clause 22.2 (Financial condition) on each Quarter Date up to and including the date of the relevant Escrow Account Certificate provided that, in the case of each Quarter Date occurring after the Relevant Quarter Date (as defined in Clause 22.4(a)), no account is taken of any New Shareholder Injection; and
- (c) the Parent will comply with each of the financial covenants set out in Clause 22.2 (Financial condition) on each of the two Quarter Dates falling after the Quarter Date as at which the relevant Escrow Account Certificate is given, and attaching such supporting evidence as the Agent may reasonably request in connection with such confirmation provided that in the case of each of such Quarter Dates no account is taken of any New Shareholder Injection.

"Event of Default" means any event or circumstance specified as such in Clause 24 (Events of Default).

"Excess Cashflow" has the meaning given to that term in Clause 22.1 (Financial definitions).

"Existing Retail Facilities" means the retail and merchant service facilities currently provided by Lloyds Cardnet to the Group.

"Facility" means a Term Facility and "Facilities" means all or any of them as the context requires.

"Facility A" means the term loan facility made available under this Agreement as described in paragraph (a)(i) of Clause 2.1 (The Facilities).

"Facility A Commitment" means:

- (a) in relation to an Original Lender, the amount set opposite its name under the heading "Facility A Commitment" in Part II of Schedule 1 (The Original Parties) and the amount of any other Facility A Commitment transferred to it under this Agreement or assumed by it in accordance with Clause 2.2 (Increase); and
- (b) in relation to any other Lender, the amount of any Facility A Commitment transferred to it under this Agreement or assumed by it in accordance with Clause 2.2 (Increase),

to the extent not cancelled, reduced or transferred by it under this Agreement.

"Facility A Loan" means a loan made or to be made under Facility A or the principal amount outstanding for the time being of that loan.

"Facility A Repayment Date" means each date set out in paragraph (a) of Clause 6.1 (Repayment of Term Loans).

"Facility A Repayment Instalment" means each of the repayment instalments set out in paragraph (a) of Clause 6.1 (Repayment of Term Loans) opposite the relevant Facility A Repayment Date.

"Facility B" means the term loan facility made available under this Agreement as described in paragraph (a)(ii) of Clause 2.1 (The Facilities).

"Facility B Commitment" means:

- (a) in relation to an Original Lender, the amount set opposite its name under the heading "Facility B Commitment" in Part II of Schedule 1 (The Original Parties) and the amount of any other Facility B Commitment transferred to it under this Agreement or assumed by it in accordance with Clause 2.2 (Increase); and
- (b) in relation to any other Lender, the amount of any Facility B Commitment transferred to it under this Agreement or assumed by it in accordance with Clause 2.2 (Increase),

to the extent not cancelled, reduced or transferred by it under this Agreement.

"Facility B Loan" means a loan made or to be made under Facility B or the principal amount outstanding for the time being of that loan.

"Facility B Repayment Date" means 31 October 2015.

"Facility C" means the term loan facility made available under this Agreement as described in paragraph (a)(iii) of Clause 2.1 (The Facilities).

"Facility C Commitment" means:

(C) in relation to an Original Lender, the amount set opposite its name under the heading "Facility C Commitment" in Part II of Schedule 1 (The Original Parties) and the amount of any other Facility C Commitment transferred to it under this Agreement or assumed by it in accordance with Clause 2.2 (Increase); and

(d) in relation to any other Lender, the amount of any Facility C Commitment transferred to it under this Agreement or assumed by it in accordance with Clause 2.2 (Increase),

to the extent not cancelled, reduced or transferred by it under this Agreement.

"Facility C Loan" means a loan made or to be made under Facility C or the principal amount outstanding for the time being of that loan.

"Facility C Repayment Date" means each date set out in paragraph (c) of Clause 6.1 (Repayment of Term Loans).

"Facility C Repayment Instalment" means each of the repayment instalments set out in paragraph (a) of Clause 6.1 (Repayment of Term Loans) opposite the relevant Facility C Repayment Date.

"Facility Office" means:

- (a) in respect of a Lender, the office or offices notified by that Lender to the Agent in writing on or before the date it becomes a Lender (or, following that date, by not less than five Business Days' written notice) as the office or offices through which it will perform its obligations under this Agreement; or
- (b) in respect of any other Finance Party, the office in the jurisdiction in which it is resident for tax purposes.

"Fee Letter" means any letter or letters dated on or about the date of this Agreement between the Arranger and the Parent (or the Agent and the Parent or the Security Trustee and the Parent) setting out any of the fees referred to in Clause 13 (Fees).

"**Finance Document**" means this Agreement, any Accession Letter, any Compliance Certificate, any Fee Letter, any Hedging Agreement, any Resignation Letter, any Selection Notice, the Subordination Agreement, any Transaction Security Document, any Utilisation Request, the Working Capital Facility Letter, the Ranking Agreement, the Amendment and Restatement Agreement, the Second Amendment and Restatement Agreement, any document entered into in respect of the Lloyds Retail Facilities and any other document designated as a "**Finance Document**" by the Agent and the Parent.

"Finance Lease" means any lease or hire purchase contract which would, in accordance with the Accounting Principles, be treated as a finance or capital lease.

"**Finance Party**" means the Agent, the Arranger, the Security Trustee, a Lender, the Working Capital Lender and Lloyds Bank plc (or any other member of the LBG Group) as provider of the Lloyds Retail Facilities.

"Financial Due Diligence Report" means the report by KPMG dated on or around the date of this Agreement relating to the Parent and its subsidiaries and addressed to, and/or capable of being relied upon by, Lloyds Bank plc as original lender.

"Financial Indebtedness" means any indebtedness for or in respect of:

- (a) moneys borrowed and debit balances at banks or other financial institutions;
- (b) any acceptance under any acceptance credit or bill discounting facility (or dematerialised equivalent);
- (C) any note purchase facility or the issue of bonds, notes, debentures, loan stock or any similar instrument;

- (d) the amount of any liability in respect of Finance Leases;
- (e) receivables sold or discounted (other than any receivables to the extent they are sold on a non-recourse basis and meet any requirement for de-recognition under the Accounting Principles);
- (f) any Treasury Transaction (and, when calculating the value of that Treasury Transaction, only the marked to market value (or, if any actual amount is due as a result of the termination or close-out of that Treasury Transaction, that amount) shall be taken into account);
- (g) any counter-indemnity obligation in respect of a guarantee, bond, standby or documentary letter of credit or any other instrument issued by a bank or financial institution;
- (h) any amount raised by the issue of redeemable shares which are redeemable (other than at the option of the issuer) before 30 September 2019 or are otherwise classified as borrowings under the Accounting Principles;
- (i) any amount of any liability under an advance or deferred purchase agreement if (i) one of the primary reasons behind entering into the agreement is to raise finance or to finance the acquisition or construction of the asset or service in question or (ii) the agreement is in respect of the supply of assets or services and payment is due more than 90 days after the date of supply;
- (j) any amount raised under any other transaction (including any forward sale or purchase, sale and sale back or sale and leaseback agreement) having the commercial effect of a borrowing or otherwise classified as borrowings under the Accounting Principles; and
- (k) the amount of any liability in respect of any guarantee for any of the items referred to in paragraphs (a) to (j) above.

"Financial Quarter" has the meaning given to that term in Clause 22.1 (Financial definitions).

"Financial Year" has the meaning given to that term in Clause 22.1 (Financial definitions).

"Flotation" means:

- a successful application being made for the admission of any part of the share capital of any member of the Group to the Official List of the UK Listing Authority and the admission of any part of the share capital of any member of the Group to trading on the London Stock Exchange plc; or
- (ii) the grant of permission to deal with any part of the issued share capital of any member of the Group on AIM or the Main Board or Growth Board of the ICAP Securities & Derivatives Exchange Limited or on any recognised investment exchange (as that term is used in the Financial Services and Markets Act 2000) or in or on any exchange or market replacing the same or any other exchange or market in any country.

"Funds Flow" means the funds flow delivered by the Parent to the Agent on or around the date of this Agreement in connection with the facilities being made available to the Group by the Finance Parties.

"Genesco Closing Schedule" means the schedule in the agreed form setting out the dates to which the Group will prepare Monthly Financial Statements and Quarterly Financial Statements after the Restatement Date.

"Group Structure Chart" means the group structure chart in the agreed form.

"Guarantor" means a Guarantor listed in Part I of Schedule 1 (the Original Parties) as a Guarantor or an Additional Guarantor, unless it has ceased to be a Guarantor in accordance with Clause 26 (Changes to the Obligors).

"**Hedging Agreement**" means any master agreement, confirmation, schedule or other agreement in agreed form entered into or to be entered into by the Parent and a Lender for the purpose of hedging interest rate liabilities in relation to the Term Facilities in accordance with any Hedging Policy Letter.

"**Hedging Policy Letter**" means any letter entered into at any time by the Parent to the Agent setting out the proposed policy of the Group in relation to the hedging of its exposure to floating rates of interest.

"Holding Account" means an account:

- (a) held in Scotland by a member of the Group with the Agent or Security Trustee;
- (b) identified in a letter between the Parent and the Agent as a Holding Account; and
- (c) subject to Security in favour of the Security Trustee which Security is in form and substance satisfactory to the Security Trustee,

(as the same may be redesignated, substituted or replaced from time to time).

"Holding Company" means, in relation to a company or corporation, any other company or corporation in respect of which it is a Subsidiary.

"Holdings" means Schuh Holdings Limited (Registered No. SC265833).

"Investor" means the UK Acquisition Company.

"IFRS" means international accounting standards within the meaning of IAS Regulation 1606/2002 to the extent applicable to the relevant financial statements.

"Impaired Agent" means the Agent at any time when:

- (a) (if the Agent is also a Lender) it is a Defaulting Lender under paragraph (a) or (b) of the definition of "Defaulting Lender";
- (b) an Insolvency Event has occurred and is continuing with respect to the Agent; or
- (c) the Agent rescinds or repudiates a Finance Document.

"Increase Confirmation" means a confirmation substantially in the form set out in Schedule 10 (Form of Increase Confirmation).

"Information Package" means the Reports, the Business Plan and the Funds Flow.

"Insolvency Event" in relation to a Finance Party means:

- (a) a Finance Party being dissolved (other than pursuant to a consolidation, amalgamation or merger);
- (b) a Finance Party admits in writing its inability generally to pay its debts as they become due; or

(c) the appointment of a liquidator, receiver, administrator, compulsory manager or similar officer in respect of that Finance Party or all or any material part of that Finance Party's assets or any analogous procedure or any formal step being taken in respect of any such appointment or procedure other than the presentation of a petition for any such appointment or procedure which is dismissed, stayed or discharged within 30 days.

"Insurance Adequacy Letter" means the letter from Aon addressed to the Finance Parties dated on or around the date of this Agreement in respect of the insurance provision of the Group.

"Intellectual Property" means:

- (a) any patents, trade marks, service marks, designs, business names, copyrights, database rights, design rights, domain names, moral rights, inventions, confidential information, know how and other intellectual property rights and interests (which may now or in the future subsist), whether registered or unregistered; and
- (b) the benefit of all applications and rights to use such assets of each member of the Group (which may now or in the future subsist).

"Interest Period" means, in relation to a Loan, each period determined in accordance with Clause 11 (Interest Periods) and, in relation to an Unpaid Sum, each period determined in accordance with Clause 10.3 (Default interest).

"Interpolated Screen Rate" means, in relation to LIBOR for any Loan, the rate which results from interpolating on a linear basis between:

- (a) the applicable Screen Rate for the longest period (for which that Screen Rate is available) which is less than the Interest Period of that Loan; and
- (b) the applicable Screen Rate for the shortest period (for which that Screen Rate is available) which exceeds the Interest Period of that Loan,

each as of the Specified Time on the Quotation Day for the currency of that Loan.

"**ITA**" means the Income Tax Act 2007.

"Joint Venture" means any joint venture entity, whether a company, unincorporated firm, undertaking, association, joint venture or partnership or any other entity.

"LBG Group" means Lloyds Banking Group plc and its Subsidiaries from time to time.

"Legal Reservations" means:

- (a) the principle that equitable remedies may be granted or refused at the discretion of a court and the limitation of enforcement by laws relating to insolvency, reorganisation and other laws generally affecting the rights of creditors;
- (b) the time barring of claims under the Limitation Acts, the possibility that an undertaking to assume liability for or indemnify a person against non-payment of UK stamp duty may be void and defences of set-off or counterclaim;
- (C) the principle that any additional interest imposed under any relevant agreement may be held to be unenforceable on the grounds that it is a penalty and thus void;
- (d) the principle that an English court may not give effect to an indemnity for legal costs incurred by an unsuccessful litigant;

- (e) the principle that in certain circumstances security granted by way of fixed charge may be characterised as a floating charge or that security purported to be constituted by way of an assignment may be recharacterised as a charge;
- (f) similar principles, rights and defences under the laws of any Relevant Jurisdiction; and
- (g) any other matters which are set out as qualifications or reservations as to matters of law (but not of fact) expressed in any legal opinion required to be given by this Agreement.

"Lender" means:

- (a) any Original Lender; and
- (b) any bank, financial institution, trust, fund or other entity which has become a Party as a Lender in accordance with Clause 25 (Changes to the Lenders),

which in each case has not ceased to be a Lender in accordance with the terms of this Agreement.

"LIBOR" means, in relation to any Loan:

- (a) the applicable Screen Rate; or
- (b) (if no Screen Rate is available for the Interest Period of that Loan) the Interpolated Screen Rate for that Loan; or
- (c) (if no Screen Rate is available for the Interest Period of that Loan and it is not possible to calculate an Interpolated Screen Rate for that Loan) the arithmetic mean of the rates (rounded upwards to four decimal places) as supplied to the Agent at its request quoted by the Reference Banks to leading banks in the London interbank market,

as of, in the case of paragraphs (a) and (c) above, 11.00 a.m. on the Quotation Day for the offering of deposits in Sterling and for a period comparable to the Interest Period for that Loan.

"Limitation Acts" means the Limitation Act 1980, the Foreign Limitation Periods Act 1984, the Prescription and Limitation (Scotland) Act 1973, the Prescription and Limitation (Scotland) Act 1984 and any other enactment (whenever passed) relating to the prescription and/or limitation of actions and/or claims in any part of the United Kingdom.

"Livingston Property" means the heritable property situated at 1 Neilson Square, Deans Industrial Estate, Livingston EH54 8EQ (Title Number WLN1738).

"Lloyds Retail Facilities" means the retail and merchant services facilities (if any) provided to the Group by a member of the LBG Group from time to time.

"LMA" means the Loan Market Association.

"Loan" means a Term Loan.

"**Majority Lenders**" means a Lender or Lenders whose Commitments aggregate more than $66^{2}/_{3}$ per cent. of the Total Commitments (or, if the Total Commitments have been reduced to zero, aggregated more than $66^{2}/_{3}$ per cent. of the Total Commitments immediately prior to that reduction).

"Mandatory Prepayment Account" means an interest-bearing account:

(a) held in Scotland by a Borrower with the Agent or Security Trustee;

- (b) identified in a letter between the Parent and the Agent as a Mandatory Prepayment Account;
- (c) subject to Security in favour of the Security Trustee which Security is in form and substance satisfactory to the Agent and Security Trustee; and
- (d) from which no withdrawals may be made by any members of the Group except as contemplated by this Agreement,

(as the same may be redesignated, substituted or replaced from time to time).

"Margin" means:

- (a) in relation to any Facility A Loan, two point five per cent. (2.5%) per annum;
- (b) in relation to any Facility B Loan, three point seven five per cent. (3.75%) per annum;
- (c) in relation to any Facility C Loan, two point five per cent. (2.5)% per annum;
- (d) in relation to any Unpaid Sum relating or referable to a Facility, the rate per annum specified above for that Facility;
- (e) in relation to any other Unpaid Sum, the highest rate specified above; and
- (f) if at any time a breach of the terms of Clause 22 (Financial Covenants) is continuing each of the rates referred to in (a) to (e) (inclusive) above will increase to two times the Margin which would otherwise be applicable to such amount for so long as such breach continues unwaived.

"Material Adverse Effect" means any event or circumstance which is in the opinion of the Majority Lenders (acting reasonably) materially adverse to:

- (a) the business, operations, property, condition (financial or otherwise) or prospects of the Group taken as a whole; or
- (b) the ability of an Obligor to perform its payment obligations under any of the Finance Documents and/or its obligations under Clause 22.2 (Financial condition) of this Agreement; or
- (C) the validity or enforceability of, or the effectiveness or ranking of any Security granted or purporting to be granted pursuant to any of, the Finance Documents or the rights or remedies of any Finance Party under any of the Finance Documents.

"Month" means a period starting on one day in a calendar month and ending on the numerically corresponding day in the next calendar month, except that:

- (a) (subject to paragraph (c) below) if the numerically corresponding day is not a Business Day, that period shall end on the next Business Day in that calendar month in which that period is to end if there is one, or if there is not, on the immediately preceding Business Day;
- (b) if there is no numerically corresponding day in the calendar month in which that period is to end, that period shall end on the last Business Day in that calendar month; and

(c) if an Interest Period begins on the last Business Day of a calendar month, that Interest Period shall end on the last Business Day in the calendar month in which that Interest Period is to end.

The above rules will only apply to the last Month of any period. "**Monthly**" shall be construed accordingly.

"Monthly Financial Statements" means the financial statements delivered pursuant to paragraph (b) of Clause 21.1 (Financial Statements).

"Non-Consenting Lender" has the meaning given to that term in Clause 37.3 (Replacement of Lender).

"**Obligor**" means a Borrower or a Guarantor.

"**Obligors**' **Agent**" means the Parent, appointed to act on behalf of each Obligor in relation to the Finance Documents pursuant to Clause 2.4 (Obligors' Agent).

"Original Financial Statements" means:

- (a) in relation to the Parent, the audited consolidated financial statements of the Group for its financial year ended 31 January 2013 and the unaudited management accounts for the Accounting Period which ended on the last day of P7 2013; and
- (b) in relation to any other Obligor, its audited financial statements delivered to the Agent as required by Clause 26 (Changes to the Obligors).

"**Participating Member State**" means any member state of the European Communities that adopts or has adopted the euro as its lawful currency in accordance with legislation of the European Community relating to Economic and Monetary Union.

"Party" means a party to this Agreement.

"**Perfection Requirements**" means the making or the procuring of the appropriate registrations, filings, endorsements, notarisations, stampings and/or notifications of the Transaction Security Documents and/or the Transaction Security created thereunder.

"Permitted Acquisition" means:

- (a) an acquisition permitted in terms of Clause 23.29 (Capital Expenditure);
- (b) an acquisition by a member of the Group of an asset sold, leased, transferred or otherwise disposed of by another member of the Group in circumstances constituting a Permitted Disposal;
- (c) an acquisition of shares or securities pursuant to a Permitted Share Issue;
- (d) an acquisition approved in writing by the Majority Lenders.

"**Permitted Bank Accounts**" means each of the accounts held by Schuh (ROI) Limited with Ulster Bank (or such other bank notified by the Parent to the Agent from time to time).

"Permitted Disposal" means any sale, lease, licence, transfer or other disposal which is on arms' length terms:

(a) of trading stock or cash made by any member of the Group in the ordinary course of trading of the disposing entity;

- (b) of any asset by a member of the Group (the "Disposing Company") to another member of the Group (the "Acquiring Company"), but if:
- (i) the Disposing Company is an Obligor, the Acquiring Company must also be an Obligor;
 - (ii) the Disposing Company had given Security over the asset, the Acquiring Company must give equivalent Security over that asset; and
 - (iii) the Disposing Company is a Guarantor, the Acquiring Company must be a Guarantor guaranteeing at all times an amount no less than that guaranteed by the Disposing Company;
- (C) of assets (other than shares, businesses, any Properties or any Intellectual Property) in exchange for other assets comparable or superior as to type, value or quality;
- (d) of obsolete or redundant vehicles, plant and equipment for cash;
- (e) arising as a result of any Permitted Security; and
- (f) of assets (other than shares, businesses, any Properties or any Intellectual Property) for cash where the higher of the market value and net consideration receivable (when aggregated with the higher of the market value and net consideration receivable for any other sale, lease, licence, transfer or other disposal not allowed under the preceding paragraphs) does not exceed £200,000 (or its equivalent) in any Financial Year of the Parent.

"Permitted Financial Indebtedness" means Financial Indebtedness:

- (a) arising under a foreign exchange transaction for spot or forward delivery entered into in connection with protection against fluctuation in currency rates where that foreign exchange exposure arises in the ordinary course of trade, but not a foreign exchange transaction for investment or speculative purposes;
- (b) arising under a Permitted Loan or a Permitted Guarantee or as permitted by Clause 23.30 (Treasury Transactions);
- (C) of any person acquired by a member of the Group after the date of this Agreement which is incurred under arrangements in existence at the date of acquisition, but not incurred or increased or its maturity date extended in contemplation of, or since, that acquisition, and outstanding only for a period of three months following the date of acquisition;
- (d) under finance or capital leases of vehicles, plant, equipment or computers, provided that the aggregate capital value of all such items so leased under outstanding leases by members of the Group does not exceed £100,000 (or its equivalent in other currencies) at any time;
- (e) arising under the Working Capital Facility Letter;
- (f) arising in respect of the Retail Facilities;
- (g) arising under a loan made by the UK Acquisition Company to the Parent which is subordinated to the Facilities in accordance with the Subordination Agreement;
- (h) arising by way of a New Shareholder Injection; and

(i) not permitted by the preceding paragraphs or as a Permitted Transaction and the outstanding principal amount of which does not exceed £100,000 (or its equivalent) in aggregate for the Group at any time.

"Permitted Guarantee" means:

- (a) the endorsement of negotiable instruments in the ordinary course of trade;
- (b) any performance or similar bond guaranteeing performance by a member of the Group under any contract entered into in the ordinary course of trade;
- (c) any guarantee permitted under Clause 23.19 (Financial Indebtedness);
- (d) any guarantee given in respect of the netting or set-off arrangements permitted pursuant to paragraph (b) of the definition of Permitted Security; or
- (e) the Guarantee (as defined in the Ranking Agreement).

"**Permitted Joint Venture**" means any investment in any Joint Venture where the Parent has obtained the prior written consent (not to be unreasonably withheld or delayed) of the Majority Lenders for the investment in that Joint Venture.

"Permitted Loan" means:

- (a) any trade credit extended by any member of the Group to its customers on normal commercial terms and in the ordinary course of its trading activities;
- (b) Financial Indebtedness which is referred to in the definition of, or otherwise constitutes, Permitted Financial Indebtedness (except under paragraph (c) of that definition);
- (c) a loan made by an Obligor to another Obligor or made by a member of the Group which is not an Obligor to another member of the Group;
- (d) a loan made by a member of the Group to an employee or director of any member of the Group if the amount of that loan when aggregated with the amount of all loans to employees and directors by members of the Group does not exceed £100,000 (or its equivalent) at any time;
- (e) a loan made by the Parent or any of its wholly owned subsidiaries to the UK Acquisition Company in compliance with the terms of paragraph (b) of the definition of Permitted Payment; and
- (f) any other loan so long as the aggregate amount of the Financial Indebtedness under any such loans does not exceed £100,000 (or its equivalent) at any time.

"Permitted Merger" means:

- (a) an amalgamation, demerger, merger, consolidation or corporate reconstruction on a solvent basis of an Obligor where:
 - (i) all of the business and assets of that Obligor are retained by one or more other Obligors;
 - (ii) the surviving entity of that amalgamation, demerger, merger, consolidation or corporate reconstruction is liable for the obligations of the Obligor it has merged with and is incorporated in the same jurisdiction as that Obligor; and

- (iii) the Agent and the Security Trustee are given thirty Business Days' notice by the Parent of that proposed amalgamation, demerger, merger, consolidation or corporate reconstruction and the Security Trustee, acting reasonably, is satisfied that the Finance Parties will enjoy the same or equivalent Security over the same assets and over that Obligor and the shares in it (or the shares of the surviving entity); or
- (b) an amalgamation, demerger, merger, consolidation or corporate reconstruction on a solvent basis of a member of the Group which is not an Obligor where all of the business and assets of that member remain within the Group; or
- (c) an amalgamation or merger between an Obligor and another entity in connection with a Permitted Acquisition where such Obligor is the surviving entity.

"Permitted Payment" means:

- (a) the payment of a dividend to the Parent or any of its wholly-owned Subsidiaries;
- (b) the payment of a dividend or the making of a loan by the Parent to the UK Acquisition Company provided that:
 - (i) no Event of Default would occur as a result of that loan and/or dividend;
 - (ii) the Parent provides to the Agent such information as the Agent may reasonably require so as to demonstrate to the Agent that such loan and/or dividend would not cause a breach of the financial covenants under Clause 22 (Financial Covenants) of this Agreement in the period of 12 months following the making of such loan and/or payment of such dividend;
 - (iii) such loan and/or dividend is paid by the Parent into an account of the UK Acquisition Company held with the Agent or Security Trustee and subject to Security in favour of the Security Trustee which Security is in form and substance satisfactory to the Agent and the Security Trustee; and
 - (iv) simultaneously with the payment referred to in (iii) above, the UK Acquisition Company subscribes in cash for a corresponding amount in the equity capital of the Parent;
- (c) the payment of any other dividend agreed between the Parent and the Lenders.

"Permitted Security" means:

- (a) any lien arising by operation of law and in the ordinary course of trading and not as a result of any default or omission by any member of the Group;
- (b) any netting or set-off arrangement entered into by any member of the Group with Lloyds Bank plc in the ordinary course of its banking arrangements for the purpose of netting debit and credit balances of members of the Group but only so long as (i) such arrangement does not permit credit balances of Obligors to be netted or set off against debit balances of members of the Group which are not Obligors and (ii) such arrangement does not give rise to other Security over the assets of Obligors in support of liabilities of members of the Group which are not Obligors;
- (c) any Security or Quasi-Security over or affecting any asset acquired by a member of the Group after the date of this Agreement, if:
 - (i) the Security or Quasi-Security was not created in contemplation of the acquisition of that asset by a member of the Group;

- (ii) the principal amount secured has not been increased in contemplation of or since the acquisition of that asset by a member of the Group; and
- (iii) the Security or Quasi-Security is removed or discharged within three months of the date of acquisition of such asset;
- (d) any Security or Quasi-Security over or affecting any asset of any company which becomes a member of the Group after the date of this Agreement, where the Security or Quasi-Security is created prior to the date on which that company becomes a member of the Group, if:
 - (i) the Security or Quasi-Security was not created in contemplation of the acquisition of that company;
 - (ii) the principal amount secured has not increased in contemplation of or since the acquisition of that company; and
 - (iii) the Security or Quasi-Security is removed or discharged within three months of that company becoming a member of the Group;
- (e) any Security arising under any retention of title, hire purchase or conditional sale arrangement or arrangements having similar effect in respect of goods supplied to a member of the Group in the ordinary course of trading and on the supplier's standard or usual terms and not arising as a result of any default or omission by any member of the Group;
- (f) any Security or Quasi-Security arising as a consequence of any finance or capital lease permitted pursuant to paragraph (e) of the definition of "Permitted Financial Indebtedness";
- (g) any Security in favour of Bank of Scotland plc or any other member of the LBG Group by any member of the Group;
- (h) any Security securing indebtedness the outstanding principal amount of which (when aggregated with the outstanding principal amount of any other indebtedness which has the benefit of Security given by any member of the Group other than any permitted under paragraphs (a) to (g) above) does not exceed £100,000; or
- (i) the Noteholder Securities (as defined in the Ranking Agreement).

"Permitted Share Issue" means an issue of:

- (a) ordinary shares by the Parent to employees, paid for in full in cash upon issue and which by their terms are not redeemable and where such issue does not lead to a Change of Control of the Parent;
- (b) shares by a member of the Group which is a Subsidiary to its immediate Holding Company where (if the existing shares of the Subsidiary are the subject of the Transaction Security) the newly-issued shares also become subject to the Transaction Security on the same terms.

"Permitted Transaction" means:

- (a) any disposal required, Financial Indebtedness incurred, guarantee, indemnity or Security or Quasi-Security given, or other transaction arising, under the Finance Documents;
- (b) the solvent liquidation or reorganisation of any member of the Group which is not an Obligor so long as any payments or assets distributed as a result of such liquidation or reorganisation are distributed to an Obligor; or
- (C) transactions (other than (i) any sale, lease, license, transfer or other disposal and (ii) the granting or creation of Security or the incurring or permitting to subsist of Financial Indebtedness) conducted in the ordinary course of trading on arms' length terms.

"**Properties**" means the heritable or freehold properties owned by the Group at the date of this Agreement and any other freehold property acquired by a member of the Group after the date of this Agreement. A reference to "**Property**" is a reference to any of the Properties.

"Qualifying Lender" has the meaning given to that term in Clause 14 (Tax gross-up and indemnities).

"**Quarter Date**" means the last day of a Financial Quarter.

"Quarterly Financial Statements" means the management accounts for a Financial Quarter.

"Quasi-Security" has the meaning given to that term in Clause 23.13 (Negative pledge).

"Quotation Day" means, in relation to any period for which an interest rate is to be determined, the first day of that period.

"**Ranking Agreement**" means the ranking agreement dated on or around the Effective Date and made between the Security Trustee, the Noteholder Security Trustee (as defined therein), Schuh Limited and Schuh (ROI) Limited.

"Receiver" means a receiver or receiver and manager or administrative receiver of the whole or any part of the Charged Property.

"**Reference Banks**" means, in relation to LIBOR, the principal London offices of Barclays Bank PLC, Lloyds Bank plc and The Royal Bank of Scotland plc or such other banks as may be appointed by the Agent in consultation with the Parent.

"**Related Fund**" in relation to a fund (the "**first fund**"), means a fund which is managed or advised by the same investment manager or investment adviser as the first fund or, if it is managed by a different investment manager or investment adviser, a fund whose investment manager or investment adviser is an Affiliate of the investment manager or investment adviser of the first fund.

"Relevant Jurisdiction" means, in relation to an Obligor:

- (a) its jurisdiction of incorporation;
- (b) any jurisdiction where any asset subject to or intended to be subject to the Transaction Security to be created by it is situated;
- (c) any jurisdiction where it conducts its business; and

(d) the jurisdiction whose laws govern the perfection of any of the Transaction Security Documents entered into by it.

"Relevant Period" has the meaning given to that term in Clause 22.2 (Financial definitions).

"Repayment Date" means a Facility A Repayment Date, the Facility B Repayment Date or a Facility C Repayment Date.

"**Repeating Representations**" means each of the representations set out in Clause 20.2 (Status) to Clause 20.7 (Governing law and enforcement) (inclusive), Clause 20.11 (No default), paragraphs (f) and (g) of Clause 20.13 (Original Financial Statements), Clause 20.19 (Ranking) to Clause 20.21 (Legal and beneficial ownership) (inclusive) and Clause 20.28 (Centre of main interests and establishments).

"Reports" means the Financial Due Diligence Report and the Tax Letter.

"Resignation Letter" means a letter substantially in the form set out in Schedule 8 (Form of Resignation Letter).

"Restatement Date" means 22 June 2011.

"**Retail Facilities**" means the retail and merchant services facilities used by the Group from time to time (including, without limitation, the Existing Retail Facilities and the Lloyds Retail Facilities).

"Screen Rate" means, in relation to LIBOR, the London interbank offered rate administered by British Bankers' Association (or any other person which takes over the administration of such rate) for the relevant period displayed on pages LIBOR01 or LIBOR02 of the Reuters screen (or any replacement Reuters page which displays that rate). If the agreed page is replaced or service ceases to be available, the Agent may specify another page or service displaying the appropriate rate after consultation with the Parent and the Lenders.

"Second Amendment and Restatement Agreement" means the amendment and restatement agreement in respect of this Agreement dated on the Second Restatement Date.

"Second Restatement Date" means 2013.

"Secured Obligations" means all present and future obligations and liabilities (whether actual or contingent, whether owed jointly, severally or in any other capacity whatsoever and whether originally incurred by an Obligor or by some other person) of each Obligor to the Finance Parties (or any of them) under each of the Finance Documents except for any obligation or liability which, if it were so included, would cause that obligation or liability or any of the Security in respect thereof, to be unlawful or prohibited by any applicable law.

"Secured Parties" means each Finance Party from time to time and any Receiver or Delegate.

"Security" means a mortgage, charge, pledge, lien or other security interest securing any obligation of any person or any other agreement or arrangement having a similar effect.

"Selection Notice" means a notice substantially in the form set out in Part II of Schedule 3 (Requests) given in accordance with Clause 11 (Interest Periods) in relation to a Term Facility.

"Senior Management" means each and all of Colin Temple, Mark Crutchley, Kenneth Ball and David Spencer.

"Senior Management Event" means Colin Temple and Mark Crutchley cease (in the period of 24 months following the Second Restatement Date) to be employed by the Parent (as chief executive officer and chief financial officer of the Group respectively) (the date on which both

of such individuals cease to be so employed being for these purposes the "**Trigger Date**") and, following the presentation to the Lenders of the Group's alternative arrangements for the senior management of the Group (including, without limitation, the identity of the proposed new chief executive officer and chief financial officer of the Group) within 120 days of the Trigger Date, the Agent (acting on the instructions of the Majority Lenders, acting reasonably) does not provide a written consent to such cessation within 180 days of the Trigger Date. This definition shall also apply to any replacement person approved by the Agent in accordance with the terms of this definition as if references in this Clause to Colin Temple or to Mark Crutchley were references to that replacement person.

"Standard Security" means the standard security by the Company in favour of the Security Trustee in respect of the Livingston Property dated on or around the date of this Agreement.

"Sterling" and the figure "£" means the lawful currency of the UK.

"**Subordination Agreement**" means the subordination agreement between the Borrower, UK Acquisition Company and the Security Trustee as security trustee for the Finance Parties dated 22 June, 23 June and 7 November 2011 .

"Subsidiary" means, in relation to any company, corporation or legal entity (a "holding company"), any company, corporation or legal entity:

- (a) which is controlled, directly or indirectly, by the holding company; or
- (b) more than half the issued share capital of which is beneficially owned, directly or indirectly, by the holding company; or
- (c) which is a subsidiary of another subsidiary of the holding company,

and, for these purposes, a company, corporation or legal entity shall be treated as being controlled by another if that other company, corporation or legal entity is able to direct its affairs and/or to control the composition of its board of directors or equivalent body.

"**Tax**" means any tax, levy, impost, duty or other charge or withholding of a similar nature (including any penalty or interest payable in connection with any failure to pay or any delay in paying any of the same).

"Tax Letter" means the letter dated on or around the date of this Agreement from KPMG to the Finance Parties in respect of the tax position of the Group.

"Taxes Act" means the Income and Corporation Taxes Act 1988.

"Term Facilities" means Facility A, Facility B and Facility C and "Term Facility" means any of them as the context may require.

"Termination Date" means:

- (a) in relation to Facility A, 31 October 2015;
- (b) in relation to Facility B, 31 October 2015; and
- (c) in relation to Facility C, 30 September 2019.

"Term Loan" means a Facility A Loan, a Facility B Loan or a Facility C Loan.

"**Total Commitments**" means the aggregate of the Total Facility A Commitments, the Total Facility B Commitments and the Total Facility C Commitments, being £24,357,000 at the Second Restatement Date.

"Total Facility A Commitments" means the aggregate of the Facility A Commitments, being £7,250,000 at the Second Restatement Date.

"Total Facility B Commitments" means the aggregate of the Facility B Commitments, being £4,607,000 at the Second Restatement Date.

"Total Facility C Commitments" means the aggregate of the Facility C Commitments, being £12,500,000 at the Second Restatement Date.

"Trade Sale" means a disposal (whether in a single transaction or a series of related transactions) of all or substantially all of the business and assets of the Group.

"Transaction Documents" means the Acquisition Agreement, the Constitutional Documents and the Finance Documents.

"**Transaction Security**" means the Security created or expressed to be created in favour of the Security Trustee pursuant to the Transaction Security Documents.

"**Transaction Security Documents**" means each of the documents listed as being a Transaction Security Document in paragraph 3(g) - (n) of Part I of Schedule 2 (Conditions Precedent), the Additional Standard Security, the Collateral Warranty Assignment and any document required to be delivered to the Agent under paragraph 13 of Part II of Schedule 2 (Conditions Precedent) together with any other any document entered into by any Obligor creating or expressed to create any Security over all or any part of its assets in respect of the obligations of any of the Obligors under any of the Finance Documents.

"**Transfer Certificate**" means a certificate substantially in the form set out in Schedule 5 (Form of Transfer Certificate) or any other form agreed between the Agent and the Parent.

"Transfer Date" means, in relation to an assignment or transfer, the later of:

- (a) the proposed Transfer Date specified in the relevant Assignment Agreement or Transfer Certificate; and
- (b) the date on which the Agent executes the relevant Assignment Agreement or Transfer Certificate.

"Treasury Transactions" means any derivative transaction entered into in connection with protection against or benefit from fluctuation in any rate or price.

"UK Acquisition Company" means Genesco (UK) Limited (Reg. No. 7667223).

"Unpaid Sum" means any sum due and payable but unpaid by an Obligor under the Finance Documents.

"Utilisation" means a Loan.

"Utilisation Date" means the date on which a Utilisation is made.

"Utilisation Request" means a notice substantially in the form set out in Part 1 of Schedule 3 (Requests).

"VAT" means value added tax as provided for in the Value Added Tax Act 1994 and any other tax of a similar nature.

"Working Capital Facility" means the working capital facilities provided by the Working Capital Lender to various members of the Group pursuant to the terms of the Working Capital Facility Letter.

"Working Capital Facility Letter" means the working capital facility letter between each of the Parent, Holdings and the Company and Lloyds Bank plc dated on or around the Restatement Date or any other facility letter entered into from time to time between members of the Group and the Working Capital Lender setting out the terms on which working capital facilities are available to the Group.

"Working Capital Lender" means Lloyds Bank plc (or another member of the LBG Group) as provider of the Working Capital Facility.

1.2. Construction

- (a) Unless a contrary indication appears, a reference in this Agreement to:
 - (i) the "Agent", the "Arranger", any "Finance Party", any "Lender", any "Obligor", any "Party", any "Secured Party", the "Security Trustee" or any other person shall be construed so as to include its successors in title, permitted assigns and permitted transferees to, or of, its rights and/or obligations under the Finance Documents and, in the case of the Security Trustee, any person for the time being appointed as Security Trustee or Security Trustees in accordance with the Finance Documents;
 - (ii) a document in "**agreed form**" is a document which is previously agreed in writing by or on behalf of the Parent and the Agent or, if not so agreed, is in the form specified by the Agent;
 - (iii) "assets" includes present and future properties, revenues and rights of every description;
 - (iv) a "Finance Document" or a "Transaction Document" or any other agreement or instrument is a reference to that Finance Document or Transaction Document or other agreement or instrument as amended or novated (however fundamentally) and includes (without limiting the generality of the foregoing) any variation, increase, extension or addition of or to any facility or amount made available under any such document or any variation of the purposes for which such facility or amount may be available from time to time;
 - (v) "guarantee" means (other than in Clause 19 (Guarantee and Indemnity) any guarantee, letter of credit, bond, indemnity or similar assurance against loss, or any obligation, direct or indirect, actual or contingent, to purchase or assume any indebtedness of any person or to make an investment in or loan to any person or to purchase assets of any person where, in each case, such obligation is assumed in order to maintain or assist the ability of such person to meet its indebtedness;
 - (vi) "**indebtedness**" includes any obligation (whether incurred as principal or as surety) for the payment or repayment of money, whether present or future, actual or contingent;
 - (vii) a "**person**" includes any individual, firm, company, corporation, government, state or agency of a state or any association, trust, joint venture, consortium or partnership (whether or not having separate legal personality);

- 37
- (viii) a "**regulation**" includes any regulation, rule, official directive, request or guideline (whether or not having the force of law but, if not having the force of law, being one with which it is customary to comply) of any governmental, intergovernmental or supranational body, agency, department or regulatory, self-regulatory or other authority or organisation;
- (ix) a provision of law is a reference to that provision as amended or re-enacted;
- (x) a time of day is a reference to London time; and
- (xi) the "**date of this Agreement**" shall be a reference to 10 November 2010.
- (b) Section, Clause and Schedule headings are for ease of reference only.
- (C) Unless a contrary indication appears, a term used in any other Finance Document or in any notice given under or in connection with any Finance Document has the same meaning in that Finance Document or notice as in this Agreement.
- (d) A Default (other than an Event of Default) is "**continuing**" if it has not been remedied or waived and an Event of Default is "**continuing**" if it has not been remedied or waived.

1.3. Third party rights

- (a) Unless expressly provided to the contrary in a Finance Document, a person who is not a Party has no right under the Contracts (Rights of Third Parties) Act 1999 to enforce or enjoy the benefit of any term of this Agreement.
- (b) Notwithstanding any term of any Finance Document, the consent of any person who is not a Party is not required to rescind or vary this Agreement at any time.

SECTION 2

THE FACILITIES

2. THE FACILITIES

2.1. The Facilities

- (a) Subject to the terms of this Agreement, the Lenders make available:
 - (i) a Sterling term loan facility in an aggregate amount equal to the Total Facility A Commitments which will be made available to Holdings; and
 - (ii) a Sterling term loan facility in an aggregate amount equal to the Total Facility B Commitments which shall be made available to Holdings; and
 - (iii) a Sterling term loan facility in an aggregate amount equal to the Total Facility C Commitments which shall be made available to the Company.

2.2. Increase

- (a) The Parent may by giving prior notice to the Agent by no later than the date falling 5 Business Days after the effective date of a cancellation of:
 - (i) the Available Commitments of a Defaulting Lender in accordance with Clause 7.5 (Right of cancellation in relation to a Defaulting Lender); or
 - (ii) the Commitments of a Lender in accordance with Clause 7.1 (Illegality),

request that the Commitments relating to any Facility be increased (and the Commitments relating to that Facility shall be so increased) in an aggregate amount in the Base Currency of up to the amount of the Available Commitments or Commitments so cancelled as follows:

- (iii) the increased Commitments will be assumed by one or more Lenders or other banks, financial institutions, trusts, funds or other entities (each an "Increase Lender") selected by the Company (each of which shall not be a member of the Group nor an Investor nor an Affiliate of an Investor and which is further acceptable to the Agent (acting reasonably)) and each of which confirms its willingness to assume and does assume all the obligations of a Lender corresponding to that part of the increased Commitments which it is to assume, as if it had been an Original Lender;
- (iv) each of the Obligors and any Increase Lender shall assume obligations towards one another and/or acquire rights against one another as the Obligors and the Increase Lender would have assumed and/or acquired had the Increase Lender been an Original Lender;
- (v) each Increase Lender shall become a Party as a "**Lender**" and any Increase Lender and each of the other Finance Parties shall assume obligations towards one another and acquire rights against one another as that Increase Lender and those Finance Parties would have assumed and/or acquired had the Increase Lender been an Original Lender;
- (vi) the Commitments of the other Lenders shall continue in full force and effect; and

- (vii) any increase in the Commitments relating to a Facility shall take effect on the date specified by the Company in the notice referred to above or any later date on which the conditions set out in paragraph (b) below are satisfied.
- (b) An increase in the Commitments relating to a Facility will only be effective on:
 - (i) the execution by the Agent of an Increase Confirmation from the relevant Increase Lender promptly on receipt on such Increase Confirmation; and
 - (ii) in relation to an Increase Lender which is not a Lender immediately prior to the relevant increase:
 - (A) the Increase Lender entering into any intercreditor documentation required by the Agent and the Lenders (acting reasonably); and
 - (B) the performance by the Agent of all necessary "know your customer" or other similar checks under all applicable laws and regulations in relation to the assumption of the increased Commitments by that Increase Lender, the completion of which the Agent shall promptly notify to the Company and the Increase Lender.
- (c) Each Increase Lender, by executing the Increase Confirmation, confirms (for the avoidance of doubt) that the Agent has authority to execute on its behalf any amendment or waiver that has been approved by or on behalf of the requisite Lender or Lenders in accordance with this Agreement on or prior to the date on which the increase becomes effective.
- (d) Unless the Agent otherwise agrees or the increased Commitment is assumed by an existing Lender, the Company shall, on the date upon which the increase takes effect, pay to the Agent (for its own account) a fee of £5,000 and the Company shall promptly on demand pay the Agent and the Security Trustee the amount of all costs and expenses (including legal fees) reasonably incurred by either of them and, in the case of the Security Trustee, by any Receiver or Delegate in connection with any increase in Commitments under this Clause 2.2.
- (e) The Parent may pay to the Increase Lender a fee in the amount and at the times agreed between the Parent and the Increase Lender in a Fee Letter.
- (f) Clause 25.4 (Limitation of responsibility of Existing Lenders) shall apply *mutatis mutandis* in this Clause 2.2 in relation to an Increase Lender as if references in that Clause to:
 - (i) an "Existing Lender" were references to all the Lenders immediately prior to the relevant increase;
 - (ii) the "New Lender" were references to that "Increase Lender"; and
 - (iii) a "re-transfer" and "re-assignment" were references to respectively a "transfer" and "assignment".

2.3. Finance Parties' rights and obligations

(a) The obligations of each Finance Party under the Finance Documents are several. Failure by a Finance Party to perform its obligations under the Finance Documents does not affect the obligations of any other Party under the Finance Documents. No Finance Party is responsible for the obligations of any other Finance Party under the Finance Documents.

- (b) The rights of each Finance Party under or in connection with the Finance Documents are separate and independent rights and any debt arising under the Finance Documents to a Finance Party from an Obligor shall be a separate and independent debt.
- (c) A Finance Party may, except as otherwise stated in the Finance Documents, separately enforce its rights under the Finance Documents.

2.4. Obligors' Agent

- (a) Each Obligor (other than the Parent) by its execution of this Agreement or an Accession Letter irrevocably appoints the Parent (acting through one or more authorised signatories) to act on its behalf as its agent in relation to the Finance Documents and irrevocably authorises:
 - (i) the Parent on its behalf to supply all information concerning itself contemplated by this Agreement to the Finance Parties and to give all notices and instructions (including, in the case of a Borrower, Utilisation Requests), to execute on its behalf any Accession Letter, to make such agreements and to effect the relevant amendments, supplements and variations capable of being given, made or effected by any Obligor notwithstanding that they may affect the Obligor, without further reference to or the consent of that Obligor; and
 - (ii) each Finance Party to give any notice, demand or other communication to that Obligor pursuant to the Finance Documents to the Parent,

and in each case the Obligor shall be bound as though the Obligor itself had given the notices and instructions (including, without limitation, any Utilisation Requests) or executed or made the agreements or effected the amendments, supplements or variations, or received the relevant notice, demand or other communication.

(b) Every act, omission, agreement, undertaking, settlement, waiver, amendment, supplement, variation, notice or other communication given or made by the Obligors' Agent or given to the Obligors' Agent under any Finance Document on behalf of another Obligor or in connection with any Finance Document (whether or not known to any other Obligor and whether occurring before or after such other Obligor became an Obligor under any Finance Document) shall be binding for all purposes on that Obligor as if that Obligor had expressly made, given or concurred with it. In the event of any conflict between any notices or other communications of the Obligors' Agent and any other Obligor, those of the Obligors' Agent shall prevail.

3. PURPOSE

3.1. Purpose

- (a) The Parent shall apply all amounts borrowed by it under Facility A and Facility B towards the payment of certain sums due to the shareholders of Holdings, certain of whom are exiting on or around the date of this Agreement and Holdings shall apply all amounts borrowed by it under Facility A and Facility B towards the refinance of certain facilities previously made available to it and certain of its Subsidiaries by Bank of Scotland plc.
- (b) The Company shall apply all amounts borrowed by it under Facility C toward the financing of the acquisition of the Additional Livingston Property, a new sorter, restructuring costs and the conversion of existing warehouse premises to offices.

3.2. Monitoring

No Finance Party is bound to monitor or verify the application of any amount borrowed pursuant to this Agreement.

4. CONDITIONS OF UTILISATION

4.1. Initial conditions precedent

The Lenders will only be obliged to comply with Clause 5.4 (Lenders' participation) in relation to any Utilisation if on or before the Utilisation Date for that Utilisation, the Agent has received all of the documents and other evidence listed in Part I of Schedule 2 (Conditions precedent) in form and substance satisfactory to the Agent. The Agent shall notify the Parent and the Lenders promptly upon being so satisfied.

4.2. Further conditions precedent

Subject to Clause 4.1 (Initial conditions precedent), the Lenders will only be obliged to comply with Clause 5.4 (Lenders' participation), if on the date of the Utilisation Request and on the proposed Utilisation Date:

- (a) no Default is continuing or would result from the proposed Utilisation; and
- (b) in relation to any Utilisation on the date of this Agreement, all the representations and warranties in Clause 20 (Representations) or, in relation to any other Utilisation, the Repeating Representations, to be made by each Obligor are true.

4.3. Maximum number of Utilisations

- (a) A Borrower (or the Parent) may not deliver a Utilisation Request if as a result of the proposed Utilisation six or more Term Loans would be outstanding.
- (b) A Borrower (or the Parent) may not request that a Facility A Loan be divided if, as a result of the proposed division, three or more Facility A Loans would be outstanding.
- (c) A Borrower (or the Parent) may not request that a Facility B Loan be divided.
- (d) A Borrower (or the Parent) may not request that a Facility C Loan be divided if, as a result of the proposed division, four or more Facility C Loans would be outstanding.

UTILISATION

5. UTILISATION – LOANS

5.1. Delivery of a Utilisation Request

A Borrower (or the Parent on its behalf) may utilise a Facility by delivery to the Agent of a duly completed Utilisation Request not later than 10.00 a.m. one Business Day prior to the proposed Utilisation Date.

5.2. Completion of a Utilisation Request for Loans

- (a) Each Utilisation Request for a Loan is irrevocable and will not be regarded as having been duly completed unless:
 - (i) it identifies the Facility to be utilised;
 - (ii) the proposed Utilisation Date is a Business Day within the Availability Period applicable to that Facility;
 - (iii) the currency and amount of the Utilisation comply with Clause 5.3 (Currency and amount); and
 - (iv) the proposed Interest Period complies with Clause 11 (Interest Periods).
- (b) Multiple Utilisations may be requested in a Utilisation Request where the proposed Utilisation Date is the date of this Agreement. Only one Utilisation may be requested in each subsequent Utilisation Request.

5.3. Currency and amount

- (a) The currency specified in a Utilisation Request must be Sterling.
- (b) The amount of the proposed Utilisation must be:
 - (i) an amount equal to £7,000,000 for that part of Facility A which is to be drawn down by the Parent or, if less, the Available Facility;
 - (ii) an amount equal to £10,000,000 for that part of Facility A which is to be drawn down by Holdings or, if less, the Available Facility;
 - (iii) an amount equal to £14,000,000 for Facility B or, if less, the Available Facility; or
 - (iv) a minimum amount of £500,000 (or, if less, the Available Facility) during the Availability Period in respect of Facility C.

5.4. Lenders' participation

(a) If the conditions set out in this Agreement have been met, each Lender shall make its participation in each Loan available by the Utilisation Date through its Facility Office.

(b) The amount of each Lender's participation in each Loan will be equal to the proportion borne by its Available Commitment to the Available Facility immediately prior to making the Loan.

5.5. Limitations on Utilisations

A Term Facility may only be utilised in accordance with the terms of this Agreement.

5.6. Cancellation of Commitment

- (a) The Facility A Commitments which, at that time, are unutilised shall be immediately cancelled at the end of the Availability Period for Facility A.
- (b) The Facility B Commitments which, at that time, are unutilised shall be immediately cancelled at the end of the Availability Period for Facility B.
- (c) The Facility C Commitments which, at that time, are unutilised shall be immediately cancelled at the end of the Availability Period for Facility C.

5.7. Clean down

The Parent shall ensure that:

- (a) the amount outstanding under the Working Capital Facility Letter; LESS
- (b) any amount of cash (other than cash held in a Holding Account or Mandatory Prepayment Account) held by wholly-owned members of the Group,

(as confirmed in a certificate signed by a director of the Parent provided to the Agent within 15 Business Days after the end of each Financial Year) shall not exceed zero for a period of not less than 10 successive days in each of its Financial Years. Not less than 9 months shall elapse between two such periods.

SECTION 4

REPAYMENT, PREPAYMENT AND CANCELLATION

6. **REPAYMENT**

6.1. Repayment of Term Loans

(a) The Borrowers shall repay the Facility A Loans in instalments by repaying on each Facility A Repayment Date the amount set out opposite that Facility A Repayment Date below:

Facility A Repayment Date	Repayment Instalment (£)
31 December 2013	875,000
31 March 2014	875,000
30 June 2014	875,000
30 September 2014	875,000
31 December 2014	875,000
31 March 2015	875,000
30 June 2015	1,000,000
31 October 2015	1,000,000

- (b) Holdings shall repay the Facility B Loans on the Facility B Repayment Date.
- (c) The Borrowers shall repay the Facility C Loans in instalments by repaying on each Facility C Repayment Date the amount set out opposite that Facility C Repayment Date below:

Facility C Repayment Date

Repayment Instalment (£)

30 September 2014	312,500
31 December 2014	312,500
31 March 2015	312,500
30 June 2015	312,500
30 September 2015	312,500
31 December 2015	312,500
31 March 2016	312,500
30 June 2016	312,500
30 September 2016	312,500
31 December 2016	312,500
31 March 2017	312,500
30 June 2017	312,500
30 September 2017	312,500
31 December 2017	312,500
31 March 2018	312,500
30 June 2018	312,500
30 September 2018	312,500
31 December 2018	312,500
31 March 2019	312,500
30 June 2019	312,500
30 September 2019	6,250,000
he Borrowers may not reportow any part of a Term Facility which is repaid	

(d) The Borrowers may not reborrow any part of a Term Facility which is repaid.

6.2. Effect of cancellation and prepayment on scheduled repayments and reductions

- (a) If the Parent cancels the whole or any part of the Facility A Commitments in accordance with Clause 7.4 (Right of cancellation and repayment in relation to a single Lender) or if the Facility A Commitment of any Lender is reduced under Clause 7.1 (Illegality) then the amount of the Facility A Repayment Instalment for each Facility A Repayment Date falling after that cancellation will reduce *pro rata* by the amount cancelled.
- (b) If any of the Facility A Loans are prepaid in accordance with Clause 7.4 (Right of cancellation and repayment in relation to a single Lender) or Clause 7.1 (Illegality) then the amount of the Facility A Repayment Instalment for each Facility A Repayment Date falling after that prepayment will reduce *pro rata* by the amount of the Facility A Loan.
- (C) Subject to the terms of paragraph (d) below, if any of the Facility A Loans are prepaid in accordance with Clause 7.3 (Voluntary prepayment of Term Loans) or Clause 8.3 (Application of mandatory prepayments) then the amount of the Facility A Repayment Instalment for each Facility A Repayment Date falling after that prepayment will reduce in inverse order of maturity by the amount of the Facility A Loan prepaid.
- (d) If any of the Facility A Loans are prepaid pursuant to the terms of Clause 7.3(a), Clause 8.2(b) or Clause 8.3(c)(ii) then the amount prepaid will be applied against the Facility A Repayment Instalments falling after that prepayment in inverse order of maturity.
- (e) If the Parent cancels the whole or any part of the Facility C Commitments in accordance with Clause 7.4 (Right of cancellation and repayment in relation to a single Lender) or if the Facility C Commitment of any Lender is reduced under Clause 7.1 (Illegality) then the amount of the Facility C Repayment Instalment for each Facility C Repayment Date falling after that cancellation will reduce *pro rata* by the amount cancelled.

- (f) If any of the Facility C Loans are prepaid in accordance with Clause 7.4 (Right of cancellation and repayment in relation to a single Lender) or Clause 7.1 (Illegality) then the amount of the Facility C Repayment Instalment for each Facility C Repayment Date falling after that prepayment will reduce *pro rata* by the amount of the Facility C Loan.
- (g) Subject to the terms of paragraph (d) below, if any of the Facility C Loans are prepaid in accordance with Clause 7.3 (Voluntary prepayment of Term Loans) or Clause 8.3 (Application of mandatory prepayments) then the amount of the Facility C Repayment Instalment for each Facility C Repayment Date falling after that prepayment will reduce in inverse order of maturity by the amount of the Facility C Loan prepaid.
- (h) If any of the Facility C Loans are prepaid pursuant to the terms of Clause 7.3(a), Clause 8.2(b) or Clause 8.3(c)(ii) then the amount prepaid will be applied against the Facility C Repayment Instalments falling after that prepayment in inverse order of maturity.

7. ILLEGALITY, VOLUNTARY PREPAYMENT AND CANCELLATION

7.1. Illegality

If it becomes unlawful in any applicable jurisdiction for a Lender to perform any of its obligations as contemplated by this Agreement or to fund, issue or maintain its participation in any Utilisation:

- (a) that Lender shall promptly notify the Agent upon becoming aware of that event;
- (b) upon the Agent notifying the Parent, each Available Commitment of that Lender will be immediately cancelled; and
- (C) each Borrower shall repay that Lender's participation in the Utilisations made to that Borrower on the last day of the Interest Period for each Utilisation occurring after the Agent has notified the Parent or, if earlier, the date specified by the Lender in the notice delivered to the Agent (being no earlier than the last day of any applicable grace period permitted by law).

7.2. Voluntary cancellation

- (a) Subject to paragraph (b) below, the Parent may, if it gives the Agent not less than 5 Business Days' (or such shorter period as the Majority Lenders may agree) prior notice, cancel the whole or any part (being a minimum amount of £250,000 and an integral multiple of £250,000) of an Available Facility. Any cancellation under this Clause 7.2 shall reduce the Commitments of the Lenders rateably under that Facility.
- (b) The Parent shall not cancel any part of the Available Commitment with respect to Facility A unless there is no Available Commitment with respect to Facility B. The Parent shall not cancel any part of the Available Commitment with respect to Facility C unless there is no Available Commitment with respect to Facility A or Facility B.

7.3. Voluntary prepayment of Term Loans

- (a) Subject to paragraph (b) below, a Borrower may, if it gives the Agent not less than 5 Business Days' (or such shorter period as the Majority Lenders may agree) prior notice, prepay the whole or any part of that Term Loan (but, if in part, being an amount that reduces the amount of that Term Loan by a minimum amount of £250,000 and an integral multiple of £50,000).
- (b) A Term Loan may only be prepaid after the last day of the Availability Period for the applicable Facility (or, if earlier, the day on which the applicable Available Facility is zero).

(C) The Facility A Loans shall only be prepaid if the whole of the Facility B Loans have been prepaid or will be prepaid at the same time. A prepayment of the Facility A Loans shall be applied to reduce the Facility A Repayment Instalments in inverse order of maturity. The Facility C Loans shall only be prepaid if the whole of the Facility A Loans have been prepaid or will be prepaid at the same time. A prepayment of Facility C Loans shall be applied to reduce the Facility C Repayment Instalments in inverse order of maturity.

7.4. Right of cancellation and repayment in relation to a single Lender

- (a) If:
 - (i) any sum payable to any Lender by an Obligor is required to be increased under paragraph (i) of Clause 14.2 (Tax gross-up); or
 - (ii) any Lender claims indemnification from the Parent or an Obligor under Clause 14.3 (Tax indemnity) or Clause 15.1 (Increased costs),

the Parent may, whilst the circumstance giving rise to the requirement for indemnification continues, give the Agent notice of cancellation of the Commitment of that Lender and its intention to procure the repayment of that Lender's participation in the Utilisations.

- (b) On receipt of a notice referred to in paragraph (a) above in relation to a Lender, the Commitment of that Lender shall immediately be reduced to zero.
- (C) On the last day of each Interest Period which ends after the Parent has given notice under paragraph (a) above in relation to a Lender (or, if earlier, the date specified by the Parent in that notice), each Borrower to which a Utilisation is outstanding shall repay that Lender's participation in that Utilisation together with all interest and other amounts accrued under the Finance Documents.

7.5. Right of cancellation in relation to a Defaulting Lender

- (a) If any Lender becomes a Defaulting Lender, the Company may, at any time whilst the Lender continues to be a Defaulting Lender, give the Agent 5 Business Days' notice of cancellation of each Available Commitment of that Lender.
- (b) On the notice referred to in paragraph (a) above becoming effective, each Available Commitment of the Defaulting Lender shall immediately be reduced to zero.
- (c) The Agent shall as soon as practicable after receipt of a notice referred to in paragraph (a) above, notify all the Lenders.

8. MANDATORY PREPAYMENT

8.1. Exit

Upon the occurrence of:

- (i) any Flotation;
- (ii) a Change of Control;
- (iii) a Trade Sale; or

(iv) a Senior Management Event,

the Facilities will be cancelled and all outstanding Utilisations, together with accrued interest, and all other amounts accrued under the Finance Documents, shall become immediately due and payable.

8.2. Disposal and Insurance Proceeds and Excess Cashflow

(a) For the purposes of this Clause 8.2, Clause 8.3 (Application of mandatory prepayments) and Clause 8.4 (Mandatory Prepayment Accounts and Holding Accounts):

"**Disposal**" means a sale, lease, licence, transfer, loan or other disposal by a person of any asset, undertaking or business (whether by a voluntary or involuntary single transaction or series of transactions).

"**Disposal Proceeds**" means the consideration receivable by any member of the Group (including any amount receivable in repayment of intercompany debt) for any Disposal made by any member of the Group except for Excluded Disposal Proceeds and after deducting:

- (i) any reasonable expenses which are incurred by any member of the Group with respect to that Disposal to persons who are not members of the Group; and
- (ii) any Tax incurred and required to be paid by the seller in connection with that Disposal (as reasonably determined by the seller, on the basis of existing rates and taking account of any available credit, deduction or allowance).

"Excluded Disposal Proceeds" means any Disposal Proceeds which:

- (i) are applied in replacement of the assets in respect of which the relevant Disposal was made as soon as possible (but in any event within 90 days or, in the case of a disposal of land or buildings, within 12 months or, in any case, such longer period as the Majority Lenders may agree) after receipt; or
- (ii) do not exceed £1,000,000.

"Excluded Insurance Proceeds" means any proceeds of an insurance claim which the Parent notifies the Agent are, or are to be, applied:

- (a) to meet a third party claim;
- (b) to compensate for a loss to be covered under any business interruption insurance policies; or
- (c) to the replacement, reinstatement and/or repair of the assets or otherwise in amelioration of the loss in respect of which the relevant insurance claim was made,

as soon as possible (but in any event within 120 days or, in the case of proceeds in relation to any land or buildings, within 12 months or, in any case, such longer period as the Majority Lenders may agree) after receipt and provided that the amount of such proceeds will cease to fall within this definition if they are not so applied within such period.

"**Insurance Proceeds**" means the proceeds of any insurance claim received by any member of the Group except for Excluded Insurance Proceeds and after deducting any

reasonable expenses in relation to that claim which are incurred by any member of the Group to persons who are not members of the Group.

- (b) The Parent shall ensure that the Borrowers prepay Utilisations in the following amounts at the times and in the order of application contemplated by Clause 8.3 (Application of mandatory prepayments):
 - (i) the amount of Disposal Proceeds;
 - (ii) the amount of Insurance Proceeds;
 - (iii) the amount equal to 50% of the Excess Cashflow for any Financial Year of the Parent, commencing with the Financial Year ending on or around 31 March 2012.

8.3. Application of mandatory prepayments

- (a) A prepayment made under Clause 8.2 (Disposal and Insurance Proceeds and Excess Cashflow) shall be applied in prepayment of Term Loans as contemplated in paragraphs (b) to (e) inclusive below.
- (b) Unless the Parent makes an election under paragraph (d) below, the Borrowers shall prepay Term Loans at the following times:
 - (i) in the case of any prepayment relating to the amounts of Disposal Proceeds or Insurance Proceeds, promptly upon receipt of those proceeds; and
 - (ii) in the case of any prepayment relating to an amount of Excess Cashflow, within 7 days of delivery pursuant to Clause 21.1 (Financial Statements) of the Annual Financial Statements for the relevant Financial Year.
- (c) A prepayment under Clause 8.2 (Disposal and Insurance Proceeds and Excess Cashflow) shall prepay the Term Loans as follows:
 - (i) firstly, in prepayment of the Facility B Loan;
 - secondly, once Facility B has been repaid in full, in reducing the Facility A Repayment Instalment for each Facility A Repayment Date falling after the date of prepayment in the manner contemplated by paragraph (d) of Clause 6.2 (Effect of cancellation and prepayment on scheduled repayments and reductions); and
 - (iii) thirdly, once Facility A has been repaid in full, in reducing the Facility C Repayment Instalment for each Facility C Repayment Instalment for each Facility C Repayment Date falling after the date of prepayment in the manner contemplated by paragraph (d) of Clause 6.2 (Effect of cancellation and prepayment on scheduled repayments and reductions).
- (d) Subject to paragraph (e) below, the Parent may elect that any prepayment under Clause 8.2 (Disposal and Insurance Proceeds and Excess Cashflow) be applied in prepayment of a Term Loan on the last day of the Interest Period relating to that Term Loan. If the Parent makes that election then a proportion of the Term Loan equal to the amount of the relevant prepayment will be due and payable on the last day of its Interest Period.
- (e) If the Parent has made an election under paragraph (d) above but a Default has occurred and is continuing, that election shall no longer apply and a proportion of the Term Loan

in respect of which the election was made equal to the amount of the relevant prepayment shall be immediately due and payable (unless the Majority Lenders otherwise agree in writing).

8.4. Mandatory Prepayment Accounts and Holding Accounts

- (a) The Parent shall ensure that:
 - (i) Disposal Proceeds and Insurance Proceeds in respect of which the Parent has made an election under paragraph (d) of Clause 8.3 (Application of mandatory prepayments) are paid into a Mandatory Prepayment Account as soon as reasonably practicable after receipt by a member of the Group;
 - (ii) any amounts which represent Excluded Insurance Proceeds and/or Excluded Disposal Proceeds which are to be applied for the specific purpose with the specific periods (as set out in the definitions of Excluded Insurance Proceeds and/or Excluded Disposal Proceeds) are paid into a Holding Account as soon as reasonably practicable after receipt by a member of the Group; and
 - (iii) an amount equal to any Excess Cashflow in respect of which the Parent has made an election under paragraph (d) of Clause 8.3 (Application of mandatory prepayments) is paid into a Mandatory Prepayment Account promptly after such election.
- (b) The Parent and each Borrower irrevocably authorise the Agent to apply:
 - (i) amounts credited to the Mandatory Prepayment Account; and
 - (ii) amounts credited to the Holding Account which have not been applied within any applicable periods detailed in the definitions of Excluded Insurance Proceeds and/or Excluded Disposal Proceeds,

to pay amounts due and payable under Clause 8.3 (Application of mandatory prepayments) and otherwise under the Finance Documents. The Parent and each Borrower further irrevocably authorise the Agent to so apply amounts credited to the Holding Account whether or not the periods detailed in the definitions of Excluded Insurance Proceeds and/or Excluded Disposal Proceeds have elapsed since receipt of those proceeds if a Default has occurred and is continuing. The Parent and each Borrower also irrevocably authorise the Agent to transfer any amounts credited to the Holding Account referred to in this paragraph (b) to the Mandatory Prepayment Account pending payment of amounts due and payable under the Finance Documents (but if all such amounts have been paid any such amounts remaining credited to the Mandatory Prepayment Account may (unless a Default has occurred) be transferred back to the Holding Account.

(c) A Lender, Security Trustee or Agent with which a Mandatory Prepayment Account or Holding Account is held acknowledges and agrees that (i) interest shall accrue at normal commercial rates on amounts credited to those accounts and that the account holder shall be entitled to receive such interest (which shall be paid in accordance with the mandate relating to such account) unless a Default is continuing and (ii) each such account is subject to the Transaction Security.

8.5. Excluded proceeds

Where Excluded Disposal Proceeds and Excluded Insurance Proceeds include amounts which are intended to be used for a specific purpose within a specified period (as set out in the relevant definition of Excluded Disposal Proceeds or Excluded Insurance Proceeds), the Parent shall

50

ensure that those amounts are used for that purpose and, if requested to do so by the Agent, shall promptly deliver a certificate to the Agent at the time of such application and at the end of such period confirming the amount (if any) which has been so applied within the requisite time periods provided for in the relevant definition.

9. **RESTRICTIONS**

9.1. Notices of Cancellation or Prepayment

Any notice of cancellation, prepayment, authorisation or other election given by any Party under Clause 7 (Illegality, voluntary prepayment and cancellation), paragraph (d) of Clause 8.3 (Application of mandatory prepayments) or Clause 8.4 (Mandatory prepayment Accounts and Holding Accounts) shall (subject to the terms of those Clauses) be irrevocable and, unless a contrary indication appears in this Agreement, any such notice shall specify the date or dates upon which the relevant cancellation or prepayment is to be made and the amount of that cancellation or prepayment.

9.2. Interest and other amounts

- (a) Any prepayment under this Agreement shall be made together with accrued interest on the amount prepaid and Break Costs.
- (b) Any prepayment or cancellation under this Agreement (except for any prepayment pursuant to any refinancing of the entire Commitments in respect of the Facilities in accordance with the terms of Clause 7.1 (Illegality) or as a result of the terms of paragraph (iv) of Clause 8.1 (Exit) or any prepayment as a result of the terms of Clause 8.2 (Disposal and Insurance proceeds and Excess Cashflow) or as a consequence of an Obligor being required to pay additional amounts pursuant to Clause 14.2 (Tax Gross-Up) where the prepayment is made within 30 days of such additional amounts becoming payable) shall be made together with an early repayment or cancellation fee payable by the Parent to the Agent (for the account of the Lenders) of an amount equal to (i) three per cent. (3%) of the amount prepaid and/or cancelled in the event that any amount is prepaid or cancelled within 12 months of the date of this Agreement as a result of or in connection with a refinancing of all or part of the Facilities by any bank or financial institution (other than all of the Eacled more than 12 months after but within 24 months of the date of this Agreement as a result of the Facilities by any bank or financial institution, and (iii) one per cent. (1%) in the event that any amount is prepaid or cancelled more than 24 months after but within 36 months of the date of this Agreement as a result of or in connection with a refinancing of all or part of the Facilities by any bank or financial institution (other than all of the Lenders), a Trade Sale or a Flotation, amount is prepaid or cancelled more than 24 months after but within 36 months of the date of this Agreement as a result of or in connection with a refinancing of all or part of the Facilities by any bank or financial institution (other than all of the Lenders), a Trade Sale or a Flotation, and (iii) one per cent. (1%) in the event that any amount is prepaid or cancelled more than 24 months after but within 36 months of the date of this Agreement as a result of or in connection

9.3. No reborrowing of Term Facilities

No Borrower may reborrow any part of a Term Facility which is prepaid.

9.4. Prepayment in accordance with Agreement

No Borrower shall repay or prepay all or any part of the Utilisations or cancel all or any part of the Commitments except at the times and in the manner expressly provided for in this Agreement.

9.5. No reinstatement of Commitments

No amount of the Total Commitments cancelled under this Agreement may be subsequently reinstated.

9.6. Agent's receipt of Notices

If the Agent receives a notice under Clause 7 (Illegality, voluntary prepayment and cancellation) or an election under paragraph (d) of Clause 8.3 (Application of mandatory prepayments) it shall promptly forward a copy of that notice or election to either the Parent or the affected Lender, as appropriate.

9.7. Prepayment elections

The Agent shall notify the Lenders as soon as possible of any proposed prepayment of any Term Loan under Clause 7.3 (Voluntary prepayment of Term Loans) or Clause 8.2 (Disposal and Insurance Proceeds and Excess Cashflow).

SECTION 5

COSTS OF UTILISATION

10. INTEREST

10.1. Calculation of interest

The rate of interest on each Loan for each Interest Period is the percentage rate per annum which is the aggregate of the applicable:

- (a) Margin; and
- (b) LIBOR.

10.2. Payment of interest

The Borrower to which a Loan has been made shall pay accrued interest on that Loan on the last day of each Interest Period (and, if the Interest Period is longer than three Months, on the dates falling at three Monthly intervals after the first day of the Interest Period).

10.3. Default interest

- (a) If an Obligor fails to pay any amount payable by it under a Finance Document on its due date, interest shall accrue on the overdue amount from the due date up to the date of actual payment (both before and after judgment) at a rate which, subject to paragraph (b) below, is two times the rate which would have been payable if the overdue amount had, during the period of non-payment, constituted a Loan in the currency of the overdue amount for successive Interest Periods, each of a duration selected by the Agent (acting reasonably). Any interest accruing under this Clause 10.3 shall be immediately payable by the Obligor on demand by the Agent.
- (b) If any overdue amount consists of all or part of a Loan which became due on a day which was not the last day of an Interest Period relating to that Loan:
 - (i) the first Interest Period for that overdue amount shall have a duration equal to the unexpired portion of the current Interest Period relating to that Loan; and
 - (ii) the rate of interest applying to the overdue amount during that first Interest Period shall be two times the rate which would have applied if the overdue amount had not become due.
- (C) Default interest (if unpaid) arising on an overdue amount will be compounded with the overdue amount at the end of each Interest Period applicable to that overdue amount but will remain immediately due and payable.

10.4. Notification of rates of interest

The Agent shall promptly notify the Lenders and the relevant Borrower (or the Parent) of the determination of a rate of interest under this Agreement.

11. INTEREST PERIODS

11.1. Selection of Interest Periods and Terms

53

54

- (a) A Borrower (or the Parent on behalf of a Borrower) may select an Interest Period for a Loan in the Utilisation Request for that Loan or (if the Loan is a Term Loan and has already been borrowed) in a Selection Notice.
- (b) Each Selection Notice for a Term Loan is irrevocable and must be delivered to the Agent by the Borrower (or the Parent on behalf of the Borrower) to which that Term Loan was made not later than 10.00 a.m. on the Business Day prior to the last day of the then current Interest Period.
- (c) If a Borrower (or the Parent) fails to deliver a Selection Notice to the Agent in accordance with paragraph (b) above, the relevant Interest Period will, subject to Clause 11.2 (Changes to Interest Periods), be three Months.
- (d) Subject to this Clause 11, a Borrower (or the Parent) may select an Interest Period of three or six Months or any other period agreed between the Parent and the Agent (acting on the instructions of all the Lenders). In addition, a Borrower (or the Parent on its behalf) may select an Interest Period of (in relation to Facility A or Facility C) a period of less than three Months, if necessary to ensure that:
 - (i) there are Facility A Loans (with an aggregate amount equal to or greater than a Facility A Repayment Instalment) which have an Interest Period ending on a Facility A Repayment Date for the Borrowers to make the relevant Facility A Repayment Instalment due on that date; or
 - (ii) there are Facility C Loans (with an aggregate amount equal to or greater than a Facility C Repayment Instalment) which have an Interest Period ending on a Facility C Repayment Date for the Borrowers to make the relevant Facility C Repayment Instalment due on that date.
- (e) An Interest Period for a Loan shall not extend beyond the Termination Date applicable to its Facility.
- (f) Each Interest Period for a Term Loan shall start on the Utilisation Date or (if already made) on the last day of its preceding Interest Period.

11.2. Changes to Interest Periods

- (a) Prior to determining the interest rate for a Facility A Loan or a Facility C Loan, the Agent may shorten an Interest Period for any Facility A Loan or Facility C Loan (as applicable) to ensure there are sufficient Facility A Loans (with an aggregate amount equal to or greater than the relevant Facility A Repayment Instalment) or Facility C Loans (with an aggregate amount equal to or greater than the relevant Facility C Repayment Instalment) (as applicable) which have an Interest Period ending on a Facility A Repayment Date or Facility C Repayment Date (as applicable) for the Borrowers to make the relevant Facility A Repayment Instalment or Facility C Repayment Instalment due on that date.
- (b) If the Agent makes any of the changes to an Interest Period referred to in this Clause 11.2, it shall promptly notify the Parent and the Lenders.

11.3. Non-Business Days

If an Interest Period would otherwise end on a day which is not a Business Day, that Interest Period will instead end on the next Business Day in that calendar month (if there is one) or the preceding Business Day (if there is not).

11.4. Consolidation and division of Facility A Loans

- (a) Subject to paragraph (b) below, if two or more Interest Periods:
 - (i) relate to Facility A Loans or Facility C Loans;
 - (ii) end on the same date; and
 - (iii) are made to the same Borrower,

those Facility A Loans or Facility C Loans (as applicable) will, unless the Parent specifies to the contrary in the Selection Notice for the next Interest Period, be consolidated into, and treated as, a single Facility A Loan or Facility C Loan (as applicable) on the last day of the Interest Period.

(b) Subject to Clause 4.3 (Maximum number of Utilisations) and Clause 5.3 (Currency and amount), if the Parent requests in a Selection Notice that a Facility A Loan or Facility C Loan be divided into two or more Facility A Loans or Facility C Loans (as applicable), that Facility A Loan or Facility C Loan (as applicable) will, on the last day of its Interest Period, be so divided with amounts specified in that Selection Notice, having an aggregate amount equal to the amount of the Facility A Loan or Facility C Loan (as applicable) immediately before its division.

12. CHANGES TO THE CALCULATION OF INTEREST

12.1. Absence of quotations

Subject to Clause 12.2 (Market disruption), if LIBOR is to be determined by reference to the Reference Banks but a Reference Bank does not supply a quotation by 11.00 a.m. on the Quotation Day, the applicable LIBOR shall be determined on the basis of the quotations of the remaining Reference Banks.

12.2. Market disruption

- (a) If a Market Disruption Event occurs in relation to a Loan for any Interest Period, then the rate of interest on each Lender's share of that Loan for the Interest Period shall be the percentage rate per annum which is the sum of:
 - (i) the Margin; and
 - (ii) the rate notified to the Agent by that Lender as soon as practicable and in any event before interest is due to be paid in respect of that Interest Period, to be that which expresses as a percentage rate per annum the cost to that Lender of funding its participation in that Loan from whatever source it may reasonably select.
- (b) In this Agreement "**Market Disruption Event**" means:
 - (i) at or about noon on the Quotation Day for the relevant Interest Period the Screen Rate is not available and none or only one of the Reference Banks supplies a rate to the Agent to determine LIBOR; or
 - (ii) before close of business in London on the Quotation Day for the relevant Interest Period, the Agent receives notifications from a Lender or Lenders (whose participations in a Loan exceed twenty five (25%) per cent. of that Loan) that the cost to it of obtaining matching deposits in the Relevant Interbank Market would be in excess of LIBOR.

12.3. Alternative basis of interest or funding

- (a) If a Market Disruption Event occurs and the Agent or the Parent so requires, the Agent and the Parent shall enter into negotiations (for a period of not more than thirty days) with a view to agreeing a substitute basis for determining the rate of interest.
- (b) Any alternative basis agreed pursuant to paragraph (a) above shall, with the prior consent of all the Lenders and the Parent, be binding on all Parties.

12.4. Break Costs

- (a) Each Borrower shall, within three Business Days of demand by a Finance Party, pay to that Finance Party its Break Costs attributable to all or any part of a Loan or Unpaid Sum being paid by that Borrower on a day other than the last day of an Interest Period for that Loan or Unpaid Sum.
- (b) Each Lender shall, as soon as reasonably practicable after a demand by the Agent, provide a certificate confirming the amount of its Break Costs for any Interest Period in which they accrue.

13. FEES

13.1. Arrangement fee

The Parent shall pay to the Arranger an arrangement fee in the amount and at the times agreed in the Fee Letter.

13.2. Agency fee

The Parent shall pay to the Agent (for its own account) an agency fee in the amount and at the times agreed in the Fee Letter.

SECTION 6

ADDITIONAL PAYMENT OBLIGATIONS

14. TAX GROSS UP AND INDEMNITIES

14.1. Definitions

(a) In this Agreement:

"**Protected Party**" means a Finance Party which is or will be subject to any liability or required to make any payment for or on account of Tax in relation to a sum received or receivable (or any sum deemed for the purposes of Tax to be received or receivable) under a Finance Document.

"Qualifying Lender" means:

- (i) a Lender (other than a Lender within sub-paragraph (ii) below) which is beneficially entitled to interest payable to that Lender in respect of an advance under a Finance Document and is:
 - (A) a Lender:
 - (1) which is a bank (as defined for the purpose of Section 879 of the ITA) making an advance under a Finance Document and is within the charge to United Kingdom corporation tax as respects any payment of interest made in respect of that advance or would be within such charge as respects such payment apart from Section 18A of the CTA; or
 - (2) in respect of an advance made under a Finance Document by a person that was a bank (as defined for the purpose of Section 879 of the ITA) at the time that that advance was made and which is within the charge to United Kingdom corporation tax as respects any payments of interest made in respect of that advance; or
 - (B) a Lender which is:
 - (1) a company resident in the United Kingdom for United Kingdom tax purposes;
 - (1) a partnership each member of which is:
 - (a) a company so resident in the United Kingdom; or
 - (b) a company not so resident in the United Kingdom which carries on a trade in the United Kingdom through a permanent establishment and which brings into account in computing its chargeable profits (within the meaning of Section 19 of the CTA) the whole of any share of interest payable in respect of that advance that falls to it by reason of the CTA;

- (2) a company not so resident in the United Kingdom which carries on a trade in the United Kingdom through a permanent establishment and which brings into account interest payable in respect of that advance in computing the chargeable profits (within the meaning of Section 19 of the CTA) of that company; or
- (C) a Treaty Lender; or
- (ii) a Lender which is a building society (as defined for the purposes of section 880 of the ITA) making an advance under a Finance Document.

"**Tax Confirmation**" means a confirmation by a Lender that the person beneficially entitled to interest payable to that Lender in respect of an advance under a Finance Document is either:

- (i) a company resident in the United Kingdom for United Kingdom tax purposes;
- (ii) a partnership each member of which is:
 - (1) a company so resident in the United Kingdom; or
 - (2) a company not so resident in the United Kingdom which carries on a trade in the United Kingdom through a permanent establishment and which brings into account in computing its chargeable profits (within the meaning of Section 19 of the CTA) the whole of any share of interest payable in respect of that advance that falls to it by reason of Part 17 of the CTA; or
- (iii) a company not so resident in the United Kingdom which carries on a trade in the United Kingdom through a permanent establishment and which brings into account interest payable in respect of that advance in computing the chargeable profits (within the meaning of section 19 of the CTA) of that company.

"Tax Credit" means a credit against, relief or remission for, or repayment of, any Tax.

"Tax Deduction" means a deduction or withholding for or on account of Tax from a payment under a Finance Document.

"**Tax Payment**" means either the increase in a payment made by an Obligor to a Finance Party under Clause 14.2 (Tax gross-up) or a payment under Clause 14.3 (Tax indemnity).

"Treaty Lender" means a Lender which:

- (i) is treated as a resident of a Treaty State for the purposes of the Treaty;
- (ii) does not carry on a business in the United Kingdom through a permanent establishment with which that Lender's participation in the Loan is effectively connected; and
- (iii) fulfils any other conditions which must be fulfilled under the double taxation agreement by residents of that Treaty State for such residents to obtain exemptions from United Kingdom taxation on interest, subject to the completion of procedural formalities.

58

"**Treaty State**" means a jurisdiction having a double taxation agreement (a "**Treaty**") with the United Kingdom which makes provision for full exemption from tax imposed by the United Kingdom on interest.

(a) Unless a contrary indication appears, in this Clause 14 a reference to "**determines**" or "**determined**" means a determination made in the absolute discretion of the person making the determination.

14.2. Tax gross-up

- (a) Each Obligor shall make all payments to be made by it without any Tax Deduction, unless a Tax Deduction is required by law.
- (b) The Parent shall promptly upon becoming aware that an Obligor must make a Tax Deduction (or that there is any change in the rate or the basis of a Tax Deduction) notify the Agent accordingly. Similarly, a Lender shall notify the Agent on becoming so aware in respect of a payment payable to that Lender. If the Agent receives such notification from a Lender it shall notify the Parent and that Obligor.
- (C) If a Tax Deduction is required by law to be made by an Obligor, the amount of the payment due from that Obligor shall be increased to an amount which (after making any Tax Deduction) leaves an amount equal to the payment which would have been due if no Tax Deduction had been required.
- (d) A payment shall not be increased under paragraph (c) above by reason of a Tax Deduction on account of Tax imposed by the United Kingdom, if on the date on which the payment falls due:
 - (i) the payment could have been made to the relevant Lender without a Tax Deduction if the Lender had been a Qualifying Lender, but on that date that Lender is not or has ceased to be a Qualifying Lender other than as a result of any change after the date it became a Lender under this Agreement in (or in the interpretation, administration, or application of) any law or Treaty, or any published practice or published concession of any relevant taxing authority; or
 - (ii)
- (A) the relevant Lender is a Qualifying Lender solely by virtue of paragraph (i)(B) of the definition of Qualifying Lender;
- (B) an officer of H.M. Revenue & Customs has given (and not revoked) a direction (a "Direction") under section 931 of the ITA (as that provision has effect on the date on which the relevant Lender became a Party) which relates to the payment and that Lender has received from that Obligor or the Parent a certified copy of that Direction; and
- (C) the payment could have been made to the Lender without any Tax Deduction if that Direction had not been made; or
- (iii) the relevant Lender is a Qualifying Lender solely by virtue of paragraph (i)(B) of the definition of Qualifying Lender and:
 - (A) the relevant Lender has not given a Tax Confirmation to the Parent; and

- (B) the payment could have been made to the Lender without any Tax Deduction if the Lender had given a Tax Confirmation to the Parent, on the basis that the Tax Confirmation would have enabled the Parent to have formed a reasonable belief that the payment was an "excepted payment" for the purpose of section 930 of the ITA; or
- (iv) the relevant Lender is a Treaty Lender and the Obligor making the payment is able to demonstrate that the payment could have been made to the Lender without the Tax Deduction had that Lender complied with its obligations under paragraph (g) below.
- (e) If an Obligor is required to make a Tax Deduction, that Obligor shall make that Tax Deduction and any payment required in connection with that Tax Deduction within the time allowed and in the minimum amount required by law.
- (f) Within thirty days of making either a Tax Deduction or any payment required in connection with that Tax Deduction, the Obligor making that Tax Deduction shall deliver to the Agent for the Finance Party entitled to the payment statement under Section 975 of ITA or other evidence reasonably satisfactory to that Finance Party that the Tax Deduction has been made or (as applicable) any appropriate payment paid to the relevant taxing authority.
- (g) (i) Subject to paragraph (ii) below, a Treaty Lender and each Obligor which makes a payment to which that Treaty Lender is entitled shall co-operate in completing any procedural formalities necessary for that Obligor to obtain authorisation to make that payment without a Tax Deduction.
 - (ii) Nothing in paragraph (i) above shall require a Treaty Lender to:
 - (A) register under the HMRC DT Treaty Passport scheme;
 - (B) apply the HMRC DT Treaty Passport scheme to any Utilisation if it has so registered; or
 - (C) file Treaty forms if it has included an indication to the effect that it wishes the HMRC DT Treaty Passport scheme to apply to this Agreement in accordance with paragraph (h) below or paragraph (a) of Clause 14.6 (HMRC DT Treaty Passport scheme confirmation),

and the Obligor making that payment has not complied with its obligations under paragraph (i) below or paragraph (b) of Clause 14.6 (HMRC DT Treaty Passport scheme confirmation).

14.3. Tax indemnity

- (a) The Parent shall (within three Business Days of demand by the Agent) pay to a Protected Party an amount equal to the loss, liability or cost which that Protected Party determines will be or has been (directly or indirectly) suffered for or on account of Tax by that Protected Party in respect of a Finance Document.
- (b) Paragraph (a) above shall not apply:
 - (i) with respect to any Tax assessed on a Finance Party:
 - (A) under the law of the jurisdiction in which that Finance Party is incorporated or, if different, the jurisdiction (or jurisdictions) in which that Finance Party is treated as resident for tax purposes; or

- 61
- (B) under the law of the jurisdiction in which that Finance Party's Facility Office is located in respect of amounts received or receivable in that jurisdiction,

if that Tax is imposed on or calculated by reference to the net income received or receivable (but not any sum deemed to be received or receivable) by that Finance Party; or

- (ii) to the extent a loss, liability or cost:
 - (A) is compensated for by an increased payment under Clause 14.2 (Tax gross-up); or
 - (B) would have been compensated for by an increased payment under Clause 14.2 (Tax gross-up) but was not so compensated solely because one of the exclusions in paragraph (d) of Clause 14.2 (Tax gross-up) applied.
- (c) A Protected Party making, or intending to make a claim under paragraph (a) above shall promptly notify the Agent of the event which will give, or has given, rise to the claim, following which the Agent shall notify the Parent.
- (d) A Protected Party shall, on receiving a payment from an Obligor under this Clause 14.3, notify the Agent.

14.4. Tax Credit

If an Obligor makes a Tax Payment and the relevant Finance Party determines that:

- (a) a Tax Credit is attributable either to an increased payment of which that Tax Payment forms part or to that Tax Payment; and
- (b) that Finance Party has obtained, utilised and retained that Tax Credit,

the Finance Party shall pay an amount to the Obligor which that Finance Party determines will leave it (after that payment) in the same after-Tax position as it would have been in had the Tax Payment not been required to be made by the Obligor. Notwithstanding anything to the contrary, in no event shall any Finance Party be required to pay any amount to the Borrowers the payment of which would place such Finance Party in a less favourable net after tax position than such Finance Party would have been in if the Tax giving rise to the indemnity payments had never been paid.

14.5. Lender Status Confirmation

Each Lender which becomes a Party to this Agreement after the date of this Agreement shall indicate, in the Transfer Certificate or Assignment Agreement or Increase Confirmation which it executes on becoming a Party, and for the benefit of the Agent and without liability to any Obligor, which of the following categories it falls in:

- (a) not a Qualifying Lender;
- (b) a Qualifying Lender (other than a Treaty Lender); or
- (c) a Treaty Lender.

If a New Lender fails to indicate its status in accordance with this Clause 14.5 then such New Lender shall be treated for the purposes of this Agreement as if it is not a Qualifying Lender until such time as it notifies the Agent which category applies (and the Agent, upon receipt of such

notification, shall inform the Company). For the avoidance of doubt, a Transfer Certificate, Assignment Agreement or Increase Confirmation shall not be invalidated by any failure of a Lender to comply with this Clause 14.5.

14.6. HMRC DT Treaty Passport scheme confirmation

- (a) A New Lender or an Increase Lender that is a Treaty Lender that holds a passport under the HMRC DT Treaty Passport scheme, and which then wishes that scheme to apply to this Agreement, shall include an indication to that effect (for the benefit of the Agent and without liability to any Obligor) in the Transfer Certificate, Assignment Agreement or Increase Confirmation which it executes by including its scheme reference number and its jurisdiction of tax residence in that Transfer Certificate, Assignment Agreement or Increase Confirmation.
- (b) Where a New Lender or an Increase Lender includes the indication described in paragraph (a) above in the relevant Transfer Certificate, Assignment Agreement or Increase Confirmation:
 - (i) each Borrower which is a Party as a Borrower as at the relevant Transfer Date or the date on which the increase in Total Commitments described in the relevant Increase Confirmation takes effect shall, and to the extent that that New Lender or Increase Lender becomes a Lender under a Facility which is made available to that Borrower pursuant to Clause 2 (The Facilities), file a duly completed form DTTP2 in respect of such Lender with HM Revenue & Customs within 30 days of that Transfer Date or that date on which the increase in Total Commitments takes effect and shall promptly provide the Lender with a copy of that filing; and
 - (ii) each Additional Borrower which becomes an Additional Borrower after the relevant Transfer Date or the date on which the increase in Total Commitments described in the relevant Increase Confirmation takes effect shall, to the extent that that New Lender or Increase Lender is a Lender under a Facility which is made available to that Additional Borrower pursuant to Clause 2 (The Facilities), file a duly completed form DTTP2 in respect of such Lender with HM Revenue & Customs within 30 days of becoming an Additional Borrower and shall promptly provide the Lender with a copy of that filing.

14.7. Stamp taxes

The Parent shall pay and, within three Business Days of demand, indemnify each Secured Party and Arranger against any cost, loss or liability that Secured Party or Arranger incurs in relation to all stamp duty, registration and other similar Taxes payable in respect of any Finance Document.

14.8. Notification to Parent and Agent

Each Lender will notify the Parent and the Agent if it is not or ceases to be a Qualifying Lender.

14.9. Value added tax

(a) All consideration expressed to be payable under a Finance Document by any Party to a Finance Party which (in whole or in part) constitute the consideration for any supply for VAT purposes are deemed to be exclusive of any VAT which is charged on that supply, and accordingly, subject to paragraph (b) below, if VAT is or becomes chargeable on any supply made by any Finance Party to any Party under a Finance Document and such Finance Party is required to account to the relevant tax authority for the VAT, that Party must pay to such Finance Party (in addition to and at the same time as paying any other consideration for such supply) an amount equal to the amount of the VAT.

- (b) If VAT is or becomes chargeable on any supply made by any Finance Party (the "Supplier") to any other Finance Party (the "Recipient") under a Finance Document, and any Party other than the Recipient (the "Relevant Party") is required by the terms of any Finance Document to pay an amount equal to the consideration for that supply to the Supplier (rather than being required to reimburse or indemnify the Recipient in respect of that consideration):
 - (i) (where the Supplier is the person required to account to the relevant tax authority for the VAT) the Relevant Party must also pay to the Supplier (at the same time as paying that amount) an additional amount equal to the amount of the VAT. The Recipient must (where this paragraph (i) applies) promptly pay to the Relevant Party an amount equal to any credit or repayment the Recipient receives from the relevant tax authority which the Recipient reasonably determines relates to the VAT chargeable on that supply; and
- (ii) (where the Recipient is the person required to account to the relevant tax authority for the VAT) the Relevant Party must promptly, following demand from the Recipient, pay to the Recipient an amount equal to the VAT chargeable on that supply but only to the extent that the Recipient reasonably determines that it is not entitled to credit or repayment from the relevant tax authority in respect of that VAT.
- (c) Where a Finance Document requires any Party to reimburse a Finance Party for any costs or expense, that Party shall reimburse or indemnify (as the case may be) such Finance Party for the full amount of such cost or expense, including such part thereof as represents VAT, save to the extent that such Finance Party reasonably determines that it is entitled to credit or repayment from the relevant tax authority in respect of the VAT.
- (d) Any reference in this Clause 14.9 to any Party shall, at any time when such Party is treated as a member of a group for VAT purposes, include (where appropriate and unless the context otherwise requires) a reference to the representative member of such group at such time (the term "representative member" to have the same meaning as in the Value Added Tax Act 1994 or in the case of Ireland, the group member notified by the Revenue Commissioners in accordance with section 15(1)(a) of the Value Added Tax Consolidation Act 2010 of Ireland as being the member responsible for complying with the provisions of that act in respect of the group).
- (e) In relation to any supply made by a Finance Party to any Party under a Finance Document, if reasonably requested by such Finance Party, that Party must promptly provide such Finance Party with details of that Party's VAT registration and such other information as is reasonably requested in connection with such Finance Party's VAT reporting requirements in relation to such supply.

15. INCREASED COSTS

15.1. Increased costs

- (a) Subject to Clause 15.3 (Exceptions) the Parent shall, within three Business Days of a demand by the Agent, pay for the account of a Finance Party the amount of any Increased Costs incurred by that Finance Party or any of its Affiliates as a result of (i) the introduction of or any change in (or in the interpretation, administration or application of) any law or regulation or (ii) compliance with any law or regulation made after the date of this Agreement.
- (b) In this Agreement "**Increased Costs**" means:

- 64
- (i) a reduction in the rate of return from a Facility or on a Finance Party's (or its Affiliate's) overall capital;
- (ii) an additional or increased cost; or
- (iii) a reduction of any amount due and payable under any Finance Document,

which is incurred or suffered by a Finance Party or any of its Affiliates to the extent that it is attributable to that Finance Party having entered into its Commitment or funding or performing its obligations under any Finance Document.

15.2. Increased cost claims

- (a) A Finance Party intending to make a claim pursuant to Clause 15.1 (Increased Costs) shall notify the Agent of the event giving rise to the claim, following which the Agent shall promptly notify the Parent.
- (b) Each Finance Party shall, as soon as practicable after a demand by the Agent, provide a certificate confirming the amount of its Increased Costs.

15.3. Exceptions

- (a) Clause 15.1 (Increased Costs) does not apply to the extent any Increased Cost is:
 - (i) attributable to a Tax Deduction required by law to be made by an Obligor;
 - (ii) compensated for by Clause 14.3 (Tax indemnity) (or would have been compensated for under Clause 14.3 (Tax indemnity) but was not so compensated solely because any of the exclusions in paragraph (b) of Clause 14.3 (Tax indemnity) applied); or
 - (iii) attributable to the wilful breach by the relevant Finance Party or its Affiliates of any law or regulation.
- (b) In this Clause 15.3 reference to a "**Tax Deduction**" has the same meaning given to the term in Clause 14.1 (Definitions).

16. OTHER INDEMNITIES

16.1. Currency indemnity

- (a) If any sum due from an Obligor under the Finance Documents (a "Sum"), or any order, judgment or award given or made in relation to a Sum, has to be converted from the currency (the "First Currency") in which that Sum is payable into another currency (the "Second Currency") for the purpose of:
 - (i) making or filing a claim or proof against that Obligor; or
 - (ii) obtaining or enforcing an order, judgment or award in relation to any litigation or arbitration proceedings,

that Obligor shall as an independent obligation, within three Business Days of demand, indemnify the Arranger and each other Secured Party to whom that Sum is due against any cost, loss or liability arising out of or as a result of the conversion including any discrepancy between (A) the rate of exchange used to convert that Sum from the First Currency into the Second Currency and (B) the rate or rates of exchange available to that person at the time of its receipt of that Sum.

(b) Each Obligor waives any right it may have in any jurisdiction to pay any amount under the Finance Documents in a currency or currency unit other than that in which it is expressed to be payable.

16.2. Other indemnities

- (a) The Parent shall (or shall procure that an Obligor will), within three Business Days of demand, indemnify the Arranger and each other Secured Party against any cost, loss or liability incurred by it as a result of:
 - (i) the occurrence of any Event of Default;
 - (ii) a failure by an Obligor to pay any amount due under a Finance Document on its due date, including without limitation, any cost, loss or liability arising as a result of Clause 30 (Sharing among the Finance Parties);
 - (iii) funding, or making arrangements to fund, its participation in a Utilisation requested by a Borrower in a Utilisation Request but not made by reason of the operation of any one or more of the provisions of this Agreement (other than by reason of default or negligence by that Finance Party alone);
 - (iv) a Utilisation (or part of a Utilisation) not being prepaid in accordance with a notice of prepayment given by a Borrower or the Parent.
- (b) The Parent shall promptly indemnify each Finance Party, each Affiliate of a Finance Party and each officer or employee of a Finance Party or its Affiliate, against any cost, loss or liability incurred by that Finance Party or its Affiliate (or officer or employee of that Finance Party or Affiliate) in connection with or arising out of the Acquisition or the funding of the Acquisition (including but not limited to those incurred in connection with any litigation, arbitration or administrative proceedings or regulatory enquiry concerning the Acquisition), unless such loss or liability is caused by the gross negligence or wilful misconduct of that Finance Party or its Affiliate (or employee or officer of that Finance Party or Affiliate). Any Affiliate or any officer or employee of a Finance Party or its Affiliate may rely on this Clause 16.2.

16.3. Indemnity to the Agent

The Parent shall promptly indemnify the Agent against:

- (a) any cost, loss or liability incurred by the Agent (acting reasonably) as a result of:
 - (i) investigating any event which it reasonably believes is a Default;
 - (ii) entering into or performing any foreign exchange contract for the purposes of paragraph (b) of Clause 31.10 (Change of currency);
 - (iii) acting or relying on any notice, request or instruction which it reasonably believes to be genuine, correct and appropriately authorised; or
 - (iv) instructing lawyers, accountants, tax advisers, surveyors or other professional advisers or experts as permitted under this Agreement; and

(b) any cost, loss or liability incurred by the Agent (otherwise than by reason of the Agent's gross negligence or wilful misconduct) in acting as Agent under the Finance Documents.

16.4. Indemnity to the Security Trustee

- (a) Each Obligor shall promptly indemnify the Security Trustee and every Receiver and Delegate against any cost, loss or liability incurred by any of them as a result of:
 - (i) any failure by the Parent to comply with its obligations under Clause 18 (Costs and Expenses);
 - (ii) acting or relying on any notice, request or instruction which it reasonably believes to be genuine, correct and appropriately authorised;
 - (iii) the taking, holding, protection or enforcement of the Transaction Security,
 - (iv) the exercise of any of the rights, powers, discretions and remedies vested in the Security Trustee and each Receiver and Delegate by the Finance Documents or by law;
 - (v) any default by any Obligor in the performance of any of the obligations expressed to be assumed by it in the Finance Documents; or
 - (vi) acting as Security Trustee, Receiver or Delegate under the Finance Documents or which otherwise relates to any of the Charged Property (otherwise, in each case, than by reason of the relevant Security Trustee's, Receiver's or Delegate's gross negligence or wilful misconduct);
- (b) The Security Trustee may, in priority to any payment to the Secured Parties, indemnify itself out of the Charged Property in respect of, and pay and retain, all sums necessary to give effect to the indemnity in this Clause 16.4 and shall have a lien on the Transaction Security and the proceeds of the enforcement of the Transaction Security for all monies payable to it.

17. MITIGATION BY THE LENDERS

17.1. Mitigation

- (a) Each Finance Party shall, in consultation with the Parent, take all reasonable steps to mitigate any circumstances which arise and which would result in any amount becoming payable under or pursuant to, or cancelled pursuant to, any of Clause 7.1 (Illegality), Clause 14 (Tax gross-up and indemnities) or Clause 15 (Increased Costs) including (but not limited to) transferring its rights and obligations under the Finance Documents to another Affiliate or Facility Office.
- (b) Paragraph (a) above does not in any way limit the obligations of any Obligor under the Finance Documents.

17.2. Limitation of liability

(a) The Parent shall indemnify each Finance Party for all costs and expenses reasonably incurred by that Finance Party as a result of steps taken by it under Clause 17.1 (Mitigation).

(b) A Finance Party is not obliged to take any steps under Clause 17.1 (Mitigation) if, in the opinion of that Finance Party (acting reasonably), to do so might be prejudicial to it.

18. COSTS AND EXPENSES

18.1. Transaction expenses

The Parent shall within three Business Days of demand pay the Agent, the Arranger and the Security Trustee the amount of all costs and expenses (including legal fees) reasonably incurred by any of them (and, in the case of the Security Trustee, by any Receiver or Delegate) in connection with the negotiation, preparation, printing, execution, syndication and perfection of:

- (a) this Agreement and any other documents referred to in this Agreement and the Transaction Security;
- (b) any other Finance Documents executed after the date of this Agreement.

18.2. Amendment costs

If (a) an Obligor requests an amendment, waiver or consent or (b) an amendment is required pursuant to Clause 31.10 (Change of currency), the Parent shall, within three Business Days of demand, reimburse each of the Agent and the Security Trustee for the amount of all costs and expenses (including legal fees) reasonably incurred by the Agent and the Security Trustee (and, in the case of the Security Trustee, by any Receiver or Delegate) in responding to, evaluating, negotiating or complying with that request or requirement.

18.3. Security Trustee's ongoing costs

- (a) In the event of (i) a Default or (ii) the Security Trustee considering it necessary or expedient or (iii) the Security Trustee being requested by an Obligor or the Majority Lenders to undertake duties which the Security Trustee and the Parent agree to be of an exceptional nature and/or outside the scope of the normal duties of the Security Trustee under the Finance Documents, the Parent shall pay to the Security Trustee any additional remuneration that may be agreed between them.
- (b) If the Security Trustee and the Parent fail to agree upon the nature of the duties or upon any additional remuneration, that dispute shall be determined by an investment bank (acting as an expert and not as an arbitrator) selected by the Security Trustee and approved by the Parent or, failing approval, nominated (on the application of the Security Trustee) by the President for the time being of the Law Society of England and Wales (the costs of the nomination and of the investment bank being payable by the Parent) and the determination of any investment bank shall be final and binding upon the parties to this Agreement.

18.4. Enforcement and preservation costs

The Parent shall, within three Business Days of demand, pay to the Arranger and each other Secured Party the amount of all costs and expenses (including legal fees) incurred by it in connection with the enforcement of or the preservation of any rights under any Finance Document and the Transaction Security and any proceedings instituted by or against the Security Trustee as a consequence of taking or holding the Transaction Security or enforcing these rights.

GUARANTEE

19. GUARANTEE AND INDEMNITY

19.1. Guarantee and indemnity

Each Guarantor irrevocably and unconditionally jointly and severally:

- (a) guarantees to each Finance Party punctual performance by each other Obligor of all that Obligor's obligations under the Finance Documents;
- (b) undertakes with each Finance Party that whenever another Obligor does not pay any amount when due under or in connection with any Finance Document, that Guarantor shall immediately on demand pay that amount as if it was the principal obligor; and
- (c) agrees with each Finance Party that if any obligation guaranteed by it is or becomes unenforceable, invalid or illegal, it will, as an independent and primary obligation, indemnify that Finance Party immediately on demand against any cost, loss or liability it incurs as a result of an Obligor not paying any amount which would, but for such unenforceability, invalidity or illegality, have been payable by it under any Finance Document on the date when it would have been due. The amount payable by a Guarantor under this indemnity will not exceed the amount it would have had to pay under this Clause 19 if the amount claimed had been recoverable on this basis of a guarantee.

19.2. Continuing Guarantee

This guarantee is a continuing guarantee and will extend to the ultimate balance of sums payable by any Obligor under the Finance Documents, regardless of any intermediate payment or discharge in whole or in part.

19.3. Reinstatement

If any discharge, release or arrangement (whether in respect of the obligations of any Obligor or any security for those obligations or otherwise) is made by a Finance Party in whole or in part on the basis of any payment, security or other disposition which is avoided or must be restored in insolvency, liquidation, administration or otherwise, without limitation, then the liability of each Guarantor under this Clause 19 will continue or be reinstated as if the discharge, release or arrangement had not occurred.

19.4. Waiver of defences

The obligations of each Guarantor under this Clause 19 will not be affected by an act, omission, matter or thing which, but for this Clause 19, would reduce, release or prejudice any of its obligations under this Clause 19 (without limitation and whether or not known to it or any Finance Party) including:

- (a) any time, waiver or consent granted to, or composition with, any Obligor or other person;
- (b) the release of any other Obligor or any other person under the terms of any composition or arrangement with any creditor of any member of the Group;
- (C) the taking, variation, compromise, exchange, renewal or release of, or refusal or neglect to perfect, take up or enforce, any rights against, or security over assets of, any Obligor or other person or any non-presentation or non-observance of any formality or other

- (d) any incapacity or lack of power, authority or legal personality of or dissolution or change in the members or status of an Obligor or any other person;
- (e) any amendment, novation, supplement, extension (whether of maturity or otherwise) or restatement (in each case, however fundamental and of whatsoever nature) or replacement of a Finance Document or any other document or security;
- (f) any unenforceability, illegality or invalidity of any obligation of any person under any Finance Document or any other document or security; or
- (g) any insolvency or similar proceedings.

19.5. Guarantor Intent

Without prejudice to the generality of Clause 19.4 (Waiver of Defences), each Guarantor expressly confirms that it intends that this guarantee shall extend from time to time to any (however fundamental) variation, increase, extension or addition of or to any of the Finance Documents and/or any facility or amount made available under any of the Finance Documents for the purposes of or in connection with any of the following: acquisitions of any nature; increasing working capital; enabling investor distributions to be made; carrying out restructurings; refinancing existing facilities; refinancing any other indebtedness; making facilities available to new borrowers; any other variation or extension of the purposes for which any such facility or amount might be made available from time to time; and any fees, costs and/or expenses associated with any of the foregoing.

19.6. Immediate recourse

Each Guarantor waives any right it may have of first requiring any Finance Party (or any trustee or agent on its behalf) to proceed against or enforce any other rights or security or claim payment from any person before claiming from that Guarantor under this Clause 19. This waiver applies irrespective of any law or any provision of a Finance Document to the contrary.

19.7. Appropriations

Until all amounts which may be or become payable by the Obligors under or in connection with the Finance Documents have been irrevocably paid in full, each Finance Party (or any trustee or agent on its behalf) may:

- (a) refrain from applying or enforcing any other moneys, security or rights held or received by that Finance Party (or any trustee or agent on its behalf) in respect of those amounts, or apply and enforce the same in such manner and order as it sees fit (whether against those amounts or otherwise) and no Guarantor shall be entitled to the benefit of the same; and
- (b) hold in an interest-bearing suspense account any moneys received from any Guarantor or on account of any Guarantor's liability under this Clause 19.

19.8. Deferral of Guarantors' rights

Until all amounts which may be or become payable by the Obligors under or in connection with the Finance Documents have been irrevocably paid in full and unless the Agent otherwise directs, no Guarantor will exercise any rights which it may have by reason of performance by it of its obligations under the Finance Documents or by reason of any amount being payable, or liability arising, under this Clause 19:

- (a) to be indemnified by an Obligor;
- (b) to claim any contribution from any other guarantor of any Obligor's obligations under the Finance Documents;
- (c) to take the benefit (in whole or in part and whether by way of subrogation or otherwise) of any rights of the Finance Parties under the Finance Documents or of any other guarantee or security taken pursuant to, or in connection with, the Finance Documents by any Finance Party;
- (d) to bring legal or other proceedings for an order requiring any Obligor to make any payment, or perform any obligation, in respect of which any Guarantor has given a guarantee, undertaking or indemnity under Clause 19.1 (Guarantee and Indemnity);
- (e) to exercise any right of set off against any Obligor; and/or
- (f) to claim or prove as a creditor of any Obligor in competition with any Finance Party.

If a Guarantor receives any benefit, payment or distribution in relation to such rights it shall hold that benefit, payment or distribution to the extent necessary to enable all amounts which may be or become payable to the Finance Parties by the Obligors under or in connection with the Finance Documents to be repaid in full on trust for the Finance Parties and shall promptly pay or transfer the same to the Agent or as the Agent may direct for application in accordance with Clause 31 (Payment mechanics) of this Agreement.

19.9. Release of Guarantors' right of contribution

If any Guarantor (a "**Retiring Guarantor**") ceases to be a Guarantor in accordance with the terms of the Finance Documents for the purpose of any sale or other disposal of that Retiring Guarantor then on the date such Retiring Guarantor ceases to be a Guarantor:

- (a) that Retiring Guarantor is released by each other Guarantor from any liability (whether past, present or future and whether actual or contingent) to make a contribution to any other Guarantor arising by reason of the performance by any other Guarantor of its obligations under the Finance Documents; and
- (b) each other Guarantor waives any rights it may have by reason of the performance of its obligations under the Finance Documents to take the benefit (in whole or in part and whether by way of subrogation or otherwise) of any rights of the Finance Parties under any Finance Document or of any other security taken pursuant to, or in connection with, any Finance Document where such rights or security are granted by or in relation to the assets of the Retiring Guarantor.

19.10. Additional security

This guarantee is in addition to and is not in any way prejudiced by any other guarantee or security now or subsequently held by any Finance Party.

19.11. Guarantee Limitations

This guarantee does not apply to any liability to the extent that it would result in this guarantee constituting unlawful financial assistance within the meaning of Section 678 or 679 of the Companies Act 2006 or, as applicable, section 60 of the Companies Act 1963 of Ireland (as amended) and, with respect to any Additional Guarantor, is subject to any limitations set out in the Accession Letter applicable to such Additional Guarantor. In particular, but without limiting the generality of the foregoing provisions, it is agreed that the guarantee and indemnity by Schuh (ROI) Limited in terms of this Clause 19 shall not include the payment of any arrangement fee

payable under this Agreement (or any agreement amending or varying the terms of this Agreement) or the payment of any fees, costs or expenses payable by any member of the Group in connection with the Acquisition Agreement or with this Agreement (or any agreement amending or varying the terms of this Agreement).

SECTION 8

REPRESENTATIONS, UNDERTAKINGS AND EVENTS OF DEFAULT

20. REPRESENTATIONS

20.1. General

Each Obligor makes the representations and warranties set out in this Clause 20 to each Finance Party.

20.2. Status

- (a) It and each of its Subsidiaries is a limited liability corporation, duly incorporated and validly existing under the law of its jurisdiction of incorporation.
- (b) It and each of its Subsidiaries has the power to own its assets and carry on its business as it is being conducted.

20.3. Binding obligations

Subject to the Legal Reservations and to any of the matters specified in paragraphs (a) to (e) inclusive of Clause 20.9:

- (a) the obligations expressed to be assumed by it in each Transaction Document to which it is a party are legal, valid, binding and enforceable obligations; and
- (b) (without limiting the generality of paragraph (a) above), each Transaction Security Document to which it is a party creates the security interests which that Transaction Security Document purports to create and those security interests are valid and effective.

20.4. Non-conflict with other obligations

The entry into and performance by it of, and the transactions contemplated by, the Transaction Documents and the granting of the Transaction Security do not and will not conflict with:

- (a) any law or regulation applicable to it;
- (b) the constitutional documents of any member of the Group; or
- (c) any agreement or instrument binding upon it or any member of the Group or any of its or any member of the Group's assets or constitute a default or termination event (however described) under any such agreement or instrument.

20.5. Power and authority

- (a) It has the power to enter into, perform and deliver, and has taken all necessary action to authorise its entry into, performance and delivery of, the Transaction Documents to which it is or will be a party and the transactions contemplated by those Transaction Documents.
- (b) No limit on its powers will be exceeded as a result of the borrowing, grant of security or giving of guarantees or indemnities contemplated by the Transaction Documents to which it is a party.

20.6. Validity and admissibility in evidence

- (a) All Authorisations required or desirable:
 - (i) to enable it lawfully to enter into, exercise its rights and comply with its obligations in the Transaction Documents to which it is a party; and
 - (ii) to make the Transaction Documents to which it is a party admissible in evidence in its Relevant Jurisdictions,

have been obtained or effected and are in full force and effect.

(b) All Authorisations necessary for the conduct of the business, trade and ordinary activities of members of the Group have been obtained or effected and are in full force and effect to the extent that failure to obtain or effect those Authorisations has or would reasonably be expected to have a Material Adverse Effect.

20.7. Governing law and enforcement

- (a) Subject to the Legal Reservations, the choice of governing law of the Finance Documents will be recognised and enforced in its Relevant Jurisdictions.
- (b) Subject to the Legal Reservations, any judgment obtained in relation to a Finance Document in the jurisdiction of the governing law of that Finance Document will be recognised and enforced in its Relevant Jurisdictions.

20.8. Insolvency

No:

- (a) corporate action, legal proceeding or other procedure or step described in paragraph (a) of Clause 24.7 (Insolvency proceedings); or
- (b) creditors' process described in Clause 24.8 (Creditors' process),

has been taken or, to the knowledge of the Parent, threatened in relation to a member of the Group and none of the circumstances described in Clause 24.6 (Insolvency) applies to a member of the Group.

20.9. No filing or stamp taxes

Under the laws of its Relevant Jurisdiction it is not necessary that the Finance Documents be filed, recorded or enrolled with any court or other authority in that jurisdiction or that any stamp, registration, notarial or similar Taxes or fees be paid on or in relation to the Finance Documents or the transactions contemplated by the Finance Documents except:

- (a) registration of particulars of the Transaction Security Documents at the Companies Registration Office in Scotland under Section 878 of the Companies Act 2006 and payment of associated fees;
- (b) registration of particulars of the Debenture at the Companies Registration Office in Ireland under Section 99 of the Companies Act 1963 and payment of associated fees;
- (c) filing of the statutory declarations by directors (and, in the case of an Obligor incorporated in Ireland, section 60 of the Companies Act 1963 of Ireland) copies of

which are delivered to the Agent under Clause 4.1 (Initial conditions precedent) at the Companies Registration Office in England and Wales;

- (d) registration of the Standard Security at the Land Register of Scotland and payment of associated fees;
- (e) registration of the Debenture in the Land Registry or registry of deeds in Ireland to the extent it relates to any specified Irish real property;
- (f) registration of the Debenture at the Irish Patents Office to the extent it relates to any specified Irish trade marks or patents,

which registrations, filings, taxes and fees will be made and paid promptly after the date of the relevant Finance Document.

20.10. Deduction of Tax

It is not required to make any deduction for or on account of Tax from any payment it may make under any Finance Document.

20.11. No default

- (a) No Event of Default and, on the date of this Agreement, no Default is continuing or is reasonably likely to result from the making of any Utilisation or the entry into, the performance of, or any transaction contemplated by, any Transaction Document.
- (b) No other event or circumstance is outstanding which constitutes (or, with the expiry of a grace period, the giving of notice, the making of any determination or any combination of any of the foregoing, would constitute) a default or termination event (however described) under any other agreement or instrument which is binding on it or any of its Subsidiaries or to which its (or any of its Subsidiaries') assets are subject which has or is reasonably likely to have a Material Adverse Effect.

20.12. No misleading information

Save as disclosed in writing to the Agent prior to the date of this Agreement:

- (a) to the best of its knowledge and belief (having made due and diligent enquiry), any factual information contained in the Information Package was true and accurate in all material respects as at the date of the relevant report or document containing the information or (as the case may be) as at the date the information is expressed to be given;
- (b) the Business Plan has been prepared in accordance with the Accounting Principles as applied to the Original Financial Statements, and the financial projections contained in the Business Plan have been prepared on the basis of recent historical information, are fair and based on reasonable assumptions and have been approved by the board of directors of the Parent;
- (C) any financial projection or forecast contained in the Information Package has been prepared on the basis of recent historical information and on the basis of reasonable assumptions and was fair (as at the date of the relevant report or document containing the projection or forecast) and arrived at after careful consideration;
- (d) the expressions of opinion or intention provided by or on behalf of an Obligor for the purposes of the Information Package were made after careful consideration and (as at

the date of the relevant report or document containing the expression of opinion or intention) were fair and based on reasonable grounds;

- (e) to the best of its knowledge and belief (having made due and diligent enquiry), no event or circumstance has occurred or arisen and no information has been omitted from the Information Package and no information has been given or withheld that results in the information, opinions, intentions, forecasts or projections contained in the Information Package being untrue or misleading in any material respect;
- (f) to the best of its knowledge and belief (having made due and diligent enquiry), all material information provided to a Finance Party by or on behalf of the Parent or the Company on or before the date of this Agreement and not superseded before that date (whether or not contained in the Information Package) is accurate and not misleading in any material respect and all projections provided to any Finance Party on or before the date of this Agreement have been prepared in good faith on the basis of assumptions which were reasonable at the time at which they were prepared and supplied; and
- (g) to the best of its knowledge and belief (having made due and diligent enquiry), all other written information provided by any member of the Group (including its advisers) to a Finance Party or the provider of any Report was true, complete and accurate in all material respects as at the date it was provided and is not misleading in any respect.

20.13. Original Financial Statements

- (a) Its Original Financial Statements were prepared in accordance with the Accounting Principles consistently applied.
- (b) Its unaudited Original Financial Statements fairly represent its financial condition and results of operations for the relevant month or financial quarter unless expressly disclosed to the Agent in writing to the contrary prior to the date of this Agreement.
- (C) Its audited Original Financial Statements give a true and fair view of its financial condition and results of operations during the relevant financial year unless expressly disclosed to the Agent in writing to the contrary prior to the date of this Agreement.
- (d) To the best of its knowledge and belief (having made due and diligent enquiry), there has been no material adverse change in its assets, business or financial condition (or the assets, business or consolidated financial condition of the Group, in the case of the Parent) since the date of the Original Financial Statements.
- (e) The Original Financial Statements of the Company and the Parent do not consolidate the results, assets or liabilities of any person or business which does not form part of the Company Shares.
- (f) Its most recent financial statements delivered pursuant to Clause 21.1 (Financial Statements):
 - (i) have been prepared in accordance with the Accounting Principles as applied to the Original Financial Statements and the Business Plan; and
 - (ii) give a true and fair view of (if audited) or fairly present (if unaudited) its consolidated financial condition as at the end of, and consolidated results of operations for, the period to which they relate.
- (g) The budgets and forecasts supplied under this Agreement were arrived at after careful consideration and have been prepared in good faith on the basis of recent historical

information and on the basis of assumptions which were reasonable as at the date they were prepared and supplied.

(h) Since the date of the most recent financial statements delivered pursuant to Clause 21.1 (Financial Statements) there has been no material adverse change in the business, assets or financial condition of the Group.

20.14. No proceedings pending or threatened

Other than as disclosed to the Agent prior to the date of this Agreement, no litigation, arbitration or administrative proceedings or investigations of, or before, any court, arbitral body or agency which, if adversely determined, are reasonably likely to have a Material Adverse Effect have (to the best of its knowledge and belief (having made due and careful enquiry)) been started or threatened against it or any of its Subsidiaries.

20.15. No breach of laws

- (a) It has not (and none of its Subsidiaries has) breached any law or regulation which breach has or is reasonably likely to have a Material Adverse Effect.
- (b) No labour disputes are current or, to the best of its knowledge and belief (having made due and careful enquiry), threatened against any member of the Group which have or are reasonably likely to have a Material Adverse Effect.

20.16. Environmental laws

- (a) Each member of the Group is in compliance with Clause 23.3 (Environmental compliance) and to the best of its knowledge and belief (having made due and careful enquiry) no circumstances have occurred which would prevent such compliance in a manner or to an extent which has or is reasonably likely to have a Material Adverse Effect.
- (b) No Environmental Claim has been commenced or (to the best of its knowledge and belief (having made due and careful enquiry)) is threatened against any member of the Group where that claim has or is reasonably likely, if determined against that member of the Group, to have a Material Adverse Effect.

20.17. Taxation

- (a) It is not (and none of its Subsidiaries is) materially overdue in the filing of any Tax returns and it is not (and none of its Subsidiaries is) overdue in the payment of any amount in respect of Tax of £100,000 (or its equivalent in any other currency) or more.
- (b) No claims or investigations are being, or are reasonably likely to be, made or conducted against it (or any of its Subsidiaries) with respect to Taxes such that a liability of, or claim against, any member of the Group of £100,000 (or its equivalent in any other currency) or more is reasonably likely to arise.
- (c) It is resident for Tax purposes only in the jurisdiction of its incorporation.

20.18. Security and Financial Indebtedness

(a) No Security or Quasi-Security exists over all or any of the present or future assets of any member of the Group other than as permitted by this Agreement.

(b) No member of the Group has any Financial Indebtedness outstanding other than as permitted by this Agreement.

20.19. Ranking

The Transaction Security has or will have first ranking priority and it is not subject to any prior ranking or pari passu ranking Security.

20.20. Good title to assets

It and each of its Subsidiaries has a good, valid and marketable title to, or valid leases or licences of, and all appropriate Authorisations to use, the assets necessary to carry on its business as presently conducted.

20.21. Legal and beneficial ownership

It and each of its Subsidiaries is the sole legal and beneficial owner of the respective assets over which it purports to grant Security.

20.22. Shares

The shares of any member of the Group which are subject to the Transaction Security are fully paid and not subject to any option to purchase or similar rights. The constitutional documents of companies whose shares are subject to the Transaction Security do not and could not restrict or inhibit any transfer of those shares on creation or enforcement of the Transaction Security. Other than in relation to share options granted or to be granted to employees in terms of any share option schemes operated by the Group, there are no agreements in force which provide for the issue or allotment of, or grant any person the right to call for the issue or allotment of, any share or loan capital of any member of the Group (including any option or right of pre-emption or conversion).

20.23. Intellectual Property

It and each of its Subsidiaries:

- (a) is the sole legal and beneficial owner of or has licensed to it on normal commercial terms all the Intellectual Property which is material in the context of its business and which is required by it in order to carry on its business as it is being conducted and as contemplated in the Business Plan;
- (b) does not (nor does any of its Subsidiaries), in carrying on its businesses, infringe any Intellectual Property of any third party in any respect which has or is reasonably likely to have a Material Adverse Effect; and
- (c) has taken all formal or procedural actions (including payment of fees) required to maintain any material Intellectual Property owned by it.

20.24. Group Structure Chart

The Group Structure Chart delivered to the Agent pursuant to Part I of Schedule 2 (Conditions Precedent) is true, complete and accurate in all material respects.

20.25. Obligors

The Parent has no Subsidiaries other than the Obligors.

20.26. Accounting reference date

77

78

The Accounting Reference Date of each member of the Group is 30 March.

20.27. Equity Documents

Not restated.

20.28. Centre of main interests and establishments

For the purposes of The Council of the European Union Regulation No. 1346/2000 on Insolvency Proceedings (the "**Regulation**"), its centre of main interest (as that term is used in Article 3(1) of the Regulation) is situated in its jurisdiction of incorporation and it has no "**establishment**" (as that term is used in Article 2(h) of the Regulations) in any other jurisdiction.

20.29. Pensions

- (a) Neither it nor any of its Subsidiaries is or has at any time been an employer (for the purposes of Sections 38 to 51 of the Pensions Act 2004) of an occupational pension scheme which is not a money purchase scheme (both terms as defined in the Pensions Schemes Act 1993).
- (b) Neither it nor any of its Subsidiaries is or has at any time been "connected" with or an "associate" of (as those terms are used in Sections 39 and 43 of the Pensions Act 2004) such an employer.

20.30. Anti corruption law

Each member of the Group has conducted its business in compliance with applicable anti corruption laws and has instituted and maintained policies and procedures designed to promote and achieve compliance with such laws.

20.31. Times when representations made

- (a) All the representations and warranties in this Clause 20 are made by each Obligor on the date of this Agreement.
- (b) The Repeating Representations are deemed to be made by each Obligor on the date of each Utilisation Request, on each Utilisation Date and on the first day of each Interest Period (except that those contained in paragraphs (a) (e) of Clause 20.13 (Original Financial Statements) will cease to be so made once subsequent financial statements have been delivered under this Agreement).
- (c) All the representations and warranties in this Clause 20 except Clause 20.12 (No misleading information) and Clause 20.24 (Group Structure Chart) are deemed to be made by each Additional Obligor on the day on which it becomes (or it is proposed that it becomes) an Additional Obligor.
- (d) Each representation or warranty deemed to be made after the date of this Agreement shall be deemed to be made by reference to the facts and circumstances existing at the date the representation or warranty is deemed to be made.

21. INFORMATION UNDERTAKINGS

The undertakings in this Clause 21 remain in force from the date of this Agreement for so long as any amount is outstanding under the Finance Documents or any Commitment is in force.

21.1. Financial statements

The Parent shall supply to the Agent in sufficient copies for all the Lenders:

- (a) as soon as they are available, but in any event within 120 days after the end of each of its Financial Years:
 - (i) its audited consolidated financial statements for that Financial Year;
 - (ii) the audited financial statements (consolidated if appropriate) of each Obligor for that Financial Year; and
 - (iii) the audited financial statements of any other Subsidiary for that Financial Year if requested by the Agent;
- (b) at any time when (by reference to the level of EBITDA and Cashflow disclosed in any financial statements provided to the Agent in terms of this Agreement) the financial performance of the Group is below that anticipated in the Budget for the relevant period by more than 15%, as soon as they are available, but in any event within 30 days after the end of each Accounting Period its financial statements on a consolidated basis for that Accounting Period (to include cumulative management accounts for the Financial Year to date); and
- (c) as soon as they are available, but in any event within 30 days after each Quarter Date its Quarterly Financial Statements on a consolidated basis for that Financial Quarter (to include cumulative management accounts for the Financial Year to date).

21.2. Provision and contents of Compliance Certificate

- (a) The Parent shall supply a Compliance Certificate to the Agent with each set of its audited consolidated Annual Financial Statements and each set of its Quarterly Financial Statements.
- (b) The Compliance Certificate shall, amongst other things, set out (in reasonable detail) computations as to compliance with Clause 22 (Financial Covenants).
- (c) Each Compliance Certificate shall be signed by one director of the Parent.

21.3. Requirements as to financial statements

- (a) The Parent shall procure that each set of Annual Financial Statements, Quarterly Financial Statements and (if applicable) Monthly Financial Statements includes a balance sheet, profit and loss account and cashflow statement. In addition the Parent shall procure that:
 - (i) each set of Annual Financial Statements shall be audited by the Auditors;
 - (ii) each set of Quarterly Financial Statements and (if applicable) Monthly Financial Statements is accompanied by a narrative by the finance director of the Parent commenting on the performance of the Group for the Financial Quarter or Accounting Period to which the financial statements relate and the Financial Year to date and any material developments or proposals affecting the Group or its business.
- (b) Each set of financial statements delivered pursuant to Clause 21.1 (Financial statements):

- (i) shall be certified by a director of the relevant company as giving a true and fair view of (in the case of Annual Financial Statements for any Financial Year), or fairly representing (in other cases), its financial condition and operations as at the date as at which those financial statements were drawn up;
- (ii) in the case of consolidated financial statements of the Group, shall be accompanied by a narrative comparing actual performance for the period to which the financial statements relate to the projected performance for that period set out in the Budget; and
- (iii) shall be prepared using the Accounting Principles, accounting practices and financial reference periods consistent with those applied in the preparation of the Original Financial Statements for that Obligor unless, in relation to any set of financial statements, the Parent notifies the Agent that there has been a change in the Accounting Principles or the accounting practices and its Auditors (or, if appropriate, the Auditors of the Obligor) deliver to the Agent:
 - (A) a description of any change necessary for those financial statements to reflect the Accounting Principles or accounting practices upon which the relevant Obligor's Original Financial Statements were prepared; and
 - (B) sufficient information, in form and substance as may be reasonably required by the Agent, to enable the Lenders to determine whether Clause 22 (Financial covenants) has been complied with, to determine the amount of any prepayments to be made from Excess Cashflow under Clause 8.2 (Disposal and Insurance Proceeds and Excess Cashflow) and to make an accurate comparison between the financial position indicated in those financial statements and the relevant Obligor's Original Financial Statements.

Any reference in this Agreement to any financial statements shall be construed as a reference to those financial statements as adjusted to reflect the basis upon which the Original Financial Statements were prepared.

- (c) If the Agent wishes to discuss the financial position of any member of the Group with the Auditors, the Agent may notify the Parent, stating the questions or issues which the Agent wishes to discuss with the Auditors. In this event, the Parent must ensure that the Auditors are authorised (at the expense of the Parent):
 - (iv) to discuss the financial position of each member of the Group with the Agent on request from the Agent; and
 - (v) to disclose to the Agent for the Finance Parties any information which the Agent may reasonably request.

21.4. Budget

- (a) The Parent shall supply to the Agent, in sufficient copies for all the Lenders, as soon as the same become available but in any event within 60 days after the start of each of its Financial Years, an annual Budget for that financial year.
- (b) The Parent shall ensure that each Budget:
 - (i) is in a form reasonably acceptable to the Agent and includes a projected consolidated profit and loss, balance sheet and cashflow statement for the Group, projected financial covenant calculations and projected Capital Expenditure (including, without limitation, any such Capital Expenditure

incurred using funds provided for the purpose by the UK Acquisition Company);

- (ii) is prepared in accordance with the Accounting Principles and the accounting practices and financial reference periods applied to financial statements under Clause 21.1 (Financial statements); and
- (iii) has been approved by the board of directors of the Parent.
- (C) If the Parent updates or changes the Budget, it shall within not more than five days of the update or change being made deliver to the Agent, in sufficient copies for each of the Lenders, such updated or changed Budget together with a written explanation of the main changes in that Budget.

21.5. Presentations

Once in every financial year, or more frequently if requested to do so by the Agent if the Agent reasonably suspects a Default is continuing or may have occurred or may occur, at least two directors of the Parent (one of whom shall be the finance director) must give a presentation to the Finance Parties about:

- (a) the on-going business and financial performance of the Group; and
- (b) any other matter which a Finance Party may reasonably request.

21.6. Year-end

- (a) The Parent shall, as soon as reasonably practicable after the Restatement Date, change its Accounting Reference Date (and the Accounting Reference Date of each other member of the Group) to 31 January but shall not otherwise change its Accounting Reference Date and shall procure that (other than to 31 January) no member of the Group changes its Accounting Reference Date.
- (b) The Parent shall procure that each Accounting Period ends on an accounting date for the purposes of the preparation of the financial statements of the Group.

21.7. Information: miscellaneous

The Parent shall supply to the Agent (in sufficient copies for all the Lenders, if the Agent so requests):

- (a) at the same time as they are dispatched, copies of all documents dispatched by the Parent to its shareholders generally (or any class of them) or dispatched by the Parent or any Obligors to its creditors generally (or any class of them);
- (b) promptly upon becoming aware of them, the details of any litigation, arbitration or administrative proceedings which are current, threatened or pending against any member of the Group, and which, if adversely determined, are reasonably likely to have a Material Adverse Effect or which would involve a liability, or a potential or alleged liability, exceeding £100,000 (or its equivalent in other currencies);
- (C) promptly upon becoming aware of the relevant disposal or claim, details of any disposal or insurance claim which will require a prepayment under Clause 8.2 (Disposal and Insurance Proceeds and Excess Cashflow);

- (d) at the same time as each board pack in relation to any proposed new store opening is distributed to the board of directors of the Parent or any other member of the Group, a copy of that board pack;
- (e) promptly, such information as the Security Trustee may reasonably require about the Charged Property and compliance of the Obligors with the terms of any Transaction Security Documents; and
- (f) promptly on request, such further information regarding the financial condition, assets and operations of the Group and/or any member of the Group (including any requested amplification or explanation of any item in the financial statements, budgets or other material provided by any Obligor under this Agreement, any changes to Senior Management and an up to date copy of its shareholders' register (or equivalent in its jurisdiction of incorporation)) as any Finance Party through the Agent may reasonably request.

21.8. Notification of default

- (a) Each Obligor shall notify the Agent of any Default (and the steps, if any, being taken to remedy it) promptly upon becoming aware of its occurrence (unless that Obligor is aware that a notification has already been provided by another Obligor).
- (b) Promptly upon a request by the Agent, the Parent shall supply to the Agent a certificate signed by one of its directors or senior officers on its behalf certifying that no Default is continuing (or if a Default is continuing, specifying the Default and the steps, if any, being taken to remedy it).

21.9. "Know your customer" checks

- (a) If:
 - (i) the introduction of or any change in (or in the interpretation, administration or application of) any law or regulation made after the date of this Agreement;
 - (ii) any change in the status of an Obligor or the composition of the shareholders of an Obligor after the date of this Agreement; or
 - (iii) a proposed assignment or transfer by a Lender of any of its rights and/or obligations under this Agreement to a party that is not a Lender prior to such assignment or transfer,

obliges the Agent or any Lender (or, in the case of paragraph (iii) above, any prospective new Lender) to comply with "know your customer" or similar identification procedures in circumstances where the necessary information is not already available to it, each Obligor shall promptly upon the request of the Agent or any Lender supply, or procure the supply of, such documentation and other evidence as is reasonably requested by the Agent (for itself or on behalf of any Lender) or any Lender (for itself or, in the case of the event described in paragraph (iii) above, on behalf of any prospective new Lender) in order for the Agent, such Lender or, in the case of the event described in paragraph (iii) above, any prospective new Lender to carry out and be satisfied it has complied with all necessary "know your customer" or other similar checks under all applicable laws and regulations pursuant to the transactions contemplated in the Finance Documents.

(b) Each Lender shall promptly upon the request of the Agent supply, or procure the supply of, such documentation and other evidence as is reasonably requested by the Agent (for itself) in order for the Agent to carry out and be satisfied it has complied with all necessary

"know your customer" or similar checks under all applicable laws and regulations pursuant to the transactions contemplated in the Finance Documents.

- (C) The Parent shall, by not less than 10 Business Days' prior written notice to the Agent, notify the Agent (which shall promptly notify the Lenders) of its intention to request that one of its Subsidiaries becomes an Additional Obligor pursuant to Clause 26 (Changes to the Obligors).
- (d) Following the giving of any notice pursuant to paragraph (c) above, if the accession of such Additional Obligor obliges the Agent or any Lender to comply with "know your customer" or similar identification procedures in circumstances where the necessary information is not already available to it, the Parent shall promptly upon the request of the Agent or any Lender supply, or procure the supply of, such documentation and other evidence as is reasonably requested by the Agent (for itself or on behalf of any Lender) or any Lender (for itself or on behalf of any prospective new Lender) in order for the Agent or such Lender or any prospective new Lender to carry out and be satisfied it has complied with all necessary "know your customer" or other similar checks under all applicable laws and regulations pursuant to the accession of such Subsidiary to this Agreement as an Additional Obligor.

22. FINANCIAL COVENANTS

22.1. Financial definitions

In this Agreement:

"Cashflow" means, in respect of any Relevant Period, EBITDA for that Relevant Period after:

- (a) adding the amount of any decrease (and deducting the amount of any increase) in Working Capital for that Relevant Period (but **excluding** an amount of up to £2,000,000 utilised by Schuh Limited on or before 30 September 2013 in prepayment of loans advanced to Schuh Limited by Genesco Inc.);
- (b) adding the amount of any cash receipts (and deducting the amount of any cash payments) during that Relevant Period in respect of any Exceptional Items not already taken account of in calculating EBITDA for any Relevant Period (other than, in the case of cash receipts, Relevant Proceeds);
- (c) adding the amount of any cash receipts during that Relevant Period in respect of any Tax rebates or credits and deducting the amount actually paid or due and payable in respect of Taxes during that Relevant Period by any member of the Group;
- (d) adding the amount of any increase in provisions, other non-cash debits and other non-cash charges (which are not Current Assets or Current Liabilities) and deducting the amount of any non-cash credits (which are not Current Assets or Current Liabilities) in each case to the extent taken into account in establishing EBITDA;
- (e) deducting the amount of any Capital Expenditure actually made during that Relevant Period by any member of the Group (other than any amounts contributed for such purpose by the UK Acquisition Company in accordance with the terms of Clause 23.29(a)) less, to the extent not already taken into account in determining EBITDA, any landlords' inducements actually received in cash;
- (f) adding the amount of any New Shareholder Injections permitted to be included in Cashflow pursuant to Clause 22.4 (Equity Cure),

and so that no amount shall be added (or deducted) more than once and there shall be excluded the effect of all cash movements associated with the Deal Costs.

"Cashflow Cover" means the ratio of Cashflow to Debt Service in respect of any Relevant Period.

"**Current Assets**" means the aggregate (on a consolidated basis) of all inventory, work in progress, trade and other receivables of each member of the Group including prepayments in relation to operating items and sundry debtors (but excluding Cash) maturing within twelve months from the date of computation but **excluding** amounts in respect of:

- (a) receivables in relation to Tax;
- (b) Exceptional Items and other non-operating items;
- (c) any interest owing to any member of the Group; and
- (d) prepaid Deal Costs.

"**Current Liabilities**" means the aggregate (on a consolidated basis) of all liabilities (including trade creditors, accruals and provisions) of each member of the Group falling due within twelve months from the date of computation but **excluding** amounts in respect of:

- (a) liabilities for Financial Indebtedness and Finance Charges;
- (b) liabilities for Tax;
- (c) Exceptional Items and other non-operating items; and
- (d) liabilities in relation to dividends declared but not paid by the Parent or by a member of the Group in favour of a person which is not a member of the Group.

"Debt Service" means, in respect of any Relevant Period, the aggregate of:

- (a) Finance Charges for that Relevant Period;
- (b) the aggregate of all scheduled repayments of Financial Indebtedness falling due during that Relevant Period but excluding:
 - (i) any amounts falling due under the Working Capital Facility Letter or any other overdraft or revolving facility and which were available for simultaneous redrawing according to the terms of that facility;
 - (ii) any such obligations owed to any member of the Group; and
 - (iii) an amount of up to £2,000,000 utilised by Schuh Limited on or before 30 September 2013 in prepayment of loans advanced to Schuh Limited by Genesco Inc.;
- (c) the amount of any cash dividends or distributions paid or made by the Parent in respect of that Relevant Period;
- (d) the amount of the capital element of any payments in respect of that Relevant Period payable under any Finance Lease entered into by any member of the Group,

and so that no amount shall be included more than once.

"EBIT" means, in respect of any Relevant Period, the consolidated operating profit of the Group before taxation:

(a) **before deducting** any Finance Charges;

- (b) **not including** any accrued interest owing to any member of the Group;
- (c) **before taking into account** any Exceptional Items;
- (d) **before deducting** any Deal Costs;
- (e) **excluding** the amount of any profit of any member of the Group which is attributable to minority interests;
- (f) **before taking into account** any unrealised gains or losses on any financial instrument (other than any derivative instrument which is accounted for on a hedge accounting basis);
- (g) **before taking into account** any gain or loss arising from an upward or downward revaluation of any other asset;
- (h) **before deducting** any amount that in accordance with the Accounting Principles is required to be deducted from the operating profits of the Group but which is in fact attributable to payments to be made by the UK Acquisition Company to either of Colin Temple or Mark Crutchley in accordance with the terms of the Acquisition Agreement;

in each case, to the extent added, deducted or taken into account, as the case may be, for the purposes of determining operating profits of the Group before taxation.

"EBITDA" means, in respect of any Relevant Period, EBIT for that Relevant Period after adding back any amount attributable to the amortisation, depreciation or impairment of assets of members of the Group.

"Exceptional Items" means any material items of an unusual or non-recurring nature which represent gains or losses including those arising on:

- (a) the restructuring of the activities of an entity and reversals of any provisions for the cost of restructuring;
- (b) disposals, revaluations or impairment of non-current assets;
- (c) disposals of assets associated with discontinued operations; and
- (d) the termination of any Treasury Transaction.

"Excess Cashflow" means, for any period for which it is being calculated, Cashflow for that period less (except to the extent already deducted in calculating Cashflow):

- (a) Debt Service for that period;
- (b) the amount of any voluntary or mandatory prepayments made under the Finance Documents during that period (but assuming for these purposes that any mandatory prepayment under paragraph (b)(iii) of Clause 8.2 is made in the Financial Year to which it is referable, and not in the following Financial Year);
- (c) to the extent included in Cashflow, the cash proceeds of any New Shareholder Injection during that period;
- (d) to the extent included in Cashflow, the cash proceeds of any Capital Expenditure contributed by the UK Acquisition Company in accordance with the terms of Clause 23.29(a); and
- (e) £1,000,000.

"**Finance Charges**" means, for any Relevant Period, the aggregate amount of the accrued interest, commission, fees, discounts, prepayment fees, premiums or charges and other finance payments in respect of Financial Indebtedness whether paid or payable by any member of the Group (calculated on a consolidated basis) in respect of that Relevant Period:

- (a) **including** the interest (but not the capital) element of payments in respect of Finance Leases;
- (b) **including** any commission, fees, discounts and other finance payments payable by (and deducting any such amounts payable to) any member of the Group under any interest rate hedging arrangement;
- (c) **excluding** any Deal Costs,

and so that no amount shall be added (or deducted) more than once.

"Financial Quarter" means the period commencing on the day after one Quarter Date and ending on the next Quarter Date.

"**Financial Year**" means the annual accounting period of the Group ending on or about 31 March in each year up to and including 31 March 2010 and ending on or about 31 January in each year thereafter.

"Interest Cover" means the ratio of EBITDA to Finance Charges in respect of any Relevant Period.

"Leverage" means, in respect of any Relevant Period, the ratio of Total Gross Debt on the last day of that Relevant Period to EBITDA in respect of that Relevant Period.

"**New Shareholder Injection**" means the aggregate amount of cash subscribed for by any Holding Company of the Parent for ordinary shares in the Parent or for subordinated loan notes (or other subordinated debt instruments) in the Parent, such subordination to be on the terms set out in the Subordination Agreement or otherwise acceptable to the Lenders.

"**Quarter Date**" means 29 October 2011 and thereafter, 31 January, 30 April, 31 July and 31 October (or such other date as represents the last trading Saturday in the relevant month or the next month as set out in the Genesco Closing Schedule) in each year.

"**Relevant Period**" means the period of thirteen Accounting Periods ending on or around 29 October 2011 and thereafter the period of 12 months ending on a Quarter Date.

"Relevant Proceeds" means Disposal Proceeds or Insurance Proceeds (each as defined in Clause 8.2 (Disposal, Insurance Proceeds and Excess Cashflow)).

"Total Gross Debt" means, at any time, the aggregate amount of all obligations of members of the Group for or in respect of Financial Indebtedness at that time but:

- (a) **excluding** any such obligations to any other member of the Group;
- (b) **including**, in the case of Finance Leases only, their capitalised value;
- (c) **excluding** any amount falling within paragraph (f) (Treasury Transactions) of the definition of Financial Indebtedness; and
- (d) excluding an amount of up to £2,000,000 advanced to Schuh Limited by Genesco Inc.,

and so that no amount shall be included or excluded more than once.

"Working Capital" means, on any date, Current Assets less Current Liabilities.

22.2. Financial condition

The Parent shall ensure that:

- (a) *Cashflow Cover*: Cashflow Cover in respect of any Relevant Period shall not be less than 1.1:1.
- (b) *Interest Cover*: Interest Cover in respect of any Relevant Period ending on or around the Quarter Date specified in column 1 below shall not be less than the ratio set out in column 2 below opposite that Quarter Date:

Column 1 Date	Column 2 Ratio
29 October 2011	4.25:1
28 January 2012	4.5:1
28 April 2012	4.5:1
28 July 2012	4.5:1
27 October 2012	4.5:1
2 February 2013	4.5:1
4 May 2013	4.5:1
3 August 2013	4.5:1
2 November 2013	4.5:1
1 February 2014	4.5:1
3 May 2014	4.5:1
2 August 2014	4.5:1
1 November 2014	4.5:1
31 January 2015	4.5:1
2 May 2015	4.5:1
1 August 2015	4.5:1
31 October 2015	4.5:1
30 January 2016	4.5:1
30 April 2016	4.5:1
30 July 2016	4.5:1

88

29 October 2016	4.5:1
28 January 2017	4.5:1
29 April 2017	4.5:1
29 July 2017	4.5:1
28 October 2017	4.5:1
3 February 2018	4.5:1
28 April 2018	4.5:1
28 July 2018	4.5:1
27 October2018	4.5:1
2 February 2019	4.5:1
4 May 2019	4.5:1
3 August 2019	4.5:1

Leverage: Leverage in respect of any Relevant Period ending on the Quarter Date specified in column 1 below shall not exceed the ratio set out in column 2 below opposite that Quarter Date: (C)

Column 1 Quarter Date	Column 2 Ratio
29 October 2011	2.75:1
28 January 2012	2.5:1
28 April 2012	2.30:1
28 July 2012	2.25:1
27 October 2012	2.25:1
2 February 2013	2.25:1
4 May 2013	2.25:1
3 August 2013	2.25:1
2 November 2013	2.25:1
1 February 2014	2.25:1
3 May 2014	2.25:1
2 August 2014	2.25:1
1 November 2014	2.25:1

31 January 2015	2.25:1
2 May 2015	2.25:1
1 August 2015	2.25:1
31 October 2015	2.25:1
30 January 2016	2.25:1
30 April 2016	2.25:1
30 July 2016	2.25:1
29 October 2016	2.25:1
28 January 2017	2.25:1
29 April 2017	2.25:1
29 July 2017	2.25:1
28 October 2017	2.25:1
3 February 2018	2.25:1
28 April 2018	2.25:1
28 July 2018	2.25:1
27 October2018	2.25:1
2 February 2019	2.25:1
4 May 2019	2.25:1
3 August 2019	2.25:1

22.3. Financial testing

The financial covenants set out in Clause 22.2 (Financial condition) shall be calculated in accordance with the Accounting Principles as applied to the Original Financial Statements and tested by reference to each of the financial statements delivered pursuant to paragraphs (a)(i) and (b) of Clause 21.1 (Financial Statements) and/or each Compliance Certificate delivered pursuant to Clause 21.2 (Provision and contents of Compliance Certificate).

22.4. Equity Cure

- (a) In the event of any breach of any of the financial covenants in Clause 22.2 (Financial condition) (each a "Financial Covenant") for any Relevant Period ending on a Quarter Date (the "Relevant Quarter Date"), the Parent may, not later than 15 Business Days (the "Reference Date") after the last date for delivery of the Compliance Certificate for that Relevant Period, inject into the Group the cash proceeds of any New Shareholder Injection (the "Cure Amount") to remedy non-compliance with a Financial Covenant.
- (b) The effect of the Cure Amount shall (subject to the provisions of this Clause) be that each Financial Covenant is recalculated to give effect to the following adjustments:

- 90
- (i) for the purpose of calculating Cashflow Cover, the Cure Amount shall either (at the option of the Parent):
 - (1) be included in the calculation of Cashflow for that Relevant Period and (unless and until it is released to the Parent in terms of paragraph (e) below) the next three Relevant Periods; or
 - (2) reduce Total Gross Debt as at the start of the Relevant Period in which the non-compliance occurred and (unless and until it is released to the Parent in terms of paragraph (e) below) the next three Relevant Periods, and Debt Service shall be recalculated for such Relevant Periods on a pro forma basis as if the Total Gross Debt has been so reduced (and applied against Facility A, Facility B and Facility C in accordance with the terms of Clause 7.3(b));
- (ii) for the purpose of calculating Leverage, the Cure Amount shall reduce Total Gross Debt as at the end of that Relevant Period; and
- (iii) for the purpose of calculating Interest Cover, the Cure Amount shall be deemed to have been applied in prepayment of the Facilities at the beginning of that Relevant Period and at the beginning of the next three Relevant Periods and Finance Charges for that Relevant Period and (unless and until it is released to the Parent in terms of paragraph (e) below) the next three Relevant Periods shall be recalculated on a pro forma basis as if the Facilities had been so reduced.
- (c) If the re-testing of the Financial Covenants after giving effect to paragraphs (a) and (b) above demonstrates no breach has occurred in respect of the Relevant Period, then the relevant breach shall be deemed to have been remedied.
- (d) A Cure Amount may be injected up to four times over the duration of the Facilities and may not be injected in consecutive Financial Quarters.
- (e) The Parent shall procure that each Cure Amount is credited to the Escrow Account and shall be retained in the Escrow Account until the date of the Quarter Date falling 6 Months after the Relevant Quarter Date at which time it will be released to the Parent if the Parent delivers an Escrow Account Certificate to the Agent within 5 Business Days of such date. In the event that any Default occurs while any amount is held in the Escrow Account or the Parent fails to deliver an Escrow Account Certificate to the Agent within 5 Business Days of the date falling 6 Months after the Relevant Quarter Date, the balance of the Escrow Account may (at the discretion of the Agent, acting on the instructions of the Majority Lenders) be applied immediately in or towards permanent prepayment of Term Loans with such amount being applied against the Term Loans in the same manner as voluntary prepayments under Clause 7.3 (Voluntary Prepayment of Term Loans).
- (f) Any recalculation made under this Clause 22.4 will be solely for the purpose of curing a Financial Covenant breach and not for any other purpose such as calculation of Margin or Excess Cashflow or for determining the application of Excess Cashflow.

23. GENERAL UNDERTAKINGS

The undertakings in this Clause 23 remain in force from the date of this Agreement for so long as any amount is outstanding under the Finance Documents or any Commitment is in force.

23.1. Authorisations

Each Obligor shall promptly:

- 91
- (a) obtain, comply with and do all that is necessary to maintain in full force and effect; and
- (b) supply certified copies to the Agent of,

any Authorisation required under any law or regulation of a Relevant Jurisdiction to:

- (i) enable it to perform its obligations under the Finance Documents;
- (ii) ensure the legality, validity, enforceability or admissibility in evidence of any Finance Document; and
- (iii) carry on its business where failure to do so has or is reasonably likely to have a Material Adverse Effect.

23.2. Compliance with laws

Each Obligor shall (and the Parent shall ensure that each member of the Group will) comply in all respects with all laws to which it may be subject, if failure so to comply has or is reasonably likely to have a Material Adverse Effect.

23.3. Environmental compliance

Each Obligor shall (and the Parent shall ensure that each member of the Group will):

- (i) comply with all Environmental Law;
- (ii) obtain, maintain and ensure compliance with all requisite Environmental Permits;
- (iii) implement procedures to monitor compliance with and to prevent liability under any Environmental Law,

where failure to do so has or is reasonably likely to have a Material Adverse Effect.

23.4. Environmental claims

Each Obligor shall (through the Parent), promptly upon becoming aware of the same, inform the Agent in writing of:

- (a) any Environmental Claim against any member of the Group which is current, pending or threatened; and
- (b) any facts or circumstances which are reasonably likely to result in any Environmental Claim being commenced or threatened against any member of the Group,

where the claim, if determined against that member of the Group, has or is reasonably likely to have a Material Adverse Effect.

23.5. Taxation

- (a) Each Obligor shall (and the Parent shall ensure that each member of the Group will) pay and discharge all Taxes imposed upon it or its assets within the time period allowed without incurring penalties unless and only to the extent that:
 - (i) such payment is being contested in good faith;

- 92
- (ii) adequate reserves are being maintained for those Taxes and the costs required to contest them which have been disclosed in its latest financial statements delivered to the Agent under Clause 21.1 (Financial statements); and
- (iii) such payment can be lawfully withheld and failure to pay those Taxes does not have or is not reasonably likely to have a Material Adverse Effect.
- (b) No member of the Group may change its residence for Tax purposes.

23.6. Merger

No Obligor shall (and the Parent shall ensure that no other member of the Group will) enter into any amalgamation, demerger, merger, consolidation or corporate reconstruction other than a Permitted Transaction or a Permitted Merger.

23.7. Change of business

The Parent shall procure that no substantial change is made to the general nature of the business of the Group from that carried on by the Group at the date of this Agreement without the prior written consent of the Majority Lenders.

23.8. Acquisitions

- (a) Except as permitted under either Clause 23.29 (Capital Expenditure) or paragraph (b) below, no Obligor shall (and the Parent shall ensure that no other member of the Group will):
 - (i) acquire a company or any shares or securities or a business or undertaking (or, in each case, any interest in any of them) or any assets; or
 - (ii) incorporate a company.
- (b) Paragraph (a) above does not apply to an acquisition of a company, of shares, securities or a business or undertaking or any assets (or, in each case, any interest in any of them) or the incorporation of a company which is:
 - (i) a Permitted Acquisition; or
 - (ii) a Permitted Transaction.

23.9. Joint ventures

- (a) Except as permitted under paragraph (b) below, no Obligor shall (and the Parent shall ensure that no member of the Group will):
 - (i) enter into, invest in or acquire (or agree to acquire) any shares, stocks, securities or other interest in any Joint Venture; or
 - (ii) transfer any assets or lend to or guarantee or give an indemnity for or give Security for the obligations of a Joint Venture or maintain the solvency of or provide working capital to any Joint Venture (or agree to do any of the foregoing).
- (b) Paragraph (a) above does not apply to any acquisition (or agreement to acquire) any interest in a Joint Venture or transfer of assets (or agreement to transfer assets) to a Joint

Venture or loan made to or guarantee given in respect of the obligations of a Joint Venture if such transaction is a Permitted Joint Venture.

23.10. Dormant subsidiaries

No Obligor shall (and the Parent shall ensure no member of the Group will) cause or permit any member of the Group which is a Dormant Subsidiary to commence trading or cease to satisfy the criteria for a Dormant Subsidiary unless such Dormant Subsidiary becomes an Additional Guarantor in accordance with Clause 26.4 (Additional Guarantors).

23.11. Preservation of assets

Each Obligor shall (and the Parent shall ensure that each member of the Group will) maintain in good working order and condition (ordinary wear and tear excepted) all of its assets necessary or desirable in the conduct of its business.

23.12. Pari passu ranking

Each Obligor shall ensure that at all times any unsecured and unsubordinated claims of a Finance Party against it under the Finance Documents rank at least pari passu with the claims of all its other unsecured and unsubordinated creditors except those creditors whose claims are mandatorily preferred by laws of general application to companies.

23.13. Negative pledge

In this Clause 23.13, "Quasi-Security" means a transaction described in paragraph (b) below.

Except as permitted under paragraph (c) below:

- (a) No Obligor shall (and the Parent shall ensure that no other member of the Group will) create or permit to subsist any Security over any of its assets.
- (b) No Obligor shall (and the Parent shall ensure that no other member of the Group will):
 - sell, transfer or otherwise dispose of any of its assets on terms whereby they are or may be leased to or re-acquired by an Obligor or any other member of the Group;
 - (ii) sell, transfer or otherwise dispose of any of its receivables on recourse terms;
 - (iii) enter into any arrangement under which money or the benefit of a bank or other account may be applied, set-off or made subject to a combination of accounts; or
 - (iv) enter into any other preferential arrangement having a similar effect,

in circumstances where the arrangement or transaction is entered into primarily as a method of raising Financial Indebtedness or of financing the acquisition of an asset.

- (c) Paragraphs (a) and (b) above do not apply to any Security or (as the case may be) Quasi-Security, which is:
 - (i) a Permitted Security; or
 - (ii) a Permitted Transaction.

23.14. Disposals

- (a) Except as permitted under paragraph (b) below, no Obligor shall (and the Parent shall ensure that no member of the Group will) enter into a single transaction or a series of transactions (whether related or not) and whether voluntary or involuntary to sell, lease, transfer or otherwise dispose of any asset.
- (b) Paragraph (a) above does not apply to any sale, lease, transfer or other disposal which is:
 - (i) a Permitted Disposal; or
 - (ii) a Permitted Transaction.

23.15. Arms' length basis

- (a) Except as permitted by paragraph (b) below, no Obligor shall (and the Parent shall ensure no member of the Group will) enter into any transaction with any person except on arms' length terms and for full market value.
- (b) The following transactions shall not be a breach of this Clause 23.15:
 - (i) intra-Group loans permitted under Clause 23.16 (Loans or credit);
 - (ii) fees, costs and expenses payable under the Transaction Documents in the amounts set out in the Transaction Documents delivered to the Agent under Clause 4.1 (Initial conditions precedent) or agreed by the Agent; and
 - (iii) any Permitted Transactions.

23.16. Loans or credit

- (a) Except as permitted under paragraph (b) below, no Obligor shall (and the Parent shall ensure that no member of the Group will) be a creditor in respect of any Financial Indebtedness.
- (b) Paragraph (a) above does not apply to:
 - (i) a Permitted Loan; or
 - (ii) a Permitted Transaction.

23.17. No Guarantees or indemnities

- (a) Except as permitted under paragraph (b) below, no Obligor shall (and the Parent shall ensure that no member of the Group will) incur or allow to remain outstanding any guarantee in respect of any obligation of any person.
- (b) Paragraph (a) does not apply to a guarantee which is:
 - (i) a Permitted Guarantee; or
 - (ii) a Permitted Transaction.

23.18. Corporate fees, dividends and share redemptions

- (a) Except as permitted under paragraph (b) below, the Parent shall not (and will ensure that no other member of the Group will):
 - (i) declare, make or pay any dividend, charge, fee or other distribution (or interest on any unpaid dividend, charge, fee or other distribution) (whether in cash or in kind) on or in respect of its share capital (or any class of its share capital);
 - (ii) repay or distribute any dividend or share premium reserve;
 - (iii) pay or allow any member of the Group to pay any management charges or fees or any advisory or other fees to (or to the order of) any of the shareholders of the Parent (or to any Holding Company of any such shareholders); or
 - (iv) redeem, repurchase, defease, retire or repay any of its share capital or resolve to do so.
- (b) Paragraph (a) above does not apply to:
 - (i) a Permitted Payment;
 - (ii) a Permitted Transaction (other than one referred to in paragraph (c) of the definition of that term); or
 - (iii) management charges or fees or other advisory or other fees payable by members of the Group to the UK Acquisition Company not exceeding £500,000 in aggregate in any Financial Year.

23.19. Financial Indebtedness

- (a) Except as permitted under paragraph (b) below, no Obligor shall (and the Parent shall ensure that no member of the Group will) incur or allow to remain outstanding any Financial Indebtedness.
- (b) Paragraph (a) above does not apply to Financial Indebtedness which is:
 - (i) Permitted Financial Indebtedness; or
 - (ii) a Permitted Transaction.

23.20. Share capital

No Obligor shall (and the Parent shall ensure no member of the Group will) issue any shares except pursuant to:

- (a) a Permitted Share Issue; or
- (b) a Permitted Transaction.

23.21. Insurance

- (a) Each Obligor shall (and the Parent shall ensure that each member of the Group will) maintain insurances on and in relation to its business and assets against those risks and to the extent as is usual for companies carrying on the same or substantially similar business.
- (b) All insurances must be with reputable independent insurance companies or underwriters.

- (C) Not restated.
- (d) Within 30 days of the date of this Agreement, the Parent shall provide to the Agent written evidence that the insurance policy(ies) relating to the Charged Property contain (in form and substance reasonably satisfactory to the Security Trustee) an endorsement naming the Security Trustee as sole loss payee.

23.22. Pensions

The Parent shall ensure that no member of the Group is or has been at any time an employer (for the purposes of Sections 38 to 51 of the Pensions Act 2004) of an occupational pension scheme which is not a money purchase scheme (both terms as defined in the Pension Schemes Act 1993) or "connected" with or an "associate" of (as those terms are under in Sections 39 or 43 of the Pensions Act 2004) such an employer.

23.23. Access

If a Default is continuing or the Agent reasonably suspects a Default is continuing or may occur, each Obligor shall, and the Parent shall ensure that each Obligor will, (not more than once in every Financial Year unless the Agent reasonably suspects a Default is continuing or may occur) permit the Agent and/or the Security Trustee and/or accountants or other professional advisers and contractors of the Agent or Security Trustee free access at all reasonable times and on reasonable notice at the risk and cost of the Obligors to (a) the premises, assets, books, accounts and records of each Obligor and (b) meet and discuss matters with Senior Management.

23.24. Management

- (a) The Parent must ensure that there is in place in respect of each Obligor qualified management with appropriate skills.
- (b) If any of the Senior Management ceases (whether by reason of death, retirement at normal retiring age or through ill health or otherwise) to perform his or her duties, the Parent must as soon as reasonably practicable thereafter:
 - (i) notify the Agent; and
 - (ii) after consultation with the Agent as to the identity of such replacement person, find and appoint an adequately qualified replacement for him or her as promptly as practicable.

23.25. Intellectual Property

Each Obligor shall and the Parent shall procure that each member of the Group will:

- (a) preserve and maintain the subsistence and validity of the Intellectual Property necessary for the business of the relevant member of the Group;
- (b) use reasonable endeavours to prevent any infringement in any material respect of the Intellectual Property;
- (C) make registrations and pay all registration fees and taxes necessary to maintain the Intellectual Property in full force and effect and record its interest in that Intellectual Property;
- (d) not use or permit the Intellectual Property to be used in a way or take any step or omit to take any step in respect of that Intellectual Property which may materially and

adversely affect the existence or value of the Intellectual Property or imperil the right of any member of the Group to use such property; and

(e) not discontinue the use of the Intellectual Property,

where failure to do so, in the case of paragraphs (a) and (b) above, or, in the case of paragraphs (d) and (e) above, such use, permission to use, omission or discontinuation, is reasonably likely to have a Material Adverse Effect.

23.26. Amendments

No Obligor shall (and the Parent shall ensure that no member of the Group will) amend, vary, novate, supplement, supersede, waive or terminate any term of a Transaction Document or any other document delivered to the Agent pursuant to Clauses 4.1 (Initial conditions precedent) or Clause 26 (Changes to the Obligors) or enter into any agreement with any shareholders of the Parent or any of their Affiliates which is not a member of the Group except in writing:

- (a) in accordance with the provisions of Clause 37 (Amendments and Waivers); or
- (b) after the date of this Agreement, in a way which could not be reasonably expected materially and adversely to affect the interests of the Lenders.

The Parent shall promptly supply to the Agent a copy of any document relating to any of the matters referred to in paragraphs (a) and (b) above.

23.27. Financial assistance

Each Obligor shall (and the Parent shall procure each member of the Group will) comply in all respects with Section 63 of the Companies Act 1963 in Ireland and any equivalent legislation in other jurisdictions including in relation to the execution of the Transaction Security Documents and payment of amounts due under this Agreement.

23.28. Group bank accounts

The Parent shall ensure that all bank accounts of the Group (other than the Permitted Bank Accounts but including the Holding Account, the Mandatory Prepayment Account and the Escrow Account) shall be opened and maintained with a Finance Party or an Affiliate of a Finance Party and are subject to valid Security under the Transaction Security Documents.

23.29. Capital Expenditure

- (a) Subject to paragraph (b) below, the aggregate Capital Expenditure of the Group in respect of any Financial Year shall not exceed 115% of the amount approved by the Agent in the Budget (or in any updated or changed Budget) for that Financial Year.
- (b) Capital Expenditure of the Group in excess of the limit specified in paragraph (a) above (or any other limit agreed between the Parent and the Majority Lenders) may be incurred using funds provided for the purpose by the UK Acquisition Company, and made available by subscription for ordinary shares in the Parent or by way of loan complying with the terms of paragraph (g) of Permitted Financial Indebtedness.

23.30. Treasury Transactions

(a) No Obligor shall (and the Parent will procure that no members of the Group will) enter into any Treasury Transaction, other than (with a member of the LBG Group provided that its terms are broadly competitive):

- (i) the hedging transactions documented by the Hedging Agreements;
- (ii) spot and forward delivery foreign exchange contracts entered into in the ordinary course of business and not for speculative purposes; and
- (iii) any Treasury Transaction entered into for the hedging of actual or projected real exposures arising in the ordinary course of trading activities of a member of the Group for a period of not more than 12 months and not for speculative purposes.

(b) Not restated.

23.31. Further assurance

- (a) Each Obligor shall (and the Parent shall procure that each member of the Group will) promptly do all such acts or execute all such documents (including assignments, transfers, mortgages, charges, notices and instructions) as the Security Trustee may reasonably specify (and in such form as the Security Trustee may reasonably require in favour of the Security Trustee or its nominee(s)):
 - (i) to perfect the Security created or intended to be created under or evidenced by the Transaction Security Documents (which may include the execution of a mortgage, charge, assignment or other Security over all or any of the assets which are, or are intended to be, the subject of the Transaction Security) or for the exercise of any rights, powers and remedies of the Security Trustee or the Finance Parties provided by or pursuant to the Finance Documents or by law;
 - (ii) to confer on the Security Trustee or confer on the Finance Parties Security over any property and assets of that Obligor located in any jurisdiction equivalent or similar to the Security intended to be conferred by or pursuant to the Transaction Security Documents; and/or
 - (iii) to facilitate the realisation of the assets which are, or are intended to be, the subject of the Transaction Security.
- (b) Each Obligor shall (and the Parent shall procure that each member of the Group shall) take all such action as is available to it (including making all filings and registrations) as may be necessary for the purpose of the creation, perfection, protection or maintenance of any Security conferred or intended to be conferred on the Security Trustee or the Finance Parties by or pursuant to the Finance Documents.

23.32. Payment of Loan Notes

Not restated.

23.33. Acquisition Agreement

The Parent will not agree to amend or waive or permit the amendment or waiver of any of the material terms of the Acquisition Agreement, without the consent of the Agent, which consent will not be unreasonably withheld or delayed.

23.34. Anti-corruption law

(a) No Obligor shall (and the Parent shall ensure that no other member of the Group will) directly or indirectly use the proceeds of the Facilities for any purpose which would breach the Bribery Act 2010, the United States Foreign Corrupt Practices Act of 1977 or other similar legislation in other jurisdictions.

- (b) Each Obligor shall (and the Parent shall ensure that each other member of the Group will):
 - (i) conduct its business in compliance with applicable anti corruption laws; and
 - (ii) maintain policies and procedures designed to promote and achieve compliance with such laws.

24. EVENTS OF DEFAULT

Each of the events or circumstances set out in this Clause 24 is an Event of Default (save for Clause 24.21 (Acceleration)).

24.1. Non-payment

An Obligor does not pay on the due date any amount payable pursuant to a Finance Document (other than amounts due and payable under the Working Capital Facility Letter) at the place at and in the currency in which it is expressed to be payable unless:

- (a) its failure to pay is caused by:
 - (i) administrative or technical error; or
 - (ii) a Disruption Event; and
- (b) payment is made within three Business Days of its due date.

24.2. Financial covenants and other obligations

- (a) Any requirement of Clause 22 (Financial covenants) is not satisfied or an Obligor does not comply with the provisions of any of Clauses 21.1 (Financial Statements), 21.2 (Provision and contents of Compliance Certificate), 23.13 (Negative Pledge), 23.14 (Disposals), 23.16 (Loans or credit), 23.17 (No Guarantees or indemnities), 23.18 (Dividends and share redemption), 23.19 (Financial Indebtedness) and 23.26 (Amendments).
- (b) An Obligor does not comply with any provision of any Transaction Security Document.

24.3. Other obligations

- (a) An Obligor does not comply with any provision of the Finance Documents (other than those referred to in Clause 24.1 (Non-payment) and Clause 24.2 (Financial covenants and other obligations)).
- (b) No Event of Default under paragraph (a) above will occur if the failure to comply is capable of remedy and is remedied within 15 Business Days (or such other period (if any) as the Agent may agree) of the earlier of (i) the Agent giving notice to the Parent or relevant Obligor or (ii) the Parent or an Obligor becoming aware of the failure to comply.

24.4. Misrepresentation

(a) Any representation or statement made or deemed to be made by an Obligor in the Finance Documents or any other document delivered by or on behalf of any Obligor under or in connection with any Finance Document is or proves to have been incorrect or misleading in any material respect when made or deemed to be made.

(b) No Event of Default under paragraph (a) will occur if the matter giving rise to the breach is capable of remedy and is remedied within 15 Business Days (or such other period (if any) as the Agent may agree) of the earlier of (i) the Agent giving notice to the Parent or relevant Obligor or (ii) the Parent or an Obligor becoming aware of the breach.

24.5. Cross default

- (a) Any Financial Indebtedness of any member of the Group is not paid when due nor within any originally applicable grace period.
- (b) Any Financial Indebtedness of any member of the Group is declared to be or otherwise becomes due and payable prior to its specified maturity as a result of an event of default (however described).
- (C) Any commitment for any Financial Indebtedness of any member of the Group is cancelled or suspended by a creditor of any member of the Group as a result of an event of default (however described).
- (d) Any creditor of any member of the Group becomes entitled to declare any Financial Indebtedness of any member of the Group due and payable prior to its specified maturity as a result of an event of default (however described).
- (e) No Event of Default will occur under this Clause 24.5 if the aggregate amount of Financial Indebtedness or commitment for Financial Indebtedness falling within paragraphs (a) to (d) above is less than £100,000 (or its equivalent in any other currency or currencies) or (for a period of 15 Business Days) if the Financial Indebtedness relates to the Working Capital Facility.

24.6. Insolvency

- (a) A member of the Group is unable or admits inability to pay its debts as they fall due or is deemed to or declared to be unable to pay its debts under applicable law, suspends or threatens to suspend making payments on any of its debts or, by reason of actual or anticipated financial difficulties, commences negotiations with one or more of its creditors with a view to rescheduling any of its indebtedness.
- (b) The value of the assets of any member of the Group is less than its liabilities (taking into account contingent and prospective liabilities).
- (C) A moratorium is declared in respect of any indebtedness of any member of the Group. If a moratorium occurs, the ending of the moratorium will not remedy any Event of Default caused by that moratorium.

24.7. Insolvency proceedings

- (a) Any corporate action, legal proceedings or other procedure or step is taken in relation to:
 - (i) the suspension of payments, a moratorium of any indebtedness, winding-up, dissolution, administration, examination or reorganisation (by way of voluntary arrangement, scheme of arrangement or otherwise) of any member of the Group;
 - (ii) a composition, compromise, assignment or arrangement with any creditor of any member of the Group;

100

- 101
- (iii) the appointment of a liquidator, receiver, administrator, administrative receiver, examiner, compulsory manager or other similar officer in respect of any member of the Group or any of its assets; or
- (iv) enforcement of any Security over any assets of any member of the Group,

or any analogous procedure or step is taken in any jurisdiction.

- (b) Paragraph (a) shall not apply to:
 - (i) any winding-up petition which is frivolous or vexatious and is discharged, stayed or dismissed within 14 days of commencement or, if earlier, the date on which it is advertised; or
 - (ii) any step or procedure contemplated by paragraph (b) of the definition of Permitted Transaction.

24.8. Creditors' process

Any expropriation, attachment, sequestration, distress or execution or any analogous process in any jurisdiction affects any asset or assets of Obligor having an aggregate value of £250,000 or more (after taking into account the anticipated level of insurance proceeds (if any) which the Obligor will be entitled to receive and which in the opinion of the Agent (acting reasonably) is not being contested) and is not discharged within 10 Business Days.

24.9. Unlawfulness and invalidity

- (a) It is or becomes unlawful for an Obligor to perform any of its obligations under the Finance Documents or any Transaction Security created or expressed to be created or evidenced by the Transaction Security Documents ceases to be effective.
- (b) Any obligation or obligations of any Obligor under any Finance Documents are not (subject to the Legal Reservations) or cease to be legal, valid, binding or enforceable and the cessation individually or cumulatively materially and adversely affects the interests of the Lenders under the Finance Documents.
- (c) Any Finance Document ceases to be in full force and effect or any Transaction Security ceases to be legal, valid, binding, enforceable or effective or is alleged by a party to it (other than a Finance Party) to be ineffective.

24.10. Cessation of business

Any member of the Group suspends or ceases to carry on (or threatens to suspend or cease to carry on) all or a material part of its business except as a result of a Permitted Disposal or a Permitted Transaction.

24.11. Change of ownership

- (a) After the date of this Agreement, an Obligor (other than the Parent) ceases to be a wholly-owned Subsidiary of the Parent; or
- (b) an Obligor ceases to own at least the same percentage of shares in a Material Company as at the date of this Agreement,

except, in either case, as a result of a disposal which is a Permitted Disposal or a Permitted Transaction.

24.12. Constitutional Documents

The Parent amends, varies, supplements, supersedes, waives or terminates its Constitutional Documents in any way that has a material and adverse impact on the Lenders without the prior written consent of the Majority Lenders.

24.13. Change of management

Not restated.

24.14. Change of Key Personnel

Not restated.

24.15. Audit qualification

The Auditors of the Group qualify the audited annual consolidated financial statements of the Parent in a manner which the Majority Lenders (acting reasonably) consider material in the context of the Finance Documents.

24.16. Expropriation

The authority or ability of any member of the Group to conduct its business is limited or wholly or substantially curtailed by any seizure, expropriation, nationalisation, intervention, restriction or other action by or on behalf of any governmental, regulatory or other authority or other person in relation to any member of the Group or any of its assets.

24.17. Repudiation and rescission of agreements

An Obligor (or any other relevant party) rescinds or purports to rescind or repudiates or purports to repudiate a Finance Document or any of the Transaction Security or evidences an intention to rescind or repudiate a Finance Document or any Transaction Security.

24.18. Litigation

Any litigation, arbitration, administrative, governmental, regulatory or other investigations, proceedings or disputes are commenced or threatened in relation to the Transaction Documents or the transactions contemplated in the Transaction Documents or against any member of the Group or its assets which has or is reasonably likely to have a Material Adverse Effect.

24.19. Material adverse change

Any event or circumstance occurs which is reasonably likely to have a Material Adverse Effect.

24.20. Employee Benefit Trust

Not restated.

24.21. Acceleration

On and at any time after the occurrence of an Event of Default which is continuing the Agent may, and shall if so directed by the Majority Lenders, by notice to the Parent:

(a) cancel the Total Commitments at which time they shall immediately be cancelled;

- (b) declare that all or part of the Utilisations, together with accrued interest, and all other amounts accrued or outstanding under the Finance Documents be immediately due and payable, at which time they shall become immediately due and payable;
- (c) declare that all or part of the Utilisations be payable on demand, at which time they shall immediately become payable on demand by the Agent on the instructions of the Majority Lenders; and/or
- (d) exercise or direct the Security Trustee to exercise any or all of its rights, remedies, powers or discretions under the Finance Documents.

SECTION 9

CHANGES TO PARTIES

25. CHANGES TO THE LENDERS

25.1. Assignments and transfers by the Lenders

Subject to this Clause 25 a Lender (the "Existing Lender") may:

- (a) assign any of its rights; or
- (b) transfer by novation any of its rights and obligations,

under any Finance Document to another bank or financial institution or to a trust, fund or other entity which is regularly engaged in or established for the purpose of making, purchasing or investing in loans, securities or other financial assets (the "**New Lender**").

25.2. Conditions of assignment or transfer

- (a) An Existing Lender must consult with the Parent for no more than 10 days before it may make an assignment or transfer in accordance with Clause 25.1 (Assignments and transfers by the Lenders) unless the assignment or transfer is:
 - (i) to another Lender or an Affiliate of a Lender;
 - (ii) if the Existing Lender is a fund, to a fund which is a Related Fund of the Existing Lender; or
 - (iii) made at a time when an Event of Default is continuing.
- (b) An assignment will only be effective on:
 - (i) receipt by the Agent (whether in the Assignment Agreement or otherwise) of written confirmation from the New Lender (in form and substance satisfactory to the Agent) that the New Lender will assume the same obligations to the other Finance Parties and the other Secured Parties as it would have been under if it was an Original Lender;
 - (ii) the New Lender entering into the documentation required for it to accede as a party to the Intercreditor Agreement; and
 - (iii) the performance by the Agent of all necessary "know your customer" or other similar checks under all applicable laws and regulations in relation to such assignment to a New Lender, the completion of which the Agent shall promptly notify to the Lender and the New Lender.
- (c) A transfer will only be effective if the New Lender enters into the documentation required for it to accede as a party to the Intercreditor Agreement and if the procedure set out in Clause 25.5 (Procedure for transfer) is complied with.
- (d) If:
 - (iv) a Lender assigns or transfers any of its rights or obligations under the Finance Documents or changes its Facility Office; and

(v) as a result of circumstances existing at the date the assignment, transfer or change occurs, an Obligor would be obliged to make a payment to the New Lender or Lender acting through its new Facility Office under Clause 14 (Tax gross-up and indemnities) or Clause 15 (Increased Costs),

then the New Lender or Lender acting through its new Facility Office is only entitled to receive payment under those Clauses to the same extent as the Existing Lender or Lender acting through its previous Facility Office would have been if the assignment, transfer or change had not occurred.

(e) Each New Lender, by executing the relevant Transfer Certificate or Assignment Agreement, confirms, for the avoidance of doubt, that the Agent has the authority to execute on its behalf any amendment or waiver that has been approved by or on behalf of the requisite Lender or Lenders in accordance with this Agreement on or prior to the date on which the transfer or assignment becomes effective in accordance with this Agreement and that it is bound by that decision to the same extent as the Existing Lender would have been had it remained a Lender.

25.3. Assignment or transfer fee

Unless the Agent otherwise agrees and excluding an assignment or transfer (i) to an Affiliate of a Lender, or (ii) to a Related Fund or (iii) made in connection with primary syndication of the Facilities, the New Lender shall, on the date upon which an assignment or transfer takes effect, pay to the Agent (for its own account) a fee of £1,500.

25.4. Limitation of responsibility of Existing Lenders

- (a) Unless expressly agreed to the contrary, an Existing Lender makes no representation or warranty and assumes no responsibility to a New Lender for:
 - (i) the legality, validity, effectiveness, adequacy or enforceability of the Transaction Documents, the Transaction Security or any other documents;
 - (ii) the financial condition of any Obligor;
 - (iii) the performance and observance by any Obligor or any other member of the Group of its obligations under the Transaction Documents or any other documents; or
 - (iv) the accuracy of any statements (whether written or oral) made in or in connection with any Transaction Document or any other document,

and any representations or warranties implied by law are excluded.

- (b) Each New Lender confirms to the Existing Lender, the other Finance Parties and the Secured Parties that it:
 - (i) has made (and shall continue to make) its own independent investigation and assessment of the financial condition and affairs of each Obligor and its related entities in connection with its participation in this Agreement and has not relied exclusively on any information provided to it by the Existing Lender or any other Finance Party in connection with any Transaction Document or the Transaction Security; and
 - (ii) will continue to make its own independent appraisal of the creditworthiness of each Obligor and its related entities whilst any amount is or may be outstanding under the Finance Documents or any Commitment is in force.

- (c) Nothing in any Finance Document obliges an Existing Lender to:
 - (iii) accept a re-transfer or re-assignment from a New Lender of any of the rights and obligations assigned or transferred under this Clause 25; or
 - (iv) support any losses directly or indirectly incurred by the New Lender by reason of the non-performance by any Obligor of its obligations under the Transaction Documents or otherwise.

25.5. Procedure for transfer

- (a) Subject to the conditions set out in Clause 25.2 (Conditions of assignment or transfer) a transfer is effected in accordance with paragraph (c) below when the Agent executes an otherwise duly completed Transfer Certificate delivered to it by the Existing Lender and the New Lender. The Agent shall, subject to paragraph (b) below, as soon as reasonably practicable after receipt by it of a duly completed Transfer Certificate appearing on its face to comply with the terms of this Agreement and delivered in accordance with the terms of this Agreement, execute that Transfer Certificate.
- (b) The Agent shall only be obliged to execute a Transfer Certificate delivered to it by the Existing Lender and the New Lender once it is satisfied it has complied with all necessary "know your customer" or similar checks under all applicable laws and regulations in relation to the transfer to such New Lender.
- (c) On the Transfer Date:
 - (i) to the extent that in the Transfer Certificate the Existing Lender seeks to transfer by novation its rights and obligations under the Finance Documents and in respect of the Transaction Security each of the Obligors and the Existing Lender shall be released from further obligations towards one another under the Finance Documents and in respect of the Transaction Security and their respective rights against one another under the Finance Documents and in respect of the Transaction Security shall be cancelled (being the "Discharged Rights and Obligations");
 - (ii) each of the Obligors and the New Lender shall assume obligations towards one another and/or acquire rights against one another which differ from the Discharged Rights and Obligations only insofar as that Obligor or other member of the Group and the New Lender have assumed and/or acquired the same in place of that Obligor and the Existing Lender;
 - (iii) the Agent, the Arranger, the Security Trustee, the New Lender and the other Lenders shall acquire the same rights and assume the same obligations between themselves and in respect of the Transaction Security as they would have acquired and assumed had the New Lender been an Original Lender with the rights, and/or obligations acquired or assumed by it as a result of the transfer and to that extent the Agent, the Arranger and the Security Trustee and the Existing Lender shall each be released from further obligations to each other under the Finance Documents; and
 - (iv) the New Lender shall become a Party as a "Lender".

25.6. Procedure for assignment

(a) Subject to the conditions set out in Clause 25.2 (Conditions of assignment or transfer) an assignment may be effected in accordance with paragraph (c) below when the Agent executes an otherwise duly completed Assignment Agreement delivered to it by the

Existing Lender and the New Lender. The Agent shall, subject to paragraph (b) below, as soon as reasonably practicable after receipt by it of a duly completed Assignment Agreement appearing on its face to comply with the terms of this Agreement and delivered in accordance with the terms of this Agreement, execute that Assignment Agreement.

- (b) The Agent shall only be obliged to execute an Assignment Agreement delivered to it by the Existing Lender and the New Lender once it is satisfied it has complied with all necessary "know your customer" or similar checks under all applicable laws and regulations in relation to the assignment to such New Lender.
- (c) On the Transfer Date:
 - (i) the Existing Lender will assign absolutely to the New Lender its rights under the Finance Documents and in respect of the Transaction Security expressed to be the subject of the assignment in the Assignment Agreement;
 - (ii) the Existing Lender will be released from the obligations (the "**Relevant Obligations**") expressed to be the subject of the release in the Assignment Agreement (and any corresponding obligations by which it is bound in respect of the Transaction Security); and
 - (iii) the New Lender shall become a Party as a "Lender" and will be bound by obligations equivalent to the Relevant Obligations.
- (d) Lenders may utilise procedures other than those set out in this Clause 25.6 to assign their rights under the Finance Documents **provided that** they comply with the conditions set out in Clause 25.2 (Conditions of assignment or transfer).

25.7. Copy of Transfer Certificate or Assignment Agreement to Parent

The Agent shall, as soon as reasonably practicable after it has executed a Transfer Certificate, an Assignment Agreement or an Increase Confirmation, send to the Parent a copy of that Transfer Certificate or Assignment Agreement or Increase Confirmation.

25.8. Disclosure of information

- (a) Any Lender may disclose to any of its Affiliates and any other person:
 - (i) to (or through) whom that Lender assigns or transfers (or may potentially assign or transfer) all or any of its rights and obligations under the Finance Documents;
 - (ii) with (or through) whom that Lender enters into (or may potentially enter into) any sub-participation in relation to, or any other transaction under which payments are to be made by reference to, the Finance Documents or any Obligor; or
 - (iii) to whom, and to the extent that, information is required to be disclosed by any applicable law or regulation; or
 - (iv) for whose benefit that Lender charges, assigns or otherwise creates Security (or may do so) pursuant to Clause 25.9 (Security over Lenders' rights); and
- (b) any Finance Party may disclose to a rating agency or its professional advisers, or (with the consent of the Parent) any other person,

any information about any Obligor, the Group and the Finance Documents as that Lender or other Finance Party shall consider appropriate if in relation to paragraphs (a)(i) and (ii) above, the person to whom the information is to be given has entered into a Confidentiality Undertaking.

Any Confidentiality Undertaking signed by a Finance Party pursuant to this Clause 25.8 shall supersede any prior confidentiality undertaking signed by such Finance Party for the benefit of any member of the Group.

25.9. Security over Lenders' rights

In addition to the other rights provided to Lenders under this Clause 25, each Lender may without consulting with or obtaining consent from any Obligor, at any time charge, assign or otherwise create Security in or over (whether by way of collateral or otherwise) all or any of its rights under any Finance Document to secure obligations of that Lender including, without limitation:

- (a) any charge, assignment or other Security to secure obligations to a federal reserve or central bank; and
- (b) in the case of any Lender which is a fund, any charge, assignment or other Security granted to any holders (or trustee or representatives of holders) of obligations owed, or securities issued, by that Lender as security for those obligations or securities,

except that no such charge, assignment or Security shall:

- (i) release a Lender from any of its obligations under the Finance Documents or substitute the beneficiary of the relevant charge, assignment or Security for the Lender as a party to any of the Finance Documents; or
- (ii) require any payments to be made by an Obligor or grant to any person any more extensive rights than those required to be made or granted to the relevant Lender under the Finance Documents.

25.10. Debt Purchase Transactions

- (a) The Parent shall not, and shall procure that each other member of the Group shall not (i) enter into any Debt Purchase Transaction or (ii) beneficially own all or any part of the share capital of a company that is a Lender or a party to a Debt Purchase Transaction of the type referred to in paragraphs (ii) or (iii) of the definition of Debt Purchase Transaction.
- (b) For so long as any shareholder in the Parent (i) beneficially owns a Commitment or (ii) has entered into a sub-participation agreement relating to a Commitment or other agreement or arrangement having a substantially similar economic effect and such agreement has not been terminated:
 - (i) in ascertaining the Majority Lenders or whether any given percentage (including, for the avoidance of doubt, unanimity) of the Total Commitments has been obtained to approve any request for a consent, waiver, amendment or other vote under the Finance Documents such Commitment shall be deemed to be zero; and
 - (ii) for the purposes of Clause 37.2 (Exceptions), such shareholder in the Parent or the person with whom it has entered into such subparticipation, other agreement or arrangement shall be deemed not to be a Lender.
- (c) Each Lender shall, unless such Debt Purchase Transaction is an assignment or transfer, promptly notify the Agent in writing if it knowingly enters into a Debt Purchase

Transaction with a shareholder in the Parent (a "Notifiable Debt Purchase Transaction").

- (d) A Lender shall promptly notify the Agent if a Notifiable Debt Purchase Transaction to which it is a party:
 - (i) is terminated; or
 - (ii) ceases to be with a shareholder in the Parent.
- (e) Each shareholder in the Parent that is a Lender agrees that:
 - (i) in relation to any meeting or conference call to which all the Lenders are invited to attend or participate, it shall not attend or participate in the same if so requested by the Agent or, unless the Agent otherwise agrees, be entitled to receive the agenda or any minutes of the same; and
 - (ii) in its capacity as Lender, unless the Agent otherwise agrees, it shall not be entitled to receive any report or other document prepared at the behest of, or on the instructions of, the Agent or one or more of the Lenders.

26. CHANGES TO THE OBLIGORS

26.1. Assignment and transfers by Obligors

No Obligor or any other member of the Group may assign any of its rights or transfer any of its rights or obligations under the Finance Documents.

26.2. Additional Borrowers

- (a) Subject to compliance with the provisions of paragraphs (c) and (d) of Clause 21.9 ("Know your customer" checks), the Parent may request that any of its wholly owned Subsidiaries which is not a Dormant Subsidiary becomes a Borrower. That Subsidiary shall become a Borrower if:
 - (i) it is incorporated in the same jurisdiction as an existing Borrower and the Majority Lenders approve the addition of that Subsidiary or otherwise if all the Lenders approve the addition of that Subsidiary;
 - (ii) the Parent and that Subsidiary deliver to the Agent a duly completed and executed Accession Letter;
 - (iii) the Subsidiary is (or becomes) a Guarantor prior to becoming a Borrower;
 - (iv) the Parent confirms that no Default is continuing or would occur as a result of that Subsidiary becoming an Additional Borrower; and
 - (v) the Agent has received all of the documents and other evidence listed in Part II of Schedule 2 (Conditions precedent) in relation to that Additional Borrower, each in form and substance satisfactory to the Agent.
- (b) The Agent shall notify the Parent and the Lenders promptly upon being satisfied that it has received (in form and substance satisfactory to it) all the documents and other evidence listed in Part II of Schedule 2 (Conditions precedent).
- (C) Other than to the extent that the Majority Lenders notify the Agent in writing to the contrary before the Agent gives the notification described in paragraph (b) above, the

Lenders authorise (but do not require) the Agent to give that notification. The Agent shall not be liable for any damages, costs or losses whatsoever as a result of giving any such notification.

(d) Upon becoming an Additional Borrower that Subsidiary shall make any filings (and provide copies of such filings) as required by paragraph
 (g) of Clause 14.2 (Tax gross-up) and paragraph (b) of Clause 14.6 (HMRC DT Treaty Passport scheme confirmation) in accordance with those paragraphs.

26.3. Resignation of a Borrower

- (a) In this Clause 26.3, Clause 26.5 (Resignation of a Guarantor) and Clause 26.7 (Resignation and release of Security on disposal), "Third Party Disposal" means the disposal of an Obligor to a person which is not a member of the Group where that disposal is permitted under Clause 23.14 (Disposals) or made with the approval of the Majority Lenders (and the Parent has confirmed this is the case).
- (b) With the prior consent of all the Lenders, the Parent may request that a Borrower ceases to be a Borrower by delivering to the Agent a Resignation Letter. If a Borrower is the subject of a Third Party Disposal, the Parent may request that such Borrower (other than the Parent, Holdings or the Company) ceases to be a Borrower by delivering to the Agent a Resignation Letter.
- (c) The Agent shall accept a Resignation Letter and notify the Parent and the other Finance Parties of its acceptance if:
 - (i) the Parent has confirmed that no Default is continuing or would result from the acceptance of the Resignation Letter;
 - (ii) the Borrower is under no actual or contingent obligations as a Borrower under any Finance Documents;
 - (iii) where the Borrower is also a Guarantor (unless its resignation has been accepted in accordance with Clause 26.5 (Resignation of a Guarantor)), its obligations in its capacity as Guarantor continue to be legal, valid, binding and enforceable and in full force and effect (subject to the Legal Reservations) and the amount guaranteed by it as a Guarantor is not decreased (and the Parent has confirmed this is the case); and
 - (iv) the Parent has confirmed that it shall ensure that any relevant Disposal Proceeds will be applied in accordance with Clause 8.3 (Application of mandatory prepayments).
- (d) Upon notification by the Agent to the Parent of its acceptance of the resignation of a Borrower, that company shall cease to be a Borrower and shall have no further rights or obligations under the Finance Documents as a Borrower except that the resignation shall not take effect (and the Borrower will continue to have rights and obligations under the Finance Documents) until the date on which the Third Party Disposal takes effect.

26.4. Additional Guarantors

- (a) Subject to compliance with the provisions of paragraphs (b) and (c) of Clause 21.9 ("Know your customer" checks), the Parent may request that any of its wholly owned Subsidiaries become a Guarantor.
- (b) The Parent shall procure that any other member of the Group which is not a Dormant Subsidiary shall, as soon as possible after becoming a member of the Group (or ceasing

to be a Dormant Subsidiary), shall become an Additional Guarantor and grant Security as the Agent may require.

- (c) A member of the Group shall become an Additional Guarantor if:
 - (i) the Parent and the proposed Additional Guarantor deliver to the Agent a duly completed and executed Accession Letter; and
 - (ii) the Agent has received all of the documents and other evidence listed in Part II of Schedule 2 (Conditions Precedent) in relation to that Additional Guarantor, each in form and substance satisfactory to the Agent.
- (d) The Agent shall notify the Parent and the Lenders promptly upon being satisfied that it has received (in form and substance satisfactory to it) all the documents and other evidence listed in Part II of Schedule 2 (Conditions precedent).
- (e) Other than to the extent that the Majority Lenders notify the Agent in writing to the contrary before the Agent gives the notification described in paragraph (c) above, the Lenders authorise (but do not require) the Agent to give that notification. The Agent shall not be liable for any damages, costs or losses whatsoever as a result of giving any such notification.

26.5. Resignation of a Guarantor

- (a) The Parent may request that a Guarantor (other than the Parent or the Company) ceases to be a Guarantor by delivering to the Agent a Resignation Letter if:
 - (i) that Guarantor is being disposed of by way of a Third Party Disposal (as defined in Clause 26.3 (Resignation of a Borrower)) and the Parent has confirmed this is the case; or
 - (ii) all the Lenders have consented to the resignation of that Guarantor.
- (b) The Agent shall accept a Resignation Letter and notify the Parent and the Lenders of its acceptance if:
 - (i) the Parent has confirmed that no Default is continuing or would result from the acceptance of the Resignation Letter;
 - (ii) no payment is due from the Guarantor under Clause 19.1 (Guarantee and indemnity);
 - (iii) where the Guarantor is also a Borrower, it is under no actual or contingent obligations as a Borrower and has resigned and ceased to be a Borrower under Clause 26.3 (Resignation of a Borrower); and
 - (iv) the Parent has confirmed that it shall ensure that the Disposal Proceeds will be applied, in accordance with Clause 8.3 (Application of mandatory prepayments).
- (C) The resignation of that Guarantor shall not be effective until the date of the relevant Third Party Disposal at which time that company shall cease to be a Guarantor and shall have no further rights or obligations under the Finance Documents as a Guarantor.

111

26.6. Repetition of Representations

Delivery of an Accession Letter constitutes confirmation by the relevant Subsidiary that the representations and warranties referred to in paragraph (c) of Clause 20.31 (Times when representations made) are true and correct in relation to it as at the date of delivery as if made by reference to the facts and circumstances then existing.

26.7. Resignation and release of security on disposal

If a Borrower or Guarantor is or is proposed to be the subject of a Third Party Disposal then:

- (a) where that Borrower or Guarantor created Transaction Security over any of its assets or business in favour of the Security Trustee, or Transaction Security in favour of the Security Trustee was created over the shares (or equivalent) of that Borrower or Guarantor, the Security Trustee may, at the cost and request of the Parent, release those assets, business or shares (or equivalent) and issue certificates of non-crystallisation;
- (b) the resignation of that Borrower or Guarantor and related release of Transaction Security referred to in paragraph (a) above shall not become effective until the date of that disposal; and
- (C) if the disposal of that Borrower or Guarantor is not made, the Resignation Letter of that Borrower or Guarantor and the related release of Transaction Security referred to in paragraph (a) above shall have no effect and the obligations of the Borrower or Guarantor and the Transaction Security created or intended to be created by or over that Borrower or Guarantor shall continue in full force and effect.

SECTION 10

THE FINANCE PARTIES

27. ROLE OF THE AGENT, THE ARRANGER AND OTHERS

27.1. Appointment of the Agent

- (a) Each of the Arranger and the Lenders appoints the Agent to act as its agent under and in connection with the Finance Documents.
- (b) Each of the Arranger and the Lenders authorises the Agent to perform the duties, obligations and responsibilities and to exercise the rights, powers, authorities and discretions specifically given to the Agent under or in connection with the Finance Documents together with any other incidental rights, powers, authorities and discretions.

27.2. Instructions

(f)

- (a) The Agent shall:
 - (i) unless a contrary indication appears in a Finance Document, exercise or refrain from exercising any right, power, authority or discretion vested in it as Agent in accordance with any instructions given to it by:
 - (A) all Lenders if the relevant Finance Document stipulates the matter is an all Lender decision; and
 - (B) in all other cases, the Majority Lenders; and
 - (ii) not be liable for any act (or omission) if it acts (or refrains from acting) in accordance with paragraph (i) above.
- (b) The Agent shall be entitled to request instructions, or clarification of any instruction, from the Majority Lenders (or, if the relevant Finance Document stipulates the matter is a decision for any other Lender or group of Lenders, from that Lender or group of Lenders) as to whether, and in what manner, it should exercise or refrain from exercising any right, power, authority or discretion and the Agent may refrain from acting unless and until it receives those instructions or that clarification.
- (c) Save in the case of decisions stipulated to be a matter for any other Lender or group of Lenders under the relevant Finance Document and unless a contrary indication appears in a Finance Document, any instructions given to the Agent by the Majority Lenders shall override any conflicting instructions given by any other Parties and will be binding on all Finance Parties save for the Security Trustee.
- (d) The Agent may refrain from acting in accordance with any instructions of any Lender or group of Lenders until it has received any indemnification and/or security that it may in its discretion require (which may be greater in extent than that contained in the Finance Documents and which may include payment in advance) for any cost, loss or liability which it may incur in complying with those instructions.
- (e) In the absence of instructions, the Agent may act (or refrain from acting) as it considers to be in the best interest of the Lenders.
- The Agent is not authorised to act on behalf of a Lender (without first obtaining that Lender's consent) in any legal or arbitration proceedings relating to any Finance Document. This paragraph (f) shall not apply to any legal or arbitration proceeding

relating to the perfection, preservation or protection of rights under the Transaction Security Documents or enforcement of the Transaction Security or Transaction Security Documents.

27.3. Duties of the Agent

- (a) The Agent's duties under the Finance Documents are solely mechanical and administrative in nature.
- (b) The Agent shall promptly forward to a Party the original or a copy of any document which is delivered to the Agent for that Party by any other Party.
- (C) Except where a Finance Document specifically provides otherwise, the Agent is not obliged to review or check the adequacy, accuracy or completeness of any document it forwards to another Party.
- (d) If the Agent receives notice from a Party referring to this Agreement, describing a Default and stating that the circumstance described is a Default, it shall promptly notify the other Finance Parties.
- (e) If the Agent is aware of the non-payment of any principal, interest, commitment fee or other fee payable to a Finance Party (other than the Agent, the Arranger or the Security Trustee) under this Agreement it shall promptly notify the other Finance Parties.
- (f) . The Agent shall only have the duties, obligations and responsibilities expressly specified in the Finance Documents to which it is expressed to be a party (and no others shall be implied).

27.4. Role of the Arranger

Except as specifically provided in the Finance Documents, the Arranger has no obligations of any kind to any other Party under or in connection with any Finance Document.

27.5. No fiduciary duties

- (a) Nothing in this Agreement constitutes the Agent and/or the Arranger as a trustee or fiduciary of any other person.
- (b) None of the Agent, the Security Trustee or the Arranger shall be bound to account to any Lender for any sum or the profit element of any sum received by it for its own account.

27.6. Business with the Group

The Agent, the Security Trustee and the Arranger may accept deposits from, lend money to and generally engage in any kind of banking or other business with any member of the Group.

27.7. Rights and discretions

- (a) The Agent may:
 - (i) rely on any representation, communication, notice or document notice or document believed by it to be genuine, correct and appropriately authorised; and
 - (ii) assume that:

- 115
- (A) any instructions received by it from the Majority Lenders, and Lenders or any group of Lenders are duly given in accordance with the terms of the Finance Documents; and
- (B) unless it has received notice of revocation, that those instructions have not been revoked; and
- (iii) rely on a certificate from any person:
 - (A) as to any matter of fact or circumstance which might reasonably be expected to be within the knowledge of that person; or
 - (B) to the effect that such person approves of any particular dealing, transaction, step, action or thing,

as sufficient evidence that that is the case and, in the case of paragraph (A) above, may assume the truth and accuracy of that certificate; and

- (iv) rely on any statement made by a director, authorised signatory or employee of any person regarding any matters which may reasonably be assumed to be within his knowledge or within his power to verify.
- (b) The Agent may assume (unless it has received notice to the contrary in its capacity as agent for the Lenders) that:
 - (ii) no Default has occurred (unless it has actual knowledge of a Default arising under Clause 24.1 (Non-payment));
 - (iii) any right, power, authority or discretion vested in any Party or the Majority Lenders has not been exercised; and
 - (iv) any notice or request made by the Parent (other than a Utilisation Request or Selection Notice) is made on behalf of and with the consent and knowledge of all the Obligors.
- (C) The Agent may engage, pay for and rely on the advice or services of any lawyers, accountants, surveyors or other experts (whether obtained by the Agent or by any other Party) and shall not be liable for any damages, costs or losses to any person, any diminution in value or any liability whatsoever arising as a result of its so relying.
- (d) Without prejudice to the generality of paragraph (c) above, the Agent may at any time engage and pay for the services of any lawyers to act as independent counsel to the Agent (and so separate from any lawyers instructed by the Lenders) if the Agent in its reasonable opinion deems this desirable.
- (e) The Agent may act in relation to the Finance Documents through its officers, employees and agents and shall not:
 - (v) be liable for any error of judgement made by any such person; or
 - (vi) be bound to supervise, or be in any way responsible for any loss incurred by reason of misconduct, omission or default on the part of any such person, unless such error or such loss was directly caused by the Agent's gross negligence or wilful misconduct.

- (f) The Agent may disclose to any other Party any information it reasonably believes it has received as agent under this Agreement.
- (g) Notwithstanding any other provision of any Finance Document to the contrary, neither the Agent nor the Arranger is obliged to do or omit to do anything if it would or might in its reasonable opinion constitute a breach of any law or regulation or a breach of a fiduciary duty or duty of confidentiality.
- (h) The Agent is not obliged to disclose to any Finance Party any details of the rate notified to the Agent by any Lender for the purpose of paragraph (a)(ii) or Clause 12.2 (Market Disruption).
- (i) Notwithstanding any provision of any Finance Document to the contrary, the Agent is not obliged to expend or risk its own funds or otherwise incur any financial liability in the performance of its duties, obligations or responsibilities or the exercise of any right, power, authority or discretion if it has grounds for believing the repayment of such funds or adequate indemnity against, or security for, such risk or liability is not reasonably assured to it.

27.8. Majority Lenders' instructions

- (a) Unless a contrary indication appears in a Finance Document, the Agent shall (i) exercise any right, power, authority or discretion vested in it as Agent in accordance with any instructions given to it by the Majority Lenders (or, if so instructed by the Majority Lenders, refrain from exercising any right, power, authority or discretion vested in it as Agent) and (ii) not be liable for any act (or omission) if it acts (or refrains from taking any action) in accordance with an instruction of the Majority Lenders.
- (b) Unless a contrary indication appears in a Finance Document, any instructions given by the Majority Lenders will be binding on all the Finance Parties other than the Security Trustee.
- (C) The Agent may refrain from acting in accordance with the instructions of the Majority Lenders (or, if appropriate, the Lenders) until it has received such security as it may require for any cost, loss or liability (together with any associated VAT) which it may incur in complying with the instructions.
- (d) In the absence of instructions from the Majority Lenders, (or, if appropriate, the Lenders) the Agent may act (or refrain from taking action) as it considers to be in the best interest of the Lenders.
- (e) The Agent is not authorised to act on behalf of a Lender (without first obtaining that Lender's consent) in any legal or arbitration proceedings relating to any Finance Document. This paragraph (e) shall not apply to any legal or arbitration proceeding relating to the perfection, preservation or protection of rights under the Transaction Security Documents or enforcement of the Transaction Security or Transaction Security Documents.

27.9. Responsibility for documentation

Neither of the Agent or the Arranger:

(a) is responsible or liable for the adequacy, accuracy and/or completeness of any information (whether oral or written) supplied by the Agent, the Arranger, an Obligor or any other person given in or in connection with any Finance Document or the Reports or the transactions contemplated in the Finance Documents or any other agreement,

arrangement or document entered into, made or executed in anticipation of, under or in connection with any Finance Document;

- (b) is responsible or liable for the legality, validity, effectiveness, adequacy or enforceability of any Finance Document or the Transaction Security or any other agreement, arrangement or document entered into, made or executed in anticipation of or in connection with any Finance Document or the Transaction Security; or
- (C) is responsible or liable for any determination as to whether any information provided or to be provided to any Finance Party is non public information the use of which may be regulated or prohibited by applicable law or regulation relating to insider dealing or otherwise.

27.10. No duty to monitor

The Agent shall not be bound to enquire:

- (a) whether or not any Default has occurred;
- (b) as to the performance, default or any breach by any Party of its obligations under any Finance Document; or
- (c) whether any other event specified in any Finance Document has occurred.

27.11. Exclusion of liability

- (a) Without limiting paragraph (b) below (and without prejudice to any other provision of any Finance Document excluding or limiting the liability of the Agent), the Agent will not be liable for:
 - any damages, costs or losses to any person, any diminution in value, or any liability whatsoever arising as a result of taking or not taking any action under or in connection with any Finance Document or the Transaction Security, unless directly caused by its gross negligence or wilful misconduct;
 - (ii) exercising, or not exercising, any right, power, authority or discretion given to it by, or in connection with, any Finance Document, the Transaction Security or any other agreement, arrangement or document entered into, made or executed in anticipation of, under or in connection with, any Finance Document or the Transaction Security; or
 - (iii) without prejudice to the generality of paragraphs (i) and (ii) above, any damages, costs or losses to any person, any diminution in value or any liability whatsoever arising as a result of:
 - (A) any act, event or circumstance not reasonably within its control; or
 - (B) the general risks of investment in, or the holding of assets in, any jurisdiction,

including (in each case and without limitation) such damages, costs, losses, diminution in value or liability arising as a result of: nationalisation, expropriation or other governmental actions; any regulation, currency restriction, devaluation or fluctuation; market conditions affecting the execution or settlement of transactions or the value of assets (including any Disruption Event); breakdown, failure or malfunction of any third party transport, telecommunications, computer services or systems; natural disasters or acts of God; war, terrorism, insurrection or revolution; or strikes or industrial action.

- (b) No Party (other than the Agent) may take any proceedings against any officer, employee or agent of the Agent, in respect of any claim it might have against the Agent or in respect of any act or omission of any kind by that officer, employee or agent in relation to any Finance Document or any Transaction Document and any officer, employee or agent of the Agent may rely on this Clause subject to Clause 1.3 (Third party rights) and the provisions of the Third Parties Act.
- (C) The Agent will not be liable for any delay (or any related consequences) in crediting an account with an amount required under the Finance Documents to be paid by the Agent if the Agent has taken all necessary steps as soon as reasonably practicable to comply with the regulations or operating procedures of any recognised clearing or settlement system used by the Agent for that purpose.
- (d) Nothing in this Agreement shall oblige the Agent or the Arranger to carry out:
 - (i) any "know your customer" or other checks in relation to any person; or
 - (ii) any check on the extent to which any transaction contemplated by this Agreement might be unlawful for any Lender,

on behalf of any Lender and each Lender confirms to the Agent and the Arranger that it is solely responsible for any such checks it is required to carry out and that it may not rely on any statement in relation to such checks made by the Agent or the Arranger.

(e) Without prejudice to any provision of any Finance Document excluding or limiting the Agent's liability, any liability of the Agent arising under or in connection with any Finance Document or the Transaction Security shall be limited to the amount of actual loss which has been finally judicially determined to have been suffered (as determined by reference to the date of default of the Agent or, if later, the date on which the loss arises as a result of such default) but without reference to any special conditions or circumstances known to the Agent at any time which increase the amount of that loss. In no event shall the Agent be liable for any loss of profits, goodwill, reputation, business opportunity or anticipated saving, or for special, punitive, indirect or consequential damages, whether or not the Agent has been advised of the possibility of such loss or damages.

27.12. Lenders' indemnity to the Agent

Each Lender shall (in proportion to its share of the Total Commitments or, if the Total Commitments are then zero, to its share of the Total Commitments immediately prior to their reduction to zero) indemnify the Agent, within three Business Days of demand, against any cost, loss or liability incurred by the Agent (otherwise than by reason of the Agent's gross negligence or wilful misconduct) in acting as Agent under the Finance Documents (unless the Agent has been reimbursed by an Obligor pursuant to a Finance Document).

27.13. Resignation of the Agent

(a) The Agent may resign and appoint one of its Affiliates acting through an office in the United Kingdom as successor by giving notice to the Lenders and the Parent.

- (b) Alternatively the Agent may resign by giving notice to the Lenders and the Parent, in which case the Majority Lenders (after consultation with the Parent) may appoint a successor Agent.
- (C) If the Majority Lenders have not appointed a successor Agent in accordance with paragraph (b) above within 30 days after notice of resignation was given, the Agent (after consultation with the Parent) may appoint a successor Agent (acting through an office in the United Kingdom).
- (d) If the Agent wishes to resign because (acting reasonably) it has concluded that it is no longer appropriate for it to remain as agent and the Agent is entitled to appoint a successor Agent under paragraph (c) above, the Agent may (if it concludes (acting reasonably) that it is necessary to do so in order to persuade the proposed successor Agent to become a party to this Agreement as Agent) agree with the proposed successor Agent amendments to this Clause 27 and any other term of this Agreement dealing with the rights or obligations of the Agent consistent with then current market practice for the appointment and protection of corporate trustees together with any reasonable amendments to the agency fee payable under this Agreement which are consistent with the successor Agent's normal fee rates and those amendments will bind the Parties.
- (e) The retiring Agent shall, at its own cost, make available to the successor Agent such documents and records and provide such assistance as the successor Agent may reasonably request for the purposes of performing its functions as Agent under the Finance Documents.
- (f) The Agent's resignation notice shall only take effect upon the appointment of a successor.
- (g) Upon the appointment of a successor, the retiring Agent shall be discharged from any further obligation in respect of the Finance Documents but shall remain entitled to the benefit of this Clause 27. Its successor and each of the other Parties shall have the same rights and obligations amongst themselves as they would have had if such successor had been an original Party.
- (h) After consultation with the Parent, the Majority Lenders may, by notice to the Agent, require it to resign in accordance with paragraph (b) above. In this event, the Agent shall resign in accordance with paragraph (b) above.

27.14. Replacement of the Agent

- (a) After consultation with the Company, the Majority Lenders may, by giving 30 days' notice to the Agent (or, at any time the Agent is an Impaired Agent, by giving any shorter notice determined by the Majority Lenders) replace the Agent by appointing a successor Agent (acting through an office in the United Kingdom).
- (b) The retiring Agent shall (at its own cost if it is an Impaired Agent and otherwise at the expense of the Lenders) make available to the successor Agent such documents and records and provide such assistance as the successor Agent may reasonably request for the purposes of performing its functions as Agent under the Finance Documents.
- (c) The appointment of the successor Agent shall take effect on the date specified in the notice from the Majority Lenders to the retiring Agent. As from this date, the retiring Agent shall be discharged from any further obligation in respect of the Finance Documents but shall remain entitled to the benefit of this Clause 27.14 (and any agency fees for the account of the retiring Agent shall cease to accrue from (and shall be payable on) that date).

- 120
- (d) Any successor Agent and each of the other Parties shall have the same rights and obligations amongst themselves as they would have had if such successor had been an original Party.

27.15. Confidentiality

- (a) In acting as agent for the Finance Parties, the Agent shall be regarded as acting through its agency division which shall be treated as a separate entity from any other of its divisions or departments.
- (b) If information is received by another division or department of the Agent, it may be treated as confidential to that division or department and the Agent shall not be deemed to have notice of it.
- (C) Notwithstanding any other provision of any Finance Document to the contrary, neither the Agent nor the Arranger are obliged to disclose to any other person (i) any confidential information or (ii) any other information if the disclosure would or might in its reasonable opinion constitute a breach of any law or a breach of a fiduciary duty.

27.16. Relationship with the Lenders

- (a) The Agent may treat the person shown in its records as Lender at the opening of business (in the place of the Agent's principal office as notified to the Finance Parties from time to time) as the Lender acting through its Facility Office:
 - (i) entitled to or liable for any payment due under and Finance Document on that day; and
 - (ii) entitled to receive and act upon any notice, request, document or communication or make any decision or determination under any Finance Document made or delivered on that day, unless it has received not less than five Business Days' prior notice from that Lender to the contrary in accordance with the terms of this Agreement.
- (b) Any Lender may by notice to the Agent appoint a person to receive on its behalf all notices, communications, information and documents to be made or despatched to that Lender under the Finance Documents. Such notice shall contain the address, fax number and (where communication by electronic mail or other electronic means is permitted under Clause 33.6 (Electronic communication)) electronic mail address and/or any other information required to enable the sending and receipt of information by that means (and, in each case, the department or officer, if any, for whose attention communication is to be made) and be treated as a notification of a substitute address, fax number, electronic mail address, department and officer by that Lender for the purposes of Clause 33.2 (Addresses) and paragraph (a)(ii) of Clause 33.6 (Electronic communication) and the Agent shall be entitled to treat such person as the person entitled to receive all such notices, communications, information and documents as though that person were that Lender.
- (C) Each Lender shall supply the Agent with any information that the Security Trustee may reasonably specify (through the Agent) as being necessary or desirable to enable the Security Trustee to perform its functions as Security Trustee. Each Lender shall deal with the Security Trustee exclusively through the Agent and shall not deal directly with the Security Trustee.

27.17. Credit appraisal by the Lenders

Without affecting the responsibility of any Obligor for information supplied by it or on its behalf in connection with any Finance Document, each Lender confirms to the Agent and the Arranger that it has been, and will continue to be, solely responsible for making its own independent appraisal and investigation of all risks arising under or in connection with any Finance Document including but not limited to:

- (a) the financial condition, status and nature of each member of the Group;
- (b) the legality, validity, effectiveness, adequacy or enforceability of any Finance Document and the Transaction Security and any other agreement, arrangement or document entered into, made or executed in anticipation of, under or in connection with any Finance Document or the Transaction Security;
- (C) whether that Secured Party has recourse, and the nature and extent of that recourse, against any Party or any of its respective assets under or in connection with any Finance Document, the Transaction Security, the transactions contemplated by the Finance Documents or any other agreement, arrangement or document entered into, made or executed in anticipation of, under or in connection with any Finance Document;
- (d) the adequacy, accuracy and/or completeness of the Reports and any other information provided by the Agent, any Party or by any other person under or in connection with any Finance Document, the transactions contemplated by the Finance Documents or any other agreement, arrangement or document entered into, made or executed in anticipation of, under or in connection with any Finance Document; and
- (e) the right or title of any person in or to, or the value or sufficiency of any part of the Charged Property, the priority of any of the Transaction Security or the existence of any Security affecting the Charged Property.

27.18. Reference Banks

If a Reference Bank (or, if a Reference Bank is not a Lender, the Lender of which it is an Affiliate) ceases to be a Lender, the Agent shall (in consultation with the Parent) appoint another Lender or an Affiliate of a Lender to replace that Reference Bank.

27.19. Agent's management time

Any amount payable to the Agent under Clause 16.3 (Indemnity to the Agent), Clause 18 (Costs and expenses) and Clause 27.12 (Lenders' indemnity to the Agent) shall include the cost of utilising the Agent's management time or other resources and will be calculated on the basis of such reasonable daily or hourly rates as the Agent may notify to the Parent and the Lenders, and is in addition to any fee paid or payable to the Agent under Clause 13 (Fees).

27.20. Deduction from amounts payable by the Agent

If any Party owes an amount to the Agent under the Finance Documents the Agent may, after giving notice to that Party, deduct an amount not exceeding that amount from any payment to that Party which the Agent would otherwise be obliged to make under the Finance Documents and apply the amount deducted in or towards satisfaction of the amount owed. For the purposes of the Finance Documents that Party shall be regarded as having received any amount so deducted.

27.21. Reliance and engagement letters

Each Finance Party and Secured Party confirms that each of the Arranger and the Agent has authority to accept on its behalf (and ratifies the acceptance on its behalf of any letters or reports

already accepted by the Arranger or Agent) the terms of any reliance letter or engagement letters relating to the Reports or any reports or letters provided by accountants in connection with the Finance Documents or the transactions contemplated in the Finance Documents (including any net asset letter in connection with the financial assistance procedures) and to bind it in respect of those Reports, reports or letters and to sign such letters on its behalf and further confirms that it accepts the terms and qualifications set out in such letters.

28. ROLE OF THE SECURITY TRUSTEE

28.1. Trust

The Security Trustee declares that it shall hold the Transaction Security on trust for the Finance Parties on the terms contained in this Agreement. Each of the parties to this Agreement agrees that the Security Trustee shall have only those duties, obligations and responsibilities expressly specified in this Agreement or in the Transaction Security Documents (and no others shall be implied).

28.2. No independent power

The Finance Parties shall not have any independent power to enforce, or have recourse to, any of the Transaction Security or to exercise any rights or powers arising under the Transaction Security Documents except through the Security Trustee.

28.3. Security Trustee's instructions

The Security Trustee shall:

- (a) unless a contrary indication appears in a Finance Document, act in accordance with any instructions given to it by the Agent and shall be entitled to assume that (i) any instructions received by it from the Agent are duly given by or on behalf of the Majority Lenders or, as the case may be, the Lenders in accordance with the terms of the Finance Documents and (ii) unless it has received actual notice of revocation that any instructions or directions given by the Agent have not been revoked;
- (b) be entitled to request instructions, or clarification of any direction, from the Agent as to whether, and in what manner, it should exercise or refrain from exercising any rights, powers and discretions and the Security Trustee may refrain from acting unless and until those instructions or clarification are received by it; and
- (c) be entitled to, carry out all dealings with the Lenders through the Agent and may give to the Agent any notice or other communication required to be given by the Security Trustee to the Lenders.

28.4. Security Trustee's actions

Subject to the provisions of this Clause 28:

- (a) the Security Trustee may, in the absence of any instructions to the contrary, take such action in the exercise of any of its powers and duties under the Finance Documents which in its absolute discretion it considers to be for the protection and benefit of all the Finance Parties; and
- (b) at any time after receipt by the Security Trustee of notice from the Agent directing the Security Trustee to exercise all or any of its rights, remedies, powers or discretions under any of the Finance Documents, the Security Trustee may, and shall if so directed by the Agent, take any action as in its sole discretion it thinks fit to enforce the Transaction Security.

28.5. Security Trustee's discretions

The Security Trustee may:

- (a) assume (unless it has received actual notice to the contrary in its capacity as Security Trustee for the Finance Parties) that (i) no Default has occurred and no Obligor is in breach of or default under its obligations under any of the Finance Documents and (ii) any right, power, authority or discretion vested in any person has not been exercised;
- (b) if it receives any instructions or directions from the Agent to take any action in relation to the Transaction Security, assume that all applicable conditions under the Finance Documents for taking that action have been satisfied;
- (c) engage, pay for and rely on the advice or services of any lawyers, accountants, surveyors or other experts (whether obtained by the Security Trustee or by any other Finance Party) whose advice or services may at any time seem necessary, expedient or desirable;
- (d) rely upon any communication or document believed by it to be genuine and, as to any matters of fact which might reasonably be expected to be within the knowledge of a Finance Party or an Obligor, upon a certificate signed by or on behalf of that person; or
- (e) refrain from acting in accordance with the instructions of the Agent or Lenders (including bringing any legal action or proceeding arising out of or in connection with the Finance Documents) until it has received any indemnification and/or security that it may in its absolute discretion require (whether by way of payment in advance or otherwise) for all costs, losses and liabilities which it may incur in bringing such action or proceedings.

28.6. Security Trustee's obligations

The Security Trustee shall promptly inform the Agent of:

- (a) the contents of any notice or document received by it in its capacity as Security Trustee from any Obligor under any Finance Document; and
- (b) the occurrence of any Default or any default by an Obligor in the due performance of or compliance with its obligations under any Finance Document of which the Security Trustee has received notice from any other party to this Agreement.

28.7. Excluded obligations

The Security Trustee shall not:

- (a) be bound to enquire as to the occurrence or otherwise of any Default or the performance, default or any breach by an Obligor of its obligations under any of the Finance Documents;
- (b) be bound to account to any other Party for any sum or the profit element of any sum received by it for its own account;
- (c) be bound to disclose to any other person (including any Finance Party) (i) any confidential information or (ii) any other information if disclosure would, or might in its reasonable opinion, constitute a breach of any law or be a breach of fiduciary duty;
- (d) be under any obligations other than those which are specifically provided for in the Finance Documents; or

- 124
- (e) have or be deemed to have any duty, obligation or responsibility to, or relationship of trust or agency with, any Obligor.

28.8. Exclusion of Security Trustee's liability

Unless caused directly by its gross negligence or wilful default, the Security Trustee shall not accept responsibility or be liable for:

- (a) the adequacy, accuracy and/or completeness of any information supplied by the Security Trustee or any other person in connection with the Finance Documents or the transactions contemplated in the Finance Documents, or any other agreement, arrangement or document entered into, made or executed in anticipation of, under or in connection with the Finance Documents;
- (b) the legality, validity, effectiveness, adequacy or enforceability of any Finance Document or the Transaction Security or any other agreement, arrangement or document entered into, made or executed in anticipation of, under or in connection with any Finance Document or the Transaction Security;
- (c) any losses to any person or any liability arising as a result of taking or refraining from taking any action in relation to any of the Finance Documents or the Transaction Security or otherwise, whether in accordance with an instruction from the Agent or otherwise;
- (d) the exercise of, or the failure to exercise, any judgment, discretion or power given to it by or in connection with any of the Finance Documents, the Transaction Security or any other agreement, arrangement or document entered into, made or executed in anticipation of, under or in connection with the Finance Documents or the Transaction Security; or
- (e) any shortfall which arises on the enforcement of the Transaction Security.

28.9. Own responsibility

It is understood and agreed by each Finance Party that that Finance Party has at all times itself been, and will continue to be, solely responsible for making its own independent appraisal of and investigation into all risks arising under or in connection with the Finance Documents including but not limited to:

- (a) the financial condition, creditworthiness, condition, affairs, status and nature of each of the Obligors;
- (b) the legality, validity, effectiveness, adequacy and enforceability of each of the Finance Documents and the Transaction Security and any other agreement, arrangement or document entered into, made or executed in anticipation of, under or in connection with the Finance Documents or the Transaction Security;
- (c) whether that Finance Party has recourse, and the nature and extent of that recourse, against any Obligor or any other person or any of their respective assets under or in connection with the Finance Documents, the transactions contemplated in the Finance Documents or any other agreement, arrangement or document entered into, made or executed in anticipation of, under or in connection with the Finance Documents;
- (d) the adequacy, accuracy and/or completeness of any information provided by any person in connection with the Finance Documents, the transactions contemplated in the Finance Documents or any other agreement, arrangement or document entered into, made or executed in anticipation of, under or in connection with the Finance Documents; and

(e) the right or title of any person in or to, or the value or sufficiency of any part of, the Charged Property, the priority of any of the Transaction Security or the existence of any security interest affecting the Charged Property,

and each Finance Party warrants to the Security Trustee that it has not relied on, and will not at any time rely on, the Security Trustee in respect of any of these matters.

28.10. No responsibility to perfect Transaction Security

The Security Trustee shall not be liable for any failure to:

- (a) require the deposit with it of any deed or document certifying, representing or constituting the title of any Obligor to any of the Charged Property;
- (b) obtain any licence, consent or other authority for the execution, delivery, legality, validity, enforceability or admissibility in evidence of any of the Finance Documents or the Transaction Security;
- (c) register, file or record or otherwise protect any of the Transaction Security (or the priority of any of the Transaction Security) under any applicable laws in any jurisdiction or to give notice to any person of the execution of any of the Finance Documents or of the Transaction Security;
- (d) take, or to require any of the Obligors to take, any steps to perfect its title to any of the Charged Property or to render the Transaction Security effective or to secure the creation of any ancillary security interest under the laws of any jurisdiction; or
- (e) require any further assurances in relation to any of the Transaction Security Documents.

28.11. Insurance by Security Trustee

- (a) The Security Trustee shall not be under any obligation to insure any of the Charged Property, to require any other person to maintain any insurance or to verify any obligation to arrange or maintain insurance contained in the Finance Documents. The Security Trustee shall not be responsible for any loss which may be suffered by any person as a result of the lack of or inadequacy of any such insurance.
- (b) Where the Security Trustee is named on any insurance policy as an insured party, it shall not be responsible for any loss which may be suffered by reason (directly or indirectly) of its failure to notify the insurers of any material fact relating to the risk assumed by the insurers or any other information of any kind, unless any Finance Party has requested it to do so in writing and the Security Trustee has failed to do so within fourteen days after receipt of that request.

28.12. Custodians and Nominees

The Security Trustee may appoint and pay any person to act as a custodian or nominee on any terms in relation to any assets of the trust as the Security Trustee may determine, including for the purpose of depositing with a custodian this Agreement or any document relating to the trust created under this Agreement and the Security Trustee shall not be responsible for any loss, liability, expense, demand, cost, claim or proceedings incurred by reason of the misconduct, omission or default on the part of any person appointed by it under this Agreement or be bound to supervise the proceedings or acts of any person.

28.13. Acceptance of Title

The Security Trustee shall be entitled to accept without enquiry, and shall not be obliged to investigate, the right and title as each of the Obligors may have to any of the Charged Property and shall not be liable for or bound to require any Obligor to remedy any defect in its right or title.

28.14. Refrain from Illegality

The Security Trustee may refrain from doing anything which in its opinion will or may be contrary to any relevant law, directive or regulation of any jurisdiction which would or might otherwise render it liable to any person, and the Security Trustee may do anything which is, in its opinion, necessary to comply with any law, directive or regulation.

28.15. Business with the Obligors

The Security Trustee may accept deposits from, lend money to, and generally engage in any kind of banking or other business with any of the Obligors.

28.16. Releases

Upon a disposal of any of the Charged Property:

- (a) pursuant to the enforcement of the Transaction Security by a Receiver or the Security Trustee; or
- (b) if that disposal is permitted under the Finance Documents,

the Security Trustee shall (at the cost of the Obligors) release that property from the Transaction Security and is authorised to execute, without the need for any further authority from the Finance Parties, any release of the Transaction Security or other claim over that asset and to issue any certificates of non-crystallisation of floating charges that may be required or desirable.

28.17. Winding up of Trust

If the Security Trustee, with the approval of the Majority Lenders, determines that (a) all of the Secured Obligations and all other obligations secured by any of the Transaction Security Documents have been fully and finally discharged and (b) none of the Finance Parties is under any commitment, obligation or liability (actual or contingent) to make advances or provide other financial accommodation to any Obligor pursuant to the Finance Documents, the trusts set out in this Agreement shall be wound up and the Security Trustee shall release, without recourse or warranty, all of the Transaction Security and the rights of the Security Trustee under each of the Transaction Security Documents.

28.18. Perpetuity Period

The perpetuity period under the rule against perpetuities, if applicable to this Agreement, shall be the period of one hundred and twenty five years from the date of this Agreement.

28.19. Powers Supplemental

The rights, powers and discretions conferred upon the Security Trustee by this Agreement shall be supplemental to the Trustee Acts 1925 and 2000 and in addition to any which may be vested in the Security Trustee by general law or otherwise.

28.20. Trustee division separate

In acting as trustee for the Finance Parties, the Security Trustee shall be regarded as acting through its trustee division which shall be treated as a separate entity from any of its other divisions or departments and any information received by any other division or department of the Security Trustee may be treated as confidential and shall not be regarded as having been given to the Security Trustee's trustee division.

28.21. Disapplication

Section 1 of the Trustee Act 2000 shall not apply to the duties of the Security Trustee in relation to the trusts constituted by this Agreement. Where there are any inconsistencies between the Trustee Acts 1925 and 2000 and the provisions of this Agreement, the provisions of this Agreement shall, to the extent allowed by law, prevail and, in the case of any inconsistency with the Trustee Act 2000, the provisions of this Agreement shall constitute a restriction or exclusion for the purposes of that Act.

28.22. Resignation of Security Trustee

- (a) The Security Trustee may resign and appoint one of its Affiliates as successor by giving notice to the other Parties (or to the Agent on behalf of the Lenders).
- (b) Alternatively the Security Trustee may resign by giving notice to the other Parties (or to the Agent on behalf of the Lenders) in which case the Majority Lenders may appoint a successor Security Trustee.
- (c) If the Majority Lenders have not appointed a successor Security Trustee in accordance with paragraph (b) above within 30 days after the notice of resignation was given, the Security Trustee (after consultation with the Agent) may appoint a successor Security Trustee.
- (d) The retiring Security Trustee shall, at its own cost, make available to the successor Security Trustee such documents and records and provide such assistance as the successor Security Trustee may reasonably request for the purposes of performing its functions as Security Trustee under the Finance Documents.
- (e) The Security Trustee's resignation notice shall only take effect upon (i) the appointment of a successor and (ii) the transfer of all of the Transaction Security to that successor.
- (f) Upon the appointment of a successor, the retiring Security Trustee shall be discharged from any further obligation in respect of the Finance Documents but shall remain entitled to the benefit of Clauses 27 (Role of the Agent, the Arranger and others) and this Clause 28 (Role of Security Trustee). Its successor and each of the other Parties shall have the same rights and obligations amongst themselves as they would have had if such successor had been an original Party.

- 128
- (g) The Majority Lenders may, by notice to the Security Trustee, require it to resign in accordance with paragraph (b) above. In this event, the Security Trustee shall resign in accordance with paragraph (b) above.

28.23. Delegation

- (a) The Security Trustee may, at any time, delegate by power of attorney or otherwise to any person for any period, all or any of the rights, powers and discretions vested in it by any of the Finance Documents.
- (b) The delegation may be made upon any terms and conditions (including the power to sub-delegate) and subject to any restrictions as the Security Trustee may think fit in the interests of the Finance Parties and it shall not be bound to supervise, or be in any way responsible for any loss incurred by reason of any misconduct or default on the part of any delegate or sub-delegate.

28.24. Additional trustees

- (a) The Security Trustee may at any time appoint (and subsequently remove) any person to act as a separate trustee or as a co-trustee jointly with it (i) if it considers that appointment to be in the interests of the Finance Parties or (ii) for the purposes of conforming to any legal requirements, restrictions or conditions which the Security Trustee deems to be relevant or (iii) for obtaining or enforcing any judgment in any jurisdiction, and the Security Trustee shall give prior notice to the Parent and the Agent of that appointment.
- (b) Any person so appointed shall have the rights, powers and discretions (not exceeding those conferred on the Security Trustee by this Agreement) and the duties and obligations that are conferred or imposed by the instrument of appointment.
- (c) The remuneration that the Security Trustee may pay to any person, and any costs and expenses incurred by that person in performing its functions pursuant to that appointment shall, for the purposes of this Agreement, be treated as costs and expenses incurred by the Security Trustee.

29. CONDUCT OF BUSINESS BY THE FINANCE PARTIES

No provision of this Agreement will:

- (a) interfere with the right of any Finance Party to arrange its affairs (tax or otherwise) in whatever manner it thinks fit;
- (b) oblige any Finance Party to investigate or claim any credit, relief, remission or repayment available to it or the extent, order and manner of any claim; or
- (c) oblige any Finance Party to disclose any information relating to its affairs (tax or otherwise) or any computations in respect of Tax.

30. SHARING AMONG THE FINANCE PARTIES

30.1. Payments to Finance Parties

If a Finance Party (a "**Recovering Finance Party**") receives or recovers any amount from an Obligor other than in accordance with Clause 31 (Payment mechanics) and applies that amount to a payment due under the Finance Documents then:

- 129
- (a) the Recovering Finance Party shall, within three Business Days, notify details of the receipt or recovery, to the Agent;
- (b) the Agent shall determine whether the receipt or recovery is in excess of the amount the Recovering Finance Party would have been paid had the receipt or recovery been received or made by the Agent and distributed in accordance with Clause 31 (Payment mechanics), without taking account of any Tax which would be imposed on the Agent in relation to the receipt, recovery or distribution; and
- (c) the Recovering Finance Party shall, within three Business Days of demand by the Agent, pay to the Agent an amount (the "Sharing Payment") equal to such receipt or recovery less any amount which the Agent determines may be retained by the Recovering Finance Party as its share of any payment to be made, in accordance with Clause 31.6 (Partial payments).

30.2. Redistribution of payments

The Agent shall treat the Sharing Payment as if it had been paid by the relevant Obligor and distribute it between the Finance Parties (other than the Recovering Finance Party) in accordance with Clause 31.6 (Partial payments).

30.3. Recovering Finance Party's rights

- (a) On a distribution by the Agent under Clause 30.2 (Redistribution of payments), the Recovering Finance Party will be subrogated to the rights of the Finance Parties which have shared in the redistribution.
- (b) If and to the extent that the Recovering Finance Party is not able to rely on its rights under paragraph (a) above, the relevant Obligor shall be liable to the Recovering Finance Party for a debt equal to the Sharing Payment which is immediately due and payable.

30.4. Reversal of redistribution

If any part of the Sharing Payment received or recovered by a Recovering Finance Party becomes repayable and is repaid by that Recovering Finance Party, then:

- (a) each Finance Party which has received a share of the relevant Sharing Payment pursuant to Clause 30.2 (Redistribution of payments) shall, upon request of the Agent, pay to the Agent for account of that Recovering Finance Party an amount equal to the appropriate part of its share of the Sharing Payment (together with an amount as is necessary to reimburse that Recovering Finance Party for its proportion of any interest on the Sharing Payment which that Recovering Finance Party is required to pay); and
- (b) that Recovering Finance Party's rights of subrogation in respect of any reimbursement shall be cancelled and the relevant Obligor will be liable to the reimbursing Finance Party for the amount so reimbursed.

30.5. Exceptions

- (a) This Clause 30 shall not apply to the extent that the Recovering Finance Party would not, after making any payment pursuant to this Clause, have a valid and enforceable claim against the relevant Obligor.
- (b) A Recovering Finance Party is not obliged to share with any other Finance Party any amount which the Recovering Finance Party has received or recovered as a result of taking legal or arbitration proceedings, if:

- 130
- (i) it notified the other Finance Party of the legal or arbitration proceedings; and
- (ii) the other Finance Party had an opportunity to participate in those legal or arbitration proceedings but did not do so as soon as reasonably practicable having received notice and did not take separate legal or arbitration proceedings.

SECTION 11

ADMINISTRATION

31. PAYMENT MECHANICS

31.1. Payments to the Agent

- (a) On each date on which an Obligor or a Lender is required to make a payment under a Finance Document, that Obligor or Lender shall make the same available to the Agent (unless a contrary indication appears in a Finance Document) for value on the due date at the time and in such funds specified by the Agent as being customary at the time for settlement of transactions in the relevant currency in the place of payment.
- (b) Payment shall be made to such account in Edinburgh as the Agent specifies.

31.2. Distributions by the Agent

Each payment received by the Agent under the Finance Documents for another Party shall, subject to Clause 31.3 (Distributions to an Obligor) and Clause 31.4 (Clawback) be made available by the Agent as soon as practicable after receipt to the Party entitled to receive payment in accordance with this Agreement (in the case of a Lender, for the account of its Facility Office), to such account as that Party may notify to the Agent by not less than five Business Days' notice with a bank in London.

31.3. Distributions to an Obligor

The Agent may (with the consent of the Obligor or in accordance with Clause 32 (Set-Off)) apply any amount received by it for that Obligor in or towards payment (on the date and in the currency and funds of receipt) of any amount due from that Obligor under the Finance Documents or in or towards purchase of any amount of any currency to be so applied.

31.4. Clawback and pre-funding

- (a) Where a sum is to be paid to the Agent under the Finance Documents for another Party, the Agent is not obliged to pay that sum to that other Party (or to enter into or perform any related exchange contract) until it has been able to establish to its satisfaction that it has actually received that sum.
- (b) Unless paragraph (c) below applies, if the Agent pays an amount to another Party and it proves to be the case that the Agent had not actually received that amount, then the Party to whom that amount (or the proceeds of any related exchange contract) was paid by the Agent shall on demand refund the same to the Agent together with interest on that amount from the date of payment to the date of receipt by the Agent, calculated by the Agent to reflect its cost of funds.
- (c) If the Agent is willing to make available amounts for the account of a Borrower before receiving funds from the Lenders then if and to the extent that the Agent does so but it proves to be the case that it does not then receive funds from a Lender in respect of a sum which it paid to a Borrower:
 - (i) the Agent shall notify the Parent of that Lender's identity and the Borrower to whom that sum was made available shall on demand refund it to the Agent; and
 - (ii) the Lender by whom those funds should have been made available or, if that Lender fails to do so, the Borrower to whom that sum was made available,

shall on demand pay to the Agent the amount (as certified by the Agent) which will indemnify the Agent against any funding cost incurred by it as a result of paying out that sum before receiving those funds from that Lender.

31.5. Impaired Agent

- (a) If, at any time, the Agent becomes an Impaired Agent, an Obligor or a Lender which is required to make a payment under the Finance Documents to the Agent in accordance with Clause 31.1 (Payments to the Agent) may instead either:
 - (i) pay that amount direct to the required recipient(s); or
 - (ii) if it its absolute discretion it considers that it is not reasonable practicable to pay that amount direct to the required recipient(s), pay that amount or the relevant part of that amount to an interest-bearing account held with an Acceptable Bank within the meaning of paragraph (a) of the definition of "Acceptable Bank" and in relation to which no Insolvency Event has occurred and is continuing, in the name of the Obligor or the Lender making the payment (the "Paying Party") and designated as a trust account for the benefit of the Party or Parties beneficially entitled to that payment under the Finance Documents (the "Recipient Party") or "Recipient Parties"). In each case such payments must be made on the due date for payment under the Finance Documents.
- (b) All interest accrued on the amount standing to the credit of the trust account shall be for the benefit of Recipient Party or the Recipient Parties pro rata to their respective entitlements.
- (c) A Party which has made a payment in accordance with this Clause 31.5 shall be discharged of the relevant payment obligation under the Finance Documents and shall not take any credit risk with respect to the amounts standing to the credit of the trust account.
- (d) Promptly upon the appointment of a successor Agent in accordance with Clause 27.14 (Replacement of the Agent), each Paying Party shall (other to the extent that that Party has given an instruction pursuant to paragraph (e) below) give all requisite instructions to the bank with whom the trust account is held to transfer the amount (together with any accrued interest) to the successor Agent for distribution to the relevant Recipient Party or Recipient Parties in accordance with Clause 31.2 (Distributions by the Agent).
- (e) A Paying Party shall, promptly upon request by a Recipient Party and to the extent:
 - (i) that it has not given an instruction pursuant to paragraph (d) above; and
 - (ii) that it has been provided with the necessary information by that Recipient Party,

give all requisite instructions to the bank with whom the trust account is held to transfer the relevant amount (together with any accrued interest) to that Recipient Party.

31.6. Partial payments

(a) If the Agent receives a payment for application against amounts due in respect of any Finance Documents that is insufficient to discharge all the amounts then due and payable by an Obligor under those Finance Documents, the Agent shall apply that payment towards the obligations of that Obligor under those Finance Documents in the following order:

132

- 133
- (i) first, in or towards payment pro rata of any unpaid fees, costs and expenses of the Agent and the Security Trustee under those Finance Documents;
- (ii) secondly, in or towards payment pro rata of any accrued interest, fee or commission due but unpaid under those Finance Documents;
- (iii) thirdly, in or towards payment pro rata of any principal due but unpaid under those Finance Documents; and
- (iv) fourthly, in or towards payment pro rata of any other sum due but unpaid under the Finance Documents.
- (b) The Agent shall, if so directed by the Majority Lenders, vary the order set out in paragraphs (a)(ii) to (iv) above.
- (c) Paragraphs (a) and (b) above will override any appropriation made by an Obligor.

31.7. No set-off by Obligors

All payments to be made by an Obligor under the Finance Documents shall be calculated and be made without (and free and clear of any deduction for) set-off or counterclaim.

31.8. Business Days

- (a) Any payment under the Finance Documents which is due to be made on a day that is not a Business Day shall be made on the next Business Day in the same calendar month (if there is one) or the preceding Business Day (if there is not).
- (b) During any extension of the due date for payment of any principal or Unpaid Sum under this Agreement interest is payable on the principal or Unpaid Sum at the rate payable on the original due date.

31.9. Currency of account

Sterling is the currency of account and payment for any sum due from an Obligor under any Finance Document.

31.10. Change of currency

- (a) Unless otherwise prohibited by law, if more than one currency or currency unit are at the same time recognised by the central bank of any country as the lawful currency of that country, then:
 - (i) any reference in the Finance Documents to, and any obligations arising under the Finance Documents in, the currency of that country shall be translated into, or paid in, the currency or currency unit of that country designated by the Agent (after consultation with the Parent); and
 - (ii) any translation from one currency or currency unit to another shall be at the official rate of exchange recognised by the central bank for the conversion of that currency or currency unit into the other, rounded up or down by the Agent (acting reasonably).
- (b) If a change in any currency of a country occurs, this Agreement will, to the extent the Agent (acting reasonably and after consultation with the Parent) specifies to be necessary, be amended to comply with any generally accepted conventions and market practice in the relevant interbank market and otherwise to reflect the change in currency.

31.11. Disruption to Payment Systems, etc.

If either the Agent determines (in its discretion) that a Disruption Event has occurred or the Agent is notified by the Parent that a Disruption Event has occurred:

- (a) the Agent may, and shall if requested to do so by the Parent, consult with the Parent with a view to agreeing with the Parent such changes to the operation or administration of the Facilities as the Agent may deem necessary in the circumstances;
- (b) the Agent shall not be obliged to consult with the Parent in relation to any changes mentioned in paragraph (a) if, in its opinion, it is not practicable to do so in the circumstances and, in any event, shall have no obligation to agree to such changes;
- (c) the Agent may consult with the Finance Parties in relation to any changes mentioned in paragraph (a) but shall not be obliged to do so if, in its opinion, it is not practicable to do so in the circumstances;
- (d) any such changes agreed upon by the Agent and the Parent shall (whether or not it is finally determined that a Disruption Event has occurred) be binding upon the Parties as an amendment to (or, as the case may be, waiver of) the terms of the Finance Documents notwithstanding the provisions of Clause 37 (Amendments and Waivers);
- (e) the Agent shall not be liable for any damages, costs or losses to any person, any diminution in value or any liability whatsoever (including, without limitation for negligence, gross negligence or any other category of liability whatsoever but not including any claim based on the fraud of the Agent) arising as a result of its taking, or failing to take, any actions pursuant to or in connection with this Clause 31.11; and
- (f) the Agent shall notify the Finance Parties of all changes agreed pursuant to paragraph (d) above.

32. SET-OFF

A Finance Party may set off any matured obligation due from an Obligor under the Finance Documents (to the extent beneficially owned by that Finance Party) against any matured obligation owed by that Finance Party to that Obligor, regardless of the place of payment, booking branch or currency of either obligation. If the obligations are in different currencies, the Finance Party may convert either obligation at a market rate of exchange in its usual course of business for the purpose of the set-off.

33. NOTICES

33.1. Communications in writing

Any communication to be made under or in connection with the Finance Documents shall be made in writing and, unless otherwise stated, may be made by fax or letter.

33.2. Addresses

The address and fax number (and the department or officer, if any, for whose attention the communication is to be made) of each Party for any communication or document to be made or delivered under or in connection with the Finance Documents is:

(a) in the case of the Parent, Holdings, Schuh (ROI) Limited or the Company, that identified with its name below;

- (b) in the case of each Lender or any other Obligor, that notified in writing to the Agent on or prior to the date on which it becomes a Party; and
- (c) in the case of the Agent or the Security Trustee, that identified with its name below,

or any substitute address, fax number or department or officer as the Party may notify to the Agent (or the Agent may notify to the other Parties, if a change is made by the Agent) by not less than five Business Days' notice.

33.3. Delivery

- (a) Any communication or document made or delivered by one person to another under or in connection with the Finance Documents will only be effective:
 - (i) if by way of fax, when received in legible form; or
 - (ii) if by way of letter, when it has been left at the relevant address or five Business Days after being deposited in the post postage prepaid in an envelope addressed to it at that address,

and, if a particular department or officer is specified as part of its address details provided under Clause 33.2 (Addresses), if addressed to that department or officer.

- (b) Any communication or document to be made or delivered to the Agent or the Security Trustee will be effective only when actually received by the Agent or Security Trustee and then only if it is expressly marked for the attention of the department or officer identified with the Agent's or Security Trustee's signature below (or any substitute department or officer as the Agent or Security Trustee shall specify for this purpose).
- (c) All notices from or to an Obligor shall be sent through the Agent.
- (d) Any communication or document made or delivered to the Parent in accordance with this Clause 33.3 will be deemed to have been made or delivered to each of the Obligors.
- (e) Any communication or document which becomes effective, in accordance with paragraphs (a) to (d) above, after 5.00 p.m. in the place of receipt shall be deemed only to become effective on the following day.

33.4. Notification of address and fax number

Promptly upon receipt of notification of an address, and fax number or change of address or fax number pursuant to Clause 33.2 (Addresses) or changing its own address or fax number, the Agent shall notify the other Parties.

33.5. Communication when Agent is Impaired Agent

If the Agent is an Impaired Agent the Parties may, instead of communicating with each other through the Agent, communicate with each other directly and (while the Agent is an Impaired Agent) all the provisions of the Finance Documents which require communications to be made or notices to be given to or by the Agent shall be varied so that communications may be made and notices given to or by the relevant Parties directly. This provision shall not operate after a replacement Agent has been appointed unless such replacement Agent becomes an Impaired Agent.

33.6. Electronic communication

(a) Any communication to be made between the Agent or the Security Trustee and a Lender under or in connection with the Finance Documents may be made by electronic mail or other electronic means, if the Agent, the Security Trustee and the relevant Lender:

- (i) agree that, unless and until notified to the contrary, this is to be an accepted form of communication;
- (ii) notify each other in writing of their electronic mail address and/or any other information required to enable the sending and receipt of information by that means; and
- (iii) notify each other of any change to their address or any other such information supplied by them.
- (b) Any electronic communication made between the Agent and a Lender or the Security Trustee will be effective only when actually received in readable form and in the case of any electronic communication made by a Lender to the Agent or the Security Trustee only if it is addressed in such a manner as the Agent or Security Trustee shall specify for this purpose.
- (c) Any electronic communication which becomes effective, in accordance with paragraphs (a) to (d) above, after 5.00 p.m. in the place of receipt shall be deemed only to become effective on the following day.

33.7. English language

- (a) Any notice given under or in connection with any Finance Document must be in English.
- (b) All other documents provided under or in connection with any Finance Document must be:
 - (i) in English; or
 - (ii) if not in English, and if so required by the Agent, accompanied by a certified English translation and, in this case, the English translation will prevail unless the document is a constitutional, statutory or other official document.

34. CALCULATIONS AND CERTIFICATES

34.1. Accounts

In any litigation or arbitration proceedings arising out of or in connection with a Finance Document, the entries made in the accounts maintained by a Finance Party are prima facie evidence of the matters to which they relate.

34.2. Certificates and determinations

Any certification or determination by a Finance Party of a rate or amount under any Finance Document is, in the absence of manifest error, conclusive evidence of the matters to which it relates.

34.3. Day count convention

Any interest, commission or fee accruing under a Finance Document will accrue from day to day and is calculated on the basis of the actual number of days elapsed and a year of 365 days.

35. PARTIAL INVALIDITY

If, at any time, any provision of a Finance Document is or becomes illegal, invalid or unenforceable in any respect under any law of any jurisdiction, neither the legality, validity or

enforceability of the remaining provisions nor the legality, validity or enforceability of such provision under the law of any other jurisdiction will in any way be affected or impaired.

36. REMEDIES AND WAIVERS

No failure to exercise, nor any delay in exercising, on the part of any Finance Party or Secured Party, any right or remedy under a Finance Document shall operate as a waiver of any such right or remedy or constitute an election to affirm any Finance Document. No election to affirm any Finance Document on the part of any Finance Party or Secured Party shall be effective unless it is in writing. No single or partial exercise of any right or remedy prevent any further or other exercise or the exercise of any other right or remedy. The rights and remedies provided in each Finance Document are cumulative and not exclusive of any rights or remedies provided by law.

37. AMENDMENTS AND WAIVERS

37.1. Required consents

- (a) Subject to Clause 37.2 (Exceptions) any term of the Finance Documents may be amended or waived only with the consent of the Majority Lenders and the Parent and any such amendment or waiver will be binding on all Parties.
- (b) The Agent may effect, on behalf of any Finance Party, any amendment or waiver permitted by this Clause 37.
- (c) Without prejudice to the generality of paragraphs (c), (d) and (e) of Clause 27.7 (Rights and Discretions), the Agent may engage, pay for and rely on the services of lawyers in determining the consent level required for and effecting any amendment, waiver or consent under this Agreement.
- (d) Each Obligor agrees to any such amendment or waiver permitted by this Clause 37 which is agreed to by the Parent. This includes any amendment or waiver which would, but for this paragraph (c), require the consent of all of the Guarantors.

37.2. Exceptions

- (a) An amendment or waiver that has the effect of changing or which relates to:
 - (i) the definition of "**Majority Lenders**" in Clause 1.1 (Definitions);
 - (ii) an extension to the date of payment of any amount under the Finance Documents;
 - (iii) a reduction in the Margin or a reduction in the amount of any payment of principal, interest, fees or commission payable;
 - (iv) a change in currency of payment of any amount under the Finance Documents;
 - (v) an increase in any Commitment or the Total Commitments, any extension of any Availability Period or any requirement that a cancellation of Commitments reduces the Commitments of the Lenders rateably under the relevant Facility;
 - (vi) a change to the Borrowers or Guarantors other than in accordance with Clause 26 (Changes to the Obligors);
 - (vii) any provision which expressly requires the consent of all the Lenders;

- 138
- (viii) Clause 2.3 (Finance Parties' rights and obligations), Clause 8 (Mandatory prepayment), Clause 25 (Changes to the Lenders), this Clause 37, Clause 40 (Governing law) or 41.1 (Jurisdiction of English courts);
- (ix) the nature or scope of:
 - (1) the guarantee and indemnity granted under Clause 19 (Guarantee and Indemnity);
 - (2) the Charged Property; or
 - (3) the manner in which the proceeds of enforcement of the Transaction Security are distributed (except in the case of paragraphs (ii) and (iii) above, insofar as it relates to a sale or disposal of an asset which is the subject of the Transaction Security where such sale or disposal is expressly permitted under this Agreement or any other Finance Document); or
- (x) the release of any Transaction Security unless permitted under this Agreement or any other Finance Document or relating to a sale or disposal of an asset which is the subject of the Transaction Security where such sale or disposal is expressly permitted under this Agreement or any other Finance Document,

shall not be made without the prior consent of all the Lenders.

(b) An amendment or waiver which relates to the rights or obligations of the Agent, the Arranger or the Security Trustee may not be effected without the consent of the Agent, the Arranger or the Security Trustee.

37.3. Replacement of a Lender

- (a) If at any time:
 - (i) any Lender becomes a Non-Consenting Lender (as defined in paragraph (c) below); or
 - (ii) an Obligor becomes obliged to repay any amount in accordance with Clause 7.1 (Illegality) or to pay additional amounts pursuant to Clause 15.1 (Increased Costs) or Clause 14.2 (Tax gross-up) to any Lender in excess of amounts payable to the other Lenders generally,

then the Parent may, on 5 Business Days' prior written notice to the Agent and such Lender, replace such Lender by requiring such Lender to (and such Lender shall) transfer pursuant to Clause 25 (Changes to the Lenders) all (and not part only) of its rights and obligations under this Agreement to a Lender or other bank, financial institution, trust, fund or other entity (a "**Replacement Lender**") selected by the Parent, and which is acceptable to the Agent (acting reasonably) which confirms its willingness to assume and does assume all the obligations of the transferring Lender (including the assumption of the transferring Lender's participations on the same basis as the transferring Lender) for a purchase price in cash payable at the time of transfer equal to the outstanding principal amount of such Lender's participation in the outstanding Utilisations and all accrued interest and/or Break Costs and other amounts payable in relation thereto under the Finance Documents.

- 139
- (b) The replacement of a Lender pursuant to this Clause shall be subject to the following conditions:
 - (i) the Parent shall have no right to replace the Agent or Security Agent;
 - (ii) neither the Agent nor the Lender shall have any obligation to the Parent to find a Replacement Lender;
 - (iii) in the event of a replacement of a Non-Consenting Lender such replacement must take place no later than 90 days after the date the Non-Consenting Lender notifies the Parent and the Agent of its failure or refusal to give a consent in relation to, or agree to any waiver or amendment to the Finance Documents requested by the Parent; and
 - (iv) in no event shall the Lender replaced under this paragraph (b) be required to pay or surrender to such Replacement Lender any of the fees received by such Lender pursuant to the Finance Documents.
- (c) In the event that:
 - (i) the Parent or the Agent (at the request of the Parent) has requested the Lenders to give a consent in relation to, or to agree to a waiver or amendment of, any provisions of the Finance Documents;
 - (ii) the consent, waiver or amendment in question requires the approval of all the Lenders; and
 - (iii) Lenders whose Commitments aggregate more than 85 per cent. of the Total Commitments (or, if the Total Commitments have been reduced to zero, aggregated more than 85 per cent. of the Total Commitments prior to that reduction) have consented or agreed to such waiver or amendment,

then any Lender who does not and continues not to consent or agree to such waiver or amendment shall be deemed a "**Non-Consenting** Lender".

37.4. Disenfranchisement of Defaulting Lenders

- (a) For so long as a Defaulting Lender has any Available Commitment, in ascertaining the Majority Lenders or whether any given percentage (including, for the avoidance of doubt, unanimity) of the Total Commitments has been obtained to approve any request for a consent, waiver, amendment or other vote under the Finance Documents, that Defaulting Lender's Commitments will be reduced by the amount of its Available Commitments.
- (b) If a Defaulting Lender fails to respond to a request for a consent, waiver, amendment or other vote under the Finance Documents or any other vote of the Lenders under the terms of this Agreement within 10 Business Days in relation to consents, waivers, amendments or votes which require Majority Lender consent, and within 15 Business Days in relation to consents, waivers, amendments or votes which require all Lender consent (unless the Parent and the Agent agree to a longer time period) of that request being made, its Commitment and/or participation shall not be included for the purpose of calculating the Total Commitments or participations under the Facility when ascertaining whether any relevant percentage of Total Commitments and/or participations has been obtained to approve that request.
- (b) For the purposes of this Clause 37.4, the Agent may assume that the following Lenders are Defaulting Lenders:

- (1) any Lender which has notified the Agent that it has become a Defaulting Lender;
- (2) any Lender in relation to which it is aware that any of the events of circumstances referred to in paragraphs (a), (b) or (c) of the definition of "Defaulting Lender" has occurred and none of the exceptions in paragraphs (i) to (vi) of the definition of "Defaulting Lender" apply.

unless it has received notice to the contrary from the Lender concerned or the Agent is otherwise aware that the Lender has ceased to be a Defaulting Lender.

37.5. Replacement of a Defaulting Lender

- (a) The Parent may, at any time a Lender has become and continues to be a Defaulting Lender, by giving 5 Business Days' prior written notice to the Agent and such Lender, replace such Lender by requiring such Lender to (and such Lender shall) transfer pursuant to Clause 26 (Changes to the Lenders) all (and not part only) of its rights and obligations under this Agreement to a Lender or other bank, financial institution, trust, fund or other entity (a "**Replacement Lender**") selected by the Parent, and which (unless the Agent is an Impaired Agent) is acceptable to the Agent (acting reasonably), which confirms its willingness to assume and does assume all the obligations or all the relevant obligations of the transferring Lender (including the assumption of the transferring Lender's participations or unfunded participations (as the case may be) on the same basis as the transferring Lender) for a purchase price in cash payable at the time of transfer equal to the outstanding principal amount of such Lender's participation in the outstanding Utilisations and all accrued interest Break Costs and other amounts payable in relation thereto under the Finance Documents.
- (b) Any transfer of rights and obligations of a Defaulting Lender pursuant to this Clause shall be subject to the following conditions:
 - (i) the Parent shall have no right to replace the Agent or Security Trustee;
 - (ii) neither the Agent nor the Defaulting Lender shall have any obligation to the Parent to find a Replacement Lender;
 - (iii) the transfer must take place no later than 5 days after the notice referred to in paragraph (a) above; and
 - (iv) in no event shall the Defaulting Lender be required to pay or surrender to the Replacement Lender any of the fees received by the Defaulting Lender pursuant to the Finance Documents.

38. CONFIDENTIALITY

38.1. Confidential Information

Each Finance Party agrees to keep all Confidential Information confidential and not to disclose it to anyone, save to the extent permitted by Clause 38.2 (Disclosure of Confidential Information) and Clause 38.3 (Disclosure to numbering service providers), and to ensure that all Confidential Information is protected with security measures and a degree of care that would apply to its own confidential information.

140

38.2. Disclosure of Confidential Information

Any Finance Party may disclose:

- (a) to any of its Affiliates and Related Funds and any of its or their officers, directors, employees, professional advisers, auditors, partners and Representatives such Confidential Information as that Finance Party shall consider appropriate if any person to whom the Confidential Information is to be given pursuant to this paragraph (a) is informed in writing of its confidential nature and that some or all of such Confidential Information may be price-sensitive information except that there shall be no such requirement to so inform if the recipient is subject to professional obligations to maintain the confidentiality of the information or is otherwise bound by requirements of confidentiality in relation to the Confidential Information;
- (b) to any person:
 - to (or through) whom it assigns or transfers (or may potentially assign or transfer) all or any of its rights and/or obligations under one or more Finance Documents or which succeeds (or which may potentially succeed) it as Agent or Security Trustee and, in each case, and to any of that person's Affiliates, Related Funds, Representatives and professional advisers;
 - (ii) with (or through) whom it enters into (or may potentially enter into), whether directly or indirectly, any sub-participation in relation to, or any other transaction under which payments are to be made or may be made by reference to, one or more Finance Documents and/or one or more Obligors and to any of that person's Affiliates, Related Funds, Representatives and professional advisers;
 - (iii) appointed by any Finance Party or by a person to whom paragraph (b)(i) or (ii) above applies to receive communications, notices, information or documents delivered pursuant to the Finance Documents on its behalf (including, without limitation, any person appointed under paragraph (d) of Clause 27.16 (Relationship with the Lenders));
 - (iv) who invests in or otherwise finances (or may potentially invest in or otherwise finance), directly or indirectly, any transaction referred to in paragraph (b)(i) or (b)(ii) above;
 - to whom information is required or requested to be disclosed by any court of competent jurisdiction or any governmental, banking, taxation or other regulatory authority or similar body, the rules of any relevant stock exchange or pursuant to any applicable law or regulation;
 - (vi) to whom or for whose benefit that Finance Party charges, assigns or otherwise creates Security (or may do so) pursuant to Clause 25.9 (Security over Lenders' rights);
 - (vii) to whom information is required to be disclosed in connection with, and for the purposes of, any litigation, arbitration, administrative or other investigations, proceedings or disputes;
 - (viii) who is a Party; or
 - (ix) with the consent of the Company;

in each case, such Confidential Information as that Finance Party shall consider appropriate if:

- (A) in relation to paragraphs (b)(i), (b)(ii) and b(iii) above, the person to whom the Confidential Information is to be given has entered into a Confidentiality Undertaking except that there shall be no requirement for a Confidentiality Undertaking if the recipient is a professional adviser and is subject to professional obligations to maintain the confidentiality of the Confidential Information;
- (B) in relation to paragraph (b)(iv) above, the person to whom the Confidential Information is to be given has entered into a Confidentiality Undertaking or is otherwise bound by requirements of confidentiality in relation to the Confidential Information they receive and is informed that some or all of such Confidential Information may be price-sensitive information;
- (C) in relation to paragraphs (b)(v), (b)(vi) and (b)(vii) above, the person to whom the Confidential Information is to be given is informed of its confidential nature and that some or all of such Confidential Information may be price-sensitive information except that there shall be no requirement to so inform if, in the opinion of that Finance Party, it is not practicable so to do in the circumstances;
- (c) to any person appointed by that Finance Party or by a person to whom paragraph (b)(i) or (b)(ii) above applies to provide administration or settlement services in respect of one or more of the Finance Documents including without limitation, in relation to the trading of participations in respect of the Finance Documents, such Confidential Information as may be required to be disclosed to enable such service provider to provide any of the services referred to in this paragraph (c) if the service provider to whom the Confidential Information is to be given has entered into a confidentiality agreement substantially in the form of the LMA Master Confidentiality Undertaking for Use With Administration/Settlement Service Providers or such other form of confidentiality undertaking agreed between the Parent and the relevant Finance Party;
- (d) to any rating agency (including its professional advisers) such Confidential Information as may be required to be disclosed to enable such rating agency to carry out its normal rating activities in relation to the Finance Documents and/or the Obligors if the rating agency to whom the Confidential Information is to be given is informed of its confidential nature and that some or all of such Confidential Information may be price-sensitive information.

38.3. Disclosure to numbering service providers

- (a) Any Finance Party may disclose to any national or international numbering service provider appointed by that Finance Party to provide identification numbering services in respect of this Agreement, the Facilities and/or one or more Obligors the following information:
 - (i) names of Obligors;
 - (ii) country of domicile of Obligors;
 - (iii) place of incorporation of Obligors;
 - (iv) date of this Agreement;
 - (v) Clause 40 (Governing law);
 - (vi) the names of the Agent and the Arranger;

142

- 143
- (vii) date of each amendment and restatement of this Agreement;
- (viii) amounts of, and names of, the Facilities (and any tranches);
- (ix) amount of Total Commitments;
- (x) currencies of the Facilities;
- (xi) type of Facilities;
- (xii) ranking of Facilities;
- (xiii) Termination Date for Facilities;
- (xiv) changes to any of the information previously supplied pursuant to paragraphs (i) to (xiii) above; and
- (xv) such other information agreed between such Finance Party and the Parent,

to enable such numbering service provider to provide its usual syndicated loan numbering identification services.

- (b) The Parties acknowledge and agree that each identification number assigned to this Agreement, the Facilities and/or one or more Obligors by a numbering service provider and the information associated with each such number may be disclosed to users of its services in accordance with the standard terms and conditions of that numbering service provider.
- (c) Each Obligor represents that none of the information set out in paragraphs (i) to (xiii) of paragraph (a) above is, nor will at any time be, unpublished price-sensitive information.
- (d) The Agent shall notify the Parent and the other Finance Parties of:
 - (i) the name of any numbering service provider appointed by the Agent in respect of this Agreement, the Facilities and/or one or more Obligors; and
 - (ii) the number or, as the case may be, numbers assigned to this Agreement, the Facilities and/or one or more Obligors by such numbering service provider.

38.4. Entire agreement

This Clause 38 (Confidentiality) constitutes the entire agreement between the Parties in relation to the obligations of the Finance Parties under the Finance Documents regarding Confidential Information and supersedes any previous agreement, whether express or implied, regarding Confidential Information.

38.5. Inside information

Each of the Finance Parties acknowledges that some or all of the Confidential Information is or may be price-sensitive information and that the use of such information may be regulated or prohibited by applicable legislation including securities law relating to insider dealing and market abuse and each of the Finance Parties undertakes not to use any Confidential Information for any unlawful purpose.

38.6. Notification of disclosure

Each of the Finance Parties agrees (to the extent permitted by law and regulation) to inform the Company:

- (a) of the circumstances of any disclosure of Confidential Information made pursuant to paragraph (b)(v) of Clause 38.2 (Disclosure of Confidential Information) except where such disclosure is made to any of the persons referred to in that paragraph during the ordinary course of its supervisory or regulatory function; and
- (b) upon becoming aware that Confidential Information has been disclosed in breach of this Clause 38 (Confidentiality).

38.7. Continuing obligations

The obligations in this Clause 38 (Confidentiality) are continuing and, in particular, shall survive and remain binding on each Finance Party for a period of twelve months from the earlier of:

- (a) the date on which all amounts payable by the Obligors under or in connection with the Finance Documents have been paid in full and all Commitments have been cancelled or otherwise cease to be available; and
- (b) the date on which such Finance Party otherwise ceases to be a Finance Party.

39. DISCLOSURE OF LENDER DETAILS BY AGENT

39.1. Supply of Lender details to Parent

The Agent shall provide to the Parent within 10 Business Days of a request by the Parent (but no more frequently than once per calendar month), a list (which may be in electronic form) setting out the names of the Lenders as at the date of that request, their respective Commitments, the address and fax number (and the department or officer, if any, for whose attention any communication is to be made) of each Lender for any communication to be made or document to be delivered under or in connection with the Finance Documents, the electronic mail address and/or any other information required to enable the sending and receipt of information by electronic mail or other electronic means to and by each Lender to whom any communication under or in connection with the Finance Documents may be made by that means and the account details of each Lender for any payment to be distributed by the Agent to that Lender under the Finance Documents.

39.2. Supply of Lender details at Parent's direction

- (a) The Agent shall, at the request of the Parent, disclose the identity of the Lenders and the details of the Lenders' Commitments to any:
 - (i) other Party or any other person if that disclosure is made to facilitate, in each case, a refinancing of the Financial Indebtedness arising under the Finance Documents or a material waiver or amendment of any term of any Finance Document; and
 - (ii) member of the Group.
- (b) Subject to paragraph (c) below, the Parent shall procure that the recipient of information disclosed pursuant to paragraph (a) above shall keep such information confidential and shall not disclose it to anyone and shall ensure that all such information is protected with security measures and a degree of care that would apply to the recipient's own confidential information.
- (c) The recipient may disclose such information to any of its officers, directors, employees, professional advisers, auditors and partners as it shall consider appropriate if any such person is informed in writing of its confidential nature, except that there shall be no such requirement to so inform if that person is subject to professional obligations to maintain the confidentiality of the information or is otherwise bound by duties of confidentiality in relation to the information.

39.3. Supply of Lender details to other Lenders

- (a) If a Lender (a "**Disclosing Lender**") indicates to the Agent that the Agent may do so, the Agent shall disclose that Lender's name and Commitment to any other Lender that is, or becomes, a Disclosing Lender.
- (b) The Agent shall, if so directed by the Requisite Lenders, request each Lender to indicate to it whether it is a Disclosing Lender.

39.4. Lender enquiry

If any Lender believes that any entity is, or may be, a Lender and:

- (a) that entity ceases to have an Investment Grade Rating; or
- (b) an Insolvency Event occurs in relation to that entity,

the Agent shall, at the request of that Lender, indicate to that Lender the extent to which that entity has a Commitment.

39.5. Lender details definitions

In this Clause 39:

"Investment Grade Rating" means, in relation to an entity, a rating for its long-term unsecured and non credit-enhanced debt obligations of BBBor higher by Standard & Poor's Rating Services or Fitch Ratings Ltd or Baa3 or higher by Moody's Investors Service Limited or a comparable rating from an internationally recognised credit rating agency.

"**Requisite Lenders**" means a Lender or Lenders whose Commitments aggregate 15 per cent. (or more) of the Total Commitments (or if the Total Commitments have been reduced to zero, aggregated 15 per cent. (or more) of the Total Commitments immediately prior to that reduction).

40. COUNTERPARTS

Each Finance Document may be executed in any number of counterparts, and this has the same effect as if the signatures on the counterparts were on a single copy of the Finance Document.

SECTION 12

GOVERNING LAW AND ENFORCEMENT

41. GOVERNING LAW

This Agreement is governed by English law.

42. ENFORCEMENT

42.1. Jurisdiction of English courts

- (a) The courts of England have exclusive jurisdiction to settle any dispute arising out of or in connection with this Agreement (including a dispute regarding the existence, validity or termination of this Agreement) (a "**Dispute**").
- (b) The Parties agree that the courts of England are the most appropriate and convenient courts to settle Disputes and accordingly no Party will argue to the contrary.
- (c) This Clause 42.1 is for the benefit of the Finance Parties and Secured Parties only. As a result, no Finance Party or Secured Party shall be prevented from taking proceedings relating to a Dispute in any other courts with jurisdiction. To the extent allowed by law, the Finance Parties and Secured Parties may take concurrent proceedings in any number of jurisdictions.

42.2. Service of process

- (a) Without prejudice to any other mode of service allowed under any relevant law, each Obligor (other than an Obligor incorporated in the United Kingdom):
 - (i) irrevocably appoints Morton Fraser LLP, St Martin's House, 16 St. Martin's le Grand, London EC1A 4EN as its agent for service of process in relation to any proceedings before the English courts in connection with any Finance Document; and
 - (ii) agrees that failure by an agent for service of process to notify the relevant Obligor of the process will not invalidate the proceedings concerned.
- (b) If any person appointed as an agent for service of process is unable for any reason to act as agent for service of process, the Parent (on behalf of all the Obligors) must immediately (and in any event within 10 days of such event taking place) appoint another agent on terms acceptable to the Agent. Failing this, the Agent may appoint another agent for this purpose.

The Parent expressly agrees and consents to the provisions of this Clause 42 and Clause 41 (Governing law).

This Agreement has been entered into on the date stated at the beginning of this Agreement.

SCHEDULE 1

THE ORIGINAL PARTIES

Part I

	The Obligors
Name of Borrower	Registration number (or equivalent, if any)
Schuh Group Limited	SC379625
Schuh (Holdings) Limited	SC265833
Schuh Limited	SC125327
Name of Guarantor	Registration number (or equivalent, if any)
Schuh Group Limited	SC379625
Schuh (Holdings) Limited	SC265833
Schuh Limited	SC125327
Schuh (ROI) Limited	272987

Part II

The Original Lenders

Name of Original Lender	Facility A Commitment	Facility B Commitment	Facility C Commitment
Lloyds Bank plc	£7,250,000	£	£12,500,000

SCHEDULE 2

CONDITIONS PRECEDENT

Part I

Conditions precedent to signing of Agreement

4. Obligors

- (a) A copy of the Constitutional Documents and of the constitutional documents of each other Obligor.
- (b) A copy of a resolution of the board or, if applicable, a committee of the board of directors of each Obligor:
 - (i) approving the terms of, and the transactions contemplated by, Transaction Documents to which it is a party and resolving that it execute, deliver and perform Transaction Documents to which it is a party;
 - (ii) authorising a specified person, on its behalf to sign and/or despatch all documents and notices to be signed and/or despatched by it under or in connection with Transaction Documents to which it is a party.
- (C) A certificate of the Parent (signed by a director) and each other Obligor (in each case signed by a director) certifying that each copy document relating to it specified in this Schedule 2 is true and complete and in full force and effect as at a date no earlier than the date of this Agreement.
- (d) A specimen of the signature of each person authorised by the resolution referred to in paragraph (b) above in relation to the Transaction Documents and related documents.
- (e) A certificate of the Parent (signed by a director) and each other Obligor (in each case signed by a director) and each other Obligor (in each case signed by a director) confirming that borrowing or guaranteeing or securing, as appropriate, the Total Commitments would not cause any borrowing, guarantee, security or similar limit binding on any Obligor to be exceeded.

5. Acquisition Documents

(a) A copy of each of the Acquisition Documents executed by the parties to those documents.

- 150
- (b) Evidence satisfactory to the Agent that Schuh Group Limited has acquired all of the shares in Holdings.

6. Finance Documents

- (a) This Agreement executed by the members of the Group party to this Agreement.
- (b) Deliberately left blank.
- (c) The Intercreditor Agreement executed by the parties thereto.
- (d) The Fee Letter executed by the Parent.
- (e) The Working Capital Facility Letter.
- (f) The Hedging Policy Letter.
- (g) A floating charge over all of the assets of the Parent.
- (h) A floating charge over all of the assets of Schuh (Holdings) Limited.
- (i) A floating charge over all of the assets of Schuh Limited.
- (j) A debenture over all of the assets of Schuh (ROI) Limited.
- (k) The Standard Security.
- (l) A shares pledge governed by Scots law by Schuh Group Limited in respect of all of the shares in Schuh (Holdings) Limited.
- (m) A shares pledge governed by Scots law by Schuh (Holdings) Limited in respect of all of the shares in Schuh Limited.
- (n) A shares pledge governed by Irish law by Schuh Limited in respect of all of the shares in Schuh (ROI) Limited.
- (o) Evidence satisfactory to the Agent that Schuh (ROI) Limited has done all that is necessary under Section 60 of the Companies Act 1963 (as amended) in order to enable it to enter into the Finance Documents and perform its obligations under the Finance Documents.
- (p) A certificate of Schuh (ROI) Limited (signed by a director) certifying that the provisions of Section 31 of the Irish Companies Act 1990 do not prohibit the execution by Schuh (ROI) Limited of any of the Finance Documents which it is intended that Schuh (ROI) Limited will execute by reason of the fact that Schuh (ROI) Limited and each other company whose liabilities are thereby guaranteed or secured are members of a group of companies consisting of a holding company and its subsidiaries for the purpose of Section 35 of the Irish Companies Act 1990.
- (q) Letter of reliance from Morton Fraser LLP addressed to Dickson Minto W.S. in terms entitling the Original Lender to rely on the report on title by Morton Fraser LLP in respect of the Livingston Property addressed to the Bank of Scotland plc.
- (r) A search in the Register of Charges and Company File in respect of Schuh Limited as the granter of the Standard Security, such Search disclosing no entries prejudicial to the grant of the Standard Security and brought down to a date being as practicable to the date of completion.

- 151
- (s) A form 12 over the Livingston Property brought down to a date as near practicable to the date of this Agreement disclosing no entries adverse to the grantor of the Standard Security's interest in the property secured.
- (t) A cheque made payable to Dickson Minto W.S. in reimbursement of Registers of Scotland registration dues payable on the Standard Security, which will have been paid on behalf of Schuh Limited.

7. Insurance

- (a) The Insurance Adequacy Letter.
- (b) All insurance policies subject to or expressed to be subject to the Transaction Security relating to the Charged Property.
- (c) Written evidence that the insurance policy(ies) relating to the Charged Property contain (in form and substance reasonably satisfactory to the Security Trustee) an endorsement naming the Security Trustee as sole loss payee.

8. Other documents and evidence

- (a) The Group Structure Chart.
- (b) The Business Plan.
- (c) The Financial Due Diligence Report.
- (d) The Funds Flow and evidence that the payments referred to in the Funds Flow have been instructed.
- (e) The Tax Letter.
- (f) A copy, certified by an authorised signatory of the Parent to be a true copy, of the Original Financial Statements of each Obligor, in the case of the unaudited management accounts being in a format agreed between the Parent and the Agent.
- (g) A letter of engagement with Lloyds Bank plc as Original Lender from the authors of the Financial Due Diligence Report.
- (h) A copy of any other Authorisation or other document, opinion or assurance which the Agent notifies the Parent is necessary or desirable in connection with the entry into and performance of the transactions contemplated by any Finance Document or for the validity and enforceability of any Finance Document.
- (i) Evidence that the fees, costs and expenses then due from the Parent pursuant to Clause 13 (Fees), Clause 14.7 (Stamp taxes) and Clause 18 (Costs and expenses) have been paid or will be paid on or around the date of this Agreement.
- (j) A Certificate of the Parent (signed by a director) detailing the estimated Deal Costs.
- (k) Utilisation Requests relating to any Utilisations to be made on the date of this Agreement.
- (l) A legal opinion from Matheson Ormsby Prentice addressed to the Finance Parties.
- (m) Signed copies of the ISDA master agreement and schedule between the Original Lender and the Parent.

- (n) Evidence satisfactory to the Agent that all of the share options issued in respect of Schuh (Holdings) Limited have been rolled over.
- (o) Evidence satisfactory to the Agent that all of the existing debt of Schuh (Holdings) Limited and it subsidiaries is to be repaid from the facilities being made available under the terms of this Agreement.
- (p) A counter indemnity from Schuh Limited in favour of the Working Capital Lender in respect of the indemnity by it in favour of Bank of Scotland plc.
- (q) A copy of the Shareholders Agreement and the termination agreement in respect of the shareholders agreement dated 2 February 2008.

Part II

Conditions precedent required to be delivered by an Additional Obligor

- 1. An Accession Letter executed by the Additional Obligor and the Parent.
- 2. A copy of the constitutional documents of the Additional Obligor.
- 3. A copy of a resolution of the board or, if applicable, a committee of the board of directors of the Additional Obligor:
 - (a) approving the terms of, and the transactions contemplated by, the Accession Letter and the Finance Documents and resolving that it execute, deliver and perform the Accession Letter and any other Finance Document to which it is party;
 - (b) authorising a specified person or persons to execute the Accession Letter and other Finance Documents on its behalf;
 - (C) authorising a specified person or persons, on its behalf, to sign and/or despatch all other documents and notices (including, in relation to an Additional Borrower, any Utilisation Request or Selection Notice) to be signed and/or despatched by it under or in connection with the Finance Documents to which it is a party; and
 - (d) authorising the Parent to act as its agent in connection with the Finance Documents
- 4. If applicable, a copy of a resolution of the board of directors of the Additional Obligor, establishing the committee referred to in paragraph 3 above.
- 5. A specimen of the signature of each person authorised by the resolution referred to in paragraph 3 above.
- 6. A copy of a resolution signed by all the holders of the issued shares of the Additional Guarantor, approving the terms of, and the transactions contemplated by, the Finance Documents to which the Additional Guarantor is a party.
- 7. A copy of a resolution of the board of directors of each corporate shareholder of each Additional Guarantor approving the terms of the resolution referred to in paragraph 6 above.
- 8. A certificate of the Additional Obligor (signed by a director) confirming that borrowing or guaranteeing or securing, as appropriate, the Total Commitments would not cause any borrowing, guarantee, security or similar limit binding on it to be exceeded.
- 9. A certificate of an authorised signatory of the Additional Obligor certifying that each copy document listed in this Part II of Schedule 2 is correct, complete and in full force and effect and has not been amended or superseded as at a date no earlier than the date of the Accession Letter.
- 10. A copy of any other Authorisation or other document, opinion or assurance which the Agent considers to be necessary or desirable in connection with the entry into and performance of the transactions contemplated by the Accession Letter or for the validity and enforceability of any Finance Document.
- 11. If available, the latest audited financial statements of the Additional Obligor.
- 12. The following legal opinions, each addressed to the Agent, the Security Trustee and the Lenders:

- (a) a legal opinion of the legal advisers to the Agent in England, as to English law in the form distributed to the Lenders prior to signing the Accession Letter; and
- (b) if the Additional Obligor is incorporated in or has its "centre of main interest" or "establishment" (as referred to in Clause 20.28 (Centre of main interests and establishments)) in a jurisdiction other than England and Wales or is executing a Finance Document which is governed by a law other than English law, a legal opinion of the legal advisers to the Agent in the jurisdiction of its incorporation, "centre of main interest" or "establishment" (as applicable) or, as the case may be, the jurisdiction of the governing law of that Finance Document (the "Applicable Jurisdiction") as to the law of the Applicable Jurisdiction and in the form distributed to the Lenders prior to signing the Accession Letter.
- 13. If the proposed Additional Obligor is incorporated in a jurisdiction other than England and Wales, evidence that the process agent specified in Clause 42.2 (Service of process), if not an Obligor, has accepted its appointment in relation to the proposed Additional Obligor.
- 14. Any security documents which are required by the Agent to be executed by the proposed Additional Obligor.
- 15. Any notices or documents required to be given or executed under the terms of those security documents.
- 16. In relation to financial assistance (or overseas equivalent), such documentary evidence as legal counsel to the Agent may require, that such Additional Obligor has complied with any law in its jurisdiction relating to financial assistance or analogous process.

SCHEDULE 3

REQUEST

Part 1

Utilisation Request

Loans

From: [Borrower] [Parent]

To: Lloyds Bank plc

Dated:

Dear Sirs

Schuh Group Limited – £24,357,000 Senior Facilities Agreement dated [] (the "Facilities Agreement")

- 1. We refer to the Facilities Agreement. This is a Utilisation Request. Terms defined in the Facilities Agreement have the same meaning in this Utilisation Request unless given a different meaning in this Utilisation Request.
- 2. We wish to borrow a Loan on the following terms:
 - (a) Borrower: []

- 155
- (b) Proposed Utilisation Date: [] (or, if that is not a Business Day, the next Business Day)
- (c) Facility to be utilised: [Facility A]/[Facility B]/[Facility C]**
- (d) Amount: [] or, if less, the Available Facility
- (e) Interest Period: []
- 3. We confirm that each condition specified in Clause 4.2 (Further conditions precedent) is satisfied on the date of this Utilisation Request.
- 4. [The proceeds of this Loan should be credited to [account]].
- 5. This Utilisation Request is irrevocable.

Yours faithfully

.....

authorised signatory for

[the Parent on behalf of [insert name of relevant Borrower]]/ [insert name of Borrower]

NOTES:

- * Amend as appropriate. The Utilisation Request can be given by the Borrower or by the Parent.
- ** Select the Facility to be utilised and delete references to the other Facilities.

Part II

Selection Notice

Applicable to a Term Loan

From: Schuh Group Limited

To: Lloyds Bank plc

Dated:

Dear Sirs

Schuh Group Limited - £24,357,000 Senior Facilities Agreement dated [] (the "Facilities Agreement")

- 1. We refer to the Facilities Agreement. This is a Selection Notice. Terms defined in the Facilities Agreement have the same meaning in this Selection Notice unless given a different meaning in this Selection Notice.
- 2. We refer to the following Facility [A]/[B] Loan[s] with an Interest Period ending on []*.
- 3. [We request that the above Facility [A/C] Loan[s] be divided into I Facility [A/C] Loans with the following amounts and Interest Periods:]**

[We request that the next Interest Period for the above Facility [A]/[B]/[C] Loan[s] is [].]***

4. This Selection Notice is irrevocable.

Yours faithfully

authorised signatory for the Parent

.....

NOTES:

- * Insert details of all Term Loans for the relevant Facility which have an Interest Period ending on the same date.
- ** Use this option if division of Facility A Loans or Facility C Loans is requested.
- *** Use this option if sub-division is not required or if Selection Notice relates to Facility B Loans.

SCHEDULE 4

PLAN REFERRED TO IN THE DEFINITION OF ADDITIONAL LIVINGSTON PROPERTY

SCHEDULE 5

FORM OF TRANSFER CERTIFICATE

To: Lloyds Bank plc as Agent and Lloyds Bank plc as Security Trustee

From: [The Existing Lender] (the "Existing Lender") and [The New Lender] (the "New Lender")

Schuh Group Limited – £24,357,000 Senior Facilities Agreement dated [] (the "Facilities Agreement")

- 1. We refer to the Facilities Agreement. This agreement (the "**Agreement**") shall take effect a Transfer Certificate for the purpose of the Facilities Agreement. Terms defined in the Facilities Agreement have the same meaning in this Agreement unless given a different meaning in this Agreement.
- 2. We refer to Clause 25.5 (Procedure for transfer) of the Facilities Agreement:
 - (a) The Existing Lender and the New Lender agree to the Existing Lender transferring to the New Lender by novation and in accordance with Clause 25.5 (Procedure for Transfer) all of the Existing Lender's rights and obligations under the Facilities Agreement which relate to that portion (s) and participations in Utilisations under the Facilities Agreement as specified in the Schedule.
 - (b) The proposed Transfer Date is [].
 - (c) The Facility Office and address, fax number and attention details for notices of the New Lender for the purposes of Clause 33.2 (Addresses) are set out in the Schedule.
- 3. The New Lender expressly acknowledges the limitations on the Existing Lender's obligations set out in paragraph (c) of Clause 25.4 (Limitation of responsibility of Existing Lenders).
- 4. The New Lender confirms, for the benefit of the Agent and without any liability to any Obligor, that it is:
 - (a) [a Qualifying Lender (other than a Treaty Lender);]
 - (b) [a Treaty Lender;]
 - (c) [not a Qualifying Lender].
- 5. [The New Lender confirms that the person beneficially entitled to interest payable to that Lender in respect of an advance under a Finance Document is either:
 - (a) a company resident in the United Kingdom for United Kingdom tax purposes; or
 - (b) a partnership each member of which is:
 - (i) a company so resident in the United Kingdom; or
 - (ii) a company not so resident in the United Kingdom which carries on a trade in the United Kingdom through a permanent establishment and which brings into account in computing its chargeable profits (within the meaning of section 19 of the CTA) the whole of any share of interest payable in respect of that advance that falls to it by reason of Part 17 of the CTA; or
 - (c) a company not so resident in the United Kingdom which carries on a trade in the United Kingdom through a permanent establishment and which brings into account interest payable in respect of that advance in computing the chargeable profits (within the meaning of section Part 17 of the CTA) of that company; or]
 - (d) [a Treaty Lender].

- 6. [The New Lender confirms that it holds a passport under the HMRC DT Treaty Passport scheme (reference number []) and is tax resident in [], so that interest payable to it by borrowers is generally subject to full exemption from UK withholding tax and requests that the Parent notify:
 - (a) each Borrower which is a Party as a Borrower as at the Transfer Date; and
 - (b) each Additional Borrower which becomes an Additional Borrower after the Transfer Date,

that it wishes that scheme to apply to the Facilities Agreement.]

[7/8]. This Agreement may be executed in any number of counterparts and this has the same effect as if the signatures on the counterparts were on a single copy of this Agreement.[8/9] This Agreement and any non contractual obligations arising out of or in connection with it are is governed by English law.

[9/10] This Agreement has been entered into on the date stated at the beginning of this Agreement.

Note: The execution of this Transfer Certificate may not transfer a proportionate share of the Existing Lender's interest in the Transaction Security in all jurisdictions. It is the responsibility of the New Lender to ascertain whether any other documents or other formalities are required to perfect a transfer of such a share in the Existing Lender's Transaction Security in any jurisdiction and, if so, to arrange for execution of those documents and completion of those formalities.

THE SCHEDULE

Commitment/rights and obligations to be transferred

[insert relevant details]

[Facility Office address, fax number and attention details for notices and account details for payments,]

[Existing Lender]

By:

By:

[New Lender]

This Agreement is accepted as a Transfer Certificate for the purposes of the Facilities Agreement by the Agent and the Transfer Date is confirmed as [].

- [Agent]
- By:

SCHEDULE 6

FORM OF ASSIGNMENT AGREEMENT

To: Lloyds Bank plc as Agent

From: [the Existing Lender] (the "Existing Lender") and [the New Lender] (the "New Lender")

Dated:

Schuh Group Limited - £24,357,000 Senior Facilities Agreement dated [] (the "Facilities Agreement")

- 1. We refer to the Facilities Agreement. This is an Assignment Agreement. This agreement (the "**Agreement**") shall take effect as an Assignment Agreement for the purposes of the Facilities Agreement. Terms defined in the Facilities Agreement have the same meaning in this Agreement unless given a different meaning in this Agreement.
- 2.
- (a) We refer to Clause 25.6 (Procedure for assignment) of the Facilities Agreement.
- (b) The Existing Lender assigns absolutely to the New Lender all the rights of the Existing Lender under the Facilities Agreement, the other Finance Documents and in respect of the Transaction Security which correspond to that portion of the Existing Lender's Commitment(s) and participations in Utilisations under the Facilities Agreement as specified in the Schedule.

- (c) The Existing Lender is released from all the obligations of the Existing Lender which correspond to that portion of the Existing Lender's Commitment(s) and participations in Utilisations under the Facilities Agreement specified in the Schedule.
- (d) The New Lender becomes a Party as a Lender and is bound by obligations equivalent to those from which the Existing Lender is released under paragraph (c) above.
- 3. The proposed Transfer Date is [].
- 4. On the Transfer Date the New Lender becomes:
 - (a) Party to the Finance Documents as a Lender; and
 - (b) Party to [other relevant agreements in other relevant capacity].
- 5. The New Lender expressly acknowledges the limitations on the Existing Lender's obligations set out in paragraph (c) of Clause 25.4 (Limitation of responsibility of Existing Lenders).
- 6. The New Lender confirms, for the benefit of the Agent and without any liability to any Obligor, that it is:
 - (a) [a Qualifying Lender (other than a Treaty Lender);]
 - (b) [a Treaty Lender;]
 - (c) [not a Qualifying Lender].
- 7. The Facility Office and address, fax number and attention details for notices of the New Lender for the purposes of Clause 33.2 (Addresses) are set out in the Schedule.
- 8. [The New Lender confirms that the person beneficially entitled to interest payable to that Lender in respect of an advance under a Finance Document is either:
 - (a) a company resident in the United Kingdom for United Kingdom tax purposes; or
 - (b) a partnership each member of which is:
 - (i) a company so resident in the United Kingdom; or
 - (ii) a company not so resident in the United Kingdom which carries on a trade in the United Kingdom through a permanent establishment and which brings into account in computing its chargeable profits (within the meaning of section 19 of the CTA) the whole of any share of interest payable in respect of that advance that falls to it by reason of Part 17 of the CTA; or
 - (c) a company not so resident in the United Kingdom which carries on a trade in the United Kingdom through a permanent establishment and which brings into account interest payable in respect of that advance in computing the chargeable profits (within the meaning of section 19 of the CTA) of that company; or]
- [8/9]. [The New Lender confirms that it holds a passport under the HMRC DT Treaty Passport scheme (reference number []) and is tax resident in [], so that interest payable to it by borrowers is generally subject to full exemption from UK withholding tax and requests that the Parent notify:
 - (a) each Borrower which is a Party as a Borrower as at the Transfer Date; and
 - (b) each Additional Borrower which becomes an Additional Borrower after the Transfer Date,

that it wishes that scheme to apply to the Facilities Agreement.]

- [9/10] This Agreement acts as notice to the Agent (on behalf of each Finance Party) and, upon delivery in accordance with Clause 25.7 (Copy of Transfer Certificate), to the Parent (on behalf of each Obligor) of the assignment referred to in this Agreement. This Agreement may be executed in any number of counterparts and this has the same effect as if the signatures on the counterparts were on a single copy of this Agreement.
- [9/10]. This Agreement and any non contractual obligations arising out of or in connection with it is governed by English law.
- [10/11]. This Agreement has been [executed and delivered as a deed] [entered into] on the date stated at the beginning of this Assignment Agreement.
- Note: The execution of this Assignment Agreement may not transfer a proportionate share of the Existing Lender's interest in the Transaction Security in all jurisdictions. It is the responsibility of the New Lender to ascertain whether any other documents or other formalities are required to perfect a transfer of such a share in the Existing Lender's Transaction Security in any jurisdiction and, if so, to arrange for execution of those documents and completion of those formalities.

THE SCHEDULE

Commitment/rights and obligations to be transferred by assignment, release and accession

[insert relevant details]

[Facility office address, fax number and attention details for notices and account details for payments]

[Existing Lender] [New Lender]

By:

By:

This Agreement is accepted as an Assignment Agreement for the purposes of the Facilities Agreement by the Agent and the Transfer Date is confirmed as [].

[Signature of this Assignment Agreement by the Agent constitutes confirmation by the Agent of receipt of notice of the assignment referred to in this Agreement, which notice the Agent receives on behalf of each Finance Party.]

[Agent]

By:

SCHEDULE 7

FORM OF ACCESSION LETTER

To: Lloyds Bank plc as Agent

From: [Subsidiary] and [Parent]

Dated:

Dear Sirs

Schuh Group Limited – £24,357,000 Senior Facilities Agreement dated [] (the "Facilities Agreement")

- 1. We refer to the Facilities Agreement. This letter (the "Accession Letter") shall take effect as an Accession Letter for the purposes of the Facilities Agreement. Terms defined in the Facilities Agreement have the same meaning in this Accession Letter unless given a different meaning in this Accession Letter.
- 2. [Subsidiary] agrees to become an Additional [Borrower]/[Guarantor] and to be bound by the terms of the Facilities Agreement and the other Finance Documents as an Additional [Borrower]/[Guarantor] pursuant to Clause [26.2 (Additional Borrowers)]/[Clause 26.4 (Additional Guarantors)] of the Facilities Agreement. [Subsidiary] is a company duly incorporated under the laws of [name of relevant jurisdiction] and is a limited liability company and registered number [].
- 3. [The Parent confirms that no Default is continuing or would occur as a result of [Subsidiary] becoming an Additional Borrower].
- 4. [Subsidiary's] administrative details are as follows:

Address:

Fax No.:

Attention:

5. This Accession Letter is governed by English law.

[This Accession Letter is entered into by deed.]

[Parent]

[Subsidiary]

SCHEDULE 8

FORM OF RESIGNATION LETTER

To: Lloyds Bank plc as Agent

From: [resigning Obligor] and [Parent]

Dated:

Dear Sirs

Schuh Group Limited - £24,357,000 Senior Facilities Agreement dated [] (the "Facilities Agreement")

- 1. We refer to the Facilities Agreement. This is a Resignation Letter. Terms defined in the Facilities Agreement have the same meaning in this Resignation Letter unless given a different meaning in this Resignation Letter.
- 2. Pursuant to [Clause 26.3 (Resignation of a Borrower)]/[Clause 26.5 (Resignation of a Guarantor)], we request that [resigning Obligor] be released from its obligations as a [Borrower]/[Guarantor] under the Facilities Agreement and the Finance Documents.
- 3. We confirm that:
 - (a) no Default is continuing or would result from the acceptance of this request; and
 - (b) *[[this request is given in relation to a Third Party Disposal of [resigning Obligor];
 - (c) [the Disposal Proceeds have been or will be applied in accordance with Clause 8.2 (Disposal and Insurance Proceeds and Excess Cashflow);]**]
 - (d) []***
- 4. This letter is governed by English law.

[Parent]

[resigning Obligor]

By:

By:

NOTES:

- * Insert where resignation only permitted in case of a Third Party Disposal.
- ** Amend as appropriate, e.g. to reflect agreed procedure for payment of proceeds into a specified account.
- *** Insert any other conditions required by the Facilities Agreement.

SCHEDULE 9

FORM OF COMPLIANCE CERTIFICATE

To: Lloyds Bank plc as Agent

From: [Parent]

Dear Sirs

Schuh Group Limited - £24,357,000 Senior Facilities Agreement dated [] (the "Facilities Agreement")

- 1. We refer to the Facilities Agreement. This is a Compliance Certificate. Terms defined in the Facilities Agreement have the same meaning when used in this Compliance Certificate unless given a different meaning in this Compliance Certificate.
- 2. We confirm that:

[Insert details of covenants to be certified].

3. [We confirm that no Default is continuing.]**

Signed

Director of [Parent] Director of [Parent]

SCHEDULE 10

FORM OF INCREASE CONFIRMATION

To: Lloyds Bank plc as Agent, Lloyds Bank plc as Security Trustee and Schuh Group Limited as Parent, for and on behalf of each Obligor

From: [the Increase Lender] (the "Increase Lender")

Dated:

Schuh Group Limited – £24,357,000 Senior Facilities Agreement dated [] (the "Facilities Agreement")

- 1. We refer to the Facilities Agreement. This agreement (the "**Agreement**") shall take effect as an Increase Confirmation for the purpose of the Facilities Agreement. Terms defined in the Facilities Agreement have the same meaning in this Agreement unless given a different meaning in this Agreement.
- 2. We refer to clause 2.2 (Increase) of the Facilities Agreement.
- 3. The Increase Lender agrees to assume and will assume all of the obligations corresponding to the Commitment specified in the Schedule (the "**Relevant Commitment**") as if it was an Original Lender under the Facilities Agreement.

- 4. The proposed date on which the increase in relation to the Increase Lender and the Relevant Commitment is to take effect (the "**Increase Date**") is [].
- 5. On the Increase Date, the Increase Lender becomes party to the relevant Finance Documents.
- 6. The Facility Office and address, fax number and attention details for notices to the Increase Lender for the purposes of Clause 33 (Notices) are set out in the Schedule.
- 7. The Increase Lender expressly acknowledges the limitations on the Lenders' obligations referred to in paragraph (f) of Clause 2.2 (Increase).
- 8. The Increase Lender confirms, for the benefit of the Agent and without liability to any Obligor, that it is:
 - (a) [a Qualifying Lender (other than a Treaty Lender);]
 - (b) [a Treaty Lender;]
 - (c) [not a Qualifying Lender].
- 9. [The Increase Lender confirms that the person beneficially entitled to interest payable to that Lender in respect of an advance under a Finance Document is either:
 - (a) a company resident in the United Kingdom for United Kingdom tax purposes; or
 - (b) a partnership each member of which is:
 - (i) a company so resident in the United Kingdom; or
 - (ii) a company not so resident in the United Kingdom which carries on a trade in the United Kingdom through a permanent establishment and which brings into account in computing its chargeable profits (within the meaning of section 19 of the CTA) the whole of any share of interest payable in respect of that advance that falls to it by reason of Part 17 of the CTA; or
 - (c) a company not so resident in the United Kingdom which carries on a trade in the United Kingdom through a permanent establishment and which brings into account interest payable in respect of that advance in computing the chargeable profits (within the meaning of section 19 of the CTA) of that company.]
- 10. [The Increase Lender confirms (for the benefit of the Agent and without liability to any Obligor) that it is a Treaty Lender that holds a passport under the HMRC DT Treaty Passport scheme (reference number []) and is tax resident in [], so that interest payable to it by borrowers is generally subject to full exemption from UK withholding tax and notifies the Parent that:
 - (a)[each Borrower which is a Party as a Borrower as at the Increase Date must, to the extent that the Increase Lender becomes a Lender under a Facility which is made available to that Borrower pursuant to clause 2 (The Facilities) of the Facilities Agreement, make an application to HM Revenue & Customs under form DTTP2 within 30 days of the Increase Date; and]^{*}
 - (b)each Additional Borrower which becomes an Additional Borrower after the Increase Date must, to the extent that the Increase Lender is a Lender under a Facility which is made available to that Additional Borrower pursuant to clause 2.1 (The Facilities) of

the Facilities Agreement, make an application to HM Revenue & Customs under form DTTP2 within 30 days of becoming an Additional Borrower.]***

- [12]. This Agreement may be executed in any number of counterparts and this has the same effect as if the signatures on the counterparts were on a single copy of this Agreement.
- [13]. This Agreement is governed by English law.
- [14]. This Agreement has been entered into on the date stated at the beginning of this Agreement.

Note: The execution of this Increase Confirmation may not be sufficient for the Increase Lender to obtain the benefit of the Transaction Security in all jurisdictions. It is the responsibility of the Increase Lender to ascertain whether any other documents or other formalities are required to obtain the benefit of the Transaction Security in any jurisdiction and, if so, to arrange for execution of those documents and completion of those formalities.

THE PARENT

SCHUH GROUP LIMITED

By: Mark Crutchley Adrian Bell (witness)

Address: 1 Neilson Square Deans Industrial Estate Livingston EH54 8RQ

Attention: Mark Crutchley

Fax: 01506 460 250

THE BORROWERS

SCHUH GROUP LIMITED

By: Mark Crutchley Adrian Bell (witness)	By:	Mark Crutchley	Adrian Bell (witness)
--	-----	----------------	-----------------------

Address: As above

Attention: As above

Fax: As above

SCHUH (HOLDINGS) LIMITED

By: Mark Crutchley Adrian Bell (witness)

Address: As above

Attention: As above

Fax: As above

THE GUARANTORS

SCHUH GROUP LIMITED

- By: Mark Crutchley Adrian Bell (witness)
- Address: As above
- Attention: As above
- Fax: As above

SCHUH (HOLDINGS) LIMITED

- By: Mark Crutchley Adrian Bell (witness)
- Address: As above
- Attention: As above
- Fax: As above

SCHUH LIMITED

- By: Mark Crutchley Adrian Bell (witness)
- Address: As above
- Attention: As above
- Fax: As above

PRESENT WHEN THE COMMON SEAL OF SCHUH (ROI) LIMITED WAS AFFIXED HERETO:

By: Mark Crutchley Adrian Bell (witness) Director Address: As above

- Attention: As above
- Fax: As above

David Gillan Reid COMPANY SECRETARY

THE ARRANGER

LLOYDS BANK PLC

By: Simon Sweeney

Address: 2nd Floor

1 Lochrin Square 92 Fountainbridge Edinburgh EH3 9QA

Fax: 0131 229 5198

Attention: [Sarah Lamb]

THE AGENT

LLOYDS BANK PLC

By: Simon Sweeney

Address: As above

Fax: As above

Attention: As above

THE SECURITY TRUSTEE

LLOYDS BANK PLC

By: Simon Sweeney

Address: As above

Fax: As above

Attention: As above

THE LENDER

LLOYDS BANK PLC

By: Simon Sweeney

Address: As above

Fax: As above

Attention: As above

THE PARENT

SCHUH GROUP LIMITED

By:

Address: 1 Neilson Square Deans Industrial Estate Livingston

Fax: 01506 460250

THE BORROWERS

SCHUH GROUP LIMITED

By:

Address: As above

Fax: As above

SCHUH (HOLDINGS) LIMITED

By:

Address: As above

Fax: As above

SCHUH LIMITED

By:

Address: As above

Fax: As above

THE GUARANTORS

SCHUH GROUP LIMITED

By:

Address: As above

Fax: As above

SCHUH (HOLDINGS) LIMITED

By:

Address: As above

Fax: As above

SCHUH LIMITED

By:

Address: As above

Fax: As above

SCHUH (ROI) LIMITED

Given under the Common Seal of

Schuh (ROI) Limited and delivered

as deed

By:

Director

Director/Company Secretary

Address: As above

Fax: As above

THE ARRANGER

LLOYDS BANK PLC

By:

Address:	2nd Floor	
	1 Lochrin Square	
	92 Fountainbridge	
	Edinburgh	
	EH3 9QA	

Fax: 0131 229 5198

THE AGENT

LLOYDS BANK PLC

By:

Address: As above

Fax: As above

THE ORIGINAL LENDER

LLOYDS BANK PLC

By:

Address: As above

Fax: As above

THE SECURITY TRUSTEE

LLOYDS BANK PLC

By:

Address: As above

Fax: As above

Exhibit (21)

SUBSIDIARIES OF THE REGISTRANT

Subsidiaries of the Company:

Names of Subsidiary ⁽¹⁾	Place of
	Incorporation
Flagg Bros. of Puerto Rico, Inc.	Delaware
Genesco Brands, LLC	Delaware
GVI, Inc.	Delaware
Hat World Corporation	Delaware
GCO Canada Inc.	Canada Minnesota
Hat World, Inc.	Minnesota
Hat World Services Co., Inc.	Delaware
Keuka Footwear, Inc.	Delaware
SIOPA Sports of America, LLC (50% owned)	Delaware
Genesco (UK) Limited	United Kingdom
Lids Retail Limited	United Kingdom
Schuh Group Limited	United Kingdom
Schuh (Holdings) Limited	United Kingdom
Schuh Limited	United Kingdom
Schuh (ROI) Limited	Republic of Ireland
Genesco GP, LLC	United Kingdom
Genesco Scot LP	United Kingdom
Genesco (Jersey) Limited	Jersey
(1) Except as otherwise indicated, 100% of the equity of each listed subsidiary is owned either	by the registrant or by a wholly-

(1) Except as otherwise indicated, 100% of the equity of each listed subsidiary is owned either by the registrant or by a wholly owned subsidiary of the registrant.

Exhibit (24)

POWER OF ATTORNEY

The undersigned, certain of the officers and directors of Genesco Inc., a Tennessee corporation ("Genesco"), do hereby constitute and appoint Roger G. Sisson and James S. Gulmi, and any one of them, to act severally as attorneys-in-fact for and in their respective names, places and steads, with full power of substitution, to execute, sign and file with the Securities and Exchange Commission the Annual Report on Form 10-K of Genesco for the fiscal year ended February 1, 2014, and any and all amendments thereto; granting to said attorneys-in-fact, and each of them, full power and authority to do and perform every act and thing whatsoever requisite or necessary to be done in and about the premises as fully to all intents and purposes as the undersigned or any of them might or could do if personally present, and the undersigned do hereby ratify and confirm all that said attorney-in-fact or any of them, or their substitute or substitutes, may lawfully do or cause to be done by virtue hereof.

Executed this 27th day of February, 2014

/s/ Robert J. Dennis	/s/ James S. Gulmi		
Robert J. Dennis, Chairman, President and Chief Executive Officer and a DirectorJames S. Gulmi, Senior Vice President-Finance (Principal Financial Officer)			
/s/ Joanna Barsh	/s/ Matthew C. Diamond		
Joanna Barsh, Director	Matthew C. Diamond, Director		
/s/ James S. Beard	/s/ Marty G. Dickens		
James S. Beard, Director	Marty G. Dickens, Director		
/s/ Leonard L. Berry	/s/ Thurgood Marshall, Jr.		
Leonard L. Berry, Director	Thurgood Marshall, Jr., Director		
/s/ William F. Blaufuss, Jr.	/s/ Kathleen Mason		
William F. Blaufuss, Jr., Director	Kathleen Mason, Director		
/s/ James W. Bradford			
James W. Bradford, Director			

CERTIFICATIONS

I, Robert J. Dennis, certify that:

1. I have reviewed this annual report on Form 10-K of Genesco Inc.;

2. Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this report;

3. Based on my knowledge, the financial statements, and other financial information included in this report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this report;

4. The registrant's other certifying officer(s) and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(e)) and internal control over financial reporting (as defined in Exchange Act Rules 13a-15(f) and 15d-15(f)) for the registrant and have:

a) Designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this report is being prepared;

b) Designed such internal control over financial reporting, or caused such internal control over financial reporting to be designed under our supervision, to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles;

c) Evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report based on such evaluation; and

d) Disclosed in this report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter (the registrant's fourth fiscal quarter in the case of an annual report) that has materially affected, or is reasonably likely to materially affect, the registrant's internal control over financial reporting; and

5. The registrant's other certifying officer(s) and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of the registrant's board of directors (or persons performing the equivalent functions):

a) All significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial information; and

b) Any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal control over financial reporting.

Date: April 2, 2014

/s/ Robert J. Dennis Robert J. Dennis Chief Executive Officer

CERTIFICATIONS

I, James S. Gulmi, certify that:

1. I have reviewed this annual report on Form 10-K of Genesco Inc.;

2. Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this report;

3. Based on my knowledge, the financial statements, and other financial information included in this report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this report;

4. The registrant's other certifying officer(s) and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(e)) and internal control over financial reporting (as defined in Exchange Act Rules 13a-15(f) and 15d-15(f)) for the registrant and have:

a) Designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this report is being prepared;

b) Designed such internal control over financial reporting, or caused such internal control over financial reporting to be designed under our supervision, to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles;

c) Evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report based on such evaluation; and

d) Disclosed in this report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter (the registrant's fourth fiscal quarter in the case of an annual report) that has materially affected, or is reasonably likely to materially affect, the registrant's internal control over financial reporting; and

5. The registrant's other certifying officer(s) and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of the registrant's board of directors (or persons performing the equivalent functions):

a) All significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial information; and

b) Any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal control over financial reporting.

Date: April 2, 2014

/s/ James S. Gulmi James S. Gulmi Chief Financial Officer

CERTIFICATION PURSUANT TO 18 U.S.C. SECTION 1350, AS ADOPTED PURSUANT TO SECTION 906 OF THE SARBANES-OXLEY ACT OF 2002

In connection with the Annual Report of Genesco Inc. (the "Company") on Form 10-K for the period ending February 1, 2014, as filed with the Securities and Exchange Commission on the date hereof (the "Report"), I, Robert J. Dennis, Chief Executive Officer of the Company, certify, pursuant to 18 U.S.C. Section 1350, as adopted pursuant to Section 906 of the Sarbanes-Oxley Act of 2002, that:

(1) The Report fully complies with the requirements of Section 13(a) or 15(d) of the Securities Exchange Act of 1934; and

(2) The information contained in the Report fairly presents, in all material respects, the financial condition and results of operations of the Company.

/s/ Robert J. Dennis Robert J. Dennis Chief Executive Officer April 2, 2014

CERTIFICATION PURSUANT TO 18 U.S.C. SECTION 1350, AS ADOPTED PURSUANT TO SECTION 906 OF THE SARBANES-OXLEY ACT OF 2002

In connection with the Annual Report of Genesco Inc. (the "Company") on Form 10-K for the period ending February 1, 2014, as filed with the Securities and Exchange Commission on the date hereof (the "Report"), I, James S. Gulmi, Chief Financial Officer of the Company, certify, pursuant to 18 U.S.C. Section 1350, as adopted pursuant to Section 906 of the Sarbanes-Oxley Act of 2002, that:

(1) The Report fully complies with the requirements of Section 13(a) or 15(d) of the Securities Exchange Act of 1934; and

(2) The information contained in the Report fairly presents, in all material respects, the financial condition and results of operations of the Company.

/s/ James S. Gulmi James S. Gulmi Chief Financial Officer April 2, 2014

Exhibit (99)

AUDITED FINANCIAL STATEMENTS

Genesco Employee Stock Purchase Plan

As of February 1, 2014 and February 2, 2013 and for Each of the Three Fiscal Years in the Period Ended February 1, 2014 with Report of Independent Registered Public Accounting Firm

Audited Financial Statements

As of February 1, 2014 and February 2, 2013 and for Each of the Three Fiscal Years in the Period Ended February 1, 2014

Contents

Report of Independent Registered Public Accounting Firm	1
Audited Financial Statements:	
Statements of Financial Condition	2
Statements of Changes in Plan Equity	3
Notes to Financial Statements	4

Report of Independent Registered Public Accounting Firm

To the Participants and Administrator of the Genesco Employee Stock Purchase Plan

We have audited the accompanying statements of financial condition of the Genesco Employee Stock Purchase Plan as of February 1, 2014 and February 2, 2013, and the related statements of changes in plan equity for each of the three fiscal years in the period ended February 1, 2014. These financial statements are the responsibility of the Plan's management. Our responsibility is to express an opinion on these financial statements based on our audits.

We conducted our audits in accordance with the standards of the Public Company Accounting Oversight Board (United States). Those standards require that we plan and perform the audit to obtain reasonable assurance about whether the financial statements are free of material misstatement. We were not engaged to perform an audit of the Plan's internal control over financial reporting. Our audits included consideration of internal control over financial reporting as a basis for designing audit procedures that are appropriate in the circumstances, but not for the purpose of expressing an opinion on the effectiveness of the Plan's internal control over financial reporting. Accordingly, we express no such opinion. An audit also includes examining, on a test basis, evidence supporting the amounts and disclosures in the financial statements, assessing the accounting principles used and significant estimates made by management, and evaluating the overall financial statement presentation. We believe that our audits provide a reasonable basis for our opinion.

In our opinion, the financial statements referred to above present fairly, in all material respects, the financial condition of the Genesco Employee Stock Purchase Plan at February 1, 2014 and February 2, 2013, and the changes in plan equity for each of the three fiscal years in the period ended February 1, 2014, in conformity with U. S. generally accepted accounting principles.

/s/ Ernst & Young LLP

Nashville, Tennessee April 2, 2014

Statements of Financial Condition

	February 1,		
Assets	2014	Febru	uary 2, 2013
Due from Genesco Inc.	\$ 78,470	\$	87,427
Total Assets	\$ 78,470	\$	87,427
Liabilities and Plan Equity			
Plan equity	\$ 78,470	\$	87,427
Total Liabilities and Plan Equity	\$ 78,470	\$	87,427

See accompanying notes.

Statements of Changes in Plan Equity

	For the Fiscal Year Ended					
		February 1, 2014		February 2, 2013		January 28, 2012
Employee contributions	\$	216,174	\$	210,090	\$	175,363
Options exercised		(195,996)		(157,242)		(132,997)
Distributions to withdrawn participants		(29,135)		(36,806)		(32,128)
Net (decrease) increase in plan equity		(8,957)		16,042		10,238
Plan equity at beginning of year		87,427		71,385		61,147
Plan Equity at End of Year	\$	78,470	\$	87,427	\$	71,385

See accompanying notes.

Note 1 <u>Summary of Significant Accounting Policies</u>

Basis of Accounting

The records of the Genesco Employee Stock Purchase Plan (the "Plan") are prepared on the accrual basis of accounting.

Administrative Expenses

All expenses incurred in administration of the Plan are paid by Genesco Inc. (the "Company") and are excluded from these financial statements.

Use of Estimates

The preparation of financial statements in conformity with U.S. generally accepted accounting principles requires Plan management to make estimates that affect the reported amounts of Plan assets and liabilities and disclosure of any contingent assets and liabilities at the date of the financial statements and the reported amounts of changes in Plan equity during the period. Actual results could differ from those estimates and the differences could be material.

Note 2 <u>The Plan</u>

Background and Summary

The following description of the Plan provides only general information. Participants should refer to the Plan document for a more complete description of the Plan's provisions.

The Plan became effective October 1, 1995 to advance the interests of the Company and its shareholders by attracting and retaining qualified employees and by encouraging them to identify with shareholder interests through the acquisition of shares of the Company's common stock.

Eligibility

Each employee who is not a highly compensated employee as described in Section 414(a) of the Internal Revenue Code of 1986 ("the Code"), and whose customary employment is greater than 20 hours per week and greater than five months per year is eligible to participate in the Plan if the employee has been employed by the Company for at least six months prior to the grant date. The Plan excludes statutory insiders and five percent shareholders.

Note 2 <u>The Plan, Continued</u>

Contributions

Contributions to the Plan are solely from participating employees of the Company who, through after-tax payroll deductions, may use their contributions to purchase common stock of the Company at the end of a one-year option period. The maximum number of shares available to any participant is the lesser of 2,000 a year or that number of shares equal to \$10,000 divided by the closing market price of the common stock on the grant date or the exercise date. The maximum contribution is the lesser of \$9,500 a year or 15% of the participant's base pay as of October 1. The minimum contribution is \$250 per participant per year. Shares will be purchased September 30 of the year following the October 1 grant date. The initial estimated grants were determined on October 1, 1995.

The participant will purchase shares of the Company's common stock at 95% of the market price on the exercise date. Options are to be granted each year, unless the Board of Directors of the Company, at its discretion, determines in advance that no options are to be granted. The cumulative number of shares which may be purchased under the Plan is 1,000,000. The options granted and rights thereto may not be sold, assigned, pledged or otherwise transferred.

Participant Accounts

Periodically throughout the year, each participant is provided with statements reflecting the value of his or her account. Participant contributions are held by the Company, which has an unsecured obligation to the Plan.

At the exercise date, the Company issues stock that is transferred to a brokerage firm and distributed according to the number of options exercised by each participant.

Vesting

Participants are 100% vested in the value of their account and may withdraw from the Plan at any time except during the period September 15 through September 30, which is the time that preparations are made for the issuance of the stock each year.

If a participant is terminated for any reason other than retirement, disability or death, the participant's involvement in the Plan and any unexercised options automatically terminate, and the participant will receive the account balance in cash.

Termination of the Plan

The Company reserves the right to terminate the Plan at any time. In the event of Plan termination, the balance of each participant's account shall be paid in cash as soon as is reasonably practical.

Plan Administrator

The Plan is to be administered by the compensation committee of the Company's Board of Directors or another designee of the Board of Directors.

Notes to Financial Statements (Continued)

Note 2 <u>The Plan, Continued</u>

Income Tax Status

The Plan is intended to qualify as an employee stock purchase plan within the meaning of Section 423 of the Code, as amended. Issuance of shares under this Plan are not intended to result in taxable income to participants in the Plan based on provisions of the Code. Accordingly, no income will result for federal income tax purposes when an option is granted or exercised; however, income may result upon disposition of the stock. Management believes that the Plan is operating in compliance with the Code and, therefore, no provision for income taxes has been reflected in the accompanying financial statements

Accounting principles generally accepted in the United States require plan management to evaluate uncertain tax positions taken by the Plan. The financial statement effects of a tax position are recognized when the position is more likely than not, based on the technical merits, to be sustained upon examination by the IRS. The plan administrator has analyzed the tax positions taken by the Plan, and has concluded that as of February 1, 2014, there are no uncertain positions taken or expected to be taken. The Plan has recognized no interest or penalties related to uncertain tax positions. The Plan is subject to routine audits by taxing jurisdictions; however, there are currently no audits for any tax periods in progress. The plan administrator believes it is no longer subject to income tax examinations for years prior to 2010.

Note 3 Options to Purchase Company Stock

At the beginning of each option period, the Company estimates the number of options to be granted based on participant contributions and the current stock price. At the end of the option period, the Company grants options to each Plan participant. In the event Plan contributions, withdrawals or the stock price are different than originally estimated, additional or fewer options may be granted at the end of the option period (exercise date).

		Option Period			
		10/01/13	10/01/12	10/01/11	
Options to Purchase Common Stock	Total	to 09/30/14	to 09/30/13	to 09/30/12	
Estimated options to be granted, January 28, 2012	3,605	0	0	3,605	
Estimated options to be granted - October 1, 2012	3,972	0	3,972	0	
Adjustment to estimated options to be granted	(318)	0	0	(318)	
Options exercised	(2,463)	0	0	(2,463)	
Options withdrawn	(901)	0	(77)	(824)	
Estimated options to be granted, February 2, 2013	3,895	0	3,895	0	
Estimated options to be granted - October 1, 2013	3,635	3,635	0	0	
Adjustment to estimated options to be granted	(22)	0	(22)	0	
Options exercised	(3,146)	0	(3,146)	0	
Options withdrawn	(860)	(133)	(727)	0	
Estimated options to be granted, February 1,					
2014	3,502	3,502	0	0	

The cumulative options exercised as of February 1, 2014, are 689,708.

Note 3 **Options to Purchase Company Stock, Continued**

	(Option Period			
	10/01/13 to 09/30/14	10/01/12 to 09/30/13	10/01/11 to 09/30/12		
Enrollment date	10/01/13	10/01/12	10/01/11		
Exercise date	09/30/14	09/30/13	09/30/12		
95% of fair market value of stock at date of exercise	NA	\$62.30	\$63.39		

The table below includes a rollforward of participants based on actual exercises and withdrawals for the option period.

	-			
Number of Participants	Total	10/01/13 to 09/30/14	10/01/12 to 09/30/13	10/01/11 to 09/30/12
Active, January 28, 2012	223	0	0	223
Enrollment - October 1, 2012	280	0	280	0
Exercised options	(172)	0	0	(172)
Withdrawn	(56)	0	(5)	(51)
Active, February 2, 2013	275	0	275	0
Enrollment - October 1, 2013	258	258	0	0
Exercised options	(198)	0	(198)	0
Withdrawn	(92)	(15)	(77)	0
Active, February 1, 2014	243	243	0	0